

# High-performance Embedded Workshop V.4.09 User's Manual

All information contained in these materials, including products and product specifications, represents information on the product at the time of publication and is subject to change by Renesas Electronics Corporation without notice. Please review the latest information published by Renesas Electronics Corporation through various means, including the Renesas Electronics Corporation website (http://www.renesas.com).

#### Notice

- 1. All information included in this document is current as of the date this document is issued. Such information, however, is subject to change without any prior notice. Before purchasing or using any Renesas Electronics products listed herein, please confirm the latest product information with a Renesas Electronics sales office. Also, please pay regular and careful attention to additional and different information to be disclosed by Renesas Electronics such as that disclosed through our website.
- 2. Renesas Electronics does not assume any liability for infringement of patents, copyrights, or other intellectual property rights of third parties by or arising from the use of Renesas Electronics products or technical information described in this document. No license, express, implied or otherwise, is granted hereby under any patents, copyrights or other intellectual property rights of Renesas Electronics or others.
- 3. You should not alter, modify, copy, or otherwise misappropriate any Renesas Electronics product, whether in whole or in part.
- 4. Descriptions of circuits, software and other related information in this document are provided only to illustrate the operation of semiconductor products and application examples. You are fully responsible for the incorporation of these circuits, software, and information in the design of your equipment. Renesas Electronics assumes no responsibility for any losses incurred by you or third parties arising from the use of these circuits, software, or information.
- 5. When exporting the products or technology described in this document, you should comply with the applicable export control laws and regulations and follow the procedures required by such laws and regulations. You should not use Renesas Electronics products or the technology described in this document for any purpose relating to military applications or use by the military, including but not limited to the development of weapons of mass destruction. Renesas Electronics products and technology may not be used for or incorporated into any products or systems whose manufacture, use, or sale is prohibited under any applicable domestic or foreign laws or regulations.
- 6. Renesas Electronics has used reasonable care in preparing the information included in this document, but Renesas Electronics does not warrant that such information is error free. Renesas Electronics assumes no liability whatsoever for any damages incurred by you resulting from errors in or omissions from the information included herein.
- 7. Renesas Electronics products are classified according to the following three quality grades: "Standard", "High Quality", and "Specific". The recommended applications for each Renesas Electronics product depends on the product's quality grade, as indicated below. You must check the quality grade of each Renesas Electronics product before using it in a particular application. You may not use any Renesas Electronics product for any application categorized as "Specific" without the prior written consent of Renesas Electronics. Further, you may not use any Renesas Electronics. Renesas Electronics product for any application for which it is not intended without the prior written consent of Renesas incurred by you or third parties arising from the use of any Renesas Electronics product for an application categorized as "Specific" or for which the product is not intended where you have failed to obtain the prior written consent of Renesas Electronics. The quality grade of each Renesas Electronics product is "Standard" unless otherwise expressly specified in a Renesas Electronics data sheets or data books, etc.
  - "Standard": Computers; office equipment; communications equipment; test and measurement equipment; audio and visual equipment; home electronic appliances; machine tools; personal electronic equipment; and industrial robots.
  - "High Quality": Transportation equipment (automobiles, trains, ships, etc.); traffic control systems; anti-disaster systems; anticrime systems; safety equipment; and medical equipment not specifically designed for life support.
  - "Specific": Aircraft; aerospace equipment; submersible repeaters; nuclear reactor control systems; medical equipment or systems for life support (e.g. artificial life support devices or systems), surgical implantations, or healthcare intervention (e.g. excision, etc.), and any other applications or purposes that pose a direct threat to human life.
- 8. You should use the Renesas Electronics products described in this document within the range specified by Renesas Electronics, especially with respect to the maximum rating, operating supply voltage range, movement power voltage range, heat radiation characteristics, installation and other product characteristics. Renesas Electronics shall have no liability for malfunctions or damages arising out of the use of Renesas Electronics products beyond such specified ranges.
- 9. Although Renesas Electronics endeavors to improve the quality and reliability of its products, semiconductor products have specific characteristics such as the occurrence of failure at a certain rate and malfunctions under certain use conditions. Further, Renesas Electronics products are not subject to radiation resistance design. Please be sure to implement safety measures to guard them against the possibility of physical injury, and injury or damage caused by fire in the event of the failure of a Renesas Electronics product, such as safety design for hardware and software including but not limited to redundancy, fire control and malfunction prevention, appropriate treatment for aging degradation or any other appropriate measures. Because the evaluation of microcomputer software alone is very difficult, please evaluate the safety of the final products or system manufactured by you.
- 10. Please contact a Renesas Electronics sales office for details as to environmental matters such as the environmental compatibility of each Renesas Electronics product. Please use Renesas Electronics products in compliance with all applicable laws and regulations that regulate the inclusion or use of controlled substances, including without limitation, the EU RoHS Directive. Renesas Electronics assumes no liability for damages or losses occurring as a result of your noncompliance with applicable laws and regulations.
- 11. This document may not be reproduced or duplicated, in any form, in whole or in part, without prior written consent of Renesas Electronics.
- 12. Please contact a Renesas Electronics sales office if you have any questions regarding the information contained in this document or Renesas Electronics products, or if you have any other inquiries.
- (Note 1) "Renesas Electronics" as used in this document means Renesas Electronics Corporation and also includes its majorityowned subsidiaries.
- (Note 2) "Renesas Electronics product(s)" means any product developed or manufactured by or for Renesas Electronics.

# Introduction

The High-performance Embedded Workshop is a powerful development environment for embedded applications targeted at Renesas micro-controllers. The main features are:

- A configurable build engine that allows you to set-up compiler, assembler and linker options by using GUI.
- An integrated text editor with user customizable syntax coloring to improve code readability.
- A configurable environment, which allows you to run your own tools.
- An integrated debugger, which allows you to build and debug in the same application.
- Version control support.

#### **Operating Environment for the High-performance Embedded Workshop**

This user's manual, online help, and release notes do not indicate that the correct operation of the High-performance Embedded Workshop is guaranteed for any types of host computers or peripheral devices.

#### Note on the Sample Source Files Generated by the High-performance Embedded Workshop

When a new workspace is created, sample source files for the microcomputer in use are automatically generated. These files contain sample code and this does not indicate that the operations of all programs based on that code are guaranteed. Please read the hardware manual for the microcomputer to check if the generated source code is applicable and modify the code as required.



#### About This User's Manual

This user's manual describes the High-performance Embedded Workshop system. This user's manual describes information on the basic "look and feel" of the High-performance Embedded Workshop and customizing the High-performance Embedded Workshop environment and detail the build and the debugging functions common to the High-performance Embedded Workshop products. The figures in this document show the High-performance Embedded Workshop operating with a debugger for the SuperH family. For details on the debugger, refer to the user's manual or help information for the emulator or simulator included in the package.

This user's manual does not intend to explain how to write C/C++ or assembly language programs, how to use any particular operating system or how best to tailor code for the individual devices. These issues are left to the respective user's manuals.

The High-performance Embedded Workshop is customized in various languages. This user's manual gives descriptions on the English version of the High-performance Embedded Workshop application.

#### **Document Conventions**

This user's manual uses the following typographic conventions:

Convention	Meaning
[Menu -> Menu Option]	'->' is used to indicate menu options (for example, [File -> Save As]).
FILENAME.C	Uppercase names are used to indicate filenames.
Key + Key	Used to indicate required key presses.
	For example, CTRL+N means press the CTRL key and then, whilst holding the CTRL key down, press the N key.

#### Figures

Some figures in this user's manual may differ from the objects they represent.

#### Trademarks

Microsoft, MS-DOS, Visual SourceSafe, Windows and Windows Vista are either registered trademarks or trademarks of Microsoft Corporation in the United States and/or other countries.

All other company and product names are registered trademarks or trademarks of their respective companies.

#### Website and Support

Renesas Electronics Tools Website

http://www.renesas.com/tools

Inquiries

http://www.renesas.com/inquiry



# Contents

1. Ov	1. Overview			
1.1	1.1 Workspace and project(s)			
1.2	Main	window	1	
1.2	2.1	Title bar	2	
1.2	2.2	Menu bar	2	
1.2	2.3	Toolbars	3	
1.2	2.4	Workspace window	6	
1.2	2.5	Editor window	8	
1.2	2.6	Output window	9	
1.2	2.7	Status bar	. 13	
1.3	Help	system	. 14	
1.4	Laund	ching the High-performance Embedded Workshop	. 15	
1.5	Creat	ing a new workspace	. 16	
1.6	Open	ing a workspace	. 16	
1.7	Using	g old workspaces	. 17	
1.8	Savin	g a workspace	. 17	
1.9	Closi	ng a workspace	. 19	
1.10	Exitir	ng the High-performance Embedded Workshop	. 19	
1.11	Comp	oonent system overview	. 19	
1.12	Mana	gement information files of High-performance Embedded Workshop	. 19	
1.13	Overv	view of Configurations and Sessions	. 20	
1.14	Overv	view of Macro-Recording Support facility and Test Support facility	. 24	
1.1	4.1	Example of test procedures	. 28	
1.1	4.2	Step 1: Recording a macro	. 30	
1.1	4.3	Step 2: Editing a macro (viewing records)	. 32	
1.1	4.4	Step 3: Playing a macro	. 33	
1.1	4.5	Step 4: Creating a test suite	. 33	
1.1	4.6	Step 5: Editing a test suite	. 34	
1.1	4.7	Step 6: Creating a test image file	. 36	
1.1	4.8	Step 7: Modifying the program before the test	. 37	
1.1	4.9	Step 8: Viewing the test result (unmatched)	. 39	
1.1	4.10	Step 9: Modifying the program back and executing the test again	. 39	
1.1	4.11	Step 10: Viewing the test result (matched)	. 40	
• •				
		asics		
2.1		uild process		
2.2		guring the Projects tab of the Workspace window		
2.3		ct files		
2.3		Adding files to a project		
2.3		Drag and drop of files and folders		
2.3		Removing files from a project		
2.3		Excluding a project file from build		
	2.3.5 Including a project file in build			
	2.4 User folders in the workspace			
2.5				
	2.5.1 Associating an application with a file group			
2.5		Creating a new file extension and file group		
2.5		Creating a new file extension		
2.6	Settin	g build options	. 57	



2.7	Build	l configurations	57
2.7	7.1	Selecting a build configuration	58
2.7	7.2	Adding a new build configuration	58
2.7	7.3	Removing a build configuration	
2.8	Build	ling a project	
2.8	3.1	Using Parallel Build	59
2.8	3.2	Building individual files	60
2.8	3.3	Building a project	60
2.8	3.4	Building multiple projects	61
2.8	3.5	Stopping tool execution	62
2.8	8.6	Deleting intermediate and output files produced in building	63
2.8	3.7	Configuring the Build tab of the Output window	65
2.8	3.8	Controlling the content of the Build tab of the Output window	
2.8	3.9	Displaying out of date files in the Workspace window	
2.9	File o	dependencies	
2.10	Conf	iguring the Workspace window	71
2.11	Inser	ting a project into the workspace	73
2.12		ng the current project	
2.13	Spec	ifying dependencies between projects	75
2.14	-	oving a project from the workspace	
2.15		tive projects paths in the workspace	
3. Ad	lvanc	ed Build Features	77
3.1	The l	puild process revisited	77
3.1	.1	What is a build?	77
3.2	Creat	ting a custom build phase	78
3.3	Orde	ring build phases	82
3.3	3.1	Build Order tab	83
3.3	3.2	Build File Order tab	86
3.3	3.3	File Mappings tab	87
3.4	Setti	ng custom build phase options	88
3.4	I.1	Options tab	89
3.4	1.2	Output Files tab	
3.4		Dependent Files tab	
3.5	Cont	rolling the build	
3.6	Logg	ing build output	
3.7	Chan	ging toolchain version	
3.8	Gene	rating a makefile	
3.9	Usin	g a makefile inside the High-performance Embedded Workshop system	96
3.10	Custo	omizing the High-performance Embedded Workshop linkage order	
4. Ed	itor		102
4.1		or window	
4.2		king with multiple files	
4.3		lard file operations	
4.3		Creating a new file	
4.3		Editing a file	
4.3		Saving a file	
4.3		Opening a file	
4.3		Closing files	
4.3	3.6	Pop-up menu to close the window	108



		100
4.4	Searching and navigating through files	
4.4		
4.4		
4.4	· · · · · · · · · · · · · · · · · · ·	
4.4	······································	
4.5	Bookmarks	
4.6	Printing a file	
4.7	Configuring text layout	
4.7	1.1 Page set-up	115
4.7	.2 Changing tabs	116
4.7	Auto indentation	117
4.8	Splitting a window	117
4.9	Changing the editor font	118
4.10	Syntax coloring	118
4.1	0.1 Changing text colors	119
4.1	0.2 Creating new keywords	120
4.1	0.3 Enabling/disabling syntax coloring	121
4.11	Templates	121
4.1	1.1 Defining a template	122
4.1	1.2 Deleting a template	123
4.1	1.3 Inserting a template	
4.12	Brace matching	
4.13	Setting the read-only attribute for a file	
4.14	Preventing modification of files while debugging	
4.15	Managing the editor columns	
4.16	Showing/hiding the column header	
4.17	Opening a file within the editor	
4.18	Tooltip watch	
4.19	Evaluate an expression	
4.17		120
5 To	ols Administration	130
5.1	Tool locations	
5.2	High-performance Embedded Workshop registration files	
5.3	Registering a component	
5.5 5.4		
	Unregistering a component	
5.5	Viewing and editing component properties	
5.6	Technical support	
5.7	Using On-Demand components	
5.8	Custom project types	138
( (	-t-mi-in- the Euclidean ment	1 4 0
	stomizing the Environment	
6.1	Customizing the toolbars	
6.2	Customizing the Tools menu	
6.3	Using custom placeholders	
6.4	Using the workspace and project log facilities	
6.5	Configuring the help system	
6.6	Keyboard shortcut customization	
6.7	Scope of a control in the setup	
6.7		149
6.7		
6.8	Specifying workspace options	150



6.	.8.1 Opening the last workspace at start-up	
6.	.8.2 Restoring files on opening a workspace	
6.	.8.3 Displaying workspace information on opening a workspace	
6.	.8.4 Saving the workspace before executing any tools	
6.	8.5 Prompting before saving a workspace	
6.	.8.6 Prompting before saving a session	
6.	8.7 Enabling auto-backup facilities	
6.	.8.8 Setting the projects to load on workspace open	
6.	.8.9 Specifying a default directory for new workspaces	
6.9	Using an external editor	
6.10	Customizing the font in your views	
6.11	Using the virtual desktop	
7. Ve	ersion Control	
7.1	Selecting a Version Control System	
7.2	Importing and exporting a set-up	
8. Cı	ustom Version Control System	
8.1	Defining Version Control menu options	
	.1.1 System-defined menu options	
8.	1.2 User-defined menu options	
8.2	Defining Version Control commands	
8.3	Specifying arguments	
8.4	Specifying comments	
8.5	Executable return code	
8.6	Specifying file locations	
8.7	Specifying file locations example	
8.8	Specifying environment	
8.9	Controlling execution of a Version Control System	
8.10	Specifying a user name and password	
8.11	Usage example of the Custom Version Control System	
	.11.1 Connecting the High-performance Embedded Workshop with RCS	
8.	.11.2 Connecting the High-performance Embedded Workshop with CVS	
	isual SourceSafe Version Control System	
9.1	Attaching Visual SourceSafe to a workspace	
9.2	Visual SourceSafe commands	
	2.1 Adding files to Visual SourceSafe	
	2.2 Removing files from Visual SourceSafe	
	2.3 Getting read-only copy of files from Visual SourceSafe	
	2.4 Checking out writable copy of files from Visual SourceSafe	
	2.5 Checking in writable copy of files into Visual SourceSafe	
	2.6 Viewing the status of files in Visual SourceSafe	
	2.7 Undoing a check out command in Visual SourceSafe	
	2.8 Viewing the history of files in Visual SourceSafe Visual SourceSafe command options	
9.3 9.4	Visual SourceSale command options Version Control setup	
9.4 9.5	Specifying file locations	
9.5 9.6	Adding Visual SourceSafe commands	
7.0	Auting visual Sourcesare commands	
10. S	Sharing Projects by Network Facilities	



10.1	Sharing projects by network facilities in a PC with Windows® XP Service Pack 2 or later		
10.2			
10.3	Setting the administrator user's password		
10.4	Adding new users to the system		
10.5	Changing your password		
10.6	Note on using the network facilities to share projects		
11. C	omparing Files		
11.1	Opening the Difference window		
12. N	avigation Facilities		
12.1	C function and #define navigation component		
12.2	C++ navigation component		
12.3	Jump to a definition from the editor		
12.4	Drag and drop navigation items		
12.5	Smart edit capability		
13. N	lap		
13.1	Managing section settings		
	1.1 Opening the Map Section Information window		
	1.2 Entering/exiting the edit mode		
	1.3 Adding a section group		
	1.4 Adding a section		
13	1.5 Adding an overlay group		
13	1.6 Automatically registering the unregistered section		
13	1.7 Editing a selected item		
13	1.8 Setting the primary section		
13	1.9 Setting a memory map		
13	1.10 Automatically allocating the memory resource		
13	1.11 Printing out the section settings tree		
13	1.12 Viewing unallocated areas		
13	1.13 Viewing sections of size 0		
13	1.14 Viewing the source code for the address		
13	1.15 Printing out the section list		
13.2	Viewing symbols		
13	2.1 Opening the Map Symbol Information window		
13	2.2 Printing out the map list		
13	2.3 Finding symbols		
13	2.4 Filtering the symbol information		
13	2.5 Viewing the source code for the address		
13	2.6 Printing out the symbol information		
14. C	ommand Line		
14.1	Opening the Command Line window		
14.2	Specifying a batch file		
14.3	Executing a batch file		
14.4	Stopping command execution		
14.5	Specifying a log file		
14.6	Starting or stopping logging		
14.7	Entering a full path to the file		
14.8	Pasting a placeholder		



14.9	Selecting all the window contents	
14.10	Copying the selection onto the clipboard	
14.11	Cutting out the selection to the clipboard	
14.12	Pasting the contents of the clipboard	255
14.13	Clearing the contents of the Command Line window	
14.14	Undoing the last operation	
14.15	Checking brace matching	
14.16	Resetting the status of a batch file	
14.17	Single-stepping in a batch file	
14.18	Setting a breakpoint in a batch file	
15. N	Iacro-Recording Support Facility	
15.1	Macro menu and toolbar	
15.2	Using the Macro dialog box	
15.3	Importing a macro file with existing macros	
15.4	Recording a macro	
15.5	Functions that can be recorded into macro files	
-		
15.6	Playing a macro	
15.7	Editing a macro	
15.8	Assigning a macro	
15.9	Configuring the Macro tab of the Output window	
1 C T		205
	est Support Facility	
16.1	Creating a test suite	
16.2	Opening and closing test suites	
16.3	Editing a test suite	
16.4	Adding tests to the test suite	
16.5	Creating a test image file	
16.6	Functions that can be saved as test-image data into test-image files	
	.6.1 Functions that can be saved into test-image files (common to all High-performance Embedded W	orkshop
	oducts) 292	
	.6.2 Functions that can be saved into test-image files (dependent on the debugger)	
16.7	Comparing a test image file	
16.8	Running tests	
16.9	Using the test browser	
16.10	Configuring the Test tab of the Workspace window	
16.11	Configuring the Test tab of the Output window	319
17. D	Pebugging Facility	
17.1	Preparations for debugging	321
17	.1.1 Compiling for debug	321
17	.1.2 Selecting a debugger	
17	.1.3 Editing project configuration	
17	.1.4 Configuring the debugger	
	.1.5 Downloading modules	
	1.6 Debugger sessions	
17.2	Viewing a program	
	.2.1 Opening the Editor window	
	2.2 Opening the Disassembly window	
- /	1 0 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	



17.2.3	Looking at the current PC position	
17.2.4	Highlighting the line at the PC	
17.3 Operating memory		
17.3.1	Opening the Memory window	
17.3.2	Setting data at a desired address in the Memory window	
17.3.3	Selecting a memory range	
17.3.4	Filling an area of memory with constant data	
17.3.5	Copying an area of memory	
17.3.6	Comparing the memory contents	
17.3.7	Testing an area of memory	
17.3.8	Saving memory contents in a text file	
17.3.9	Finding a value in memory	
17.3.10	Changing the display address	
17.3.11	Changing the scroll area	
17.3.12	Starting address to value of the register	
17.3.13	Tracking the stack pointer position	
17.3.14	Changing the program display position immediately after downloading	
17.3.15	Refreshing the Memory window	
17.3.16	Disabling refresh of the Memory window	
17.3.17	Regularly refreshing the Memory window	
17.3.18	Specifying the refresh interval	
17.3.19	Changing the data length	
17.3.20	Changing the radix	
17.3.21	Changing the code	
17.3.22	Setting the layout	
17.3.23	Changing the number of digits displayed	
17.3.24	Switching display or non-display of measurement result	
17.3.25	Saving an area of memory	
17.3.26	Loading a memory area from a file	
17.3.27	Splitting up the window display	
17.3.28	Verifying a memory area	
17.3.29	Changing text colors	
17.4 Displ	aying memory contents as an Image	
17.4.1	Opening the Image window	
17.4.2	Regularly refreshing the Image window	
17.4.3	Refreshing the Image window	
17.4.4	Specifying the refresh interval	
17.4.5	Viewing Images as Consecutive Frames	
17.4.6	Displaying the pixel information	
17.5 Displ	aying memory contents as Waveforms	
17.5.1	Opening the Waveform window	
17.5.2	Regularly refreshing the Waveform window	
17.5.3	Refreshing the Waveform window	
17.5.4	Specifying the refresh interval	
17.5.5	Zoom-in display	
17.5.6	Zoom-out display	
17.5.7	Resetting the zoom display	
17.5.8	Setting the zoom magnification	
17.5.9	Setting the horizontal scale	
17.5.10	Non-display of cursor	
17.5.11	Displaying the sampling information	



17.6	Looki	ing at I/O memory	407
17	7.6.1	Opening the IO window	408
17	7.6.2	Expanding an I/O register display	409
17	7.6.3	Modifying the values of I/O registers	
17	7.6.4	Refreshing the IO window	410
17	7.6.5	Disabling refresh of the IO window	410
17	7.6.6	Selecting the I/O register(s) to view	
17	7.6.7	Loading an I/O file	
17	7.6.8	Printing the currently displayed contents	
17	7.6.9	Saving the currently displayed contents	
	7.6.10	Finding an I/O register	
	7.6.11	Finding the next	
		ing at registers	
	7.7.1	Opening the Register window	
	7.7.2	Changing the register display radix	
	7.7.3	Switching Register Bank	
	1.7.4	Setting the layout	
	7.7.5	Choosing a register to be displayed	
	1.7.6	Modifying register contents	
	7.7.7	Setting the flag value	
	7.7.8	Splitting up the window display	
		Saving register contents	
	7.7.9		
	7.7.10	Refreshing the Register window	
	7.7.11	Disabling refresh of the Register window	
	7.7.12	Using register contents	
	7.7.13	Changing text colors	
17.8	-	fying the radix	
17.9		ting the target MCU	
17.10		g PC to the address at cursor	
17.11		lizing the debugger	
		ecting/disconnecting the debugger	
		iting your program	
	7.13.1	Continuing run	
	7.13.2	Running from reset	
17	7.13.3	Running program, ignoring any breakpoints	
17	7.13.4	Running to cursor	
17	7.13.5	Running from a specified address	
17	7.13.6	Continuing execution to a main function at a reset	424
17	7.13.7	Single step	425
17	7.13.8	Multiple steps	427
17.14	Stopp	ing your program	427
17	7.14.1	Stopping the program by the Halt toolbar button	427
17	7.14.2	Standard breakpoints (PC breakpoints)	428
17.15	View	ing the current status	429
17.16	View	ing the function call history	430
17	7.16.1	Opening the Stack Trace window	430
17	7.16.2	Viewing the source program	
17	7.16.3	Specifying the view	
17	7.16.4	Selecting an encoding format	
		g an external debugger	
	7.17.1	Configuring the Hitachi Debugging Interface to integrate with High-performance Embedded Worksl	



17.	17.2 Configuring the PD debugger to integrate with High-performance Embedded Workshop	433
17.	17.3 Configuring an external debugger to integrate with High-performance Embedded Workshop	434
17.18	Debugging functions dependent on the debugger	435
17.	18.1 Looking at labels	435
17.	18.2 Elf/Dwarf2 support	439
17.	18.3 Looking at variables	443
18. S	ynchronized Debugging	
18.1	The Synchronized Debug dialog box	
18	1.1 Managing configurations	
18	1.2 Defining the synchronized Debuggers	
18	1.3 Setting the synchronization options	
18	1.4 Setting the memory update option	
18	1.5 Synchronized debugging mode	
18	1.6 Start synchronized debugging	
	1.7 Update synchronized debugging	
18	1.8 Stop synchronized debugging	
18.2	Using High-performance Embedded Workshop while synchronized	
	2.1 Common functionality	
	2.2 Parallel mode functionality	
	2.3 Internal mode functionality	
18.3	Using the command line window when synchronized	
18.4	Glossary of terms	
10 т	achnical Support	490
	echnical Support	
19.1 19.2	Viewing the version information	
19.2 19.3	Check for updates Creating a bug report	
19.5	Creating a bug report	
Refer	ence	482
1. Ma	in Menus	
1.1	File Menu Options	
1.2	Edit Menu Options	
1.3	View Menu Options	
1.4	Project Menu Options	
1.5	Build Menu Options	
1.6	Debug Menu Options	
1.7	Setup Menu Options	
1.8	Tools Menu Options	
1.9	Test Menu Options	
1.10	Window Menu Options	
1.10	Help Menu Options	
2. W1	ndows	491
3. Co	mmands	
3.1	Command List (Alphabetic Order)	
3.2	Command List (Listed by Function)	494
4. Re	gular Expressions	497



High	High-performance Embedded Workshop V.4.09	
5. Pl	aceholders	
5.1	What is a placeholder?	
5.2	Inserting a placeholder	
5.3	Available placeholders	500
5.4	Placeholder tips	501
6. I/0	O File Format	
7. Sy	mbol File Format	
8. K	eyboard Shortcuts	
9. Di	rag and Drop in the Debugger	
10. U	Jsing Labels to View Your Code	
11. I	ntegrated Toolbars in a Components View	511
12. 7	To Build in Toolchain for High-performance Embedded Workshop V.1.x	514
13. H	HMAKE User Guide	
13.1	Command line	
13.2	File syntax	
13.3	Description blocks	516
13.4	Comments	
13.5	Message commands	



# 1. Overview

The functions for High-performance Embedded Workshop are explained in this manual.

This chapter describes the fundamental concepts of the High-performance Embedded Workshop.

# 1.1 Workspace and project(s)

High-performance Embedded Workshop has a concept of "Workspace" and "Project".

Workspace

Workspace is a largest unit when you make the program with High-performance Embedded Workshop. Workspace can have several projects. When you create a workspace, more than one project is needed, and one project is automatically made when you create the workspace.

• Project(s)

When you make the program, sometimes you make a certain function as a library and make hierarchical levels between several modules. In such cases, you can also create the library project and insert it to a workspace.



# 1.2 Main window

There are three main windows: the workspace window, the editor window and the output window. The workspace window shows the projects and files that are currently in the workspace; the editor window provides file viewing and editing facilities; and the output window shows the results of a various processes (e.g. build, version control commands and so on).



High-performance Embedded Workshop V.4.09



#### 1.2.1 Title bar

The title bar displays the name of current activate project and file. It also contains the standard **Minimize**, **Maximize** and **Close** buttons. Click the **Minimize** button to minimize the High-performance Embedded Workshop on the Windows® task bar. Click the **Maximize** button to force High-performance Embedded Workshop to fill the screen. Click the **Close** button to close the High-performance Embedded Workshop (this has the same effect as selecting [**File -> Exit**], or pressing ALT+F4).

#### 1.2.2 Menu bar

The menu bar initially contains eleven menus: **File**, **Edit**, **View**, **Project**, **Build** \*, **Debug**, **Setup**, **Tools**, **Test**, **Window** and **Help**. All of the menu options are grouped logically under these headings. For instance, if you want to open a file then the **File** menu is where you will find the right menu option; if you want to set-up a tool then the **Tools** menu is the correct selection.

<u>File Edit View Project Build Debug Setup Tools Test Window Help</u>

#### Note:

\*. If you use a debug-only project "Debugger only - xxxxx" created by High-performance Embedded Workshop V.4.01 or a later version, the **Build** menu will not be displayed by default. The **Build** menu is displayed, however, if the debug-only project "Debugger only - xxxxx" has been created by High-performance Embedded Workshop earlier than V.4.01.



1. Overview

#### 1.2.3 Toolbars

The toolbars provide a shortcut to the options that you will use the most often. There are twelve default toolbars: Editor, Search, Templates, Bookmarks, Default Window, Standard, Version Control, Map, Macros, Debug, Debug Run, and System Tools (as shown in the figures below).

With the default session, the High-performance Embedded Workshop does not initially display buttons of **Version Control** and peripheral functions on the toolbar.

If you use a debug-only project "Debugger only - xxxxx" created by High-performance Embedded Workshop V.4.01 or a later version, **Editor**, **Search**, **Templates**, **Bookmarks**, **Default Window**, and **Standard** will not be displayed by default.

Toolbars can be created, modified and removed via [**Tools -> Customize**] (see section 6.1, Customizing the toolbars, for further information).

#### Editor toolbar





#### Standard toolbar

	Tool Current Configuration	Current Session	Launch External Debugger
표 🧇 🛗 📇 2	Debug	SimSessionSH-4	- 21

#### Version Control toolbar

This toolbar is only available when a version control tool is being used in the current project.



Мар

MAP

#### Map toolbar



You can add a customized button, to which a created macro has been assigned, on the right to the standard macro buttons on the toolbar as shown below.



#### **Debug toolbar**

This toolbar is only available when a session is being used which has a target attached.

Enable/Disable Breakpoint										
Тодд			Dec	1						nect
Breakpoint		Hex		Oc	tal I	Co	nn	ect		
	÷	G	16	10	ļ					

#### **Debug Run toolbar**

This toolbar is only available when a session is being used which has a target attached.





#### System Tools toolbar

You can add an external tool button, to which an external tool menu has been assigned, on the right to the system tool button on the toolbar as shown below.



When the Standard toolbar is docked, it has a Control bar as shown in the figure below. If you want to move the docked Standard toolbar, click and drag its Control bar to the new location.

The figure below shows the Standard toolbar when it is docked and also when it is floating.



#### To dock a toolbar

Select one of the following operations:

- Double-click on the title bar of a floating toolbar, **OR**
- Drag the title bar of a floating toolbar and draw it toward an edge of a docked window, menu bar, toolbar or the High-performance Embedded Workshop main frame, on whose edge you would like to dock the window, until the shape of the floating bar changes.

#### To float a toolbar

Select one of the following operations:

- Double-click on the control bar of a docked toolbar, **OR**
- Drag the control bar of a docked toolbar and draw it away from the edge of the High-performance Embedded Workshop main frame and from an edge of the other docked windows, menu bar or toolbars.



#### 1.2.4 Workspace window

Workspace	×
□	
📄 💼 🚯 Tutorial	
📕 🚊 🔄 Assembly source file	
🔛 intpre.src	
vecttbl.src	
vhandler.src	
📄 🛁 🔂 C source file	
≟] dbsct.c	
🔄 🔛 resetpre.c	
sbrk.c	Ţ
Projects 🛃 Templates 🔍 Navigation 🚺 Test	

The Workspace window has four tabs (Projects, Templates, Navigation, and Test).

#### • Projects tab

Allows you to show the current workspace, projects and files. You can quickly open any project file or dependent file by double-clicking on its icon. See section 2.2, Configuring the Projects tab of the Workspace window, for more information on the Projects tab.

If you hover the mouse pointer over a file in the **Projects** tab then the file's full path will be displayed in a tool tip.

qpsdr 🔚	
- 🖆 Demoe C:\Work	space\DemoSH4\DemoSH4\dbsct.c
🖆 lowsrc.c	
🔤 🖆 resetprg.c	

#### • Templates tab

Allows you to display template settings. See section 4.11, Templates, for more information about a template.

#### Navigation tab





#### High-performance Embedded Workshop V.4.09

Allows you to provide jumps to various textual constructs within your project's files. What is actually displayed within the **Navigation** tab depends upon what components are currently installed. The figure above shows C macro definitions (C Defines) and C functions (C Functions) conforming to the ANSI standard. See Chapter 12, Navigation Facilities, for more information about navigation.

#### • Test tab

Allows you to setup or view test suites as part of the test-support facility. See Chapter 16, Test Support Facility, for more information about the test-support facility.

#### To allow the Workspace window or the Output window docking

Right-click anywhere inside the Workspace window or the Output window. Then a pop-up menu will be displayed.

<ul> <li>Allow Docking</li> </ul>	
-----------------------------------	--

If the **Allow Docking** option is checked, docking is allowed. Otherwise, docking is not allowed. Select the **Allow Docking** option to check or un-check it.

When the **Allow Docking** option is checked, you can dock a window, toolbar or menu bar to the edge of the Highperformance Embedded Workshop main window or to the edge of another docked window. You can also float them 'above' the other High-performance Embedded Workshop windows or outside the High-performance Embedded Workshop main window. Figure (i) below shows a docked "Workspace" window, and figure (ii) below shows a floating "Workspace" window.



When the **Workspace** window or the **Output** window is docked, it has a control bar as shown below.

If you want to move a docked window, click and drag its control bar to the new location.

Control bar	Close button
	X



- 1. Ensure that the **Allow Docking** option is checked on the window's pop-up menu.
- 2. Select one of the following operations:
  - Double-click on the title bar of a floating window, **OR**
  - Drag the title bar of a floating window and draw it toward an edge of a docked window, menu bar or toolbar, or the High-performance Embedded Workshop main frame, on whose edge you would like to dock the windows.

#### To float the Workspace window or the Output window

- 1. Ensure that the **Allow Docking** option is checked on the window's pop-up menu.
- 2. Select one of the following operations:
  - Double-click on the control bar of a docked window, **OR**
  - Drag the control bar of a docked window and draw it away from the edge of the High-performance Embedded Workshop main frame and from an edge of the other docked windows, menu bar or toolbar.
  - Drag the control bar of a docked window while pressing the "CTRL" key.

#### To hide the Workspace window or the Output window

Select one of the following operations:

- Click on the close button, which is located in the top right corner of the window, **OR**
- Right-click anywhere inside a floating window and select the **Hide** option on the pop-up menu.

Hide

#### To show the Workspace window or the Output window

Select [View -> Workspace] or [View -> Output] respectively.

#### 1.2.5 Editor window

The editor window is where you will work with the files of your project. The High-performance Embedded Workshop allows you to have many files open at one time, to switch between them, to arrange them and to edit them in whichever order you want to. By default, the editor window is displayed in a notebook style. This means that each file has a separate tab associated with it to aid in navigating between files (see the figure below).

Line	S.,	Source		
14		<pre>#include "typedefine.h"</pre>		
15				-
16		#pragma section \$DSEC		
17		static const struct {		
18		_UBYTE *rom_s;	/*	Start address of the
19		_UBYTE *rom_e;	/*	End address of the in
20		_UBYTE *ram_s;	/*	Start address of the
21		} DTBL[] = {		-
•				
🧼 d	Ibscl	.c 🧇 resetprg.c		

The editor contains a gutter (column) on the left-hand side of the window. The standard column allows the user to configure the position of bookmarks and software breakpoints (\*) quickly and easily. If you are unsure what purpose a column has or what the information it is displaying is if you place the mouse pointer over the column a tool tip (pop-up window) is displayed showing its identity.

The editor window can be customized via the **Format Views** dialog box, which can be invoked via [**Setup -> Format Views**]. This dialog box allows you to configure fonts, colors, tabs and so on for the editor window. It also allows the user to change the look of other views, which have been installed by High-performance Embedded Workshop. If you would prefer to use your favorite editor rather than the High-performance Embedded Workshop internal editor then specify your alternative in the **Options** dialog box, which can be invoked via [**Setup -> Options**]. For further details on how to use the editor, see Chapter 4 Editor.

#### Note:

\*. Allows you to set software breakpoints only when the debugger is connected.

#### 1.2.6 Output window

The **Output** window, by default, has seven tabs (**Build**, **Debug**, **Find in Files 1**, **Find in Files 2**, **Macro**, **Test**, and **Version Control**) on display.





#### **Build tab**

The Build tab shows the output from any build process (e.g. compiler, assembler and so on). If an error is encountered in a source file, an icon and the error message will be displayed in the Build tab, along with the source file name and line number.

If you click on a toolbar button or pop-up menu option associated with display of error messages, the line of the error message will be highlighted and the editor will show the source code of that line (if the line has any source code).

Double-clicking the line of an error message will also show the source code in the editor.

When you attempt an operation to display an error or warning, the status bar shows this error or warning message.

Right-clicking displays a pop-up menu containing available options.

A basic operation is allocated to the toolbar.

The functions of **Toolbar display** and **Customize toolbar** are also included in the pop-up menu displayed by rightclicking the toolbar area.

Pop-up Menu Option	Toolbar Button Function				
-	<b>0</b> 1	Highlights the next occurrence of error and shows the source code in the editor.			
-	<b>9</b> 1	Highlights the previous occurrence of error and shows the source code in the editor.			
-	<b>∆</b> ↓	Highlights the next occurrence of warning and shows the source code in the editor.			
-	Ɣ	Highlights the previous occurrence of warning and shows the source code in the editor.			
Display next Error/Warning/Info	<mark>%</mark> 1	Displays the editor that generated the next build error or warning.			
Display previous Error/Warning/Info	<u>&amp;</u> 1	Displays the editor that generated the previous build error or warning.			
Help	2	Shows the help information about the line.			
Go to Error/Warning/Info	-	Goes to the associated source line.			
Clear Window	0	Clears the contents of the window.			
Save		Saves the contents of the window into a text file.			
Сору	₿ <u>₽</u>	Copies the selected contents onto the Windows® clipboard.			
Toolbar display	-	Shows or hides the toolbar.			
Customize toolbar	-	Customizes toolbar buttons.			

The Build tab also shows an icon corresponding to the error message output by build execution.

Icon Name	lcon	Error Message Level	
Build Error	•	Error	
Build Warning	Δ	Warning	
Information	0	Information	



#### Debug tab

Shows the output from any debugger process. Any debug component that needs to display information will send its output to this window.

Right-clicking displays a pop-up menu containing available options.

A basic operation is allocated to the toolbar.

The functions of **Toolbar display** and **Customize toolbar** are also included in the pop-up menu displayed by rightclicking the toolbar area.

Pop-up Menu Option	<b>Toolbar Button</b>	Function
Clear Window	a	Clears the contents of the window.
Save		Saves the contents of the window into a text file.
Сору		Copies the selected contents onto the Windows® clipboard.
Toolbar display	-	Shows or hides the toolbar.
Customize toolbar	-	Customizes toolbar buttons.

#### Find in Files 1 and Find in Files 2 tab

Shows the results of the last **Find in Files** action. To activate find in files, select [**Edit -> Find In Files**], or click the **Find In Files** toolbar button. For further details on using Find in Files, see section 4.4.2, Finding text in multiple files.

Right-clicking displays a pop-up menu containing available options.

A basic operation is allocated to the toolbar.

The functions of **Toolbar display** and **Customize toolbar** are also included in the pop-up menu displayed by rightclicking the toolbar area.

Pop-up Menu Option	<b>Toolbar Button</b>	Function
Go to Occurrence	-	Go to the associated source line.
Clear Window	a	Clears the contents of the window.
Save		Saves the contents of the window into a text file.
Сору	₽ <u>₽</u>	Copies the selected contents onto the Windows® clipboard.
Toolbar display	-	Shows or hides the toolbar.
Customize toolbar	-	Customizes toolbar buttons.

#### Macro tab

Shows the current records of macros (macro-recording support facility). You can view information such as Highperformance Embedded Workshop command-line commands recorded into a High-performance Embedded Workshop macro file from execution of [**Tools -> Macro Recording**] to [**Tools -> Stop Macro**]. It is also possible to view this information while recording. For details on the macro-recording support facility, see Chapter 15, Macro-Recording Support Facility.



Right-clicking displays a pop-up menu containing available options.

A basic operation is allocated to the toolbar.

The functions of **Toolbar display** and **Customize toolbar** are also included in the pop-up menu displayed by rightclicking the toolbar area.

Pop-up Menu Option	<b>Toolbar Button</b>	Function
Clear Window	a	Clears the contents of the window.
Save		Saves the contents of the window into a text file.
Сору		Copies the selected contents onto the Windows® clipboard.
Toolbar display	-	Shows or hides the toolbar.
Customize toolbar	-	Customizes toolbar buttons.

#### Test tab

Shows the results and progress of the current test execution (test support facility). The test execution progress shows the current test being executed and the number of tests remaining. If errors occur then these are displayed in this window. For details on the test-support facility, see Chapter 16, Test Support Facility.

Right-clicking displays a pop-up menu containing available options.

A basic operation is allocated to the toolbar.

The functions of **Toolbar display** and **Customize toolbar** are also included in the pop-up menu displayed by rightclicking the toolbar area.

Pop-up Menu Option Too	Ibar Button	Function
------------------------	-------------	----------

Clear Window	0	Clears the contents of the window.
Save		Saves the contents of the window into a text file.
Сору		Copies the selected contents onto the Windows® clipboard.
Toolbar display	-	Shows or hides the toolbar.
Customize toolbar	-	Customizes toolbar buttons.

#### Version Control tab

Shows the results of version control actions. The tab is only displayed if a version control system is in use. For further details on version control, see Chapter 7, Version Control.

Right-clicking displays a pop-up menu containing available options.

A basic operation is allocated to the toolbar.



The functions of **Toolbar display** and **Customize toolbar** are also included in the pop-up menu displayed by rightclicking the toolbar area.

Pop-up Menu Option	<b>Toolbar Button</b>	Function
Clear Window	a	Clears the contents of the window.
Save		Saves the contents of the window into a text file.
Сору		Copies the selected contents onto the Windows® clipboard.
Toolbar display	-	Shows or hides the toolbar.
Customize toolbar	-	Customizes toolbar buttons.

Press the "SHIFT+ESC" key, and the **Output** window closes.

The color of text or background and font shown in the Output window can be customized in the same manner as in other windows. You can also customize the **Build** tab of the Output window so that the texts in the lines of error messages will be highlighted in a color different from that of the texts in other lines.

#### To customize the current colors

- 1. Select [Setup -> Format Views] to open the Format Views dialog box.
- 2. Select the **Output** item from the tree in the left-hand section of the dialog box and expand it.
- 3. Select the category for which you wish to change the color from the table below.

Category	Foreground Color of the Color Tab (Default)	Background Color of the Color Tab (Default)	Tab(s) to be Applied	Type of Output to be Applied
Text	SYSTEM	SYSTEM	All tabs	All
Build Error Text	Black	White	Build	Error
Build Warning Text	Black	White	Build	Warning
Information Text	Black	White	Build	Information

- 4. Change the selection in the **Foreground** and **Background** lists of the **Color** tab.
- 5. Click the OK button.

#### 1.2.7 Status bar

The status bar displays various information about the current state of the High-performance Embedded Workshop. The figure below shows the status bar.



Displays HEW status message	s Virtual d	lesktop	Line nu	ımber/l	Lines Ir	sertion/0	verwrite n	node	
/GUI command description	ו ו		Attributes		Column r	number	Caps I		n Lock
Ready	2 23 24	Default1 desktop	Read-write	62/87	4	INS		NU	M //

The customizable status bar feature allows the user to customize the display of the status bar area and its contents. The status bar will also now show target-related information where required by the specific target being used.

The bottom section of the status bar area shows the standard application-based information as in previous version of High-performance Embedded Workshop. Above this is the target-related information. The information displayed here is completely target dependant, and in some cases there may not be any target-related information to display.

The status bar can be switched on and off via [**View -> Status Bar**]. When any part of the status bar area is visible, this main menu option will completely switch off the status bar area. When the status bar is completely hidden, this main menu option will switch on the application-based section of the status bar.

The status bar can also be switched off via the status bar pop-up menu. The pop-up menu allows individual status bar items to be toggled on and off; including both application and target related areas as well as individual items on each.

The pop-up menu can be displayed by selecting the right-mouse button over the status bar area.

Showing/hiding the entire target related area (*)	Debugger	📑 📰 📰 Default1 desktop
Showing/hiding the entire application related area –	Application	

#### Note:

\*. The entire target related area of the status bar is hidden depending on the target being used.

All target related status bar items are switched off, the entire target related area of the status bar will be hidden by default. This can be re-shown by switching on one or more target related individual status bar items.

## 1.3 Help system

The **Help** menu is the rightmost menu on the High-performance Embedded Workshop menu bar. It contains the **Help Topics** menu option, which, when selected, takes you to the main High-performance Embedded Workshop help window.

To obtain help on specific dialogs click the context-sensitive help button, which is located in the top right-hand corner of each dialog (as shown in the figure below).





When this is clicked, the mouse pointer will change to a pointer with a question mark above it. Whilst the mouse pointer is in this state, click on the part of the dialog that you require assistance on. Alternatively, select the control for which you require help, and press the F1 key.

# 1.4 Launching the High-performance Embedded Workshop

To initiate the High-performance Embedded Workshop, open the **Start** menu of Windows<sup>®</sup>, select **Programs**, select **Renesas**, select **High-performance Embedded Workshop**, and then select the shortcut of the High-performance Embedded Workshop. The **Welcome!** dialog box will be displayed after the start-up screen.

Welcome	9	? ×
2	O <u>C</u> reate a new project workspace	OK Cancel
	<u>O</u> pen a recent project workspace:	
	C:\Workspace\DemoSH4\DemoSH4.hws	Administration
	O Browse to another project workspace	
P#4	O <u>S</u> tart synchronized session:	
<b>₩</b> ↔ <b>₩</b>	Default	

If you have recently used a workspace, **Open a recent project workspace** is selected by default. Otherwise **Create a new project workspace** is selected.

To create a new workspace select the **Create a new project workspace** button and click the OK button.

To open one of the recent project workspaces select the **Open a recent project workspace** button, select a workspace from the drop-down list, and click the OK button. The **Recent Workspace** list displays the same content as that seen in the workspace most recently used **File** list. This list also appears on the **File** menu.

To open a workspace by specifying a workspace file (".HWS" file), select the **Browse to another project workspace** button, and click the OK button.

To open the Synchronized Debug dialog box select the **Start synchronized session** button and select a configuration from the drop-down list and click the OK button. This item is only available when a synchronized debugging facility has been used.

To register or un-register a tool from the High-performance Embedded Workshop, click the Administration button.

Click the Cancel button to use the High-performance Embedded Workshop without opening a workspace.



If you do not wish to open the **Welcome!** dialog box next time you launch the High-performance Embedded Workshop, select [**Setup -> Options**]. The **Options** dialog box opens. Remove a tick mark from the **Display Welcome Dialog** checkbox in the **Confirmation** tab.

If you do not wish to view the start-up screen when launching the High-performance Embedded Workshop, remove a tick mark from the **Display Splash Screen** checkbox in the **Confirmation** tab of the **Options** dialog box.

# 1.5 Creating a new workspace

#### To create a new workspace

- 1. Select the **Create a new project workspace** option from the **Welcome!** dialog box and click the OK button or select [**File -> New Workspace**]. The **New Project Workspace** dialog box will be displayed.
- 2. Enter the name of the new workspace into the Workspace Name field. This can be up to 32 characters in length and contain letters, numbers, and the underscore character. Especially, do not use a minus sign, or a space. As you enter the workspace name, the High-performance Embedded Workshop will add a sub-directory and Project Name for you automatically. This can be changed if desired. This allows the workspace and project name to be different. To select the directory in which you would like to create the workspace, use the Browse button or type the directory into the Directory field manually.
- 3. Select the **CPU family** and **Tool chain** upon which you would like to base the workspace.
- 4. When a new workspace is created, the High-performance Embedded Workshop will also automatically create a project with the name specified in the **Project Name** field and place it inside the new workspace. The project types list displays all of the available project types (e.g. Application, Library etc.). Select the type of project that you want to create from this list. The project types displayed will be all valid types for the current pair of **CPU** family and **Tool chain**. The project types are classified in three classes: toolchain-only, debug-only (Debugger only xxxxx), and full project generator that configures both the debugger and toolchain aspect of the High-performance Embedded Workshop.
- 5. Click the OK button to create the new workspace and project. This then launches the wizard you have selected to guide you through the creation process.

#### Note:

It is not possible to create a workspace if one already exists in the same directory.

# 1.6 Opening a workspace

#### To open a workspace

- 1. Select **Browse to another project workspace** option from the **Welcome!** dialog box and click the OK button or select [**File -> Open Workspace**]. The **Open Workspace** dialog box will be displayed.
- 2. Select the workspace file (".HWS" file) that you want to open.



3. Click the **Select** button to open the workspace. If the High-performance Embedded Workshop is set up to display information when a workspace is opened, the Workspace Properties dialog box will be displayed. Otherwise, the workspace will be opened.

Note that whether the Workspace Properties dialog box is shown depends on the setting of either the **Show workspace information on workspace open** check box in the Workspace Properties dialog box or the Display workspace information dialog on opening workspace check box on the **Workspace** tab of the **Options** dialog box. The **Options** dialog box can be invoked via [**Setup -> Options**]. Click the OK button in the Workspace Properties dialog box to open the workspace.

Show workspace information on workspace open

The High-performance Embedded Workshop keeps track of the last workspaces that you have opened and adds them to the **File** menu under the **Recent Workspaces** submenu. This gives you a shortcut to opening workspaces, which you have used recently.

#### To open a recently used workspace

Select **Open a recent project workspace** in the **Welcome!** dialog box, select the name of the workspace from the dropdown list, and then click the OK button.

Another way is to select [File -> Recent Workspaces], and then from this submenu select the name of the workspace.

#### Note:

The High-performance Embedded Workshop only permits one workspace to be open at a time. Consequently, if you attempt to open a second workspace, the first will be closed before the new one is opened.

## 1.7 Using old workspaces

The High-performance Embedded Workshop can open any workspace that was created on a previous version of the High-performance Embedded Workshop. This should be automatically upgraded when the workspace is opened. A back-up version of the initial workspace or project file must be saved in the current directory of the file that has been upgraded.

## **1.8** Saving a workspace

The amount of information on workspaces (and projects) to be saved increases as the High-performance Embedded Workshop is upgraded. Thus the format in which workspaces can be saved is slightly different depending on the version of the High-performance Embedded Workshop.



To save a High-performance Embedded Workshop workspace in the current version's format, select [File -> Save Workspace].

To save the workspace in a previous version's format, select [**File** -> **Save Workspace As**]. If the current workspace has changed since it was last changed, you will be prompted to save it (to the current file and version). The **Save Workspace As** dialog box will then be launched:

Save Workspace As	? ×
Files:	
Saved Filename	Directory
Workspace_Name.hws Project_Name.hwp DefaultSession.hsf SimSessionSH-1.hsf	C:\WorkSpace\Workspace_Name C:\WorkSpace\Workspace_Name\Project_Name C:\WorkSpace\Workspace_Name\Project_Name C:\WorkSpace\Workspace_Name\Project_Name
F <u>o</u> rmat:	<u>P</u> refix:
HEW 4.09.00 Format (cur	rent)
HEW 4.09.00 Format (curr	ent)
HEW 4.08.00 Format	el
HEW 4.07.00 Format HEW 4.06.00 Format	
HEW 4.05.00 Format	
HEW 4.04.00 Format	
HEW 4.03.00 Format	
HEW 4.02.00 Format	
HEW 4.01.00 Format HEW 4.00.00 Format	

This contains a list of all the files currently associated with the workspace, a drop-list control containing a list of available versions (the current version is selected by default) and an edit control where you can enter a short prefix that will be used to change the name of all the associated files. When the prefix is changed, the File list is updated to display the new names.

# For example, to save the workspace in a format that can be loaded from High-performance Embedded Workshop V.4.02.00

- 1. Set the **Format** drop-down list: HEW 4.02.00 Format
- 2. Enter Prefix: 402\_
- 3. Click OK.



A copy of the workspace will be saved into the current directory (so all the source files, etc. will remain in use) but with the entered prefix, e.g.:

C:\Workspace\Workspace\_Name\402\_Workspace\_Name.hws

C:\Workspace\Workspace\_Name\Projet\_Name\402\_Projet\_Name.hwp

 $C: \workspace \workspace \noised \end{tabular} Name \end{tabular} Projet \noised \noised\n$ 

 $C: \workspace \workspace \noised \workspace \noised \noised\noised \noised \noised\n$ 

Note that the current workspace remains loaded as the working copy.

# 1.9 Closing a workspace

To close a High-performance Embedded Workshop workspace, select the [File -> Close Workspace] menu option. If there are any outstanding changes to the workspace or any of its projects you will be requested whether or not you wish to save them.

# 1.10 Exiting the High-performance Embedded Workshop

The High-performance Embedded Workshop can be exited by selecting [**File** -> **Exit**], pressing ALT+F4, or by selecting the close option from the system menu (which is opened by clicking the icon at the upper-left corner of the High-performance Embedded Workshop title bar).

# 1.11 Component system overview

The High-performance Embedded Workshop allows the user to extend the High-performance Embedded Workshop functionality by adding additional components to the system. This is achieved by registering the component in the **Tools Administration** dialog box. These components can add windows, menus and toolbars to the High-performance Embedded Workshop system. Examples of the components are the debugger and builder components of High-performance Embedded Workshop. The debugger component adds all of the menus and toolbars associated with the debugger and the builder component does the same for the build functionality. The components you have registered in the system will modify the look and feel of High-performance Embedded Workshop. In some cases you may not have some of the menus which you can see in this manual. For instance if the builder component is not installed you will not have the toolchain menu option in the **Build** menu.

# 1.12 Management information files of High-performance Embedded Workshop

The High-performance Embedded Workshop has the following files containing management information necessary for its correct operation.



- A file in which application-level setting is stored
- A file in which the default positions of windows are recorded
- High-performance Embedded Workshop database files
- Component's data files \*
- Keyboard shortcut layout files
- Macro information file and newly added macro files
- Custom project-type wizard files
- Syntax highlighting files
- Templates files
- Synchronized debug option configuration files \*

#### Note:

\*. Support for this function depends on the debugger.

These files are specific to each user account on the host computer. When multiple versions of the High-performance Embedded Workshop are installed, the files specific to each High-performance Embedded Workshop are saved for each user account.

## **1.13** Overview of Configurations and Sessions

Standard toolbar	Current Configuration	Current Session	
Standard			×
📇 🧇 🛗 🚟 👗	Debug 💌	SimSessionSH-4	<b>•</b> >
	SimDebug_SH-4	SimSessionSH-4	
	Debug	DefaultSession	
	Release		
	Configuration	Session	

The drop-down list box on the **Standard** toolbar shows the current configuration and session.

This example shows the configuration and session in which the Renesas SuperH Standard toolchain was selected at creation of a project and "SH-4 Simulator" was selected as the target debugger.

You can switch between different configurations and sessions by selecting one in the drop-down list box.

It is also possible to add or delete configurations or sessions as necessary.



#### • Configuration

#### What is configuration?

Configurations are the build option settings (e.g., output of debug information or optimization) having their own names. The term "configuration" can also be referred to as "build configuration".

In the figure of the **Standard** toolbar shown as an example, configurations "SimDebug\_SH-4", "Debug", and "Release" are available.

#### Configurations created by the High-performance Embedded Workshop

When a project is created after selection of the toolchain, the High-performance Embedded Workshop automatically creates configurations "Debug" and "Release".

#### Selecting "SuperH RISC engine" from CPU family and "Renesas SuperH RISC engine" from Tool chain respective

If you have selected a target debugger at creation of the project, the High-performance Embedded Workshop creates a configuration suitable for the debugger. If you have selected "SH-4 Simulator" in Targets at creation of the project, configuration "SimDebug\_SH-4" is created.



and the second second second second	et System for Debugging
	SH-4 Functional Simulator SH-4 Functional Simulator[Little endian] SH-4 Simulator(Little endian) SH-4 with BSC Simulator SH-4 with BSC Simulator(Little endian) SH-4 (SH7750R) Functional Simulator SH-4(SH7750R) Functional Simulator(Litt SH-4(SH7750R) Simulator SH-4(SH7750R) Simulator(Little endian)
	Target type : SH-4

Configuration names can be changed when creating a project. Directories corresponding to each of the configurations are created under the project directory \*. These directories have the names of the configurations.

#### Note:

\*. The project directory having the project name is created under the workspace directory used for creation of a new workspace.



#### How to set up build options

To set up build options kept with a configuration, select the configuration from the drop-down list box in the dialog box for set up build option opened from the Build menu. Options such as the compiler, assembler, and optimizing linkage editor can be set up. The figure below shows the dialog box used for setup of build options for the Renesas SuperH Standard toolchain.


. .

....

such as the compiler and SuperH RISC engine Standard Too	
Configuration : SimDebug_SH-4 Debug Release All Configurations Multiple Configurations	C/C++ Assembly Link/Library Standard Library CPU  Category: Source Show entries for: Include file directories Add Intert . Bemove Move.gp Move.gown Options C/C++: -cpu=sh4 -object="\$(CONFIGDIR)\\$(FILELEAF).ob]" -debug -gbr=auto -chgincpath -errorpath -global_volatile=0 -opt_range=all-infinite_loop=0.del_vacant_loop=0 OK Cancel

See section 2.7, Build configurations, for more information about a configuration.

#### • Session

#### What is session?

Sessions, which have their own names, are the units to manage various settings such as High-performance Embedded Workshop debugger options \* used for debugging with connection to the debugger and options specific to the debugger, and information on the Memory and Register windows and their positions. The term "session" can also be referred to as "debugger session".

In the figure of the Standard toolbar shown as an example, sessions "SimSessionSH-4" and "DefaultSession" are available. Information on each session is saved in an individual file in the High-performance Embedded Workshop project.

#### Note:

\*. There are various option settings such as the debugger to be connected with the High-performance Embedded Workshop, the object format for debugging, download modules, and the timing of connecting the High-performance Embedded Workshop with the debugger or executing a command batch by the High-performance Embedded Workshop.



#### Sessions created by the High-performance Embedded Workshop

When a project is created after selection of the toolchain, the High-performance Embedded Workshop automatically creates session "DefaultSession". If you have selected a target debugger at creation of the project, a session is automatically created for connecting the High-performance Embedded Workshop with the selected debugger. For example, if you have selected "SH-4 Simulator" in "Target" at creation of a project, session "SimSessionSH-4" is automatically created.

#### How to set up High-performance Embedded Workshop debugger options

To set up High-performance Embedded Workshop debugger options kept with a session, select the session from the drop-down list box in the Debug Settings dialog box opened from the Build menu. Options can be set up in the Target and Options tabs.

ebug Settings		I
SimSessionSH-4	Target Options	
SinSessionSH-4 DefaultSession	Larget	
All Sessions	SH-4 Simulator	-
	<u>C</u> ore:	
	Einde Dors Target	2
	Debug format:	-
	Ell/Dwarf2	1
	Download modules	
	Filename Offiset Address Format	<u>A</u> dd
	\$(CONFIGDIR)\\$(PRO 00000000 EII/Dwarf2	<u>⊠</u> ndb,
		Lenave
		<u>_</u> h
		Down

See section 17.1.6, Debugger sessions, for more information about a Debugger sessions.

# 1.14 Overview of Macro-Recording Support facility and Test Support facility

When coding is finished, the program must be tested. If a problem is found in testing, you will need to correct the problem and re-test the program. Testing takes a significant number of steps in development of programs.

To ease the process of testing, the High-performance Embedded Workshop provides the macro-recording support and test support facilities.



#### • Macro-recording support facility

The macro-recording support facility allows you to record operations, which are associated with the High-performance Embedded Workshop system applications **\*1**, build **\*2**, and debugging **\*3**, as High-performance Embedded Workshop command-line commands or to execute these recorded commands.

Files to record the operations (High-performance Embedded Workshop macro files) are command line batch files that have "hdc" as the extension and can be modified. These files are stored in the "Macros" folder within the High-performance Embedded Workshop management information folder under the application data folder for each user profile. By default, the file Default.hdc is created.

Note, however, that High-performance Embedded Workshop is not capable of recording all operations in the High-performance Embedded Workshop system corresponding to the High-performance Embedded Workshop command-line commands. For the operations that can be recorded, a macro record icon ( $\bigcirc$ ) is shown in the "Macro Recording" column of the menu list. This indicates that this operation can be recorded into a High-performance Embedded Workshop macro file.

#### Notes:

- \*1. Changing a project, session, or configuration
- **\*2**. Compilation and build
- **\*3**. Downloading a module, changing a memory value or register value, setting/deleting a software breakpoint, and running a program

The macro-recording support facility is available in the **Tools** menu and on the **Macros** toolbar.

Macros toolbar				
Mac	×			
•	Þ	00		

If the **Macros** toolbar is hidden, select [**Setup -> Customize**] to open the **Customize** dialog box. Select the **Macros** checkbox on the **Toolbars** tab of the **Customize** dialog box.

Tools Menu	Macros Toolba	Ibar Function		
Macros	-	Opens the Macro dialog box. This dialog box has the following features and lists		
		the macro names recorded in each of the selected High-performance Embedded		
		Workshop macro files.		
		- Creating a new blank macro file		
		- Importing an existing macro file		
		- Starting a macro record operation		
		- Starting a macro play function		
		- Editing a current macro file		
		- Assigning the selected macro		
		- Removing the selected macro		
Macro Recordine	g 🔍	Starts a macro record operation.		



High-performance Embedded	Workshop V.4.09
nigh pononnance Embodada	1101 Nonop 1.1.00

Tools Menu	Macros Too	Ibar Function
Play Macro	Þ	The Select Macro Function dialog box opens. Choose which macro you wish to
		execute.
-	00	Pauses the current macro operation.
Stop Macro		Stops the current macro operation.

The procedure from selection of **Record Macro** ( $\bigcirc$ ) to that of **Stop Macro** ( $\square$ ) consists one macro and one Highperformance Embedded Workshop macro file can record multiple macros (a set of High-performance Embedded Workshop command-line commands). A macro includes multiple High-performance Embedded Workshop commandline commands.

The figure below shows the High-performance Embedded Workshop macro file and macros.



The current records of macros are shown in the **Macro** tab of the **Output** window. The "Build" macro in the figure above is outputted as follows.



See section 1.14.1, Example of test procedures, for an operation procedure.

#### • Test support facility

The test support facility allows you to create a test image file of the selected test items (test image data), execute a macro created by the macro-recording support facility or an existing High-performance Embedded Workshop command batch file, and compare a test image file with the current High-performance Embedded Workshop system. Test image data can be saved into a test image file (.HIF).

Note, however, that High-performance Embedded Workshop is not capable of acquiring test-image information of all functions in the High-performance Embedded Workshop system. For the items from which test-image information can be acquired, see section 16.6, Functions that can be saved as test-image data into test-image files.



The typical test procedure is outlined in the figure below. See section 1.14.1, Example of test procedures, for a test procedure.



The test support facility is available in the **Test** menu and in the pop-up menu of the **Test** tab of the workspace window.

The first step towards using the test support facility is to create a test suite (a set of tests). This is available on the **Test** menu. Once a test suite is created, following operations can also be done in the **Test** tab of the workspace window.

Modify the test suite and specify macros to be executed during the test and the test image file for comparison.

Then select some test items to be saved a test image file. Acquire and save test image of the selected test items, which you wish to use for comparison.

After completion of the test (macros) the selected test items are compared. The result is displayed in the test browser window.

If there is any difference between the test result and the saved the test image file, the result of the test item is shown as "FAIL". You can view the details of the error by double-clicking the line of this test item.



Test Browser			×
Test Name Description	Result		
TEST_A Load and Reset and Step	FAIL		
<b>▲</b>			►
Component	Result	Details	
CPU - Memory-SimSessionSH-4	FAIL	Failed in memory range: 0x00000000 - 0x0000000f	
•			Þ

#### 1.14.1 Example of test procedures

This page introduces an example of test procedures, which is a sequence through build and debugging using the macrorecording support facility and test support facility.

This example uses a program that sorts ten random data items written in C language, which are typically included in the Tutorial project in an emulator debugger or in a project for which "Demonstration" has been selected as the project type in the SuperH family simulator debugger, in ascending/descending order. This example lets you see a difference in data stored in an array.

Before starting the test in this example, array "a" in the program code must be changed from a local to global variable.

#### • Before test

First, open a workspace and connect the High-performance Embedded Workshop to the debugger.

- 1. Create a new project workspace or open an existing project workspace. For creation of a project workspace, see section 17.1.2, Selecting a debugger.
- 2. Select a session, which you wish to use for connection with the debugger, from the session list on the **Standard** toolbar. For selection of a session, see section 17.1.6 (1), Selecting a session.

After some operations, a **Confirmation Request** dialog box may appear. If you have executed build all on the entire program, a **Confirmation Request** dialog box appears by default asking if you wish to download the program. To download the program, click the **Yes** button.

Since the action of clicking on the buttons in this confirmation dialog box cannot be recorded into macros, this dialog box will open next time you execute a macro. To continue the macro execution, you should click one of the buttons. The execution must be continued if you are using the test-support facility, which is used to test a sequence of operations. For this reason, make the following setting so that the confirmation dialog box will not appear.

In this example, select the **Don't ask this question again** check box in **Confirmation Request** dialog box.





Save the High-performance Embedded Workshop environment setting that you wish to use for tests so that the environment for the tests will always be the same.

- 1. Select [File -> Save Workspace].
- 2. Select [File -> Save Session].

You can always use the same environment to start a test by loading the High-performance Embedded Workshop environment setting that has been saved.

#### • Preparing a test script to be used in the example of test procedures

In this example, use the macro-recording support facility to record the following combination of operations as Highperformance Embedded Workshop command-line commands into a macro. This macro should be used as a test script at execution of a test.

Classification	Operation		
Preparation before the running the program - Initialize values	Reload the session. (See "Step 1 (2)".)		
- Prepare a program	Initialize register values *. (See "Step 1 (3)".)		
	Build the program for demonstration *. (See "Step 1 (4)".)		
Download the program	Download the program for demonstration. (See "Step 1 (5)".)		
Add variables, run the program, and check the result	Use the memory fill function to set the memory value of array "a" to 0. (See "Step 1 (6)".		
	Reset the CPU. (See "Step 1 (7)".)		
	Run the program until a selected line within a main function is reached. (See "Step 1 (9)".)		
	Add the array, which stores random data, to the <b>Watch</b> window *. (See "Step 1 (10)".)		

#### Note:

\*. Support for this function depends on the debugger.



#### • Example of test procedures

Invoke the test support facility and select the data in an array (memory content at the address of an array) as the target of the test. Save test-image information to the test image file that you wish to use for comparison and modify the program so that the number of times to store data into the array will be reduced. Execute the test and see that the data stored in the array has changed. Then modify the program again to restore the number of times to store data and execute the test. Make sure that data in the array is the same that in the saved test image file.

Step 1: Recording a macro

Step 2: Editing a macro (viewing records)

Step 3: Playing a macro

Step 4: Creating a test suite

Step 5: Editing a test suite

Step 6: Creating a test image file

Step 7: Modifying the program before the test

Step 8: Viewing the test result (unmatched)

Step 9: Modifying the program back and executing the test again

Step 10: Viewing the test result (matched)

#### 1.14.2 Step 1: Recording a macro

In this example, use the macro-recording support facility to record the following combination of operations as Highperformance Embedded Workshop command-line commands.

Recorded High-performance Embedded Workshop command-line commands are output to the **Macro** tab of the **Output** window. However, some operations open another tab.

- 1. Click the **Record Macro** button (●) on the **Macros** toolbar. The mouse pointer is modified to include the record icon ( ▶●). The message "Recording..." is shown in the **Macro** tab of the **Output** window.
- 2. Select [File -> Refresh Session]. ■This records the refresh\_session command.
- Click the Registers button (E1) on the CPU toolbar to open the Register window. Double-click the register, of which you wish to change the value, to invoke the Set Value dialog box. Enter 0 in Value.
   This records the register\_set command.
- 4. Click the **Build** button () on the **Standard** toolbar to build the program for demonstration. This records the build command.
- 5. Double-click to download the program for demonstration within the Download modules folder in the Projects tab of the workspace window.
   This records the file\_load command.



6. Click the **Memory** button () on the **CPU** toolbar to invoke the **Display Address** dialog box. Enter the address of the data field in **Display Address**. Then select **Fill** from the pop-up menu. Enter 0 as the data in the **Fill** dialog box.

This records the memory\_fill command.

- Click the Reset CPU button () on the Debug Run toolbar to reset the CPU.
   This records the reset command.
- 8. Open a source file including a main function in the editor window.
- 9. Right-click on the last line of the main function and select **Go To Cursor** from the pop-up menu. **This records the go\_till command.**
- 10. Add the array element "a[6]" storing random data to the **Watch** window. **This records the watch\_add command.**
- 11. Click the **Stop Macro** button ( $\square$ ) on the **Macros** toolbar. The mouse pointer returns to be an arrow.
- 12. The **Add New Macro Function** dialog box opens. Enter a macro name (e.g., Demo). The message "Finished recording 'Demo'." is shown in the **Macro** tab of the **Output** window.

To view the recorded High-performance Embedded Workshop command-line commands, be sure to select the **Macro** tab of the **Output** window.

× •	01 01 AJ AT 21 21 0 m 2 ?
	Recording
	refresh session
	register set RO OxO
	register set R1 0x0
	register set R2 0x0
	register set R3 OxO
:	register set R4 OxO
:	register_set R5 0x0
	register set R6 0x0
1	register_set R7 OxO
	register_set R8 OxO
1	register_set R9 OxO
:	register_set R10 OxO
:	register_set R11 OxO
11	register_set R12 OxO
1	register_set R13 OxO
11	register_set R14 OxO
1	build wait
	file_load Elf/Dwarf2 \"\$(CONFIGDIR)\\\$(PROJECTNAME).abs\" OxO p byte
1	memory_fill 0x70000000 0x70000027 0x0000 long noverify
11	reset
UI 1	go_till go Ox2OE4
1	watch_add Watch1 {a[6]}
	Finished recording 'Demo'.

\Macro /

# 1.14.3 Step 2: Editing a macro (viewing records)

- 1. Select [**Tools -> Macros**] to invoke the **Macro** dialog box. The High-performance Embedded Workshop macro file "Default", in which a macro has been recorded, is listed in **Current macro file**.
- 2. Select the macro "Demo" in **Macro functions** and click the **Edit** button. The editor window shows the Highperformance Embedded Workshop macro file "Default.hdc", in which a macro has been recorded.

Within a High-performance Embedded Workshop macro file, the High-performance Embedded Workshop command-line commands of the operations listed below are included between the lines showing information to identify macros.

When a macro includes operations such as running a program, the **Debug** tab of the **Output** window shows information including the cause of a stop. This may cause difference in comparison of test results. To clear the contents of the tabs in the **Output** window, the High-performance Embedded Workshop command-line command clear\_output\_window is available. It is possible to add the line at the next line of "#Macro Demo-Start" to clear the information in the **Debug** tab of the **Output** window.

clear\_output\_window debug

C:\Documents and Settings\\Macros\Default.hdc					
Line	Source	C	S.,	Source	
1					
2				#Macro Demo -Start 🔶	
3				refresh_session	
4				clear_output_window_debug	
5				register_set R0 0x0	
6				register_set Rl 0x0	
7				register_set R2 0x0	
8				register_set R3 0x0	
9				register_set R4 0x0	
10				register_set R5 0x0	
11				register_set R6 0x0	
12				register_set R7 0x0	
13				register_set R8 0x0	
14				register_set R9 0x0	
15				register_set R10 0x0	
16				register_set R11 0x0	
17				register_set R12 0x0	
18 19				register_set R13 0x0	
20				register_set R14 0x0 build all wait	
21				file_load Elf/Dwarf2 \"\$(CONFIGDIR)\\\$(PROJECTNAME)	
22				memory fill 0x70000000 0x70000027 0x0000 long nover	
23				reset	
24				go_till go 0x20E4	
25				watch add Watchl {a[6]}	
26				#Macro Demo -End	
•					

Modify the description as required and click the **Save File** button ( $\square$ ) on the **Editor** toolbar. Close the Highperformance Embedded Workshop macro file "Default.hdc".

#### 1.14.4 Step 3: Playing a macro

- 1. Click the **Play Macro** button ( $\triangleright$ ) on the **Macros** toolbar. The **Select Macro Function** dialog box opens.
- 2. Selecting the macro "Demo" starts playing the recorded commands. The mouse pointer is modified to include the play icon  $(\mathbf{A}^{\mathbf{D}})$ .
- 3. When the playback of a macro is completed, the mouse cursor returns to be an arrow.

# 1.14.5 Step 4: Creating a test suite

1. Select [Test -> Create New Test Suite] to invoke the Create New Test Suite dialog box.

Create New Test Suite	? ×
Test suite name:	ОК
Test_Suite_Demo	Cancel
Directory for test suite:	
C:\Workspace\DemoSH4	<u>B</u> rowse

- 2. Enter "Test\_Suite\_Demo" in Test suite name.
- 3. Initially the workspace directory is shown in **Directory for test suite**. This can be modified as required.
- 4. Click the OK button.
- 5. Displays the **Test** tab of the workspace window. Shows the "Test\_Suite\_Demo" test-suite icon.

Test_S	Suite_Demo		X
Projects	🕫 Templates	< Navigation	📝 Test

#### 1.14.6 Step 5: Editing a test suite

Select the macro recorded at step 1 for running a test.

- 1. Right-click on the "Test\_Suite\_Demo" test-suite icon to display a pop-up menu.
- 2. Select Edit Test Suite to open the Modify Test Suite dialog box.
- 3. Click the **Add** button. The **Add New Test** dialog box opens.
- 4. Enter "Test\_Demo" in **Test name**.
- 5. Enter "Demonstration" in **Test description**.
- 6. Click the Add button. The Add Test Script dialog box opens.
- 7. Select the "Demo" checkbox in Select macros to add to test.
- 8. Click the **OK** button.

Add Test Script			? ×
Script type:			ОК
HEW Macro		<b></b>	Cancel
Select <u>m</u> acros to add	to test:		
Test Name	File Path		
🗹 Demo	Default		
-			

9. **Test image file** shows the test suite name entered at step 4, '\_', and test name "Test\_Demo" entered in **Test name** (test suite name\_test name). This can be modified as required.



High-performance Embedded Workshop V.4.09

Add New Test		? ×
Test <u>n</u> ame:		ок
Test_Demo		
T <u>e</u> st description:		Cancel
Demonstration		
Tests to e <u>x</u> ecute:		
Macro Name	File Path	<u>A</u> dd
Demo	Default	<u>R</u> emove
		Move to <u>t</u> op
		Move <u>u</u> p
		Move <u>d</u> own
		Move to <u>b</u> ottom
Test jmage file:		
C:\Workspace\Dem	oSH4\Test_Suite_Demo_Test_Demo.hif	Bro <u>w</u> se

10. Click the **OK** button. The test "Test\_Demo" is added to the **Test cases** list.

Modify Test Suite		? ×
Test cases:		ОК
Test Name Test_Demo	Description Demonstration	Cancel
		Add Modify <u>R</u> emove

11. Click the **OK** button. The "Test\_Demo" test icon appears under the "Test\_Suite\_Demo" test-suite icon.

	Suite_Demo st_Demo		
Projects	🛃 Templates	🔄 Navigation	📝 Test

#### 1.14.7 Step 6: Creating a test image file

To compare the Memory content at the address of an array or the contents of the **Watch** window, the range of data acquisition must be selected.

The test image of a test item is saved to the test image file "Test\_Suite\_Demo\_Test\_Demo.hif" specified at step 5.

- 1. Right-click on the "Test\_Demo" test icon to display a pop-up menu.
- 2. Select **Edit Test Image File** to invoke the **Edit Test Image File** dialog box. The title of the dialog box includes the test name "Test\_Demo" and test-image file name "Test\_Suite\_Demo\_Test\_Demo.hif".
- 3. Select the checkboxes **Memory-xxxxx** (under **CPU** in **Available components**) and **Watch-xxxxx** (under **Symbol**).

Edit Test Image File - Test_Demo - Test_Suite_Demo_Te	est_Demo <mark>?</mark> 🗙
Available components:	ОК
🗾 🔤 StackTrace-SimSessionSH-4 📃 🔺	
🛄 🛄 Trace-SimSessionSH-4	Cancel
✓ ✓ ☑ ৣ Memory-SimSessionSH-4	<u>C</u> heck All
- 🗖 🚝 Register-SimSessionSH-4	
🔤 🗖 📰 Status-SimSessionSH-4	<u>S</u> ettings
🗖 🔊 Output	
🔤 🗖 🔊 Build Output Window	Import
Symbol	
Cocals-SimSessionSH-4	
■ (▼ Watch-SimSessionSH-4 )	
· · · · ·	1

- 4. The Memory content at the address of an array is saved into the test-image file. Double-click **Memory-xxxxxx** under **CPU** in **Available components**. The **Edit Test Memory Ranges** dialog box opens.
- 5. Click the Add button to display the Add Memory Range dialog box.
- 6. Enter the start address of an array in **Start address** and end address in **End address** and select the size in **Access size**.
- 7. Click the **OK** button.



High-performance Embedded Workshop V.4.09

Add Memory Range	? ×
Start address:	ОК
_a 🗾	Cancel
End address: _a+0x27	
Access size:	

- 8. Click the **OK** button in the **Edit Test Memory Ranges** dialog box.
- 9. The content of array element "a[6]" in the "Watch1" sheet is saved into the test-image file. Double-click Watchxxxxxx under Symbol in Available components of the Edit Comparison Settings dialog box. The Edit Test Watch dialog box opens.
- 10. Select the checkbox of "a[6]" in **Symbols**.
- 11. Click the **OK** button.

Edit Test Watch	? ×
Watch Sheet:	
Watch1 💌	OK
<u>S</u> ymbols:	Cancel
<b>▼</b> a6)	
<u>C</u> urrent Setting:	
Watch1 : a[6]	
Lgnore address	

12. Click the **OK** button in the **Edit Test Image File** dialog box.

In the state of current High-performance Embedded Workshop system, the test image specified in test item is acquired, and the test image is saved to the test image file created at step 5.

#### 1.14.8 Step 7: Modifying the program before the test

In this example, the target of the test is a selected range of memory data in arrays. Then modify the program to reduce the number of times to store data into the array, so that the saved test-image information and the actual memory data will not match.



Line	Source Ad	C.,	S.,	Source
21				
22				long a[10]; —
23				
24	00002000			void main(void)
25				(
26				long min, max;
27				long j;
28				int i;
29				
30	00002006			printf("### Data Input ###\n");
31				_
32	0000201a			for( i=0; i<10; i++ ){
33	00002022			$j = rand(); \rightarrow 8$
34	00002032			if(j < 0)(
35	00002036			j = -j;
36				}
37	00002038			a[i] = j;
38	00002044			printf("a[%d]=%ld\n",i,a[i]);
39				<u>ل</u> ے (
•				

- 1. Open a source file including a main function in the editor window and modify the program to reduce the number of times to store data into the array.
- 2. Click the **Save File** button ( $\square$ ) on the **Editor** toolbar.
- 3. Right-click on the "Test\_Demo" test icon to display a pop-up menu.
- 4. Select **Run Test Case** to start the test. The progress and result of the test are shown in the **Test** tab of the **Output** window.



5. When the test is completed, the **Test Browser** window opens.

# 1.14.9 Step 8: Viewing the test result (unmatched)

You can view the test result in the Test Browser window.

× •	<b>F</b>			
$ \Pi $	Test Name	Description		Result
	Test_Demo	Demonstration		FAIL
	•			<u> </u>
	Component		Result	Details
	CPU - Memory-SimSe	ssionSH-4	FAIL	Failed in memory range: 0x70000000 - 0x70000027
	Symbol - Watch-SimS	essionSH-4	FAIL	Symbol data is not matching.
	•			Þ

- 1. To view the details of unmatched data, double-click a test item with a red icon in the bottom tab.
- 2. When **Symbol Watch-xxxxx** is double-clicked, for example, the **Test Comparison Details** dialog box appears as shown below. This dialog box shows the values of both the saved test-image information (Src=xxxxxx) and the result of this test (Dest=xxxxxx) regarding elements in arrays.

Test Comparison Details	? ×
Test_Suite_Demo	OK 1
Symbol - Watch-SimSessionSH-4	
Information:	<u>Export</u>
Failed at symbol a[6], Src = (long)H'00002704{70000018}, Dest =	(long)H'0000( 🔺

- 3. Clicking the **Export** button saves the test result into a text file.
- 4. Click the **OK** button.

# 1.14.10 Step 9: Modifying the program back and executing the test again

Modify the program to restore the number of times to store data in the array and then execute the test.

- 1. Return the number from 8 to 10 and build the program.
- 2. Click the **Save File** button ( $\square$ ) on the **Editor** toolbar.
- 3. Right-click on the "Test\_Demo" test icon to display a pop-up menu.



- 4. Select **Run Test Case** to start the test.
- 5. When the test is completed, the **Test Browser** window opens.

# 1.14.11 Step 10: Viewing the test result (matched)

You can view the test result in the Test Browser window. Make sure that the test result now matches the saved test-image information.

× • • •				
Test Name	Description	B	esult	
Test_Demo	Demonstration	SI	UCCESS	
			1	Þ
Component		Result	Details	
CPU - Memory-SimSe	essionSH-4	SUCCESS		
Symbol - Watch-Sim9	iessionSH-4	SUCCESS		
II				►



# 2. Build Basics

This chapter explains the general basic functions of the High-performance Embedded Workshop whilst the more advanced features can be found in chapter 3, Advanced Build Features.

# 2.1 The build process

The typical build process is outlined in the figure below. This may not be the exact build process that your installation of High-performance Embedded Workshop will use, as it depends upon the tools that were provided with your installation of High-performance Embedded Workshop (you may not have a compiler, for instance). In any case, the principles are the same - each phase of the build takes a set of project files and builds them; if every file builds successfully then the next phase is executed.



In the example shown in the figure above, the Compiler is the first phase, the Assembler is the second phase and the Linker is the third and final phase. During the Compiler phase, the C/C++ source files from the project are compiled in turn. During the Assembler phase, the assembler source files are assembled in turn. During the Linker phase, all library files and output files from the Compiler and Assembler phases are linked together to produce the load module.

The build process can be customized in several ways. For instance, you can add your own phases, disable phases, delete phases and so on. These advanced build issues are left to chapter 3, Advanced Build Features.



# 2.2 Configuring the Projects tab of the Workspace window

The **Projects** tab shows the current workspace, projects and files. You can quickly open any project file or dependent file by double-clicking on its icon.



Right-clicking on a workspace opens a pop-up menu containing the available options.

Insert Project       -       Adds a project to workspace.         Load All Projects       -       Loads all projects to workspace.         Collapse       -       Collapses the tree below a workspace icon.	
Collapse - Collapses the tree below a workspace icon	
Clean All Projects   Deletes intermediate and output files from or projects in this workspace.	configurations in all
Version Control - Executes a version control system.	
Configure View - Configures the workspace view.	
Properties - Displays workspace properties.	

Right-clicking on a project opens a pop-up menu containing the available options.

Pop-up Menu Option Macro Recording		Macro Recording	Function	
Build	d Build 🔴		Builds out of date project files	
	Build All	•	Builds project files, regardless of whether the project files are out of date.	
	Clean Current Project	•	Deletes intermediate and output files from the current configuration in this project.	
	Update All Dependencies	-	Updates all dependencies.	
Set as	Current Project	•	Sets this project as the current project.	



Pop-up Menu Option	Macro Recording	Function
Remove Project	-	Removes a project from workspace.
Unload Project	-	Unloads a project
OR		OR
Load Project		Loads a project.
Add Files	-	Adds files to a project.
Remove Files	-	Removes files from project.
Add Folder	-	Add folder to a project.
Expand/Collapse	-	Expands or collapses the tree below a project icon.
Version Control	-	Executes a version control system.
Configure View	-	Configures the workspace view.
Properties	-	Displays project properties.

Right-clicking on a folder (other than **Download module** and **Dependencies**) opens a pop-up menu containing the available options.

Pop-up Menu Option	Macro Recording	Function
Add Folder	-	Add a custom folder.
Rename Folder	-	Rename a folder.
Remove Folder	-	Remove a folder.
Configure View	-	Configures the workspace view.

Right-clicking on a project file opens a pop-up menu containing the available options.

Pop-up Menu Option Macro Recording		Function	
Open <file name=""></file>	-	Opens a file in the High-performance Embedded Workshop editor.	
Open <file name=""> in external editor *</file>	-	Opens a file in the external editor.	
Build <file name=""></file>	•	Builds a file.	
Build Options	-	Sets build options.	
Add File	-	Adds files to a project.	
Remove File	-	Removes files from project.	
Exclude Build OR Include Build	-	Excludes a project file from build OR Includes a project file in build.	
Version Control	-	Executes a version control system.	
Configure View	-	Configures the workspace view.	
Show Differences	-	Compares files.	
Properties	-	Displays file properties.	

# Note:

\*. This option is available only when the Use external editor checkbox is selected in the Editor tab of the Options dialog box opened via [Setup -> Options].



Right-clicking on the **Download modules** folder opens a pop-up menu containing the available options (Available when the debugger is connected).

Pop-up Menu Option Macro Recording		Function	
Download all module		Downloads all modules in the Download modules list on the	
		Target tab of the Debug Settings dialog box (i.e. download	
		modules shown under the Download Modules folder in the	
		Projects tab of the workspace window).	
Download A New Module		Opens the Download Module dialog box, which allows you to add	
		download modules.	
Debug Settings	-	Opens the Debug Settings, which allows you to modify the debug	
		settings.	
Configure View	-	Configures the workspace view.	

Right-clicking on a download module opens a pop-up menu containing the available options (Available when the debugger is connected).

Pop-up Menu Option Macro Recording		Function	
Download	•	Downloads modules.	
Download (Debug Data Only)	•	Downloads modules (debug data only).	
Unload	•	Unloads modules.	
Download A New Module	•	Opens the Download Module dialog box, which allows you to add	
		download modules.	
Remove	-	Removes the selected modules.	
Debug Settings -		Opens the Debug Settings dialog box, which allows you to modify	
		the debug settings.	
Configure View	-	Configures the workspace view.	
Relocate Module *	-	Opens a dialog box which allows you to select one directory for	
		relocating all files retrieved from the download module.	
Properties	-	Shows the selected download module setting in the Download	
		Module dialog box. This download module setting can be modified.	
		The download module will be unloaded if it has already been	
		downloaded.	

#### Note:

\*. If you use a debug-only project (i.e., "Debugger only - xxxxxx"), this menu option will be displayed.

Right-clicking on a file retrieved from the download module opens a pop-up menu containing the available options (Available only when you use the debug-only project).

Pop-up Menu Option	Macro Recording	ng Function	
Open <file name=""></file>	-	Opens a file in the High-performance Embedded Workshop editor.	
Add File	-	Adds files to a project.	
Configure View	-	Configures the workspace view.	
Relocated file(s) - Opens a dialog box which allows you to select the s		Opens a dialog box which allows you to select the same short	
		filename retrieved from the download module at a new location.	
Properties	-	Displays file properties.	



Right-clicking on the Dependencies folder opens a pop-up menu containing the available options.

Pop-up Menu Option	Macro Recording	Function
Configure View	-	Configures the workspace view.

Right-clicking on a dependence file opens a pop-up menu containing the available options.

Pop-up Menu Option	Macro Recording	Function
Version Control	-	Executes a version control system.
Configure View	-	Configures the workspace view.
Show Differences	-	Compares files.
Properties	-	Displays file properties.

For details on "Current project", "Loaded project" and "Unloaded project", see section 2.12, Setting the current project.

# 2.3 Project files

In order for the High-performance Embedded Workshop to be able to build your application, you must first tell it which files should be in the project and how each file should be built (see the figure below).





# 2.3.1 Adding files to a project

Before building an application, select files to configure the application.

#### To add files to a project

- 1. Files can be added to the active project only. Select one of the following operations to open the **Add files to project 'project name'** dialog box:
  - Select [**Project -> Add Files**], **OR**
  - Right-click on the project within the **Projects** tab of the workspace window and select **Add Files** from the pop-up menu, **OR**
  - Press the "Insert" key when the **Projects** tab of the **Workspace** window is selected.
- 2. Select the file(s) you wish to add to the project.
- 3. A **Relative Path** checkbox is available at the bottom of this dialog box. Selecting this checkbox allows the files to be relative project files. These project files are relative to the workspace file and can also be placed outside the workspace structure. Even if you re-allocate the entire source tree, the High-performance Embedded Workshop is able to check the relative position of the files to find them. By default, this checkbox is selected.
- 4. A **Hide Project Files** checkbox is available at the bottom of this dialog box. Selecting this checkbox only shows the files not added to the current project. By default, this checkbox is not selected.
- 5. Click the **Add** button.

#### There are other ways to add files to a project

- Right-clicking on an open file in the **Editor** window displays a pop-up menu. If the file is already in the project then the **Add File To Project** menu option is disabled. Select the **Add File To Project** menu option to add the file to the current project.
- In the High-performance Embedded Workshop it is also possible to 'drag and drop' files from Windows<sup>®</sup> Explorer onto the Projects tab of the workspace window. For details, see section 2.3.2, Drag and drop of files and folders.

#### Notes:

- If you add a file that has an unrecognized file type to the project, then the file will be added to the project, but certain functions will be disabled for this file. When you double-click on a file with an unrecognized file type in the Workspace window, the open operation is passed to the Windows<sup>®</sup> operating system (instead of opening the file in the editor). The default 'Open' operation is then carried out as if the file was double-clicked in Windows<sup>®</sup> Explorer. To view the currently defined extensions, use the **File Extensions** dialog box. See section 2.5, File extensions and file groups, for further information.
- Hide Project Files is not selectable under the Windows Vista® or Windows® 7 operating system.



# 2.3.2 Drag and drop of files and folders

It is possible to 'drag and drop' files or folders from Windows® Explorer onto the **Projects** tab of the workspace window. These files or folders can only be dropped onto a project in the active project or a user folder in the active project. Note, however, that you cannot drag and drop files or folders onto multiple folders.

#### When you drag and drop files

The behavior depends on the destination.

- A file dropped onto the active project is added into the group folder of the file type. If a suitable group folder does not exist, the High-performance Embedded Workshop creates it. For example, if you drop the file "test.c" onto the active project, this file will be added to the group folder "C source file".
- A file dropped onto a user folder is directly added to this folder. Even if a file with the same name is already in the folder, the new file can be added to the folder only when these two files have different paths.
- If a file of an unrecognized file type is added to the project, this file is directly added to the project.

#### To drag and drop folders

Drag and drop of a folder from the Windows<sup>®</sup> Explorer opens the **Add Files From Folder** dialog box, which allows you to select the files in the folder that you wish to add to the project. This dialog box shows the names of all files included in the folder that was dropped. **Filename** and **File Type** show the full paths of files and file types, respectively.

Filename	File Type 🔸		
C:\Workspace\DemoSH4\DemoSH4\intprg.src	Assembly source file		Cancel
C:\Workspace\DemoSH4\DemoSH4\lowlvl.src	Assembly source file		
C:\Workspace\DemoSH4\DemoSH4\vecttbl	Assembly source file		Uncheck <u>A</u> ll
C:\Workspace\DemoSH4\DemoSH4\vhandl	Assembly source file		
C:\Workspace\DemoSH4\DemoSH4\dbsct.c	C source file		
C:\Workspace\DemoSH4\DemoSH4\DemoS	C source file		
C:\Workspace\DemoSH4\DemoSH4\lowsrc.c	C source file		
C:\Workspace\DemoSH4\DemoSH4\resetpr	C source file		
C:\Workspace\DemoSH4\DemoSH4\sbrk.c	C source file		
C:\Workspace\DemoSH4\DemoSH4\Debug	Library file		
C:\Workspace\DemoSH4\DemoSH4\Debug	Object file	<b>_</b>	
C:\Workspace\DemoSH4\DemoSH4\Debug	Library file		

1. In the dialog box, the files are initially sorted by the file type in alphabetical order. If you click a column header (of file names or types), the files will be sorted by the file name or type.

- 2. You can select multiple files by clicking. If you then click on one of the checkboxes for the selected files or press the Space key, checkboxes for all of the selected files will be switched ON or OFF (depending on the previous state).
- 3. If the **Show files in subfolders** checkbox is selected, the dialog box also shows the files within subfolders under the folder that was dropped. Otherwise the dialog box only shows the files within the folder that was dropped. By default, this checkbox is selected.
- 4. If the **Show source files only** checkbox is selected, the dialog box only shows the files having the extension specified for the source files in this project (e.g. C source files). Otherwise the dialog box shows all files. If the High-performance Embedded Workshop cannot recognize the file type, "Unknown type" will be shown in the **File Type** column. By default, this checkbox is selected.
- 5. If the Create folder structure in project checkbox is selected, files are added to the tree where the folder was dropped. For example, when a folder "data" containing some files is dropped onto the tree, the files are added into a user folder "data", which is newly created. If the Create folder structure in project checkbox is not selected, when you drop a folder onto a user folder, files in the folder are added into this user folder. When a folder is dropped onto the active project tree, files in the folder are added into the group folder of the file type (e.g. a file "main.c" is added into the group folder "C source file"). If a suitable group folder does not exist, the High-performance Embedded Workshop creates it. Files of an unknown type dropped onto the project are directly added to the project.

By default, this checkbox is not selected.

- 6. Clicking the **Uncheck All** button deselects all checkboxes, while clicking the **Check All** button selects all checkboxes.
- 7. Clicking the OK button adds all selected files, which have tick marks in their checkboxes, to the project. Next time this dialog box is opened, it shows the most recently used setting.

#### 2.3.3 Removing files from a project

#### There are three ways of removing files from a project

- Files can be individually removed from a project,
- A selection of files can be removed,
- All files can be removed.

#### To remove a file(s) from a project

- 1. Select one of the following operations to open the **Remove Project Files** dialog will be displayed:
  - Select [**Project -> Remove Files**], **OR**
  - Right-click on the project within the **Projects** tab of the workspace window and select **Remove Files** from the pop-up menu.
- 2. Select the file(s) that you want to remove from the **Project files** list.
- 3. Click the **Remove** button to remove the file(s), or click the **Remove All** button to remove all files from the list.

4. Click the OK button to remove the files from the project.

Project files:       OK         Idbsct.c       [C:\Workspace\DemoSH4\DemoSH4         DemoSH4.c       [C:\Workspace\DemoSH4\DemoSH4         intprg.src       [C:\Workspace\DemoSH4\DemoSH4         lowlvl.src       [C:\Workspace\DemoSH4\DemoSH4         lowsrc.c       [C:\Workspace\DemoSH4\DemoSH4         resetprg.c       [C:\Workspace\DemoSH4\DemoSH4         sbrk.c       [C:\Workspace\DemoSH4\DemoSH4         vecttbl.src       [C:\Workspace\DemoSH4\DemoSH4         vhandler.src       [C:\Workspace\DemoSH4\DemoSH4	ΥX
	cel
× F	

To remove selected files from a project using the Projects tab of the Workspace window

- 1. Select the files that you want to remove in the **Projects** tab of the **Workspace** window. Multiple files can be selected by holding down the SHIFT or CTRL key.
- 2. Press Delete.
- 3. A confirmation dialog box opens for you to select whether or not to delete the selected files from the project. To delete the selected files, select Yes. Otherwise select No.



If you do not wish to open this confirmation dialog box, select the **Don't ask this question again** checkbox. To open this dialog box again, select [**Setup -> Options**] to open the **Options** dialog box. Select the **Delete file from project** checkbox on the **Confirmation** tab. By default, this checkbox is selected.



# 2.3.4 Excluding a project file from build

A file in a project can be excluded from build on a configuration by configuration basis.

#### To exclude a project file from build

- 1. Click on the file that you want excluded from build, in the **Projects** tab of the **Workspace** window. It is also possible to select several files by using a mouse or inputs through the keyboard as follows:
  - **To select several files** Click the files while pressing the CTRL key.
  - Select several files as a range Click a file as the start of the range. Then keep pressing the SHIFT key and click another file as the end of the selected range.
- 2. Take either of the two ways listed below. A red cross (**\***) will appear on the file's icon, and the file will be excluded from build.
  - Select **Exclude Build <file>**, where <file> is the name of the selected file, from the pop-up menu opened by right-clicking. <file> is not displayed if several files are selected.
  - Select [Build -> Include/Exclude Build].

# 2.3.5 Including a project file in build

A file that has been excluded from build can be included again.

#### To include a project file in build

- 1. Click on a file that has previously been excluded from build, in the **Projects** tab of the **Workspace** window. It is also possible to select several files by using a mouse or inputs through the keyboard as follows:
  - **To select several files** Click the files while pressing the CTRL key.
  - Select several files as a range Click a file as the start of the range. Then keep pressing the SHIFT key and click another file as the end of the selected range.
- 2. Take either of the two ways listed below. The red cross will be removed from the file's icon (≚), and the file will be included in build.
  - Select **Include Build <file>**, where <file> is the name of the selected file, from the pop-up menu opened by clicking the right-mouse button. <file> is not displayed if several files are selected.
  - Select [Build -> Include/Exclude Build].



# 2.4 User folders in the workspace

In the High-performance Embedded Workshop it is possible to add folders to the Projects tab of the workspace window. This allows you to logically group your files into certain areas within a project. The folder can be set to any name and this is entered in a dialog box. The operations listed below are not allowed, however, for folders **Download modules** and **Dependencies** that are automatically displayed.

#### To add a user folder

- 1. Select the project on the **Projects** tab of the **Workspace** window.
- 2. Right-click and select Add Folder.
- 3. Enter the name in Folder of the **Add Folder** dialog box.
- 4. Click OK.
- 5. You can now drag and drop files into this folder to group them logically.

#### To add a sub-folder

- 1. Select the folder on the **Projects** tab of the **Workspace** window.
- 2. Right-click and select Add Folder.
- 3. Enter the name in Folder of the Add Folder dialog box
- 4. Click OK.
- 5. You can now drag and drop files into this folder to group them logically.

#### To drag and drop a folder

Drag a folder in the active project from the **Projects** tab of the **Workspace** window.

- If the folder is dropped onto the active project, the folder is directly added into the project.
- If the folder is dropped onto a user folder in the active project, the folder is added into the user folder.

You can also drag folders from the Windows® Explorer and drop it into the Projects tab of the workspace window. For details, see section 2.3.2, Drag and drop of files and folders.

#### To remove a user folder

- 1. Select the folder on the **Projects** tab of the **Workspace** window.
- 2. Select one of the following operations (the folder must be empty):



- Press the Delete key.
- Right-click and select **Remove Folder** from the pop-up menu.

#### To modify a user folder name

- 1. Select the folder on the **Projects** tab of the **Workspace** window.
- 2. Right-click and select Rename Folder from the pop-up menu.
- 3. Enter the new name in Folder of the Rename Folder dialog box.
- 4. Click OK.

# 2.5 File extensions and file groups

The High-performance Embedded Workshop can identify files by their extension. The system defines certain extensions depending upon the tools that are being used. For example, if you are using a compiler then the .c extension will be in the 'C source file' group and will be used as input to the compiler phase. Additionally, the High-performance Embedded Workshop allows you to define your own extensions. For example, if the project you are developing uses assembler source files the default extension may be .src. If you would like to use a different extension instead of .src (e.g. .asm) then you can define a new extension and request that the High-performance Embedded Workshop treats it in the same way as a .src file.

File extensions and file groups can be viewed and modified via the **File Extensions** dialog box, which is invoked by selecting [**Project -> File Extensions**]. This dialog box displays all the extensions and file groups that are defined within the current workspace.

F	ile Extensions	:	? ×
	Extension *.abs *.inc *.lis *.a30 *.asm *.src *.bin *.bin *.lst	Group Absolute file Assembly include file Assembly list file Assembly source file Assembly source file Binary file C header file C list file	OK Cancel <u>A</u> dd <u>R</u> emove <u>O</u> pen with
	≣ <b>*</b> .c	C source file	
	🖹 *.hpp	C++ header file	
	≣]*lnn ∢	G++ list file	

The **File Extensions** list is divided into two columns. On the left are the file extensions, and on the right are the file groups associated with the extensions. Many file extensions can belong to the same group. For example, assembler source files may have several extensions in a single project (e.g. .src, .asm, .mar etc.).



# 2.5.1 Associating an application with a file group

In addition to opening a file with the editor, the **File Extensions** dialog box allows you to associate any application with any file group so that when you double-click on a file in the **Projects** tab of the **Workspace** window, the appropriate application is launched with the file.

Fi	le Extensions		? X
[	Extension	Group	ОК
	🖹 *.lpp	C++ list file	
Text editor	🗄 *.cpp	C++ source file	Cancel
	🗋 *.cpu	CPU information file	
Word processor	w *.doc	Documents	<u>A</u> dd
	🖹 *.exp	Expanded assembly source file	Remove
	🖹 *.hex	Hex file	
Librarian	😪 *.lib	Library file	Open with
	🗐 *.lbp	Library information file	
	🖹 *.map	Linkage map file	
	🖹 *.fsy	Linkage symbol file	
No application	—]*.obj	Object file	1
	•		
l	•	Þ	

# To associate an application with a file group

- 1. Select [Project -> File Extensions]. The File Extensions dialog box opens.
- 2. Select the file group that you want to associate from the file extensions list.
- 3. Click the **Open with** button. The **Modify File Extension** dialog box opens.



Modify File Extension		? ×
<u>File group</u> : Linkage map file		ОК
Associated application: Open "Linkage map file" with: Editor		Cancel
	<u>A</u> dd <u>M</u> odify <u>R</u> emove	

4. Select **None** to remove any association, **Editor** to open this type of file in the internal/external editor, or select **Other** to open this type of file with a specific application. If you select **Other**, you can either specify a new application, or select any previously defined application from the drop-down list. Click the **Add** button to define a new application. The **Add Application** dialog box opens.

Add Application	? ×
<u>N</u> ame:	ОК
	Cancel
<u>C</u> ommand (excluding parameters):	
	<u>B</u> rowse
<u>P</u> arameters:	
Initial directory:	
	B <u>r</u> owse

Enter the name of the tool into the **Name** field. Enter the full path to the tool in the **Command** field (do not include any parameters). Enter the parameters that are required to open a file into the **Parameters** field. Be sure to use the \$(FULLFILE) placeholder to specify the location of the file (see Reference 5, Placeholders, for more information on placeholders). Enter the initial directory in which you would like the application to run into the **Initial directory** field. Click the **OK** button to finish creating the application. Click the **Modify** button to modify an application. The **Modify Application** dialog box will be displayed. This dialog is the same as the **Add Application** dialog box described above except that the **Name** field is read-only. Modify the settings as desired and then click the OK button.

5. Click the OK button to set the application for the selected file group.



#### 2.5.2 Creating a new file extension and file group

If you want to manage files that are not, by default, recognized by the High-performance Embedded Workshop (e.g. documents) then you need to create a new extension and a new file group.

#### To create a new file extension in a new file group

- 1. Select [**Project -> File Extensions**]. The **File Extensions** dialog box opens.
- 2. Click the Add button. The Add File Extension dialog box opens.

Add File Extension	? ×
Eile extension:  *. txt  File group:  C Extension belongs to an existing group:  Absolute file  Extension belongs to a <u>n</u> ew group:  Text	Cancel
Associated application: Open "Text" with: Windows default Add Add Modify Remove	

- 3. Enter the extension that you want to define into the **File extension** field. Use only alphanumeric and an underscore as characters of a file extension string. The drop-down list contains all extensions that are undefined in the current project. Selecting one of these extensions will add the text to the file extension field automatically.
- 4. Select the **Extension belongs to new group** option and enter a description that defines this new file group.
- 5. At this stage it is possible to change the associated application. There are four available choices in the **Open** "**<extension group>**" with drop-down list:
  - Editor
  - None
  - Other
  - Windows default

If **Editor** is selected, the **Open File** function in the workspace window causes the file to be opened in the Highperformance Embedded Workshop editor. If **None** is selected then the **Open File** operation is disabled when it is attempted. Selecting **Other** allows you to configure another tool for the **Open File** operation. See section 2.5.1,



Associating an application with a file group, for more details. If **Windows default** is selected then the **Open File** function in the **Workspace** window passes the **Open File** operation to the Windows<sup>®</sup> operating system. This then selects the default behavior for this file extension as defined in Windows<sup>®</sup> Explorer.

6. Click the OK button to add the extension to the **File Extensions** list.

#### 2.5.3 Creating a new file extension

If your files use a different extension from those accepted by the High-performance Embedded Workshop for a given phase (e.g. your assembler source files are .asm but the High-performance Embedded Workshop only recognizes .src), then you need to create a new extension and add it to an existing file group. This process is described below.

#### To create a new file extension in an existing file group

- 1. Select [**Project -> File Extensions**]. The **File Extensions** dialog box opens.
- 2. Click the Add button. The Add File Extension dialog box opens.

Add File Extension	? ×
File extension: *. asm File group: • Extension belongs to an existing Assembly source file • Extension belongs to a <u>n</u> ew group	<b>•</b>
	<u>A</u> dd Modify Remove

- 3. Enter the extension that you want to define into the **File extension** field. Use only alphanumeric and an underscore as characters of a file extension string. The drop-down list contains all extensions that are undefined in the current project. Selecting one of these extensions will add the text to the file extension field automatically.
- 4. Select the **Extension belongs to an existing group** option and select the group to which you would like to add this new extension.
- 5. Click the OK button to add the extension to the File Extensions list.

# 2.6 Setting build options

Once you have added the necessary files to the project, the next step is to instruct the High-performance Embedded Workshop on how to build each file. To do this, you will need to select a menu option from the **Build** menu. The contents of this menu depend upon which tools you are using.

#### To set options for a build phase

- 1. Select the **Build** menu and select the phase whose options you would like to modify.
- 2. A dialog box will be displayed allowing you to specify the options.
- 3. After making your selections, click the OK button to set them.

To obtain further information, use the context-sensitive help button or select the area in which you need assistance and press F1.

# 2.7 Build configurations

The High-performance Embedded Workshop allows you to store all of your build options into a build configuration, i.e. you can 'freeze' all of the options and give them a name. Later on, if you select that configuration, all options for all of the build phases will be restored. These configurations also allow the user to specify debugger settings for a build configuration. This means that each configuration can be targeted at a different end platform.

The figure below shows three configurations: **Default**, **MyDebug** and **MyOptimized**. In the first configuration, **Default**, each phase (compile and assemble) is set to its standard settings. In the second configuration, **MyDebug**, each file is being built with debug information switched on. In the third configuration, **MyOptimized**, each file is being built with optimization on full and without any debug information. The developer of this project can select any of those configurations and build them without having to return to the options dialogs to set them again.



#### 2.7.1 Selecting a build configuration

#### To select the current configuration

- 1. Select [Build -> Build Configurations]. The Build Configurations dialog box opens.
- 2. Select the build configuration that you want to use from the Current Configuration drop-down list.
- 3. Click the OK button.

Debug	-
Debug	
Release	
SimDebug_SH-4	

You can also select a different build configuration by selecting it from the Current Configuration drop-down list on the **Standard** toolbar.

#### 2.7.2 Adding a new build configuration

#### To add a new build configuration

- 1. Select [Build -> Build Configurations]. The Build Configurations dialog box opens.
- 2. Click the Add button. The Add Configuration dialog box opens.
- 3. Enter the new build configuration name into the **Configuration name** field. As you enter the new build configuration name, the directory underneath changes to reflect the configuration directory that will be used.

Configuration name:	Config1
Based on configuration:	Debug

- 4. Select one of the existing build configurations, on which you want to base the new build configuration, from the **Based on configuration** drop-down list.
- 5. Click the **OK** buttons on both dialogs to complete the creation of the new build configuration.

#### 2.7.3 Removing a build configuration

#### To remove a build configuration

- 1. Select [Build -> Build Configurations]. The Build Configurations dialog box opens.
- 2. Select the build configuration to remove and click the **Remove** button.
- 3. Click the OK button to close the **Build Configurations** dialog box.


# 2.8 Building a project

# 2.8.1 Using Parallel Build

Parallel build allows the High-performance Embedded Workshop to use the multiple processors on your host PC for building your project. This will result in faster builds inside the High-performance Embedded Workshop.

### Why use parallel build?

Parallel build will launch multi-phase build processes (assembler, compiler) from your selected toolchain in parallel for the number of logical cores inside your PC.

For example, if you have 2 logical cores in your PC, the High-performance Embedded Workshop will launch 2 compilations in parallel.

The High-performance Embedded Workshop does not parallelize:

- Single shot phases like the linker.
- Multi-shot or single-shot custom phases.
- High-performance Embedded Workshop generated make files.

#### Parallel build option

Parallel build is enabled automatically when a workspace is created or opened that is using a toolchain supporting is parallel build.

It is possible to disable this option for your project and once this option disabled it will be saved to the project file when the workspace is saved.

#### Notes:

- When opening a workspace created with a previous version of High-performance Embedded Workshop with a toolchain that supports parallel build it is automatically switched on.
- If a toolchain is upgraded to a new version that supports parallel build, the parallel build option is automatically switched on.
- When using a workspace with multiple projects the option on the [Setup -> Options], Build tab reflects that set for the "Current project".
- When building multiple dependent projects the individual project setting is used for each project. Not that of the "Current project".

# To setup parallel build

- 1. Ensure a valid workspace is open.
- 2. Select [Setup -> Options]. The Options dialog box opens.
- 3. Select the **Build** tab.



- 4. Check the "Use parallel build for toolchain (internal build only)" option.
- 5. Click the **OK** button.

#### Notes:

- This feature is not supported in all toolchains.
- If your workspace is open and the option is disabled your current toolchain does not support parallel build.

# 2.8.2 Building individual files

The High-performance Embedded Workshop lets you build project files individually.

### To build an individual file

- 1. Select the file to build from the **Projects** tab of the **Workspace** window.
- 2. Select one of the following operations:
  - Click the **Build File** toolbar button (1986), **OR**
  - Select **Build <file>** from the pop-up menu, **OR**
  - Select the [Build -> Build File] menu option, OR
  - Press CTRL+F7.

All output is redirected to the **Build** tab of the Output window.

# 2.8.3 Building a project

The **Build** option only compiles or assembles those files that have changed since the last build. Additionally, it will rebuild source files if they depend upon a file that has changed since the last build. For instance, if the file 'TEST.C' #include's the file 'HEADER.H' and the latter has changed since the last build, the file 'TEST.C' will be recompiled.

#### To perform a build operation

Select one of the following operations:

- Click the **Build** toolbar button (**M**), **OR**
- Press F7, **OR**
- Select [Build -> Build], OR



• Right-click on a project in the **Projects** tab of the **Workspace** window and select [**Build -> Build**] from the pop-up menu.

The **Build All** option compiles and assembles all source files, irrespective of whether they have been modified or not, and links all of the new object files produced.

# To perform a build all operation

Select one of the following operations:

- Click the **Build All** toolbar button (**bin**), **OR**
- Select [Build -> Build All], OR
- Right-click on a project in the **Projects** tab of the **Workspace** window and select [**Build -> Build All**] from the pop-up menu.

All output from a build or build all operation is redirected to the **Build** tab of the Output window. Both the Build and the Build All operations will terminate if any project files produce errors.

# 2.8.4 Building multiple projects

The High-performance Embedded Workshop allows you to build multiple projects and configurations at once.

# To build multiple projects

- 1. Select [Build -> Build Multiple]. The Build Multiple dialog box opens.
- 2. The **Build Multiple** dialog box gives you the choice of which projects and configurations to build. Select the check boxes next to the projects and configurations that you want to build. For example, in figure below if you wanted to build the entire "hewtest2" project you would check the "hewtest2-Debug" and the "hewtest2-Release" selections and leave all other check boxes unchecked.



High-performance Embedded Workshop V.4.09

Project configurations:       Build         hewtest1-Debug       Build All         hewtest2-Debug       Clean         hewtest3-Debug       Check All         hewtest4-Debug       Uncheck All         hewtest4-Release       Cancel

- 3. When you are happy with your chosen selection, click the **Build** button and the High-performance Embedded Workshop will build the selected projects and configurations.
- 4. If you want to build all of the projects, click the **Build All** button. This will automatically select all projects and configurations, and build them all.
- 5. If you also wish to delete intermediate and output files of the selected project or configuration, click the **Clean** button. This does not execute the build.
- 6. Results from the build are displayed in the Build tab of the Output window in the same way as in a normal build process.
- 7. If you use the dialog again it will restore the checked status from the last time you used it.

# 2.8.5 Stopping tool execution

The High-performance Embedded Workshop allows you to halt the build process once it is under way.

#### To stop tool execution

Select one of the following operations:

- Click the Stop Tool Execution toolbar button ( ), OR
- Select [Build -> Stop Tool Execution].

The 'Build Stopped by User' message appears in the Build tab of the Output window.

#### Note:

Do NOT assume that any output from the tool you terminated is valid. It is recommended that you delete any output files produced and ensure that the phase is executed again.



#### 2.8.6 Deleting intermediate and output files produced in building

You can delete intermediate and output files (produced in building) included in the current configuration in the project.

#### To delete intermediate and output files included in the current configuration in the project

Select one of the following operations:

- Select [Build -> Clean Current Project], OR
- Right-click on a project in the Projects tab of the workspace window. Select [Build -> Clean Current Project] from the pop-up menu.

A confirmation dialog opens for you to select whether or not to delete all intermediate and output files produced in building. To delete all files, select Yes. Otherwise select No or Cancel.

Confirmation Request			? ×			
You are about to delete all intermediate files from project 'DemoSH4'. Do you wish to continue?						
Don't ask this question again						
Yes	<u>N</u> o	Cancel				

If you do not wish to open this confirmation dialog box, select the **Don't ask this question again** checkbox.

#### To open the confirmation dialog box again

- 1. Select [Setup -> Options]. The Options dialog box opens.
- 2. Select the **Confirmation** tab.
- 3. Select the **Clean project** checkbox. This checkbox is selected by default.
- 4. Click OK.

#### To delete intermediate and output files in all projects of the workspace

Select one of the following operations:

- Click the Clean All Projects toolbar button (
  .), OR
- Select [Build -> Clean All Projects], OR
- Right-click on a workspace in the **Projects** tab of the workspace window. Select [**Clean All Projects**] from the pop-up menu.



A confirmation dialog opens for you to select whether or not to delete all intermediate and output files produced in building. To delete all files, select Yes. Otherwise select No.

Confirmation Request	? ×					
You are about to delete all intermediate files from all projects in the workspace. Do you wish to continue?						
Don't ask this question again						
<u>Y</u> es <u>N</u> o	Cancel					

If you do not wish to open this confirmation dialog box, select the Don't ask this question again checkbox.

#### To open the confirmation dialog box again

- 1. Select [Setup -> Options]. The Options dialog box opens.
- 2. Select the **Confirmation** tab.
- 3. Select the **Clean workspace** checkbox. This checkbox is selected by default.
- 4. Click OK.

The **Build** tab in the Output window shows all projects and configurations in which intermediate and output files have been deleted.

You can select multiple configurations and delete intermediate and output files produced in building. For details, see section 2.8.4, Building multiple projects.

#### Note:

The types of intermediate and output files to be deleted vary with the package and version number.

- C Compiler Package for M16C Series and R8C Family V.5.45 Release 00 or later
- C Compiler Package for M32C Series V.5.42 Release 00 or later
- Compiler Package for R32C Series V.1.02 Release 00 or later

Relocatable, absolute, linkage list (".MAP"), and assembly-program list files (".LIS") are deleted.

Subcommand files of toolchains are not deleted.

The extension for absolute files varies with the specified format.

- C Compiler Package for M16C Series and R8C Family V.5.30 Release 0 to V.5.44 Release 00
- C Compiler Package for M32C Series V.5.40 Release 0 to V.5.41 Release 01A
- C Compiler Package for R32C Series V.1.01 Release 00



- C/C++ Compiler Package for SuperH Family V.5.1 to V.9.03 Release 02
- C/C++ Compiler Package for H8SX, H8S, H8 Family V.3.0A to V.7.00 Release 00
- C/C++ Compiler Package for RX Family V.1.00 Release 00
- C Compiler Package for M32R Family V.5.00 Release 00 to V.5.01 Release 01
- C Compiler Package for 740 Family V.1.00 Release 1 to V.1.01 Release 02
- Assembler Package for 740 Family V.4.10 Release 02

Only relocatable and absolute files are deleted. Linkage list (".MAP") and assembly-program list (".LIS") files, and subcommand files of toolchains are not deleted.

The extension for absolute files varies with the specified format.

# 2.8.7 Configuring the Build tab of the Output window

The **Build** tab shows the output from any build process (e.g. compiler, assembler and so on). If an error is encountered in a source file, an icon and the error message will be displayed in the **Build** tab, along with the source file name and line number.

If you click on a toolbar button or pop-up menu option associated with display of error messages, the line of the error message will be highlighted (with text in white and its background in dark blue) and the editor will show the source code of that line (if the line has any source code).

Double-clicking the line of an error message will also show the source code in the editor.

When you attempt an operation to display an error or warning, the status bar shows this error or warning message.

You can also customize the **Build** tab so that the texts in the lines of error messages will be highlighted in a color different from that of the texts in other lines. The figure below shows the error messages in red and warnings in green as an example.



Right-clicking displays a pop-up menu containing available options.

A basic operation is allocated to the toolbar.

The functions of **Toolbar display** and **Customize toolbar** are also included in the pop-up menu displayed by rightclicking the toolbar area.

Pop-up Menu Option	<b>Toolbar Button</b>	Function
-	01	Highlights the next occurrence of error and shows the source
		code in the editor.
-	<b>O</b> T	Highlights the previous occurrence of error and shows the source
		code in the editor.
-	∆↓	Highlights the next occurrence of warning and shows the source
		code in the editor.
-	<b>∆</b> ↑	Highlights the previous occurrence of warning and shows the
		source code in the editor.
Display next Error/Warning/Info	<mark>%</mark> ↓	Displays the editor that generated the next build error or warning.
Display previous Error/Warning/Info	<u>₿</u> ↑	Displays the editor that generated the previous build error or
		warning.
Help	2	Shows the help information about the line.
Go to Error/Warning/Info	-	Goes to the associated source line.
Clear Window	0	Clears the contents of the window.
Save		Saves the contents of the window into a text file.
Сору		Copies the selected contents onto the Windows® clipboard.
Toolbar display	-	Shows or hides the toolbar.
Customize toolbar	-	Customizes toolbar buttons.

The **Build** tab also shows an icon corresponding to the error message output by build execution.

Icon Name	lcon	Error Message Level	
Build Error	•	Error	
Build Warning		Warning	
Information	0	Information	

You can also customize the **Build** tab of the Output window so that the texts in the lines of error messages will be highlighted in a color different from that of the texts in other lines.

### To customize the current colors

- 1. Select [Setup -> Format Views] to open the Format Views dialog box.
- 2. Select the **Output** item from the tree in the left-hand section of the dialog box and expand it.
- 3. Select the category for which you wish to change the color from the table below.

Category	Foreground Color of the Color Tab (Default)	Background Color of the Color Tab (Default)	( )	Type of Output to be Applied
Text	SYSTEM	SYSTEM	All tabs	All
Build Error Text	Black	White	Build	Error



### High-performance Embedded Workshop V.4.09

Category	Foreground Color of the Color Tab (Default)	Background Color of the Color Tab (Default)	.,	Type of Output to be Applied
Build Warning Text	Black	White	Build	Warning
Information Text	Black	White	Build	Information

- 4. Change the selection in the **Foreground** and **Background** lists of the **Color** tab.
- 5. Click the OK button.

# 2.8.8 Controlling the content of the Build tab of the Output window

It is often useful to display extra information (such as the command line options that are being applied to a file) during a build. The High-performance Embedded Workshop allows you to specify whether or not you want such options displayed in the **Build** tab of the output window during a Build, Build All or Build File operation via the **Options** dialog box.

#### To view or hide extra information during a build

- 1. Select [Setup -> Options]. The Options dialog box opens.
- 2. Select the **Build** tab.
- 3. Set the three check boxes in the **Show** group as follows:
  - **Command line** controls whether the command line is shown as each tool is executed.
  - Environment controls whether the environment is shown as each tool is executed.
  - **Initial directory** controls whether the current directory is shown as each tool is executed.

Show:	
☑ Command line	🔽 Initial directory
✓ Environment	Mark out of date files

# 2.8.9 Displaying out of date files in the Workspace window

Files updated later than the file generated by the previous build (i.e. out of date files) are marked in the workspace window. In the window below the file "demo.c" is out of date.

When you click **Build** next time these files will be re-built. This is also displayed for dependent projects of the current project.

The view of these files is updated whenever something that affects the build occurs, e.g. options changing, file addition, dependencies changing, files modified, etc.





# To display out of date files in the workspace window

- 1. Select [Setup -> Options]. The Options dialog box opens.
- 2. Select the **Build** tab.
- 3. Check the **Mark out of date files** check box.
- 4. Click OK.

-Show:	
☐ <u>C</u> ommand line ☐ En <u>v</u> ironment	☐ Initial directory ☑ Mark out of date files

# 2.9 File dependencies

A typical project will contain dependencies between files. For example, a C file may '#include' one or more header files. In complex projects, source files will include (or depend upon) others and this can quickly become difficult to manage. However, the High-performance Embedded Workshop provides a dependency scanning mechanism whereby all files in a project are checked for dependencies. Once complete, the **Projects** tab of the Workspace window will display an up-to-date list with all the project file dependencies.

Initially, the dependencies for all files are contained within the **Dependencies** folder (although this can be modified by configuring the projects tab).



High-performance Embedded Workshop V.4.09



#### Note:

Some include files may include other files. Up to 31 nesting levels of dependent files are displayed in the Dependencies folder.

File dependencies in a project are automatically updated at the following operation timings:

Operation Timing	Specific Operation
Open workspace	Select option in the Welcome! dialog box
	Select [File -> New Workspace]
	Select [File -> Open Workspace]
Add a project to workspace	Select [Project -> Insert Project]
Set the current project	Select [Project -> Set Current Project]
Add files to a project	Select [File -> Add Files]
Save files	Select [File -> Save]
Build	Select [Build -> Build]
Automatic dependency scanning	Deselect the Disable automatic dependency scanning checkbox

#### To prevent automatic scanning of file dependencies in a project

- 1. Select [Setup -> Options]. The Options dialog box opens.
- 2. Select the **Build** tab.
- 3. Select the **Disable automatic dependency scanning** checkbox. This checkbox is not selected by default.
- 4. Click OK.



#### Note:

Selecting this option clears all file dependencies in a project. To manually update file dependencies in a project, see "To manually update file dependencies in a project".



When [**Build -> Build**] is selected, a confirmation dialog box appears to ask if you wish to update the file dependencies before executing the build.

If you have not manually updated dependency data, it may be out of date.

When you want dependency data to be updated prior to build, select Yes.

If you select No, it executes "Build" based on current dependency scanning data.



If you do not wish to open this confirmation dialog box, select the Don't ask this question again checkbox.

#### To open the confirmation dialog box again

- 1. Select [Setup -> Options]. The Options dialog box opens.
- 2. Select the **Confirmation** tab.



- 3. Select the Scan Dependencies Prior To Build checkbox. This checkbox is selected by default.
- 4. Click OK.

#### To manually update file dependencies in a project

Select one of the following operations:

- Select [Build -> Update All Dependencies], OR
- Right-click on a project in the Projects tab of the Workspace window and select [Build -> Update All Dependencies] from the pop-up menu.

# 2.10 Configuring the Workspace window

If you right-click anywhere inside the Projects tab of the Workspace window, a pop-up menu will be invoked. Select the **Configure View** menu option to modify the way in which information is displayed. The following four sections detail the effect of each option on the **Configure View** dialog box.

#### Show Dependencies Under Each File

Show dependencies under each file

If you check the **Show dependencies under each file** checkbox, the dependent files are shown under the including source file as a flat structure, i.e. the files themselves become folders (as in figure (ii) below). If this option is not selected then a separate folder contains all dependencies (as in figure (i) below).

ω	& Workspace	×	"	🚸 Workspace	_ 🗆 🗡
	⊡… 🔄 Demo			⊡… 🤁 Demo	
	Demo	н.		🗄 🕞 Demo	
	📄 🔄 Assembly source file	н.		📄 🔄 Assembly source file	
	lowlvl.src	н.		in 🔛 lowlyl.src	
	📔 📥 🔄 C source file	н.		📄 🔄 C source file	
	dbsct.c	н.		dbsct.c	
	🚽 🔛 Demo.c	н.		🚽 🔤 Demo.c	
	intprg.c	н.		intprg.c	
	l 🔄 🔛 lowsrc.c	н.		vect.h	
	resetprg.c	н.		📙 🖃 📴 lowsrc.c	
	sbrk.c	н.		l lowsrc.h	
	vecttbl.c	н.		📋 🖳 🖆 resetprg.c	
	📔 🖻 🔄 Dependencies	н.		📕 🔚 stacksct.h	
	l lowsrc.h	н.		📋 🖆 🖆 🖆	
	sbrk.h	н.		sbrk.h	
	🚽 📰 stacksct.h	н.		📋 🖃 🖄 vecttbl.c	
	vect.h	н.		📔 👘 vect.h	
	Projects		1	Projects	



#### Show Standard Library Includes

Show standard library includes

By default, any dependent files found in standard include paths will not be shown (as in figure (j) below). For example, in C code, if you write a '#include' statement, such as '#include <stdio.h>', stdio.h will not be listed as a dependent file. To view such system include files, select the **Show standard library includes** checkbox (as in figure (jj) below).



#### Show File Paths

Show file paths

By default, any files will not be shown with their full path (as in figure (k) below). If the **Show file paths** checkbox is selected, all of the files in the Projects tab of the Workspace window are shown with their full path, i.e. from a drive letter (as in figure (kk) below).

(k)	_(kk)	
⊟⊶ 🔄 C source file ↓ ↓ ↓ ↓ dbsct.c	⊡🤤 C source file   🖹 C:\Workspace\Demo\Demo\dbsct.	с

#### Sorting the workspace window into time stamp order

Sort files in time order

When **Sort files in time order** option is selected, the workspace window is then sorted into file time stamp order. The newest files are located at the top of list with the older files towards the bottom.

If files are updated after selecting this option, manually update the order of these files.



#### To manually update the file order

Select **Refresh Order** in the pop-up menu on the Projects tab of the Workspace window.

# 2.11 Inserting a project into the workspace

When a workspace is created, it contains only one project but, after it is created, you can insert new or existing projects into the workspace.

#### To insert a new project into the workspace

New project

- 1. Select [**Project -> Insert Project**]. The **Insert Project** dialog box opens.
- 2. Select the **New project** radio button.
- 3. Click the OK button. The Insert New Project dialog box opens.
- 4. Enter the name of the new workspace into the **Project Name** field. This can be up to 32 characters in length and contain letters, numbers and the underscore character. Especially, do not use a minus sign, or a space. As you enter the project name the High-performance Embedded Workshop will add a sub-directory for you automatically. This can be deleted if desired.
- 5. Use the **Browse** button to graphically select the directory in which you would like to create the project. Alternatively, you can type the directory into the **Directory** field manually.
- 6. The project type list displays all of the available project types (e.g. Application, Library etc.). Select the type of project that you want to create from this list.
- 7. Click the OK button to create the project and insert it into the workspace.

#### To insert an existing project into a workspace

<u>B</u> rowse

- 1. Select [**Project -> Insert Project**]. The **Insert Project** dialog box opens.
- 2. Select the **Existing project** radio button.
- 3. Click the Browse button to search for the project database file (".HWP" file).
- 4. Click the OK button to insert the existing project into the workspace.



# 2.12 Setting the current project

A project can be in three states - the **Current** project, a **Loaded** project or an **Unloaded** project.

Since a workspace can contain many projects, only one of them can be the **Current** project at any time. This project is the one that build actions and debug operations can be performed on (e.g. clicking the Build toolbar button will build the **Current** project).

### To set a project as the current project

Select one of the following operations:

- Select the project that you want to make active from the [Project -> Set Current Project] sub-menu, OR
- Select the project from the Projects tab of the workspace window. Right-click to display the pop-up menu and select the **Set as Current Project** option.

If the project is **Loaded**, it is possible to open the project's directory and view the files. It is also possible to change the builder or debugger options for the project. A **Loaded** project can also have tool executions performed on it from the **Tools** menu.

### To unload a project from the workspace

- 1. Select the **Loaded** project from the Projects tab of the workspace window.
- 2. Right-click to display the pop-up menu and select the **Unload Project** option. It is possible to select multiple projects in the workspace window to perform this operation.

If the project is **Unloaded**, its icon appears 'grayed' in the Projects tab of the workspace window and no actions can be performed upon it.

Unloaded projects can be loaded by the following operations.

#### To load all projects in the workspace

- 1. Select a workspace in the Projects tab of the workspace window.
- 2. Right-click to display the pop-up menu and select the Load All Projects option.

#### To load a project in the workspace

- 1. Select the **Unloaded** project from the Projects tab of the workspace window.
- 2. Right-click to display the pop-up menu and select the **Load Project** option. It is possible to select multiple projects in the workspace window to perform this operation.



# 2.13 Specifying dependencies between projects

The projects within a workspace can be dependent upon one another so that when one project is built, all its dependent projects are built first. This is useful if one project uses another in the workspace. For example, imagine that a workspace contains two projects. The first project is a library, which is included by an application project. In this case the library must have been built and up-to-date before the second application can build correctly. To achieve this situation we can specify the library as a dependent (i.e. child) project of the application project. This would then allow the library to be built first if it is out-of-date.

When a dependent project is built, the High-performance Embedded Workshop attempts to match the configuration in the dependent project with that of the current project. This means that if the current configuration is 'Debug' then the High-performance Embedded Workshop will attempt to build the 'Debug' configuration in the dependent project. If this matched configuration does not exist then the High-performance Embedded Workshop will use the configuration that was last used in the dependent project.

#### To make projects depend upon another

- 1. Select [Project -> Dependent Projects]. The Dependent Projects dialog box opens.
- 2. Select the project to which you would like to add dependents. When you do this, the **Dependent projects** list will display all of the projects in the workspace (excluding the selected project).

Dependent projects:
✓ Jepend01
☐depend02

- 3. The **Dependent projects** list has a checkbox for each project listed. Set the associated checkboxes to make those projects depend upon the selected project.
- 4. Click on the OK button.

When there is a project dependent on the current project, the High-performance Embedded Workshop checks whether building of the dependent project should be executed earlier than that of the current project.

When the current project is a parent, the dependency checking will be performed on its child project.

Even if another project is dependent on the child project, this dependent project will not be checked.

# 2.14 Removing a project from the workspace

# To remove a project from a workspace

- 1. Select the project that you wish to remove in the Projects tab of the workspace window.
- 2. Right-click on the selected project to invoke a pop-up menu.
- 3. Select **Remove Project**.



### Note:

You cannot remove the Current project from the workspace.

# 2.15 Relative projects paths in the workspace

In the High-performance Embedded Workshop when you add a project you can choose to add the project to the workspace using a relative path. This allows you to position a file above the workspace directory and it will still be relocated correctly if you relocate the High-performance Embedded Workshop workspace. The project is always relative to the workspace so if the project is one directory above the workspace before it is moved the High-performance Embedded Workshop will try to find the project in the same relative location after the relocation procedure. This is especially useful if you are using a project shared between more than one workspace.

In older versions of High-performance Embedded Workshop this project would not have been relocated and would have still tried to access the original file path. The older version of High-performance Embedded Workshop could only relocate the projects, which were in a sub-directory of the workspace directory. This is still the standard behavior for the High-performance Embedded Workshop.

### To add the project to the workspace using a relative path

- 1. Select the project in the workspace window.
- 2. Right-click on the selected project to invoke a pop-up menu.
- 3. Select Properties.
- 4. Click the **Project file specified as a relative path (in the .HWS file)** checkbox to switch the relative file path feature.
- 5. Click OK.

Project file specified as a relative path (in the .HWS file)



# 3. Advanced Build Features

This chapter explains the advanced build concepts.

# 3.1 The build process revisited

"Chapter 2, Build Basics" began by describing the build process in terms of a compiler, an assembler and a linker. This will be the case for most installations of the High-performance Embedded Workshop. However, if you want to begin changing the build process (e.g. adding and removing phases) then it is important to understand more about the way in which a build functions.

# 3.1.1 What is a build?

Building a project means applying a set of tools upon certain input files in order to produce the desired output. Thus, we apply a compiler upon C/C++ source files in order to create object files, we apply an assembler upon assembler source files in order to create object files and so forth. At each step or 'phase' of the build, we apply a different tool upon a different set of input files. The figure below presents a different view of the build process.



The High-performance Embedded Workshop provides the ability to change this build process via its **Build Phases** dialog box, which can be accessed by selecting [**Build -> Build Phases**]. On the left-hand side are the phases that are defined in the current project.



Build Order Build File Order File Mappings	
Build phase order:	
SH C/C++ Library Generator	<u>A</u> dd
✓SH C/C++ Compiler ✓SH Assembler	<u>M</u> odify
<b>✓</b> OptLinker	<u>R</u> emove
	Move <u>U</u> p
	Move <u>D</u> own
	Import

The remainder of this chapter details the various functions that the Build Phases dialog box provides.

# 3.2 Creating a custom build phase

If you want to execute another tool before, during or after a standard build process then this can be achieved by creating your own (i.e. custom) build phase.

### To create a new custom build phase

- 1. Select [Build -> Build Phases] to invoke the Build Phases dialog box.
- 2. Click the Add button. This will invoke the New Build Phase wizard dialog box.
- 3. Follow the 4 steps below. To move forward and backward between steps click the **Next** > and < **Previous** buttons respectively.

# Step 1

The first step asks whether you want to create an entirely new phase or whether you want to add a system phase. A system phase is a 'ready-made' phase which is already defined within the toolchain you are using (e.g. compiler, assembler, linker, librarian, etc.) or a utility phase (e.g. file copy, complexity analyzer etc.). The **Add an existing system phase** button is inactive if no more system phases are available. Select the **Create a new custom phase** button to create your own build phase.



New Build Phase - Step 1 of 4			
A Xnaa	What type of phase would you like to add ?		
<b></b>	C Add an existing system phase:		
PERFECT DOD	System Phase Version		
TTX	< <u>B</u> ack. <u>Next</u> > Can	cel	

# Step 2

The second step asks what type of phase you would like to create. There are two choices: multiple or single. When a multiple phase is executed, the command is applied to each file in the project of a certain file group. For example, if you set the input file group to be C source files then the command will be executed once for each C source file in the project. A single phase is executed once at most during a build.

New Build Phase - Step 2 of 4				
S Z	What type of phase would you like to create?			
	Multiple phase:			
The wo	The command is applied to each file in a file group. An example of this type of tool is a compiler or assembler.			
P	Select input file group:			
Ly	◯ <u>S</u> ingle phase:			
The The	The command is only ever executed once per build. An example of this type of tool is a linker.			
- Contrast	< <u>B</u> ack <u>N</u> ext > Car	ncel		

The input file group list contains the current file groups defined for the project. It is possible to define multiple input file groups by selecting the **Multiple Groups** entry in the input file group list.





Once this choice has been made the input file group selection is displayed as **Multiple Groups**. This dialog box allows the user to choose multiple input file groups for the custom phase being added to the project. To select a file group check the box next to the file group's name. One or more file groups can be selected in this dialog box.

# Step 3

The third step requests the fundamental information about the new build phase. Enter the name of the phase into the **Phase name** field. Enter the location of the program file into the **Command** field (do not insert any command line options as these options are specified via the **Options** menu of the High-performance Embedded Workshop menu bar). Specify the default options for the phase (i.e. what options you would like new files to take when added to the project) into the **Default Options** field. If you have a preferred directory in which you would like this program to run (i.e. where you want the current working directory to be set to before the tool is executed) then enter it into the **Initial directory** field.

New Build Phase - Step 3	of 4		? ×
S E	<u>P</u> hase name: MyPhase		
- T- MD	Command (excluding parameters): C:\Tools\MyTool.exe	▶	B <u>r</u> owse
P	Default options: \$(FULLFILE)	▶	
A A A A A A A A A A A A A A A A A A A	Initial directory: \$(CONFIGDIR)		Bro <u>w</u> se
AND	< <u>B</u> ack <u>N</u> ext >		Cancel



### Step 4

The fourth and final step allows you to specify any environment variables that the phase requires.

New Build Phase - Step 4 of 4		
	Does the command require any environment variables to be set? If so, enter them into the list below:	
	ENV=1024	
The Th	Read Output On Fly	
2000 Martin	< <u>B</u> ack <u>F</u> inish Cancel	

To add a new environment variable click the **Add** button (the **Environment Variable** dialog is displayed). Enter the variable name into the **Variable** field and the variable's value into the **Value** field and then click the OK button to add the new variable to the list. To modify an environment variable select the variable from the list and then click the **Modify** button. Make the required changes to the **Variable** and **Value** fields and then click the OK button to add the modified variable to the list. To remove environment variables select the variable that you want to remove from the list and then click the **Remove** button.

Environment Variable	? ×
Varia <u>b</u> le:	ОК
PATH	
Value:	Cancel

If the tool you are adding can display its output whilst the tool is running then use the **Read Output on Fly** option. This will display the tool output as each line of output happens. If this option is set to off then the High-performance Embedded Workshop will store all output that is being displayed by the tool, and display it in the **Output** window when the tool has finished its operation. This can be a problem when the tool is running an operation that might take many minutes, as it is difficult to see the progress of the current execution.



#### Note:

Using **Read Output on Fly** can cause problems when using certain tools on certain operating systems. If you are having problems with tools locking up or freezing in High-performance Embedded Workshop then un-check the **Read Output on Fly** option.

Click the **Finish** button to create the new phase. By default the new phase is added to the bottom of the **Build phase** order list in the **Build Order** tab of the **Build Phases** dialog box.

# 3.3 Ordering build phases

In a standard build (shown in figure below), you could add a phase at four different positions: before the compiler (1), before the assembler (2), before the linker (3) or after the linker (4).

You may place your own custom phases or move system phases to any position in the build order. It is important to remember that if the output of your custom phase can be input into another phase then the phase order must be correct if the build is to behave as intended.



Select [**Build -> Build Phases**]. The **Build Phases** dialog box will be displayed. The build phase dialog box provides facilities for ordering build phases via the **Build Phases** dialog box. It has two tabs, which are concerned with the ordering of phases: **Build Order** and **Build File Order**. And then you can click OK button.



# 3.3.1 Build Order tab

The **Build Order** tab displays the current order in which phases will be executed when the **Build** () or **Build All** () buttons is selected. The check box to the left of each phase indicates whether or not the phase is currently enabled. A phase can be toggled on/off by checking/unchecking its corresponding checkbox respectively.

	Build Order Build File Order File Mappings	
	Build phase order:	
Г	■SH C/C++ Library Generator	<u>A</u> dd
System 📕	SH C/C++ Compiler	Modify
phase 📜	— I SH Assembler	<u></u>
L	OptLinker	<u>R</u> emove
Custom —— phase	₩yPhase	Move <u>U</u> p
priase		Move <u>D</u> own
		Import

### To change the order of phases (system/custom) in a build or build all operation

- 1. Select the phase to be moved and then click the **Move Up** and **Move Down** buttons to move the phase up and down respectively.
- 2. Click the OK button to set the new ordering.

#### To view the properties of a system phase

- 1. Select the system phase that you would like to view.
- 2. Click the **Modify** button.
- 3. The **Command** tab shows general information about the phase. This may include copyright information, version information and so on.
- 4. Select the **Environment** tab to view the environment settings of the phase.
- 5. Click the OK button to close the dialog box.

#### To modify a custom phase

- 1. Select the custom phase that you would like to modify.
- 2. Click the **Modify** button. The Modify Phase dialog box will be invoked with the **Command** tab selected.



Command		
Command (excluding parameters):		
C:\Tools\MyTool.exe		Browse
Initial directory:		
\$(CONFIGDIR)		Bro <u>w</u> se
Input file group (for multi-shot phases only):		
Absolute file		
Don't check for input file(s) existence before	ore exec	uting
<u>Read output on-the-fly</u>		

- 3. Change the contents of **Command** and **Initial directory** as appropriate.
- 4. If there are multiple shot phases, select the file type to execute a phase from the **Input file group** (for multi-shot phases only) drop-down list.
- 5. Select the **Don't check for input file(s) existence before executing** checkbox if you don't want the Highperformance Embedded Workshop to abort the execution of the phase if any of the input files do not exist.
- 6. Select the **Read output on-the-fly** checkbox if you want to display build output as it happens, rather than showing the output at the end of each phase execution.
- 7. Select the **Environment** tab to edit the environment settings for the phase.

Environment	
Environment settings:	
ENV=1024 PATH=C:\Tools	<u>A</u> dd
SHOWALL=FALSE	<u>M</u> odify
	<u>R</u> emove
×	

- 8. Use the Add, Modify and Remove buttons to add, modify and remove environment variables respectively.
- 9. Select the **Output Syntax** tab and define the error, warning, and information for the custom phase.

Messages that include filenames can only be handled as syntax. Thus all syntax defined on the Output Syntax page should include filenames. Always use a placeholder to specify a filename.





When the lines showing error messages in the Build tab of the output window match the definition of **Output Syntax**, you can find icons on the left to the error messages and view the source code in the editor.

If the Renesas SuperH C/C++ compiler is used, for example, the following error message may be output to the Build tab of the output window.

C:\Workspace\SH\SH\resetprg.c(70) : C2225 (E) Undeclared name "a"

To define this error, type the following syntax.

\$(FULLFILE)(\$(LINE))\*(E)\*

The correspondence between an error message and its syntax is as shown below.

Error Message (Example)	Output Syntax (Example)	Description
C:\Workspace\SH\SH\resetprg.c	\$(FULLFILE)	Placeholder of the file name with its full path
(70)	(\$(LINE))	Placeholder of the line number enclosed with
		parentheses
: C2225	*	Wild-card for ":" and error message number
(E)	(E)	(E) indicating an error
Undeclared name "a"	*	Wild-card for error message

There are some restrictions on these regular expressions. For example, it is not possible to define two unclear strings next to each other, such as "\*\*", "\*\$(FILEDIR)", or "\$(FULLFILE)\*".

It is also not possible to define ":", "\", and ".", which are commonly used in filenames and directory names, to separate placeholders for filenames and directory names from other characters, such as "\$(FULLFILE):\$(LINE)" or "\$(FILENAME).\*".

• When "Error" has been selected as the syntax type and the output message matches the definition of the syntax:

An error icon  $(\bigcirc)$  appears on the left to the message output as the result of building. Doubleclicking on the line displays the corresponding source code in the editor.



• When "Warning" has been selected as the syntax type and the output message matches the definition of the syntax:

A warning icon ( $\triangle$ ) appears on the left to the message output as the result of building. Doubleclicking on the line displays the corresponding source code in the editor.

• When "Other" has been selected as the syntax type and the output message matches the definition of the syntax:

An information icon (O) appears on the left to the message output as the result of building. Doubleclicking on the line displays the corresponding source code in the editor.

The total number of errors and warnings will also be displayed in the Build tab of the output window at the end of a build phase.

- 10. Use the Add, Modify and Remove buttons to add, modify and remove output syntax line formats respectively.
- 11. Click the OK button when all modifications have been made.

#### Note:

You can only change the environment of a system phase via the Tools Administration dialog box.

#### To remove a custom phase

- 1. Select the phase to be removed and click the **Remove** button.
- 2. Click the OK button to confirm the new settings.

#### To import a custom phase

- 1. Click the **Import** button. The **Import Custom Phase** dialog box is displayed, which allows you to browse to an existing project, from which you want to import a custom phase.
- 2. Choose the location of the project, from which you want to import a custom phase. Once selected, the **Import Phase** dialog box is displayed, which lists the custom phases in the imported project.
- 3. Once you have decided which phase to import, highlight it in the list and click the OK button. The phase will then be added to the **Build Phases** dialog box, at the bottom of the build order.

# 3.3.2 Build File Order tab

If you were to select a C source file from the Workspace window and then activate [**Build -> Build File**] (or click the **Build File** toolbar button () you would expect the file to be compiled. Likewise, if you were to select an assembly source file from the workspace window and then activate [**Build -> Build File**] you would expect the file to be assembled. The connection between file group and which phase(s) to execute is managed by the **Build File Order** tab of the **Build Phases** dialog box. The list displays all of the current phases that will be executed when the build file operation is selected upon the file group shown in the **File group** list box. In figure below the "C source file" file group is selected and the "Compiler" and "MyPhase" phases are associated with it. Entries in the **Phase order** list, of the **Build File Order** tab, are added automatically as new entries are added to the **Build Order** tab.





For example, if you were to add a phase which takes C source files as input then this phase will be automatically added to the list of phases to execute when a build file operation is applied to a C source file. If you don't want a certain phase to execute when [**Build -> Build File**] is selected then clear the check box to the left of the phase name in the **Phase order** list.

# 3.3.3 File Mappings tab

By default, the files input to a build phase are only taken from the project, i.e. all project files of the type specified in the **Select input file group** drop-down list on the **New Build Phase** dialog box. Select [**Build -> Build Phases**]. The **Build Phases** dialog box will be displayed. If you would like a build phase to take files output from a previous build phase (these files are called intermediate files), then you must define this in the **File Mappings** tab of the **Build Phases** dialog box.

Build Order Build File Order File Mappings			
Source SH Assembler SH C/C++ Com SH C/C++ Com SH C/C++ Libra	Destination OptLinker SH Assembler OptLinker OptLinker	File Group Object file Assembly source file Object file Library file	<u>A</u> dd <u>M</u> odify <u>R</u> emove

A **File Mapping** means that you would like the output files of a certain type produced by one build phase (referred to as the **Source** phase) to another build phase (referred to as the **Destination** phase). Such intermediate files are then passed in addition to the project files.

# To add a file mapping

- 1. Click the Add button. The Define File Mapping dialog box opens.
- 2. Select an existing file group to use for the new mapping from the File group drop-down list.
- 3. Select the source phase (i.e. the phase that generates the files) from the Source phase drop-down list.
- 4. Select the destination phase (i.e. the phase that takes these files) from the **Destination phase** drop-down list.

5. Click the OK button to create the new mapping.

Define File Mapping	? ×
<u>F</u> ile group:	ОК
Object file	
Source phase:	Cancel
SH Assembler 📃 💌	
Destination phase:	
OptLinker 💌	

### To modify a file mapping

- 1. Select the mapping to be modified.
- 2. Click the **Modify** button. The **Define File Mapping** dialog box opens.
- 3. Modify the options as necessary.
- 4. Click the OK button to commit the changes.

#### To remove a file mapping

- 1. Select the mapping to be removed.
- 2. Click the **Remove** button.
- 3. Click the OK button to commit the changes.

# 3.4 Setting custom build phase options

Once you have defined a custom phase, you will want to specify the command line options that should be used when it is executed. Each defined phase has a menu option in the **Build** menu. To specify options for that phase select it. The dialog that will be displayed depends on whether the custom phase selected was multiple or single (according to the selection of phase type when it was created in the New Build Phase wizard).



MyPhase Options	? 🗙
Debug SH4demo SH4demo C source file SH4demo.c SH4demo.c Default Options	Options       Output Files       Dependent Files         Command:

If the phase selected was multiple then a list of project files is displayed on the left-hand side of the dialog to enable you to specify the build options on a file by file basis. If the phase selected was single then there is no project file list displayed. In either case, the three tabs below are available. This is where you can set the options that you want to apply to the selected file(s). You can also choose which configurations are being viewed. In the configuration list, each configuration is listed along with a special entry named **Multiple configurations**. If you select **Multiple configurations** then the **Select project configurations to modify** dialog box is displayed which allows you to select more than one configuration. This method is used throughout High-performance Embedded Workshop for modifying multiple configurations at once.

# 3.4.1 Options tab

This tab allows you to define the command line options that will be passed to the phase. The **Command** field displays the command that was entered when you defined the phase. Enter into the **Options** field the command line arguments that you would like to pass to the command. If you want to insert a placeholder, select the relevant placeholder from the **Placeholder** drop-down list and then click the **Insert** button. See Reference 5, Placeholders, for more information on placeholders.

Options Output Files Dependent Files	
<u>C</u> ommand:	
C:\Tools\MyTool.exe	
Options:	
\$(FULLFILE)	
Placeholder:	Insert button
File path + name	for
	placeholder

# Note:

An error will be returned when the total number of characters of the command line options exceeds 256.



# 3.4.2 Output Files tab

This tab is where you can specify the output files that will be produced by the phase. Before each file is passed into this phase, the High-performance Embedded Workshop checks that the output files are of a less recent date than the input file. If so, the phase will be executed for that file (i.e. input files have been modified since the output file or files were last produced). If the files are up-to-date then the phase will not be executed.

### Note:

If no output files are specified, the phase will execute regardless.

Options Output Files Dependent Files	
<u>O</u> utput files:	
	<u>A</u> dd
	<u>M</u> odify
	<u>R</u> emove

# To add an output file

- 1. Click the **Add** button. The **Add Output File** dialog box will be invoked.
- 2. Enter the file path or browse to it using the **Browse** button.
- 3. Click the OK button to add this output file to the list.

#### To modify an output file

- 1. Select the output file that you would like to modify.
- 2. Click the **Modify** button. The **Modify Output File** dialog box opens.
- 3. Modify the fields as required.
- 4. Click the OK button to add the modified entry back to the list.

#### To remove an output file

- 1. Select the output file that you would like to remove.
- 2. Click the **Remove** button.

An output file's path can include placeholders.



# 3.4.3 Dependent Files tab

This tab is where you can specify the dependent files that are needed by the phase. Before each file is passed into this phase, the High-performance Embedded Workshop checks that the dependent files are of a more recent date than the input file. If so, the phase will be executed for that file (i.e. dependent files have been modified since the input files was last modified). If not, the phase will not be executed.

Options Output Files Dependent Files	
Dependent files:	
	<u>A</u> dd
	<u>M</u> odify
	<u>R</u> emove

#### To add a dependent file

- 1. Click the Add button. The Add Dependent File dialog box opens.
- 2. Enter the file path or browse to it using the **Browse** button.
- 3. Click the OK button to add this output file to the list.

#### To modify a dependent file

- 1. Select the dependent file that you would like to modify.
- 2. Click the Modify button. The Modify Dependent File dialog box opens.
- 3. Modify the fields as required.
- 4. Click the OK button to add the modified entry back to the list.

# To remove a dependent file

- 1. Select the dependent file that you would like to remove.
- 2. Click the **Remove** button.

A dependent file's path can include placeholders.



# 3.5 Controlling the build

By default, the High-performance Embedded Workshop will execute all of the phases in a build and only stop if a fatal error is encountered. You can change this behavior by setting the controls on the **Build** tab of the **Options** dialog box.

Select [Setup -> Options] to display the Options dialog box and then select the Build tab. If you want to stop the build when a certain number of errors are exceeded then set the Stop build if the number of errors exceeds checkbox and specify the error count limit in the field to the right. If you want to stop the build when a certain number of warnings are exceeded then set the Stop build if the number of warnings exceeds checkbox and specify the warning count limit in the field to the right.



In addition to specifying error and warning count limits, the **Build** tab also allows you to request that the **Command line**, **Environment** and **Initial directory** of each execution should be displayed. Check the appropriate check boxes as necessary.

### Notes:

- Irrespective of what these controls are set to, the build will always halt if a fatal error is encountered.
- When projects are dependent on one another, building stops if an error occurs in the child project and its parent project will not be built.
- Note the following descriptions when you check the **Stop build if the number of errors exceeds** check box and specify a number to the controls.
  - i. The build will stop immediately once the maximum number of errors or warnings is exceeded. It will not continue to the end of the current phase.
  - ii. When the number of errors in one execution of a tool exceeds the specified number, the number of error messages displayed on the "Output" window is the specified number plus one. A message saying that the number of error has exceeded the specified number is NOT displayed on the **Output** window.

The same things as **Stop build if the number of errors exceeds** shown above hold also in **Stop build if the number of warnings exceeds**. There is no correlation between **Stop build if the number of errors exceeds** and **Stop build if the number of warnings exceeds**. They are independent.

# 3.6 Logging build output

The High-performance Embedded Workshop allows you to write the results of each build to file.



### To specify a log file

- 1. Select [Setup -> Customize]. The Customize dialog box opens.
- 2. Select the **Log** tab.
- 3. Select the Generate build log checkbox.
- 4. Enter the full path of the log file into the **Path** field, or browse to it graphically by clicking the **Browse** button, or click the placeholder button ( ) and select the placeholder from the pop-up menu.
- 5. Click the OK button to confirm the new log file settings.

🔽 Generate <u>b</u> uild log:		
Path:		
\$(WORKSPDIR)\\$(WORKSPNAME).txt	<u>B</u> rowse	

# 3.7 Changing toolchain version

If two or more versions of the same toolchain are registered in the High-performance Embedded Workshop, you can choose a version of the toolchain on the **Change Toolchain Version** dialog box. To invoke the dialog, select [**Tools -> Change Toolchain Version**]. Choose one of the versions from the **Toolchain version** drop-down list and click the OK button to enforce your choice.

Change Toolchain	Version		? ×
Toolchain name: Current version: <u>C</u> PU Family: <u>I</u> oolchain: Toolchain <u>v</u> ersion: Toolchain build pha	Renesas SuperH RISC engi 9.1.0.0 SuperH RISC engine Renesas SuperH RISC eng 9.2.0.0	<b></b>	OK Cancel
Build phase OptLinker SH Assembler SH C/C++ Compile SH C/C++ Library	er (	Version 9.04.00 7.01.02 9.02.00 3.00.02	Information

To show information about toolchain components, select a tool from the **Toolchain build phase** drop-down list on the **Change Toolchain Version** dialog box, and click the **Information** button. A tool information dialog box will show you information about the tool. Click the **Close** button to close the dialog box.

# 3.8 Generating a makefile

The High-performance Embedded Workshop allows you to generate a makefile associated with your workspace. Highperformance Embedded Workshop is capable of generating hmake, nmake, and gnumake compatible files. The make tool to execute makefiles is operated in the DOS command prompt and thus the use of makefiles allows you to build projects without High-performance Embedded Workshop.

The generated makefiles can be executed in the High-performance Embedded Workshop. For details, see section 3.9, Using a makefile inside the High-performance Embedded Workshop system.

### To generate a makefile

- 1. Open a workspace where you wish to generate a makefile (if the workspace includes two or more projects, the active project must be one in which you wish to generate a makefile).
- 2. Select a configuration for generating a makefile.
- 3. Select [Build -> Generate Makefile]. The Generate Makefile dialog box opens.

Generate Makefile	? ×
Generate makefile for:	OK Cancel
Makefile <u>f</u> ormat: HMake	[
Relative paths: Workspace directory and below only	ſ
Use static sub-command files           Scan dependencies whilst building make file	

- 4. Select a makefile type in Generate makefile for.
  - the current configuration in the currently active project
  - all configurations in the currently active project
  - all configurations in all loaded projects in the current workspace
- 5. Enter the file name in the **MakeFile name** edit box. To insert a placeholder, place the cursor at the position where you wish to insert the placeholder, click the placeholder button (**)**, and select Placeholder from the pop-up menu. The default makefile names are as follows.
  - When "the current configuration in the currently active project" is selected: \$(PROJECTNAME)\_\$(CONFIGNAME).mak
  - When "all configurations in the currently active project" is selected: \$(PROJECTNAME).mak


- When "all configurations in all loaded projects in the current workspace" is selected: \$(WORKSPNAME).mak
- 6. Select a makefile format in the **Makefile format** drop-down list. High-performance Embedded Workshop is capable of generating GNUMake, HMake and NMake compatible files.
- 7. In the **Relative paths** drop-down list, select how directories within the makefile should be expressed. "Workspace directory and below only" is the default option. See the table below for details.

Option	Workspace Directory and below	Outside the Workspace Directory
None	Absolute path	Absolute path
Workspace directory and below only	Relative path	Absolute path
All	Relative path	Relative path

8. Selecting the **Use static sub-command files** checkbox will generate separate command files in the make destination directory. By default, this checkbox is not selected.

### Note:

While a toolchain included in the compiler packages listed below is in use, the **Use static sub-command files** checkbox has a fixed tick if you have selected GNUMake in **Makefile format**.

- C/C++ Compiler Package for SuperH Family
- C/C++ Compiler Package for H8SX, H8S, H8 Family
- C/C++ Compiler Package for RX Family
- 9. Selecting the **Scan dependencies whilst building make file** checkbox will execute a dependency scan before creation of a makefile. This checkbox is selected by default.
- 10. Click OK.

The High-performance Embedded Workshop will create a subdirectory called 'make' within the current workspace directory and then generate the makefile into it. The executable HMAKE.EXE, located in the High-performance Embedded Workshop installation directory, is provided for you to execute the makefiles generated by HMake selected in **Makefile format**.

If you wish to modify makefiles, see Reference 13, HMAKE User Guide.

### Note:

If the name of the High-performance Embedded Workshop installation directory includes a space, the GNU Make command may not work correctly when GNU Make is selected as the makefile format with the makefile generating function.

### To execute a makefile (HMake)

1. Open a DOS Command Prompt window and move to the 'make' directory where the makefile was generated.

2. Execute HMAKE. Its command line is HMAKE.EXE <makefile>.

### Note:

The degree of portability of a generated makefile is entirely dependent upon how portable the project itself is. For example, any compiler options that include full paths to an output directory or include file directory will mean that, when given to another user with a different installation, the build will probably fail. In general use placeholders wherever possible - using a full, specific path should be avoided when possible.

## 3.9 Using a makefile inside the High-performance Embedded Workshop system

The High-performance Embedded Workshop also allows you to configure the internal build to use a particular make tool (hmake, nmake, or gmake) as well as the internal High-performance Embedded Workshop system.

The build procedures using makefiles are listed below. nmake and gmake tools must be prepared by the user, while hmake is installed in the High-performance Embedded Workshop installation directory.

### To set-up the internal makefile execution

- 1. Create a High-performance Embedded Workshop workspace.
- 2. If you want to use a High-performance Embedded Workshop generated makefile, select [**Build -> Generate Makefile**] to generate a makefile. For details, see section 3.8, Generating a makefile.
- 3. Select [Setup -> Options]. The Options dialog box opens.
- 4. Click the **Build** tab.

Makefile Build			
Command (excluding parameters):			
\$(HEWDIR)\hmake.exe	B <u>r</u> owse		
✓ Use makefile build system (for this workspace)			
	Bro <u>w</u> se		
Initial directory: \$(WORKSPDIR)\make	Brows <u>e</u>		
Makefile options:	System		
Clean options:			

- 5. Enter the make tool name to execute the makefile in the **Command (excluding parameters)** edit box. The default is hmake (\$(HEWDIR)\hmake.exe).
- 6. Select the **Use makefile build system (for this workspace)** checkbox. This means that the makefile should be executed rather than the internal build.

- 7. While the **Generate makefile before build commences** checkbox is selected, the High-performance Embedded Workshop will generate a makefile before executing the build. This means the makefile will always be up to date with the High-performance Embedded Workshop project system.
- Enter the file name in the Makefile filename edit box. The default file name is "\$(WORKSPDIR)\make\\$(PROJECTNAME)\_\$(CONFIGNAME).mak". Note: This entry can be left empty if no makefile needs to be specified on the command line.
- 9. In the **Initial directory** edit box, enter the current directory where the makefile is to be executed. When this edit box is empty, the initial directory will default to "\$(WORKSPDIR)".
- 10. Enter the options for building the makefile in the **Makefile options** edit box.
- 11. Enter the options for cleaning the makefile in the **Clean options** edit box.
- 12. Click on the **OK** button.

Once this system is setup the build button, menu and keyboard shortcuts are linked to the makefile tool execution. The clean all projects button, menu and keyboard shortcut will also be linked to the makefile execution. All output is directed to the Build tab of the output window as in the case of the normal build. If you are using a build phase supported by High-performance Embedded Workshop the line of an error message can be double-clicked to jump to the source files. The help link should also be supported. Note when using the HMAKE.exe system the Build All toolbar button () will pass a command to HMAKE to force a re-build all operation.

### To set-up the makefile system environment

By default, the makefile will be executed in the default system environment (the same environment used within Highperformance Embedded Workshop). If this is not desirable, the environment can be configured as described below:

1. In the **Build** tab, click the **System** button to the right of the **Makefile options** controls. This will launch the **Makefile System Environment** dialog box:

Makefile System Environment	? ×
☑ Inherit system environment	OK
<u>E</u> nvironment variables:	Cancel
	<u>A</u> dd
	Modify
	Remo <u>v</u> e

 The Inherit system environment checkbox can be used to alter the inheritance of the system environment: Set: The makefile will be executed in the standard environment. (Default.) Clear: The makefile will be executed in a fresh environment. Note:



Only the "SystemRoot" variable will be carried over into the new environment as it is needed to allow program to run in Windows®.

- 3. In both cases, additional environment variable can be set using the **Environment variables** controls. Any values set here will override the value in the current system environment.
- 4. Click OK to save the changes, and return to the **Options** dialog box. Values will not be committed to the Workspace until the OK button in that dialog is also used.

## 3.10 Customizing the High-performance Embedded Workshop linkage order

Modules are usually linked in the alphabetical order in High-performance Embedded Workshop. However, you can specify the linkage order if you wish to.

### To manually change the linkage order

- 1. Click [Build -> Linkage Order]. The Linkage Order dialog box opens.
- 2. Select the Use custom linkage order check box.
- 3. The **Object order** list box allows you to specify the linkage order of modules. Each module has a different icon depending on where it originated from. This is shown below:

### Note:

While a toolchain included in the compiler packages listed below is in use, the **Object order** list box shows library files along with object files.

- C/C++ Compiler Package for SuperH Family
- C/C++ Compiler Package for H8SX, H8S, H8 Family
- C/C++ Compiler Package for RX Family
- C/C++ Compiler Package for M16C Series and R8C Family V.6.00 Release 00 or later

However, only object files matter to the linkage order. The order of library files is only used in searching for undefined symbols.



## High-performance Embedded Workshop V.4.09

	Linkage Order	? ×
Intermediate file — File in project — User library from options	<ul> <li>✓ Use custom linkage order</li> <li>Object order:</li> <li>main.obj</li> <li>resetprg.obj</li> <li>intprg.obj</li> <li>runtime_library.lib</li> <li></li></ul>	OK Cancel Move to <u>t</u> op Move <u>up</u> Move <u>d</u> own Move to <u>b</u> ottom
		Import
	<u>C</u> urrent configuration: Debug	Copy to

You can change the linkage order in the following ways.

## • To move a single or consecutive modules

Move to top	Moves the selected module(s) to the top.	A     C       B     □       C     □       A     □       B     □       E     □
Move up	Switches the selected module(s) and the previous module.	A B C D E A A C D B E E
Move down	Switches the selected module(s) and the next module.	A B C D E E A D B C E E
Move to bottom	Moves the selected module(s) to the bottom.	A B C D E C A D E A D E C



To move non-consecutive modules

•

Move to top	Moves all of the selected modules to the top.	A B C D E B A D C E
Group up	Moves all of the selected modules to the position of the first one among them.	A B C D E C A B D C
Group down	Moves all of the selected modules to the position of the last one among them.	A B C D E E C C C C C C C C C C C C C C C C
Move to bottom	Moves all of the selected modules to the bottom.	A B C C D E D D C B B D D

If the top module is selected, the **Move to top** and **Move up** buttons will be disabled. If the bottom module is selected, the **Move to bottom** and **Move down** buttons will be disabled.

The current linkage order can be saved in a text format. You can modify the linkage order saved in the file and load it to apply the new linkage order.

### To load the linkage order from a text file

- 1. Click [Build -> Linkage Order]. The Linkage Order dialog box opens.
- 2. Select the Use custom linkage order check box.
- 3. Click on the Import button. The Import Linkage Order dialog box opens.
- 4. Choose a file and click on the Select button. The Linkage Order Import Report dialog box opens.
- 5. To load the linkage order saved in the file, click on the **Continue** button. The listbox will be updated with the new linkage order.

Note: This will overwrite your current linkage order.

6. Click on the **Cancel** button to close the dialog box without modifying your current linkage order.



### To save the linkage order in a text file

- 1. Click [Build -> Linkage Order]. The Linkage Order dialog box opens.
- 2. Select the Use custom linkage order check box.
- 3. Click on the **Export** button. The **Export Linkage Order** dialog box opens.
- 4. Choose a file and directory and click on the **Save** button. The current settings in the list box are saved in the specified text file.

### **Example:**

# Exported linkage order for project "Project\_name" - configuration "Debug"

c:\workspace\workspace\_name\project\_name\debug\main.obj

 $c:\!\verb|workspace|workspace_name|project_name|debug|resetprg.obj|$ 

c:\workspace\workspace\_name\project\_name\debug\intprg.obj

c:\workspace\workspace\_name\project\_name\debug\runtime\_library.lib

c:\workspace\workspace\_name\project\_name\debug\projectname.lib

c:\workspace\workspace\_name\project\_name\debug\user\_edit\_options.lib

Open the file (e.g. in the Editor), modify the linkage order, and then save it in the text format. Follow the procedures of "To import the linkage order from the text file" to load the new linkage order.

When you are using multiple configurations it is likely that the linkage order will be very similar. To do this effectively you can copy the current settings in the dialog to other configurations. This is described below:

### To copy the linkage order from one configuration to another

- 1. Click [Build -> Linkage Order]. The Linkage Order dialog box opens.
- 2. Select the Use custom linkage order check box.
- 3. Select the configuration you wish to copy in the **Current configuration** drop-down list. This defaults to the currently loaded configuration.
- 4. Click the **Copy to** button this displays the **Select Configuration To Copy To** dialog box and asks you which of the configurations in the current project you wish to copy the current linkage order to.
- 5. Select a configuration and click OK.



# 4. Editor

This chapter describes how to use the editor that is provided with the High-performance Embedded Workshop.

## 4.1 Editor window

The editor window contains the file windows that are being viewed or edited.

The title bar of the active window will appear a different color from that of the others ("dbsct.c" is the active window in figure bellow). All text operations such as typing, pasting text and so forth only affect the active window.

To switch to another source file window (i.e. to make some other window the active window) there are a number of methods:

- Click on it if it is visible, **OR**
- Press CTRL+TAB or CTRL+SHIFT+TAB to cycle through the windows one after another, OR
- Select the window by name from the Window menu, OR
- Select its tab at the bottom of the editor window.



Title barShows information on files (file name, asterisk \* while editing, and Read-Only attribute). If a<br/>file is included in a directory of the current project, the file name is displayed on the title bar of<br/>the editor window. Otherwise the title bar of the editor window displays the file name<br/>including its absolute path.

Toolbar The toolbar is only shown while the debugger is connected. When one display mode is active, clicking another toolbar button switches the display mode. If you place the mouse pointer on a toolbar button, the name of the button pops up.



Column header	You can adjust the column width by dragging the mouse on a column header. Moreover, right-
	clicking the column header displays a pop-up menu. A tick mark right next to an entry
	indicates that this column is displayed. Clicking an entry will switch showing/hiding the
	column. If you place the mouse pointer on a column header, the name of the column pops up.
	You can also select showing/hiding the column header itself.
Column	There are several spaces (columns) on the left of the source code field. If you place the mouse
	pointer on a column, the name of the column pops up.
Source code field	This field includes codes highlighting the syntax. The maximum number of characters per line
	shown in the editor is 8,192. Right-clicking within this field displays a pop-up menu
	containing available options.
Tab	Shows information on files (file name and asterisk * while editing). If you place the mouse
	pointer on a tab, the file name including its absolute path pops up.
	Right-clicking within this field displays a pop-up menu containing available options.

## 4.2 Working with multiple files

The file area is where you will work with the files of your project. The editor allows you to have many files open at one time, to switch between them, to arrange them in different configurations and to edit them in whichever order you want to. The operations that you can perform upon the windows are typical of most Windows® applications and they can be found under the **Window** menu:

Menu Option	Operation	
[Window -> Cascade]	Arrange all open windows so that they overlap, with the top left of each Editor window visible.	
[Window -> Tile Horizontally]	Arrange all open windows horizontally so that they occupy the entire Editor window, without any overlaps.	
[Window -> Tile Vertically]	Arrange all open windows vertically so that they occupy the entire Editor window, without any overlaps.	
[Window -> Arrange Icons]	Line up all minimized windows at the bottom of the Editor window.	
[Window -> Close All]	Close all open Editor windows.	

The files within the editor can be displayed in a 'notebook' style. This means that each file has a separate tab associated with it to aid in navigating between files.

### To show files in a notebook style

Show files in notebook

- 1. Select [Setup -> Options]. The Options dialog box opens.
- 2. Select the Editor tab.
- 3. Select the **Show files in notebook** checkbox as appropriate.
- 4. Click the OK button.



## 4.3 Standard file operations

### 4.3.1 Creating a new file

### To create a new editing window

Select one of the following operations:

- Click the **New File** toolbar button (**b**), **OR**
- Press CTRL+N, **OR**
- Select [File -> New].

The new window will be given an arbitrary name by default. You can provide a new name when you save the file.

### 4.3.2 Editing a file

The High-performance Embedded Workshop editor's standard editing functionality is available through the usual methods (i.e. the menu, toolbar and keyboard shortcuts) and is additionally supported via a pop-up menu that is local to each Editor window. Right-click in an open Editor window to invoke the pop-up menu.

The table below outlines the basic operations that are provided by the editor.

Operation	Effect	Action
Undo	Reverses the last editing operation	Select [Edit -> Undo]
		Press CTRL+Z
Redo	Repeats the last undone editing operation	Select [Edit -> Redo]
		Press CTRL+Y
Cut	Removes highlighted text and places it on the	Click the <b>Cut</b> toolbar button ( 👗 )
	Windows® clipboard	Press CTRL+X
		Select [Edit -> Cut]
		Select Cut from the pop-up menu
Сору	Places a copy of the highlighted text into the Windows®	Click the <b>Copy</b> toolbar button (
	clipboard	Press CTRL+C
		Select [Edit -> Copy]
		Select Copy from the pop-up menu
Paste	Copies the contents of the Windows® clipboard into the	Click the Paste toolbar button (🖺)
	active window at the position of the insertion cursor	Press CTRL+V
		Select [Edit -> Paste]
		Select Paste from the pop-up menu
Clear	Removes highlighted text (it is not copied to the	Select [Edit -> Clear]
	Windows® clipboard)	Press Delete
Select All	Selects (i.e. highlights) the entire contents of the active	Select [Edit -> Select All]
	window	Press CTRL+A



If you edit the file, the title bar of the editor window shows an asterisk (\*). (e.g. filename.c\*)

This asterisk remains until you save the file. If you undo all the changes made in the file, the asterisk disappears.

### Selecting text in the editor

It is possible to select text in the same manner as all editors. However to access column selection hold down the ALT key while you are selecting the text with the mouse. This changes the selection technique from line to column selection.

### 4.3.3 Saving a file

### To save the contents of an editing window

- 1. Ensure that the window, whose contents you want to save, is the active window.
- 2. Select one of the following operations:
  - Click the Save File toolbar button (**H**), OR
  - Press CTRL+S, **OR**
  - Select [File -> Save].
- 3. If the file has not been saved before, a **File Save** dialog box will be displayed. Enter a filename, specify a directory and then click the OK button to create the file with the name given in the directory specified. If the file has been saved before then the file will be updated (no dialog box will be displayed).

### To save the contents of an editing window under a new name

- 1. Ensure that the window, whose contents you want to save, is the active window.
- 2. Select [File -> Save As].
- 3. A **File Save** dialog box will be displayed. Enter a filename, specify a directory and then click the OK button to create the file with the name given, in the directory specified.

### To save the contents of every open editor window

- 1. Select one of the following operations:
  - Click the Save All toolbar button (
  - Select [File -> Save All].
- 2. If any of the files has not been saved before, a **File Save** dialog box will be displayed. Enter a filename, specify a directory and then click the OK button to create the file with the name given in the directory specified. If any of the files have been saved before, then that file will be updated (no dialog will be displayed).



When you attempt to save a read-only file, a confirmation dialog box appears asking if you wish to save the file as another name.



- Clicking **Yes** opens the **Save file** dialog box. Change the file name before saving.
- Clicking **No** closes the file without saving.
- Clicking **Overwrite** saves the file by overwriting the contents. The file is no longer read-only.
- Clicking **Cancel** cancels the attempt to save the file.

### 4.3.4 Opening a file

### To open a file

- 1. Select one of the following operations to open a File Open dialog box:
  - Click the **File Open** toolbar button (), **OR**
  - Press CTRL+O, **OR**
  - Select [File -> Open].
- 2. Use the directory browser to navigate to the directory in which the file you want to open is located. Use the **Files** of **Type** combo box to select the type of file you want to open (or set it to **All Files** (\* . \*) to see every file in a directory).
- 3. Once you have located the file, select it and click **Open**.

To open a file, drag the file from the Windows® Explorer and drop it into the High-performance Embedded Workshop main window.

To open a source file on the **Projects** tab of the workspace window, select one of the following ways:

- Double-click the file, **OR**
- Select the file and click the right-hand mouse button. Select **Open <file name>** from the pop-up menu, **OR**
- If the file has already been selected (focused), press Enter.



You can also use another editor (external editor) to open files. For details, see section 6.9, Using an external editor.

The High-performance Embedded Workshop keeps track of the last files that you have opened and adds them to the **File** menu under the [**Recent Files**] sub-menu.

### To open a recently used file

Select [File -> Recent Files] and from this sub-menu select the desired file.

### 4.3.5 Closing files

### To close an individual file

Select one of the following operations:

- Double-click the Editor window's system menu (located at the top left of each window), OR
- Click on the **Editor** window's system menu (located at the top left of each window) and select the **Close** menu option, **OR**
- Ensure that the window that you want to close is the active window and then press CTRL+F4, OR
- Ensure that the window that you want to close is the active window and then select [File -> Close], OR
- Click on the **Close** button (located at the top right of each window).

### To close all files

Select [Window -> Close All].

If you attempt to close an edited file that is not saved, a confirmation dialog box appears asking if you wish to save the file.



- Clicking **No** closes the file without saving.
- Clicking **Cancel** returns to the editor window.



• Clicking **Yes** closes the file by overwriting the contents. If the file has the read-only attribute, a confirmation dialog box appears asking if you wish to save the file as another name.



- Clicking Yes opens the Save file dialog box. Change the file name before saving.
- Clicking **No** closes the file without saving.
- Clicking **Overwrite** closes the file by overwriting the contents.
- Clicking **Cancel** returns to the editor window.

### 4.3.6 Pop-up menu to close the window

	e și	6 <u>7</u>			
	Line	Source Ad	C.,	S.,	Source
	14				<pre>#include "typedefi</pre>
	15				
	16				#pragma section \$D
	17				static const struc
	18				_UBYTE *rom_s;
	19				_UBYTE *rom_e;
	20				_UBYTE *ram_s;
	21				} DTBL[] = {
	•				
1	I db	sct.c* Sa	ve		
		<u>C</u> lo	)se		

In the High-performance Embedded Workshop editor window it is possible to save and close editor file window via the related tab's pop-up menu.

Right-clicking on the tab of the Editor window opens a pop-up menu containing available options.

Pop-up Menu Option	Function	
Save	Saves the contents of an editing window.	
Close	Closes an individual file.	



In the Disassembly window it is possible to close window via the related tab's pop-up menu.

Right-clicking on the tab of the Disassembly window opens a pop-up menu containing available options.

Pop-up Menu Option	Function		
Close	Closes the Disassembly window.		

## 4.4 Searching and navigating through files

### 4.4.1 Finding text

### To search for text in the current file

- 1. Ensure that the window, whose contents you want to search, is the active window.
- 2. Position the insertion cursor at the point from which you want to start your search.
- 3. Select one of the following operations to open the **Find** dialog box:
  - Click the **Find** toolbar button (**M**), **OR**
  - Press the CTRL+F key, **OR**
  - Select [Edit -> Find], OR
  - Select **Find** from the pop-up menu in the Editor window.

Find		? X
Find what: String	•	<u>F</u> ind Next
<ul> <li>Match whole word only</li> <li>Match case</li> <li>Regular expression</li> </ul>	Direction C ∐p C Down	Cancel

- 4. Enter the text that you want to search for into the **Find what** field, or select a previous search string from the drop-down list box. If you select text before invoking the find operation, the selected text will be automatically placed into the **Find what** field.
- 5. If you would like to search for character string as a whole word then click the **Match whole word only** checkbox. When this option is not selected, the search will be for any string that is matched by the search string.
- 6. If you would like your search to be case-sensitive (i.e. to distinguish between upper and lower case letters) then check the **Match case** checkbox.
- 7. If your search string uses regular expressions then check the **Regular expressions** checkbox. See Reference 4, Regular Expressions, for further information.



- 8. The **Direction** radio buttons allow you to select the direction of the search. Selecting **Down** means that the search will be performed from the insertion cursor towards the bottom of the file. Selecting **Up** means that the search will be performed from the insertion cursor towards the top of the file.
- 9. Click the **Find Next** button to begin the search.

You can also search for text in multiple files.

### 4.4.2 Finding text in multiple files

### To search for text in multiple files

- 1. Select one of the following operations to open the **Find In Files** dialog box:
  - Click the **Find In Files** toolbar button (**P**), **OR**
  - Press the F4 key, **OR**
  - Select [Edit -> Find in Files].

Find In Files	? ×
Find what:     main       In files/types:     *.abs;*.asm;*.bin;*.bls;*.c;*.cal;*.cc;*.cp;*.c	<u>F</u> ind Cancel
Match case     Match whole word only       Regular expressions     Output to 'Find in Files 2'	
Location     Directory:     C:\Workspace\DemoSH4\DemoSH4	Browse
<ul> <li>✓ Search sub-directories</li> <li>C Active project</li> </ul>	
C All loaded projects	

- 2. Enter the text that you want to search for into the **Find** field, or select a previous search string from the dropdown list box. If you selected text before invoking the find operation, the selected text will be automatically placed into the **Find what** field.
- 3. Enter the file extensions of the files you would like to search into the **In files/types** field. If several extensions are specified, be sure to separate them with a semicolon (e.g. \*.C; \*.H).
- 4. If you would like your search to be case-sensitive (i.e. to distinguish between upper and lower case letters) then check the **Match case** checkbox.

If you would like your search to match a whole word, as opposed to matching all occurrences, then check the Match whole word only checkbox.
 Note: this option is mutually-exclusive with the Regular expressions, only one of these two options can be

used at any given time.

- 6. If your search string uses regular expressions then check the **Regular expressions** checkbox. See Reference 4, Regular Expressions, for further information.
- If you would like your search results to appear in the Find in Files 2 tab in the Output Window instead of the Find in Files 1 tab, check the Output to 'Find in Files 2' checkbox (this enables two different sets of search results to be available in the Output Window).
- 8. Select a search location type.
- Directory

Selecting this option will enable searching a directory (or directory structure) for the given search string. For this options, set the **Search sub directories** checkbox if you also wish to search all sub-directories of the given directory. If you just want to search the specified single directory in Directory field then ensure that this checkbox is not checked.

## • Active project

Selecting this option will enable you to search the files belonging to the currently active project in the workspace.

• All loaded projects

Selecting this option will enable you to search the files belonging to all loaded projects in the workspace.

Note that the **Active project** and **All loaded projects** options will not search project dependencies (include files).

9. Click Find to begin the search. Any matches found will be displayed in the Find in Files 1 or Find in Files 2 tab of the Output window. To stop a Find In Files action once it is under way, select the [Edit -> Stop Find in Files] menu option. Once the Find In Files operation is complete, you may jump to an instance of the search string by double-clicking on the desired entry in the Output window.

## 4.4.3 Replacing text

Replacing text is similar to finding text, as discussed in the previous section. The difference is that when the text is found you have the option to replace it with other text.

### To replace text in a file

- 1. Ensure that the window, whose contents you want to replace, is the active window.
- 2. Position the insertion cursor at the point from which you want to start your search.



- 3. Select one of the following operations to open the **Replace** dialog box:
  - Press the CTRL+H key, **OR**
  - Select [Edit -> Replace], OR
  - Select **Replace** from the pop-up menu in the Editor window.

Replace		? ×
Find what:	tring	<u>F</u> ind Next <u>R</u> eplace
<ul> <li>Match <u>w</u>hole word</li> <li>Match <u>c</u>ase</li> <li>Regular <u>expression</u></li> </ul>	C <u>S</u> election	Replace <u>A</u> ll Cancel

- 4. Enter the text that you want to search for into the **Find what** field, or select a previous search string from the drop-down list box. If you select text before invoking the replace operation, the selected text will be automatically placed into the **Find what** field.
- 5. Enter the text that you want to replace the search string with, or select a previous replace string from the dropdown list box.
- 6. If you would like to search for character string as a whole word then click the **Match whole word only** checkbox. When this option is not selected, the search will be for any string that is matched by the search string.
- 7. If you would like your search to be case-sensitive (i.e. to distinguish between upper and lower case letters) then check the **Match case** checkbox.
- 8. If your search string uses regular expressions then check the **Regular expressions** checkbox. See Reference 4, Regular Expressions, for further information.
- 9. If you clicked the Find Next button, the editor will search for the first occurrence of the search string. Click the Replace button if you want to replace it. Click Replace All button to replace all occurrences or click the Cancel button to stop the replace action. If you select Selection in the Replace In field, the replace action will be performed in the range of the selected text. If you select Whole file, the replace action will be performed on the whole file. If you select All open files, all files that are currently open in the editor will have the replace operation carried out on them.

### 4.4.4 Jumping to a specified line

### To jump to a line in a file

- 1. Ensure that the window, whose contents you want to replace, is the active window.
- 2. Select one of the following operations to open the **Goto** dialog box:



- Press CTRL+G, OR
- Select [Edit -> Goto Line], OR
- Select **Goto Line** from the pop-up menu in the Editor window.

Goto			? ×
Line number	1		
OK		Cancel	

3. Enter the number of the line that you want to jump to into the **Line number** box and then click the OK button. The insertion cursor will be placed at the start of the line number specified.

## 4.5 Bookmarks

When working with many large files at a time, it can become difficult to locate specific lines or areas of interest. Bookmarks enable you to specify lines that you want to jump back to at a subsequent time. One example of its use is in a large C file where you may want to set a bookmark on each function definition. Once a bookmark has been set, it exists until it is removed or the file is closed.

### To set a bookmark

- 1. Place the insertion cursor on the line to mark.
- 2. Select one of the following operations:
  - Click the **Toggle Bookmark** toolbar button ( ), **OR**
  - Press CTRL+F2, **OR**
  - Select [Edit -> Bookmarks -> Toggle Bookmark], OR
  - Right-click and select [Bookmarks -> Toggle Bookmark] from the pop-up menu.
- 3. A bookmark icon () will be placed on the bookmarked line, to indicate that it is an active bookmark.

### To remove a bookmark

- 1. Place the insertion cursor on the marked line.
- 2. Select one of the following operations:
  - Click the **Toggle Bookmark** toolbar button (**C**), **OR**
  - Press CTRL+F2, **OR**
  - Select [Edit -> Bookmarks -> Toggle Bookmark], OR

- Right-click and select [Bookmarks -> Toggle Bookmark] from the pop-up menu.
- 3. The bookmark icon will be removed from the line.

### To jump to the next bookmark in a file

- 1. Ensure that the insertion cursor is somewhere within the file to be searched.
- 2. Select one of the following operations:
  - Click the Next Bookmark toolbar button (🟠), OR
  - Press F2, **OR**
  - Select [Edit -> Bookmarks -> Next Bookmark], OR
  - Right-click and select [**Bookmarks -> Next Bookmark**] from the pop-up menu.

### To jump to the previous bookmark in a file

- 1. Ensure that the insertion cursor is somewhere within the file to be searched.
- 2. Select one of the following operations:
  - Click the **Previous Bookmark** toolbar button (🟠), **OR**
  - Press SHIFT+F2, **OR**
  - Select [Edit -> Bookmarks -> Previous Bookmark], OR
  - Right-click and select [Bookmarks -> Previous Bookmark] from the pop-up menu.

### To remove all bookmarks

- 1. Ensure that the window, whose bookmarks you want to remove is the active window.
- 2. Select one of the following operations:
  - Click the Clear All Bookmarks toolbar button (📩), OR
  - Select [Edit -> Bookmarks -> Clear All Bookmarks], OR
  - Right-click and select [Bookmarks -> Clear All Bookmarks] from the pop-up menu.

## 4.6 **Printing a file**

### To print a file

- 1. Ensure that the window, whose contents you want to print, is the active window.
- 2. Select one of the following operations:
  - Click the **Print** toolbar button (🗁), **OR**
  - Press CTRL+P, **OR**
  - Select [File -> Print].

## 4.7 Configuring text layout

### 4.7.1 Page set-up

When you print a file from the High-performance Embedded Workshop editor, the settings in the print dialog affect the way the file is printed (e.g. double or single sided). Control over how the text is formatted on the page can also be controlled via the **Page Setup** option. This allows you to specify the margins (top, bottom, left and right) of your printouts. It is often necessary to set this because some printers cannot print to the edges of an A4 page. Furthermore, some users have their own layout requirements (e.g. a large left-hand margin so that code can be placed in an A4 binder).

### To set up the page margins

- 1. Select [File -> Page Setup]. The Page Setup dialog box opens.
- 2. Enter the width of the margins required into the Left, Right, Top and Bottom fields.
- 3. Set the **Inch** or **mm** option accordingly.
- 4. Click the OK button for the new settings to take effect.

Page Setup			? ×
Margins Left: Iop: Inch	<u>R</u> ight: <u>B</u> ottom:	0	
Header:			
Footer:			
I I Wrap Te>	 ОК	Can	



### To set up the header and footer information

- 1. Select [File -> Page Setup]. The Page Setup dialog box opens.
- 2. Enter into the **Header** and **Footer** edit fields the text required to be displayed. All normal placeholders are available along with page numbering, text justification and date fields. These are all expanded before the page is to be printed.
- 3. Click the OK button for the new settings to take effect.

### To set up print wrapping

- 1. Select [File -> Page Setup]. The Page Setup dialog box opens.
- 2. Click the **Wrap Text** check box. This switches on the wrap text facility when printing so no text is truncated and everything is visible.
- 3. Click OK for the new settings to take effect.

## 4.7.2 Changing tabs

🔽 <u>U</u> se spaces as tabs	Tab si <u>z</u> e:	4
------------------------------	--------------------	---

### To change tab size

- 1. Select [Setup -> Options]. The Options dialog box opens.
- 2. Select the **Editor** tab.
- 3. Enter into **Tab Size** the number of desired tabs.
- 4. Click the OK button for the new tab settings to take effect.

When the TAB key is pressed in the editor a tab character is usually stored in the file. However, sometimes it is preferable to store spaces instead. The representation of tab characters can be controlled via the **Options** dialog box.

### To use spaces as tabs

- 1. Select [Setup -> Options]. The Options dialog box opens.
- 2. Select the **Editor** tab.
- 3. Set the Use spaces as tabs checkbox as appropriate.
- 4. Click the OK button for the new tab settings to take effect.

#### 4.7.3 Auto indentation

When you press Enter the editor, the insertion cursor will move to the next line down, at the first column (i.e. against the left-hand side of the window). Auto Indentation is a feature which, when RETURN is pressed, places the insertion cursor on the next line (as before) but under the first non-whitespace character of the previous line. This enables you to type neat C/C++ or assembler code faster as you don't have to type leading spaces or tabs yourself.

The figure below illustrates two examples. The first shows the effect of pressing Enter when the Auto Indentation feature is disabled - the insertion cursor returns to the left-hand side of the window on the next line. When the 'int z = 20' line is typed, it is not aligned with the previous two lines. The second example shows the effect of pressing Enter when Auto Indentation is enabled - the insertion cursor drops underneath the 'i' of the 'int' word on the previous line. Now, when the 'int z = 20' line is typed, it is automatically aligned (i.e. automatically indented).



(i) Auto Indentation Disabled

(ii) Auto Indentation Enabled

### To enable/disable Auto Indentation

- 1. Select [Setup -> Options]. The Options dialog box opens.
- 2. Select the **Editor** tab.
- 3. Set the Enable auto indentation checkbox. By default, this checkbox is selected.
- 4. Click the OK button for the new settings to take effect.

Enable auto indentation

#### 4.8 Splitting a window

The High-performance Embedded Workshop editor allows you to split an editor window into two. The split bar button is located just underneath the maximize button at the top right-hand corner of any editor window (as shown below).



### To split a window

Double-click the split bar button to split the window in half, or click on the split bar button, keep the button pressed, drag the mouse down and then release the mouse button at the point you want to split the window.



### To adjust the position of the split bar

Click on the split bar itself, keep the button pressed then move the bar to the new position and then release the button.

### To remove the split bar

Double-click on the split bar or move the split bar to the top or bottom of the window.

## 4.9 Changing the editor font

The High-performance Embedded Workshop allows you to specify the font to be used in its internal editor. All editor windows, regardless of the file type, use the same font.

### To change the editor font

- 1. Select [Setup -> Format Views]. The Format Views dialog box opens.
- 2. Select the **Source** icon in the tree. The **Font** tab should be available on the right of the **Format Views** dialog box.
- 3. Select the font type from the **Font** list.
- 4. Select the font size from the **Point Size** list.
- 5. When this is being modified the sample text below shows what the font will look like.
- 6. Click the OK button.

Font		
	Font	
	Courier New	
	Point Size	
	HEW Sample Text	

## 4.10 Syntax coloring

To enhance code readability, the High-performance Embedded Workshop editor can display specific strings (i.e. keywords) in different colors. For instance, C source code comments could be shown in green, and C types (e.g. int) could be shown in blue.

The coloring method used can be specified on a file group by file group basis. For example, you can define different color schemes for C source file, C header file, Assembly source file, or Assembly include file.



### Note:

When you create a new file, syntax coloring will not be active, as a new file does not initially have an extension (new files are named arbitrarily by the editor without an extension). In order to activate syntax coloring, you must save the new file with a name and extension that the High-performance Embedded Workshop recognizes. See section 2.5, File extensions and file groups, for information on file extensions.

### 4.10.1 Changing text colors

### To change existing colors

- 1. Select [Setup -> Format Views]. The Format Views dialog box opens.
- 2. Select the view you are interested in changing the font for. If it is the editor expand the "Source" in the tree on the left hand side of the dialog box.
- 3. Select the file type for which you want to edit syntax coloring from the File group list and then expand and select it.
- 4. Select the category are interested in. The tabs on the right side of the dialog box change depending on the selection.
- 5. Select the **Color** tab.
- 6. Modify the **Foreground** and **Background** color lists as desired. The **System** color refers to the current window foreground and background settings in Control Panel.
- 7. Click the OK button for the new color settings to take effect.



### 4.10.2 Creating new keywords

### To create new keyword groups

- 1. Select [Setup -> Format Views]. The Format Views dialog box opens. Expand the Source view icon in the tree.
- 2. Select the file type for which you want to create a new keyword group from the tree on this dialog box.
- 3. Click the Add button underneath the tree. The Add Category dialog box then opens. Enter the name of the group into the Category Title field. Click the OK button to add a keyword group. To modify the name of the group, select the keyword group and click the Modify button underneath the tree. Modify Category dialog box then opens. Enter the name of the group into the Category Title field. To remove a keyword group from the tree, select the keyword group and click the Remove button underneath the tree.

Add Category	? ×
Category Title:	ОК
	Cancel

### To create new keywords

- 1. Select [Setup -> Format Views]. The Format Views dialog box opens.
- 2. Select the desired keyword group to be modified.
- 3. Click the **Keywords** tab.



4. Click the **Add** button to add a keyword. Then the **Add Keyword** dialog box opens. Specify a keyword in the **Keyword** field and click the OK button to close the dialog box. To remove a keyword from the Keywords list, select the keyword and click the **Remove** button.



Add Keyword	? ×
Keyword:	OK
	Cancel

### Note:

On the **Keyword** field of the **Add Keyword** dialog box, specify a keyword which consists of only alphanumeric, an underscore, and the # character.

### 4.10.3 Enabling/disabling syntax coloring

### To enable/disable syntax coloring

- 1. Select [Setup -> Options]. The Options dialog box opens.
- 2. Select the **Editor** tab.
- 3. Set the **Enable syntax coloring** checkbox as necessary and then click the OK button.

Enable syntax coloring

## 4.11 Templates

When developing software it is often necessary to enter the same text repeatedly, for instance, when typing a function definition, for loop or a comment block for a function. The High-performance Embedded Workshop editor allows you to specify a block of text (or template) which can be inserted into the currently active Editor window. Thus, once a template has been defined, it can be automatically inserted without the need to re-enter it manually.

Figure below shows a list of templates, which is located on the **Templates** tab of the Workspace window.



Any new templates, which have been added to the High-performance Embedded Workshop, are displayed under the **Templates** folder. The **Toolchain Templates** folder is for templates, which are read only and have been provided for use in the High-performance Embedded Workshop system by the current toolchain.

Templates in this view can be dragged for insertion into an editor file. It is also possible to drag an area of text from the editor into the templates folder for quick template creation.



Right-clicking on the Templates folder opens a pop-up menu containing the available options.

Pop-up Menu Option	Function
Add	Adds a new template.
Remove	Removes the current selection.
Edit	Edits the current selection.

### 4.11.1 Defining a template

### To define a template

- 1. Select one of the following operations to display the **Define Templates** dialog box:
  - Click the **Define Template** toolbar button (**T**), **OR**
  - Select [Edit -> Templates -> Define Templates], OR
  - Right-click on the **Templates** in the Templates tab of the workspace window and select **Edit** from the pop-up menu, **OR**
  - Right-click in the editor window and select [Templates -> Define Templates] from the pop-up menu.

Define Templates	?	×
	ОК	
	Cancel	Ī
	<u>A</u> dd	
	<u>R</u> emove	
	User Input	
Shortcuts:	2	
	V	

- 2. Click the **Add** button. The **Add Template** dialog box is displayed, which asks you to enter your chosen template name. This name must be unique, otherwise a duplicated template name message will be displayed and the template will not be added.
- 3. If you want to modify an existing template use the **Template Name** drop-down menu to select which template you want to modify.
- 4. Enter the desired text into the **Template Text** area. You can copy text from another Editor window and then paste it into this dialog box using CTRL+V.
- 5. Enter the following placeholder to insert special information when the template is inserted:



Menu Entry	Placeholder	Replaced With
File path + name	\$(FULLFILE)	Filename (including full path)
File name	\$(FILENAME)	Filename (excluding path, including extension)
File leaf	\$(FILELEAF)	Filename (excluding path and extension)
Workspace name	\$(WORKSPNAME)	Workspace name
Project name	\$(PROJECTNAME)	Current project name
Line number	\$(LINE)	First line number of template insertion
Time	\$(TIME)	Current time
Date, text	\$(DATE_TEXT)	Current date in text form
Date, day/month/year	\$(DATE_DMY)	Current date in dd/mm/yy form
Date, month/day/year	\$(DATE_MDY)	Current date in mm/dd/yy form
Date, year/month/day	\$(DATE_YMD)	Current date in yy/mm/dd form
User name	\$(USER)	Current Windows® user
Cursor position	\$(CURSOR)	Insertion cursor - positions the cursor in this position
		after template has been inserted

6. Enter '\$(CURSOR)' to specify where the insertion cursor is to be placed after the template has been inserted. If this is not specified then the insertion cursor will be placed after the last character in the template, as in a normal paste operation.

7. There are 10 shortcut keys reserved for templates. If you want to designate one of these select the key in the drop-down list at the bottom of the edit template dialog box. These range from ALT+0 to ALT+9.

### User input

When defining a template it is possible to define a user input field. Using the following placeholder specifies this \$(USERINPUT<n:1-10>|"<some text>"). The 'n' is a number, which identifies the user input identifier.

These placeholders can be added manually but the **User Input** button on the **Define Templates** dialog box adds these placeholders in an automated manner. When the template is inserted into a file a dialog is displayed which allows you to enter some custom text for each of these fields. This text is then inserted instead of the placeholder. You can define ten of these user input fields.

## 4.11.2 Deleting a template

### To delete a template

- 1. Select one of the following operations to display the **Define Templates** dialog box:
  - Click on the **Define Template** toolbar button (**T**), **OR**
  - Select [Edit -> Templates -> Define Templates], OR
  - Right-click on the **Templates** in the Templates tab of the workspace window and select **Edit** from the pop-up menu, **OR**
  - Right-click in the editor window and select [Templates -> Define Templates] from the pop-up menu.



- 2. Use the **Template Name** drop-down list to select the name of the template you wish to remove and then click the **Remove** button.
- 3. Clicking the OK button saves the changes and closes the dialog box.

### To remove selected templates using the Templates tab of the Workspace window

- 1. Select the templates that you want to remove in the **Templates** tab of the Workspace window. Multiple templates can be selected by holding down the SHIFT or CTRL key.
- 2. Right-click on the **Templates** within the Templates tab of the workspace window and select **Remove** from the pop-up menu.

### 4.11.3 Inserting a template

### To insert a template

- 1. Select one of the following operations to display the **Insert Template** dialog box:
  - Click the **Insert Template** toolbar button (**T**), **OR**
  - Select [Edit -> Templates -> Insert Template], OR
  - Right-click in the editor window and select [**Templates** -> **Insert Templates**] from the pop-up menu.
- 2. Use the **Template Name** drop-down list to select the name of the template to be inserted, and then click the OK button. The dialog box is closed and the chosen template is added to the current Editor window.

Alternatively, you can press ALT along with the number of the template to be inserted (e.g. ALT+4 to insert template 4). You can define these shortcuts on the **Defile Templates** dialog box. A drop-down list is available at the bottom of the **Defile Templates** dialog box.

## 4.12 Brace matching

Complicated source code can often become unwieldy, especially when blocks of C/C++ code are deeply nested within each other, or when complex logic statements are expressed within a large 'if' clause. To help in such situations, the High-performance Embedded Workshop editor provides a **Brace Matching** feature, which highlights text between braces of type { }, ( ) and [ ].

### To find a matching brace

- 1. Either highlight the open brace to match from or place the cursor before it.
- 2. Select one of the following operations:
  - Click the Match Braces toolbar button ( {-}), OR



- Press CTRL+SHIFT+M, **OR**
- Select [Edit -> Match Braces], OR
- Select Match Braces from the pop-up menu.

To check the structure of an entire file, place the cursor at its start and then repeatedly invoke the match brace operation. The editor will successively highlight each pair of braces in turn until there are no more left to match.

## 4.13 Setting the read-only attribute for a file

While using the High-performance Embedded Workshop, you can set a file to be read-only. When you attempt to save a read-only file, a confirmation dialog box appears asking if you wish to save the file as another name.

### To set a file to be read-only

- 1. Select one of the following operations:
  - Right-click on a file in the **Projects** tab of the workspace window to open a pop-up menu. You can even select multiple files, **OR**
  - Right-click within the editor window to open a pop-up menu.
- 2. Selecting **Properties** opens the **Properties** dialog box.
- 3. Select the **Read-only** checkbox. If you have selected two or more files including both read-only and writable files, the checkbox is gray (intermediate state). By default, this checkbox is not selected.
- 4. The title bar of the editor window shows "Read-Only".

When a file is open within the editor window and if you switch on/off the read-only attribute of the file via the Windows® Explorer, it does not match the attribute shown on the title bar of the editor window (because the title bar is not updated). The attribute shown on the title bar is not updated until you start modifying the contents or re-open the file.

## 4.14 Preventing modification of files while debugging

Use this option if you wish to prevent modification of files while debugging with High-performance Embedded Workshop and a debugger connected. This prevents modification of all files that are open in the High-performance Embedded Workshop editor.

### To prevent modification of files while debugging

- 1. Select [**Setup -> Options**] to open the **Options** dialog box.
- 2. Select the **Editor** tab.



- 3. Selecting the **Do not allow file editing when target is connected** checkbox prevents modification of files while the High-performance Embedded Workshop is connected to a debugger. In other words, if the High-performance Embedded Workshop is not connected to a debugger, files can be modified. By default, this checkbox is not selected.
- 4. Click OK.

## 4.15 Managing the editor columns

The editor in High-performance Embedded Workshop has the ability to manage columns (apart from the main editor column). These can be added and used by any component in the High-performance Embedded Workshop system. Examples of this functionality might include a hardware breakpoint column added by the target, or possibly an address information column added by the debugger. The global column states feature is also accessible from the main edit menu.

### To switch off a column in all source files

- 1. Select one of the following operations to display the Global Editor Column States dialog box:
  - Select [Edit -> Define Column Format], OR
  - Right-click in the editor window and select **Define Column Format** from the pop-up menu.

Global Editor Column State	? ×
<ul> <li>✓ Line Number</li> <li>✓ S / W Breakpoints</li> <li>✓ S / W Breakpoints - ASM</li> <li>✓ S / W Breakpoints - ASM</li> <li>✓ Source Address</li> </ul>	Cancel

- 2. If the column's checkbox is checked then the column is enabled; if the column's checkbox is gray then this means that the column is enabled in some files, and disabled in others.
- 3. Click the OK button for the new column settings to take effect.

### To switch off a column in one source file

- 1. Open the editor window of the file which you wish to remove a column from.
- 2. Select one of the following operations:
  - Right-clicking the column header displays a pop-up menu. A tick mark right next to an entry indicates that this column is displayed. Clicking an entry will switch showing/hiding the column, **OR**



• Right-click in the editor window and select **Columns**. The cascaded menu option appears. Each column is displayed in this pop-up menu. If the column is enabled it will have a tick next to its name. Clicking on the entry will toggle whether the column is displayed or not.

You can adjust the column width by dragging the mouse on a column header.

## 4.16 Showing/hiding the column header

The editor window has a column header. You can select to show or hide the column header.

### To switch showing/hiding the column header

- 1. Right-click within a field in the editor window showing source or disassembly codes.
- 2. A pop-up menu opens. Select Turn Header On/Off.

## 4.17 Opening a file within the editor

The High-performance Embedded Workshop allows you to open a file within the editor window.

### To open a file shown in the editor window

1. Right-click on the file name in the **Source** field in the editor window.

Source			
#include #include	"type "stae	edefine.] cksct.h"	ά.
#define SR	Init	0x000	Open 'typedefine.h'

2. Select **Open '<file name>'** from the pop-up menu.

## 4.18 Tooltip watch

Use this function to know the value of a variable defined in the source program. Open the editor window or disassembly window (in source mode) to view the source program and rest the mouse cursor over the variable name that you want to examine. A tooltip (pop-up window) will appear showing the watch information.



### To use Tooltip watch

- 1. Select [**Setup -> Options**]. The **Options** dialog box opens.
- 2. Select the **Editor** tab.
- 3. Check the **Enable tooltip watch** check box.
- 4. Click OK.

Enable tooltip watch

### To view a tooltip watch on the editor window or disassembly window (in source mode)

- 1. Open the editor window or disassembly window (in source mode) showing the variable that you want to examine.
- 2. Rest the mouse cursor over the variable name that you want to examine. A tooltip will appear near the variable containing basic watch information for that variable.

S.,	Source
	sort(a);
	printf("*** Sorting results
	<pre>for( i=0; i&lt;10; i++ ){</pre>
	printf("a[%d]=%ld\n",i,a
	}
	min = a[0];
	$\max = \{(\log[10]) \{70000000\}\}$
	$\min = 0;$
	max = 0;
	change(a);
•	min = a[9];
	max = a[0];

## 4.19 Evaluate an expression

Launches the **Evaluate** dialog box allowing the user to enter a numeric expression, e.g. "205\*2", and display the result in all currently supported radices.

### To evaluate an expressions

- 1. Select [Edit -> Evaluate]. The Evaluate dialog box opens.
- 2. Enter the expression that you wish to evaluate and click the **Evaluate** button.



Provides a calculator function, evaluating simple and complex expressions, with parentheses and symbols. All operators have the same precedence but parentheses may be used to change the order of evaluation. The operators have the same meaning as in C/C++. Expressions can also be used in any command where a number is required. The result is displayed in all supported radix types.

### Valid operators:

Addition (+)	Subtraction (-)	Multiplication (*)	Division (/)
Logical AND (&&) *	Logical OR (  ) *	Logical NOT (!) *	Equal to (==)
Bitwise AND (&)	Bitwise OR ( )	Bitwise NOT (~)	Unequal to (!=)
Left arithmetic shift (<<)	Right arithmetic shift (>>)	Less than (<)	Greater than (>)
Modulo (%) *	Bitwise exclusive OR (^)	Less than or equal to (<=)	Greater than or equal to (>=)
NT 4			

Note:

\*. Support for this function depends on the debugger.

### **Register names:**

It can be useful to be able to use the value contained in a CPU register when you are entering a value. You can do this by specifying the register name prefixed by the '#' or '%' character, e.g.: '#PC' or '%PC'.

The supported prefix depends on the debugger.

### **Character Constants:**

Characters enclosed in single quote marks (') may be use as character constants. For example, 'A', etc. These character constants are converted to ASCII code and used as 1-byte immediate values.

### **Character String Literals:**

Character strings enclosed in double quote marks (") may be use as character string literals. Examples are "abc", etc.

### Note:

Support for this function depends on the debugger.



# 5. Tools Administration

You can control the components that are used by the High-performance Embedded Workshop via the **Tools** Administration dialog box, which is invoked by selecting [**Tools -> Administration**]. The **Tools Administration** dialog box is only accessible when no workspace is open.

Component	Version	<b></b>	Cancel
🔄 Toolchains			
🛠 H8S,H8/300 Standard Toolchain	6.2.1.0		
📲 🏂 Renesas M16C Standard Toolchain	5.44.00		
🛒 🕺 🕺 🕺 🕺 🕺 🕺 🕺 🕺 🕺 Standard Toolchain	5.41.01		
🖳 薞 SuperH RISC engine Standard Toolchain	9.2.0.0		<u>R</u> egister
🗠 🔄 System Tools			
🔤 🖬 Call Walker	1.07.00		<u>U</u> nregister
🦝 H Series Librarian Interface	1.03.00		Describer
- 🌆 Mapview	1.01.00		Properties
🔲 Ütility Phases			Europh
🖂 Debugger Components			Egport
🖂 🏹 740 Family CPU	1.03.00		<u>S</u> earch disk
- 🕅 Binary File Object Reader	1.05.00		<u>s</u> earch uisk
ElfDuisef2 Object reader	2 10 01		Tool information
Show <u>all components</u>			

There are five standard types of component:

Toolchain	A Toolchain is a set of build phases (e.g. compiler, assembler, linker and librarian). These components provide the build capability.
System Tool	A System Tool is an application (.EXE) that can be launched from the <b>Tools</b> menu. These are often provided as extra applications, which support the toolchain (e.g. an interactive graphical librarian, etc.).
Utility Phase	A Utility Phase is a 'ready made' build phase, which supports some specific build functionality (e.g. analyze complexity of source code, count lines of source code, etc.). These components provide added functionality to the build that is not toolchain-specific.
Debugger Component	A Debugger Component is a component that supports some specific debugger functionality (e.g. CPU DLL, Target platform, Object reader, etc.).
Extension Component	An Extension Component is a component that provides key functionality in a certain area of the High-performance Embedded Workshop system. These components cannot be unregistered when installed (e.g. The High-performance Embedded Workshop builder, debugger and flash support).


# 5.1 Tool locations

The High-performance Embedded Workshop maintains the locations of High-performance Embedded Workshop compatible components automatically as each new tool is installed. After installation, the High-performance Embedded Workshop stores information about the component (including its location). This is referred to as **Registration**. Although initial registration is automatic, during the course of development or if you want to manage the tools being used in your projects more effectively, you may need to register components yourself.

# 5.2 High-performance Embedded Workshop registration files

When a High-performance Embedded Workshop compatible component (i.e. toolchain, system tool or utility phase) is installed, part of its installation will include a file with the extension .HRF. This file describes the component to the High-performance Embedded Workshop (see the figure below).



The process of registration refers to loading a component's .HRF file into the **Tools Administration** dialog box (see the figure below).



In order to use a component with High-performance Embedded Workshop it must first be registered (see section 5.3, Registering a component, for further information). The **Tools Administration** dialog box shows all currently registered components. To access it, ensure no workspaces are open and then select [**Tools -> Administration**]. If you attempt to access the **Tools Administration** dialog box when a workspace is open, the **Tools Administration** dialog box is opened but cannot be modified. When High-performance Embedded Workshop is installed, any new tools are automatically



registered. Day to day usage of the High-performance Embedded Workshop though, may mean you need to know more about the tools registration process.

High-performance Embedded Workshop stores tool information in a tool database file, which is stored in the root of the tool installation directory. By default this is set to the High-performance Embedded Workshop application directory, however if you are working in a network environment this directory may be set to another location. It is possible to change the tool directory location and this causes a re-scan of the tools that are registered in High-performance Embedded Workshop.

#### To change the tools directory location

- 1. Select [Tools -> Administration]. The Tools Administration dialog box opens.
- 2. Click the **Modify** button, which is next to the **Current HEW tools location** field.
- 3. Browse to the root directory of the new tool location and click the OK button.
- 4. This will switch the directory and change the tool location to the new directory. It will be necessary to scan for any new tools that may be in this location. This can be achieved by using the Search Disk or Register Tool functionality.

# 5.3 Registering a component

The High-performance Embedded Workshop will automatically attempt to register any new components installed since the last time it was invoked. However, in some circumstances you may need to register components yourself.

In some cases it is useful to search a drive for High-performance Embedded Workshop-compatible components. This is especially useful if the High-performance Embedded Workshop installation was deleted or corrupted, as it can recreate your tool information instantly.

#### To search for components and register them

- 1. Click the **Search Disk** button on the **Tools Administration** dialog box. The **Search Disk for Components** dialog box opens.
- 2. Enter the directory in which you would like to search into the top field, or browse to it graphically by clicking the **Browse** button.
- 3. Check the **Include Subfolders** checkbox if you would like to search the directory specified and all directories below it.
- 4. Click the **Start** button to begin the search. During the search, the **Start** button will change to a **Stop** button. Click the **Stop** button to halt the search at any time.
- 5. The results of the search are shown in the **Located Components** list. Select a component and click the **Register** button to register an individual component, or click the **Register All** button to register all located components.
- 6. Click the **Close** button to exit the dialog box.



Search Disk for Cor	nponents	;	? ×
Select the directory in C:\Program Files\Rei Include subfolders	nesas\Hev		<u>S</u> tart Close
Located components:	:		
Component	Version	HRF Location	<u>R</u> egister
HewTargetServer	1.07.00	C:\Program Files\Renesas\Hew\Syst	
PdTargetServer	1.00.00	C:\Program Files\Renesas\Hew\Syst	Register <u>A</u> ll
Generic TCL Too	1.03.03	C:\Program Files\Renesas\Hew\Syst	
Call Walker	1.07.00	C:\Program Files\Renesas\Hew\Tool	
H8S,H8/300 Sta	6.2.1.0	C:\Program Files\Renesas\Hew\Tool	
H Series Libraria	1.03.00	C:\Program Files\Renesas\Hew\Tool	
Mapview	1.01.00	C:\Program Files\Renesas\Hew\Tool	
Renesas M32C	5.41.01	C:\Program Files\Renesas\Hew\Tool	
Renesas M16C	5.44.00	C:\Program Files\Renesas\Hew\Tool	
SuperH RISC en	9.1.1.0	C:\Program Files\Renesas\Hew\Tool	
•		•	
Search status: 99 fil	es found		

#### To register a single component

- 1. Click the **Register** button on the **Tools Administration** dialog box.
- 2. Browse to the component's .HRF file and click the **OK** button to register that component.

#### Note:

The High-performance Embedded Workshop registration file is located in the root directory of a component's installation.

## 5.4 Unregistering a component

The components that are registered with the High-performance Embedded Workshop affect the way it behaves. For example, every compatible system tool that is registered will be added to the **Tools** menu when a new project is created. Sometimes this may not be desirable. If this is not required, open the **Tools Administration** dialog box, select the undesired component from the **Registered Components** list and click the **Unregister** button. A dialog box will be invoked, which asks you to confirm this action. Click **Yes** to unregister the component.

#### Note:

Unregistering a component does not remove its installation from the hard disk. It simply removes the information that the High-performance Embedded Workshop was storing about that component (i.e. it 'disconnects' it from the High-performance Embedded Workshop). The action can be easily reversed at any time by registering the tool manually (see section 5.3, Registering a component).



# 5.5 Viewing and editing component properties

To view information regarding a component, select it from the **Registered Components** list on the **Tools Administration** dialog box and click the **Properties** button. The **Properties** dialog box opens.

The General tab displays the name, version and location of the selected component.

SuperH RISC	engine Standard Toolchain 9.2.0.0 Properties	? ×
General Info	rmation Environment	
Name :	SuperH RISC engine Standard Toolchain	
Version :	9.2.0.0	
Location :	C:\Program Files\Renesas\Hew\Tools\Renesas\Sh\9_2_0	
	OK Can	cel

Select the **Information** tab to view any information about the component. This may include copyright information, enhancements and so on.

SuperH RISC engine Standard Toolchain 9.2.0.0 Properties	? ×
General Information Environment	
Information:	1
High-performance Embedded Workshop	<u> </u>
SuperH RISC engine Standard Toolchain (V.9.2.0.0)	
SuperH RISC engine C/C++ Standard Library Generator (V.3.00.02) SuperH RISC engine C/C++ Compiler (V.9.02.00)	<b>_</b>
OK	Cancel

If there is an issue with the component and it is working incorrectly additional information is displayed here.

Select the **Environment** tab, if it exists, to view and edit a component's environment settings. This tab is most commonly used to modify the environment of a toolchain.



## High-performance Embedded Workshop V.4.09

SuperH RISC engine Standard Toolchain 9.2.0.0 Properties			
General Information Environment			
PATH=C:\Program Files\Renesas\Hew\Tools\Renesas\Sh\?         SHC_INC=C:\Program Files\Renesas\Hew\Tools\Renesas\S         SHC_LIB=C:\Program Files\Renesas\Hew\Tools\Renesas\S         SHC_TMP=C:\DOCUME~1\User\LOCALS~1\Temp         Rem	íy		
OK (	Cancel		

### To add a new environment variable

- 1. Click the Add button. The Environment Variable dialog opens.
- 2. Enter the variable name into the **Variable** field.
- 3. Enter the variable's value into the Value field
- 4. Click the OK button to add the new variable to the **Environment** tab.

## To modify an environment variable

- 1. Select the variable that you want to modify from the **Environment** tab.
- 2. Click the **Modify** button. The **Environment Variable** dialog box opens.
- 3. Make the required changes to the **Variable** and **Value** fields.
- 4. Click the OK button to modify the environment variable.

Environment Variable	? ×
Varia <u>b</u> le: PATH	OK Cancel
⊻alue: \$(TOOLDIR)\bin	Placeholder pop-up menu

#### To remove an environment variable

- 1. Select the variable that you want to remove from the **Environment** tab.
- 2. Click the **Remove** button.



#### Note:

Placeholder pop-up menus are included to ensure that the environment can be specified as flexibly as possible. For further information about using placeholders, see Reference 5, Placeholders.

# 5.6 Technical support

The **Tools Administration** dialog box is capable of displaying information regarding 'hidden' system components. These are part of the High-performance Embedded Workshop itself, and cannot be registered/unregistered manually. If you check the **Show all components** checkbox on the **Tools Administration** dialog, extra component folders are displayed (see the figure below).

📄 🗁 🔄 Project Generators	
H8S,H8/300 Project Generator	3.2.0
🔤 🔊 Renesas SH Project Generator	4.00.00
Project Generators - Custom	
📋 🖶 🔚 Version Control System	
Custom VCS	1.4.0.0
Show all components	

When seeking technical support, you may be asked to give details about some or all of these components. To do so, open the respective folder, select a component and click the **Properties** button. The **Properties** dialog box will be invoked.

The High-performance Embedded Workshop also has a feature that outputs tool information regarding the registered components to a file. This allows you to retrieve information about the entire High-performance Embedded Workshop system. This information can then be sent to your technical support contact if you are experiencing problems with the High-performance Embedded Workshop.

#### To output tool information

- 1. Select [Tools -> Administration]. The Tools Administration dialog box opens.
- 2. Click the Tool information button. A Save tool information file dialog box is displayed.
- 3. Choose the location of the output file and click the **OK** button.
- 4. A file is created in the chosen location with the current registered tool setup of the High-performance Embedded Workshop.

If any of the components have problems these can be seen in the tools administration dialog. If the icon has an additional icon this explains the problem. There are two additional icons that can be displayed.



If a component is found but cannot be used due to it being an old version or another dependent component is not available then the icon in figure below is used to show this.

🏹	H8S,H8/300 Series Simulator Target Platform	3.0
	HMon Embedded Monitor Platform	1.0
🕅	Intel Hex Record Object Reader	1.0

Component not found icon

If the component is not located where the registration file says it is then the icon in figure below is used to show this.



**Incompatible component found** icon

#### Note:

If the tool has one of these errors then it is possible to get more information by the following method:

#### To get tool error feedback

- 1. Select [Tools -> Administration]. The Tools Administration dialog box opens.
- 2. Select the tool in the list that has an issue.
- 3. Click **Properties**.
- 4. Select the information tab and scroll the edit field to the bottom.
- 5. The reason for the problem will be displayed in this area.

# 5.7 Using On-Demand components

The High-performance Embedded Workshop version 3.0 onwards has the concept of on-demand components. These components are not automatically loaded by the application or the debugger component. These components can be loaded by the user or as part of the project generation process.



#### To load or unload an on-demand component manually

- 1. Click [Project -> Components]. The Component Gallery dialog box opens.
- 2. Select the component you wish to load. Click the **Load** button. The components image should change to the loaded state.
- 3. If you wish to unload a component. Select the component. Click the **Unload** button. The components image should change to the unloaded state.
- 4. Click OK to verify the changes.



#### Note:

Each project in your workspace can have different components loaded and unloaded. If you have multiple projects you can use the "Multiple projects" and "All projects" items to change a components load status over more than one project. If you select a combination which means the component is loaded in one project and not anther then the intermediate state icon is displayed.

# 5.8 Custom project types

The [**Project -> Create Project Type**] menu option in High-performance Embedded Workshop and allows you to create a template for your project. This menu option takes the settings of the current project and then creates a project type for you. The user can specify the name of the new type and the style of the Project Generation Wizard. Once created, these project types appear in the **Tools Administration** dialog box and are initially hidden in the System Components part of the **Tools Administration** tree.

To export one of the Custom Project Generators, select the **Export** button on the **Tools Administration** dialog box. The export functionality packages the Custom Project Generator into a binary file, which includes an executable. This can be given to another user who then runs the executable. This installs the Project Generator into the correct location on the target user's machine.



#### Note:

A project template can be created by selecting [**Project -> Create Project Type**] based on the project in use. This template includes the version information of the toolchain. When a project is created by using this template after the toolchain version has been updated in your High-performance Embedded Workshop system, check that the toolchain version of the created project matches the using environment. When the registered toolchain can be updated, the toolchain version can be changed in the dialog box that is displayed by selecting [**Tools -> Change Toolchain Version**].

Change Toolchain	Version		? ×
Toolchain name: Current version: <u>C</u> PU Family: <u>T</u> oolchain: Toolchain <u>v</u> ersion:	Renesas SuperH RISC eng 9.1.0.0 SuperH RISC engine Renesas SuperH RISC eng 9.2.0.0	•	OK Cancel
Toolchain <u>b</u> uild pha Build phase OptLinker SH Assembler SH C/C++ Compile SH C/C++ Library	er	Version 9.04.00 7.01.02 9.02.00 3.00.02	Information



# 6. Customizing the Environment

# 6.1 Customizing the toolbars

The High-performance Embedded Workshop provides standard toolbars as detailed in the Toolbars topic. In addition to these, you may also construct your own toolbars.

Toolbars		
Ioolbars: ▼Editor ▼Search ▼Templates ♥Bookmarks ♥Debug ♥Standard ♥Version Control ♥Default Window ♥Debug Run ♥Debug Run ♥Debug Run ♥Debug Run ♥Debug Run	☑ <u>S</u> how Tooltips	<u>N</u> ew <u>R</u> eset Reset <u>P</u> ositions

## To create a new toolbar

- 1. Select [Setup -> Customize]. The Customize dialog box opens.
- 2. Select the **Toolbars** tab.
- 3. Click the New button. The New Toolbar dialog box opens.
- 4. Enter the name of the new toolbar into the **Toolbar Name** field.
- 5. Click the OK button to create the new toolbar.

#### Note:

When a new toolbar is created it will appear undocked (i.e. 'floating') and empty.

#### To add buttons to a toolbar

- 1. Select [Setup -> Customize]. The Customize dialog box opens.
- 2. Select the **Commands** tab.
- 3. Browse the available buttons by selecting the button categories from the **Categories** list. Select a button from the **Buttons** area to display information on its operation.
- 4. Click and drag a button from the dialog box onto the toolbar.



Command	8
Categories: Default Windc ▲ Search Editor Standard Debug Run Debug Bookmarks ▼	Buttons
Select a category, t to any toolbar	hen click a button to see its description. Drag the button
Description	

#### To remove buttons from a toolbar

- 1. Select [**Setup -> Customize**]. The **Customize** dialog box opens.
- 2. Select the **Commands** tab.
- 3. Click and drag a button from the toolbar onto the **Buttons** area.

#### To modify the name of a user-defined toolbar

- 1. Select [Setup -> Customize]. The Customize dialog box opens.
- 2. Select the **Toolbars** tab.
- 3. In the **Toolbars** list, select the user-defined toolbar and whose name you want to modify.
- 4. Modify the name of the toolbar in the **Toolbar Name** field.
- 5. Click the OK button to save the toolbar's new name.

#### To remove a user-defined toolbar

- 1. Select [Setup -> Customize]. The Customize dialog box opens.
- 2. Select the **Toolbars** tab.
- 3. Select the user-defined toolbar from the **Toolbars** list and the **Reset** button will change to a **Delete** button. Click the **Delete** button.

#### To reset a standard toolbar back to its original state

- 1. Select [Setup -> Customize]. The Customize dialog box opens.
- 2. Select the **Toolbars** tab.
- 3. Select the standard toolbar from the **Toolbars** list and then click the **Reset** button.

#### To reset the toolbar position back to its original state

- 1. Select [Setup -> Customize]. The Customize dialog box opens.
- 2. Select the **Toolbars** tab.
- 3. Click the **Reset Positions** button.

The **Reset Positions** button reverts all toolbars back to their original 'factory' default positions. Be careful as this will reset any custom positions that you may have setup during your session.

#### To show or hide toolbar tooltips

- 1. Select [Setup -> Customize]. The Customize dialog box opens.
- 2. Select the **Toolbars** tab.
- 3. Set the **Show Tooltips** checkbox as desired.

## 6.2 Customizing the Tools menu

The **Tools** menu can be customized to include your own menu options.

#### To add a new menu option

- 1. Select [Setup -> Customize]. The Customize dialog box.
- 2. Select the **Menu** tab. The first thing you need to decide is whether you are adding a global application-wide tool (which will be available to all of your workspaces), or whether you wish to add a workspace-wide tool (which is only valid for the current workspace). Once you have chosen, choose the relevant section of the dialog box.

Menu				
Application wide tools:				
Name	Version	<u>A</u> dd		
		<u>M</u> odify		
		<u>R</u> emove		
Workspace wide tools:				
Name	Version	A <u>d</u> d		
		M <u>o</u> dify		
		R <u>e</u> move		



- 3. Click the Add button. The Add Tool dialog box opens.
- 4. Enter the name of the tool into the **Name** field.
- 5. Enter the command, excluding arguments, into the **Command** field.
- 6. Enter any arguments that you would like to pass to the command into the Arguments field.
- 7. Enter the initial directory, in which you would like the tool to run, into the Initial directory field.
- 8. Click the OK button to add the menu option to the **Tools** menu.

Add Tool	? ×
Name:	ок 1
Notepad	
Command:	Cancel
C:\WINNT\NOTEPAD.EXE	
Arguments:	
\$(FULLFILE)	
Initial directory:	
Browse	

New menu options are added to the bottom of the list (i.e. bottom of the Tools menu).

#### To modify a menu option

- 1. Select [Setup -> Customize]. The Customize dialog box opens.
- 2. Select the Menu tab.
- 3. Select the menu option that you would like to modify and then click the **Modify** button.
- 4. Make the desired changes on the **Modify Tool** dialog box and then click the OK button.



Modify Tool	? ×
Name:	OK
Notepad	Cancel
Co <u>m</u> mand:	Cancer
C:\WINNT\NOTEPAD.EXE	
Arguments:	
\$(FULLFILE)	
Initial directory:	
Browse	

## To remove a menu option

- 1. Select [Setup -> Customize]. The Customize dialog box opens.
- 2. Select the Menu tab.
- 3. Select the menu option that you would like to remove and then click the **Remove** button.

When you have one or more tools added the **System Tools** toolbar will be available in High-performance Embedded Workshop. This will have one toolbar button for each tool. Clicking on a tools button will launch the tool exactly as if it had been selected from the Tools menu. For tools that have EXE commands, the button's image will be the same as the Windows<sup>®</sup> icon for the EXE. If no such image is available the default image  $(\square)$  will be used.

# 6.3 Using custom placeholders

Throughout the High-performance Embedded Workshop the user can use a number of pre-defined placeholders for directory definitions. For example, the user can use the "\$(PROJDIR)" variable to signify the current High-performance Embedded Workshop project directory. This makes it much easier to relocate projects and keep all of the paths correct.

The High-performance Embedded Workshop also has the ability to define custom placeholders. This means you can enter your own custom placeholder definition and decide upon its directory value. Once defined this placeholder becomes available throughout the rest of the High-performance Embedded Workshop system.

The placeholders can be defined on an application-wide level so the placeholders are available to all workspaces and projects that use the High-performance Embedded Workshop. The other method of defining the placeholders is using the workspace-wide custom placeholders. This means the placeholders can only be used in the current workspace. This list is only available when you have a workspace open.



	Placeholders	
Application wide custom placeh	olders:	
Placeholder Directory		<u>A</u> dd
		<u>M</u> odify
		<u>B</u> emove
<u>W</u> orkspace wide custom place	nolders:	
Placeholder Directory		A <u>d</u> d
		M <u>o</u> dify
		R <u>e</u> move

## To add a custom placeholder

- 1. Select [Setup -> Customize]. The Tools Customize dialog box opens.
- 2. Select the **Placeholders** tab.
- 3. Choose whether you need to use an application-wide or workspace-wide placeholder.
- 4. Click the Add button which is adjacent to the list you require. The Add New Placeholder dialog box opens.
- 5. In the fields provided choose a suitable name for the placeholder and a description of what the placeholder means.
- 6. Choose a directory, which relates to this placeholder.

## Note:

In High-performance Embedded Workshop V.2.01 or later version, the user-defined directory can be set as the custom placeholder, which can be used for setting the toolchain option. When the directory path is specified, specify an absolute path in 'Directory', as shown in figure below.

New Custom Placeholder	? ×
Placeholder:	ОК
USERDIR	Cancel
Description:	
User define directory	
Directory:	
C:\User\Src	<u>B</u> rowse



## 6.4 Using the workspace and project log facilities

The High-performance Embedded Workshop has workspace and project logging facilities integrated into the application. These facilities can be switched on via the log tab on the **Customize** dialog box. This option is especially useful when the network database is in operation. This is because user names and changes are logged to this file.

	Log
☑ <u>G</u> enerate build log:	
Path: \$(WORKSPDIR)\\$(WORKSPNAME).txt	<u>B</u> rowse
<ul> <li>Generate workspace log</li> <li>Generate log for projects</li> </ul>	

When the **Generate workspace log** check box is clicked any workspace changes will be logged to a file with the same name as the workspace with a ".log" extension. This file will be located in the same directory as the workspace file.

When the **Generate log for projects** check box is clicked any projects in the current workspace that have changes made to them will be logged to a file with the same name as the project with a ".log" extension. This file will be located in the same directory as the project file.

The log file is updated when the workspace is saved.

# 6.5 Configuring the help system

The High-performance Embedded Workshop provides context sensitive help within the **Editor** window. In other words, if you select some text in the **Editor** window and then press F1, the High-performance Embedded Workshop will attempt to locate help on that selected item. The help files that will be searched are listed in the **Help** tab of the **Setup Customize** dialog box.

#### To add a new help file

- 1. Select [Setup -> Customize]. The Setup Customize dialog box opens.
- 2. Select the **Help** tab.

		Help
<u>H</u> elp files:		
Description C/C++ Standard Library Help	File Name c:\hew\tools\renesas	Add
<u>D</u> efault help file:		
C/C++ Standard Library Help	<b>•</b>	



3. Click the Add button. The Add Help File dialog box opens.

Add Help File	? ×
Description:	ОК
Programing in ANSI C	
<u>File name:</u>	Cancel
C:\FILES\HELP\ANSI.HLP	<u>B</u> rowse

- 4. Enter a description of the help file into the **Description** field.
- 5. Enter the full path of the help file into the **File name** field (or browse to it graphically by clicking the **Browse** button).
- 6. Click the **OK** button to add the new help file to the list.

To make a help file the default choice, select it from the **Default Help File** drop-down list or set it to **None** if you would like to be prompted for a help file whenever F1 is pressed.

#### To remove a help file

- 1. Select [Setup -> Customize]. The Setup Customize dialog box opens.
- 2. Select the **Help** tab.
- 3. Select the help file to be removed and then click **Remove**.
- 4. Click the **OK** button to confirm the new help file settings.

## 6.6 Keyboard shortcut customization

The High-performance Embedded Workshop allows the keyboard shortcuts to be customized to your own preferences. This means that major operations can be configured to different keys especially useful if you are migrating from a different tool.

To reach the keyboard shortcut customization dialog click [Setup -> Customize]. Then when Customize dialog box is invoked click the Keyboard tab.



		Keyboard
Default keyboard Jayout High-performance Embedded Category:	I Workshop (default) 📃 💌	Re <u>s</u> et Assign
Commands: New Open Close New Workspace Save Workspace Close Workspace Close Workspace New Session Import Session Save Session Description:	Current <u>k</u> eyboard shortcut: None N <u>e</u> w keyboard shortcut: None Key currently assigned to:	<u>R</u> emove Import Export

This dialog box allows instant selection of either the default High-performance Embedded Workshop keyboard shortcuts or the PD debugger shortcuts. To change the entire keyboard layout select an item in the **Default keyboard layout** drop-down list. By default it uses the **High-performance Embedded Workshop** settings.

A number of operations are possible on this dialog box:

#### To add a new keyboard shortcut

- 1. Select the main menu category of the command you wish to modify. It is only possible to modify the commands that have a menu. Only some cases are special that allow modification, these are named global.
- 2. Select the command you wish to modify or add a new keyboard shortcut for in the **Commands** list. The current shortcut is displayed in the **Current keyboard shortcut** field.
- 3. Enter the new shortcut in the **New keyboard shortcut** field. Various combinations of buttons can be used here. For example "CTRL+J" or "CTRL+SHIFT+O", etc. If the chosen shortcut is already in use it is displayed below the New keyboard shortcut field.
- 4. If you are happy with your selection click the **Assign** button.
- 5. Changes are not saved until the OK button is clicked on the Customize dialog box.

#### To remove a keyboard shortcut

- 1. Select the main menu category of the command you wish to modify. It is only possible to modify the commands that have a menu. Only some cases are special that allow modification, these are named global.
- 2. Select the command you wish to modify or add a new keyboard shortcut for in the **Commands** list. The current shortcut is displayed in the **Current keyboard shortcut** field.
- 3. Click the **Remove** button.



4. Changes are not saved until the OK button is clicked on the Customize dialog box.

#### To reset all the keyboard shortcuts

- 1. Click the **Reset** button. All shortcuts revert to the default settings for the currently selected keyboard layout.
- 2. Changes are not saved until the OK button is clicked on the **Customize** dialog box.

The keyboard shortcuts dialog box allows you to import and export keyboard settings to a defined file. This allows you to easily transfer settings from one machine to another.

#### To import keyboard shortcuts

- 1. Select the keyboard layout you wish to replace with your imported settings in the **Default keyboard layout** drop-down list.
- 2. Click the **Import** button. A standard file dialog box opens.
- 3. Choose the filename to load the keyboard layout from.
- 4. Click OK.

#### To export keyboard shortcuts

- 1. Click the **Export** button. A standard file dialog is displayed.
- 2. Choose the filename to save the settings of the currently selected keyboard layout to.
- 3. Click OK.

## 6.7 Scope of a control in the setup

#### 6.7.1 Scope of a control in the Customize dialog box

The scope of each control in the **Customize** dialog box, which is launched via [**Setup -> Customize**], differs. This can be confusing so these have been listed below:

Tab	Control	Scope
Toolbar	All	Each workspace
Command	All	The whole system
Menu	Application wide tools	The whole system
	Workspace wide tools	Each workspace
Placeholders	Application wide custom placeholder	The whole system
	Workspace wide placeholder	Each workspace



High-performance Embedded Workshop V.4.09

Tab	Control	Scope
Debugger	Debugger tool	Each project
	Debugger location	Default: whole system and each project basis
	Command line options	Each project
	Session file	Each project
	Download module	Each project
Log	All	Each workspace
Help	All	Each workspace
Keyboard	All	The whole system

## 6.7.2 Scope of a control in the Options dialog box

Scope of each control of each tab of the **Options** dialog box, which is launched via [**Setup -> Options**], affects the whole system.

## 6.8 Specifying workspace options

The High-performance Embedded Workshop allows you to control several aspects of a workspace via the **Options** dialog box. To invoke it select [**Setup -> Options**], and select the **Workspace** tab.

### 6.8.1 Opening the last workspace at start-up

When you exit the High-performance Embedded Workshop, the last workspace you had open is stored. On subsequently launching the High-performance Embedded Workshop, you may want the last workspace to be opened automatically.

#### To open the last workspace at start-up

- 1. Select [Setup -> Options]. The Options dialog box opens.
- 2. Select the **Workspace** tab.
- 3. Select the **Open last workspace at start-up** checkbox if you would like the High-performance Embedded Workshop to automatically open the last workspace when it is launched.
- 4. Click the OK button.

☑ Open last workspace at start-up

## 6.8.2 Restoring files on opening a workspace

When you close a workspace, the High-performance Embedded Workshop stores the names of the files that were open at that time. When you open a workspace, the High-performance Embedded Workshop can restore (i.e. open) the same files so that you can continue your session in exactly the same state as when you left it. If you would like the files associated with a workspace to be opened when the workspace is opened, then set this checkbox.



#### To restore files on opening a workspace

- 1. Select [**Setup -> Options**]. The **Options** dialog box opens.
- 2. Select the **Workspace** tab.
- 3. Select the **Restore files on opening workspace** checkbox if you would like the files associated with a workspace to be opened when the workspace is opened.
- 4. Click the OK button.

Restore files on opening workspace

#### 6.8.3 Displaying workspace information on opening a workspace

When many workspaces are being used, it is sometimes difficult to remember exactly what is contained within each workspace. The High-performance Embedded Workshop allows you to enter a textual description of each workspace. This description can be displayed whenever a workspace is opened.

#### To enter a workspace description

- 1. Select the workspace icon from the Projects tab of the workspace window.
- 2. Right-click to invoke the pop-up menu and then select **Properties**. The **Workspace Properties** dialog box opens.
- 3. Enter the description into the **Information** field.
- 4. Select the **Show workspace information on workspace open** checkbox if you want a **Workspace Properties** dialog box to be launched on opening a workspace. This checkbox has the same role as the **Display workspace information dialog on opening workspace** checkbox on the **Workspace** tab of the **Options** dialog box.
- 5. Click the OK button.

Display workspace information dialog on opening workspace

### 6.8.4 Saving the workspace before executing any tools

It is possible to force the High-performance Embedded Workshop into saving the current workspace before executing any build phases (i.e. build, build all or build file operations) or version control commands.

#### To save the workspace before executing any tools

- 1. Select [Setup -> Options]. The Options dialog box opens.
- 2. Select the **Workspace** tab.



- 3. Select the **Save workspace before executing any tools** checkbox.
- 4. Click the OK button.

Save workspace before executing any tools

#### 6.8.5 Prompting before saving a workspace

If you are using the **Save workspace before executing any phases** function, you may want the High-performance Embedded Workshop to prompt you before saving the workspace. For further information about saving a workspace before executing phases, see section 6.8.4, Save workspace before executing any phases.

#### To display a prompt before saving the workspace

- 1. Select [Setup -> Options]. The Options dialog box opens.
- 2. Select the **Workspace** tab.
- 3. Select the **Prompt before saving workspace** checkbox.
- 4. Click the OK button.

Prompt before saving workspace

#### 6.8.6 Prompting before saving a session

Checking this option will force the High-performance Embedded Workshop into displaying a prompt before the session is saved to disk.

#### To display a prompt before saving the session

- 1. Select [Setup -> Options]. The Options dialog box opens.
- 2. Select the **Workspace** tab.
- 3. Select the **Prompt before saving session** checkbox.
- 4. Click the OK button.

✓ Prompt before saving session



#### 6.8.7 Enabling auto-backup facilities

The High-performance Embedded Workshop supports the facility to backup the workspace, project and session files at regular intervals. This means that if your application or development system should fail you will not lose so much work. Any changes you have made will be saved to temporary files.

When re-opening the workspace you will be prompted and asked if you wish to auto-recover the files that were not saved during your last session.

#### To enable auto-file backup facilities

- 1. Select [Setup -> Options]. The Options dialog box opens.
- 2. Select the **Workspace** tab.
- 3. Select the **Save Auto-Recovery info** checkbox as necessary.
- 4. Select the number of minutes you wish the auto-backup facility to be launched.
- 5. Click the OK button.

Save Auto-Recovery info every	10 🗧 minutes	
-------------------------------	--------------	--

#### **Restoring your files**

If you open your workspace and the following dialog is displayed it means that the last time the workspace was used problems were encountered.

HEW Workspace Auto-R	ecovery	? ×
HEW failed to shutdown workspace. Would you use the saved versions	like to recover the follo	e you used this wing files, rather than
Filename	Directory	
demo.hws	c:\demo	
🗹 demo.hwp	c:\demo\demo	
defaultsession.hsf	c:\demo\demo	
☑ simsessionsh-3	c:\demo\demo	
•		Þ
	OK Cance	1

To recover the files check the checkbox alongside the filenames you wish to recover and click OK. Clicking Cancel will discard the auto-recovery files and load from the original files.

#### 6.8.8 Setting the projects to load on workspace open

You can configure which projects are initially loaded on workspace open. This is an application wide setting which is only relevant to workspaces that contain multiple projects.

#### To change the projects to load on workspace open

- 1. Select [Setup -> Options]. The Options dialog box opens.
- 2. Click the **Workspace** tab.

Project load on workspace open:
Last used current project
Last used current project
All projects Last loaded projects
Last loaded projects

- 3. To determine the action when a workspace is opened, change the **Project load on workspace open** drop list to one of the following options:
- Last used current project Only the current project when you last closed the workspace will be loaded.
- All projects All projects belonging to the open workspace will be loaded.
- Last loaded projects Any project which was loaded when you last closed the workspace will be loaded.
- 4. Click the OK button.

#### 6.8.9 Specifying a default directory for new workspaces

When a new workspace is created, the High-performance Embedded Workshop invokes the **New Project Workspace** dialog box. One of the fields on this dialog is the directory in which the new workspace will be created. By default, this is the Workspace directory. However, it is also possible to set this default directory to another location.

#### To change the default directory for new workspaces

- 1. Select [Setup -> Options]. The Options dialog box opens.
- 2. Select the **Workspace** tab.
- 3. Enter the directory in which to create new workspaces into the **Default directory for new workspaces** field, or browse to it graphically by clicking the **Browse** button.
- 4. Click the OK button.



Default directory for <u>n</u> ew workspaces:	
C:\Workspace	<u>B</u> rowse
·	

# 6.9 Using an external editor

The High-performance Embedded Workshop allows you to use an external editor. Once an external editor has been specified, it will be launched when the following actions are performed:

- Selecting a file in the **Open File** dialog box opened by selecting [File -> Open].
- Selecting [File -> Recent Files].
- Double-clicking on a file in the **Projects** tab of the workspace window.
- Double-clicking on an entry in the **Navigation** tab of the workspace window.
- Double-clicking on an error/warning in the **Build** tab of the output window.
- Double-clicking on an entry in the **Find in Files 1** tab of the output window.
- Double-clicking on an entry in the **Find in Files 2** tab of the output window.
- Right-clicking on a file within the Projects tab of the workspace window and selecting the **Open** <**file> in external editor** option from the pop-up menu.
- Clicking the Edit Code button in the Breakpoints dialog box opened by selecting [Edit -> Source Breakpoints].
- Drag and drop into the High-performance Embedded Workshop window: When a file is dragged from the Windows® Explorer and dropped into the High-performance Embedded Workshop window (note that, however, a drag and drop of a file onto the current project and user folder into the **Projects** tab of the workspace window only adds this file to the project and does not open the file.)

#### To specify an external editor

External editor:	
Use <u>external editor</u>	
	<u>M</u> odify

- 1. Select [Setup -> Options]. The Options dialog box opens.
- 2. Select the **Editor** tab.
- 3. Check the Use external editor checkbox. The External Editor dialog box opens.



External Editor	? ×
Command: C:\WINNT\NOTEPAD.EXE Browse	ОК
Arguments to open file:	Cancel
\$(FULLFILE)	<u>A</u> dvanced >>
Arguments to open file at line:	
\$(FULLFILE)	

- 4. Enter the path of the executable (without any arguments) into the **Command** field.
- 5. Enter the arguments required to open a file into the **Arguments to open file** field. Use the \$(FULLFILE) placeholder to represent the path of the file to be opened.
- 6. Enter the arguments required to open a file at a specific line into **Arguments to open file at line** field. Use the \$(FULLFILE) placeholder to represent the path of the file to be opened and the \$(LINE) placeholder to represent the line number at which the cursor should be initially positioned.
- 7. Clicking the **Advanced**>> button allows you to select the external or internal (High-performance Embedded Workshop) editor for use when you attempt to open a file by any of the following operations.
- File open menu

When a file is selected in the **Open File** dialog box.

By default, the internal editor is selected.

• Workspace window double click

When a file in the Projects tab or an entry in the Navigation tab is double-clicked in the workspace window

By default, the external editor is selected.

• Output window double click

When an error or warning in the Build tab or an entry in the Find in Files 1 or Find in Files 2 tab is doubleclicked in the output window

By default, the external editor is selected.

• Drag and drop file into HEW main window

When a file is dragged from the Windows® Explorer and dropped into the High-performance Embedded Workshop window (note that, however, a drag and drop of a file onto the current project and user folder into the **Projects** tab of the workspace window only adds this file to the project and does not open the file.)

By default, the internal editor is selected.

8. Click the OK button.



When the Use external editor checkbox is selected, the following actions always select the external editor for use.

- Clicking the Edit Code button in the Breakpoints dialog box opened by selecting [Edit -> Source Breakpoints].
- Selecting [File -> Recent Files].

If you select a file in the Projects tab of the workspace window and right-click to display a pop-up menu, the **Open <file name> in external editor** menu option is available under the **Open <file name>** menu option.

#### Note:

When using an external editor be aware of the following issues:

- Each time you invoke the external editor, in whichever way, a separate instance of the editor will be launched.
- You must save your own files before you perform a Build, Build All or Build File operation.

# 6.10 Customizing the font in your views

The High-performance Embedded Workshop contains many components which you may wish to make look differently. It is possible to change the font and text coloring for the views.

#### To change the look of your windows

- 1. Select [Setup -> Format Views]. The Format Views dialog box opens.
- 2. Select the view you are interested in changing the appearance of. To change all views select the "All views" category.
- 3. Expand the item in the tree to see all items you can change the look of.
- 4. Select the item. Notice the tab changes on the right of this dialog box.
- 5. Change the font or text color.
- 6. Click OK to save the changes and the views will be automatically updated with the new coloring.



Format Views	? ×
All views   Selected Text   Text   Text   Source   Source   Stack Trace   Stack Trace   Text   Text   Text   Source   Stack Trace   Text   Text <tr< td=""><td></td></tr<>	
E Difference     HEW Sample Text     Hew sample Text     From R Output	
<u>A</u> dd <u>M</u> odify <u>R</u> emove OK Cance	el

The Format Views dialog box allows you to customize the color and font of text shown in the windows.

Item Category Objects to be Customized		Objects to be Customized
Source	Selected Text	A selected text shown in the editor window and disassembly window (in source mode)
	PC Line Highlight	The line at the PC (program counter) in the editor window and disassembly window (in source mode) while the <b>Enable line highlight for PC position</b> checkbox is selected in the Editor tab of the Options dialog box
	Other keyword groups based on the syntax of respective file types	Comment or keywords* shown in the editor window and disassembly window (in source mode) Note:
		Control, Declaration, Operator, PreProcessor, Value, Directive
Difference	LHS Different Lines	Unmatched text lines shown in the left pane in the Difference window
	LHS Moved Lines	Moved text lines shown in the left pane in the Difference window
	RHS Different Lines	Unmatched text lines shown in the right pane in the Difference window
	RHS Moved lines	Moved text lines shown in the right pane in the Difference window
Register	Modified	Text sections of changed values shown in the Register window
Memory Accessed		Text sections of executed codes at coverage measurement shown in the Memory window
	Not Accessed	Text sections of codes not executed at coverage measurement shown in the Memory window
	Unknown	Text outside the coverage range shown in the Memory window
	No Memory	Text outside the memory range shown in the Memory window
	Modified	Text sections of changed values shown in the Memory window
Output	Build Error	Lines of error messages shown in the Build tab of the Output window
	Build Warning	Lines of warning messages shown in the Build tab of the Output window
	Information	Lines of information messages shown in the Build tab of the Output window



Item	Category	Objects to be Customized	
Disassembly	PC Line Highlight	The line at the PC (program counter) in the editor window and	
		disassembly window (in mixed mode or disassembly mode) or	
		Disassembly window while the Enable line highlight for PC posit	
		checkbox is selected in the Editor tab of the Options dialog box	
		Source codes shown in the editor window and disassembly window (in	
		mixed mode)	
All	Text	Text shown in the windows	

# 6.11 Using the virtual desktop

High-performance Embedded Workshop has implemented the concept of the virtual desktop. This allows window configurations to be defined that can be switched with the click of a button. When a particular button is clicked the windows are hidden or displayed depending on the current settings of that window configuration.

It is possible to have a maximum of 4 desktop configurations in use. When the session is saved the window positions for each configuration are saved to the session file. You can then switch simply between each configuration to gain access to the other windows. The toolbars and windows are dependent on the virtual desktop configuration. Source files are independent of the virtual desktop system and will remain in view.

#### To rename your configuration to a more meaningful name

- 1. Select [Window -> Virtual desktop -> Desktop Manager]. The Desktop manager dialog box opens.
- 2. Select the window configuration you wish to change the name for.
- 3. Click Rename.
- 4. Enter the new meaningful name in the edit field and click OK.
- 5. Click OK to keep the changes and revert to the High-performance Embedded Workshop main window.

#### To switch desktop configurations

There are a number of ways to switch desktop configuration. The first and easiest method is using the virtual desktop buttons located on the status bar. These are shown below.

1			
	Build desktop	Read-write	25/87 //
1			

In this example the selected desktop is number 1. This has been given the name "Build" by the user. Its description is seen in the edit box to the right on the buttons. Clicking a different desktop selects that button and changes the description control. Once clicked High-performance Embedded Workshop then loads the windows in the new configurations style.



Another method of changing the desktop configuration is as follows:

- 1. Select [Window -> Virtual Desktop].
- 2. Select is cascaded menu.
- 3. Then select the desktop configuration you wish to view on this menu. The selected option is ticked.



# 7. Version Control

The High-performance Embedded Workshop provides facilities for connecting to version control system. Some of the reasons why version control systems are used with a project are:

- To maintain the integrity of a project.
- To store each stage of a project.
- To enable different users to co-develop a project by controlling revisions to its source files.

Figure below illustrates a typical project where a version control system is in use. This shows three users who all use the same-shared network drive to exchange source code. The version control system provides access and updates to the source files.





#### • Files Supported for Version Control by the High-performance Embedded Workshop

Version control by the High-performance Embedded Workshop basically supports the following types of files.

File type	Extension Group	File Extension
Workspace file	HEW Workspace	hws
Project file	HEW Projects	hwp
File for version control	Configuration file	hvc
Source file	C source file	С
	C++ source file *	срр
	Assembly source file	a30, ms, a74, asm, mar, src *
Include file	C header file	h
	C++ header file *	hpp
	Assembly include file	inc

#### Note:

\*. Support for this file type depends on the toolchain.

Other types of files can also be supported once they are added to the project.

## • Version Control System that can be Connected with the High-performance Embedded Workshop

High-performance Embedded Workshop does not provide the version control system itself. It must be installed in your PC in advance. The High-performance Embedded Workshop can be connected to the version control system via the GUI interface.

First, select a version control system, which you wish to connect with the High-performance Embedded Workshop. Either of the followings can be selected:

Туре	Version Control System Name Description Displayed on Screen	
Version control by custom settings	Custom VCS	Any version control system in which command- line commands should be executed via the MS- DOS command prompt (such as RCS, CVS, or PVCS). The commands in the version control system must be defined as being associated with the GUI interface of the High-performance Embedded Workshop.
Version control by Microsoft® Visua SourceSafe 5.0, 6.0 and 2005	I Microsoft Visual SourceSafe	Microsoft® Visual SourceSafe 5.0, 6.0 and 2005. Main features are already defined.



# 7.1 Selecting a Version Control System

The [Tools -> Version Control] sub-menu contains eight menu options but only the [Tools -> Version Control -> Select] option is initially available. This is because a version control system is not yet active for the current workspace.

<u>S</u> elect
Configure
💼 Add to VCS
🔁 Bemove from VCS
🐣 Get from VCS
Check out from VCS
🚊 Check in to VCS
🙆 Get VCS status

#### To select a version control system

1. Select [Tools -> Version Control -> Select]. The Select Version Control System dialog box opens, which lists all of the supported version control systems.

Select Version Control System	? ×	
Version control system:	OK	
Microsoft Visual SourceSafe	Cancel	

- 2. Select the desired version control system from the Version control system list.
- 3. Click on the **OK** button.
- Once the "Custom VCS" is selected as a version control system, the [Tools -> Version Control -> Configure] option will become available. See Chapter 8, Custom Version Control System.
- Once "Microsoft Visual SourceSafe" is selected as a version control system, other options of the [Tools -> Version Control -> Select] option will become available. See Chapter 9, Visual SourceSafe Version Control System.

#### To deselect a version control system

- 1. Select [Tools -> Version Control -> Select]. The Select Version Control System dialog box opens.
- 2. Select "<None>" from the Version control systems list.
- 3. Click on the **OK** button.



# 7.2 Importing and exporting a set-up

Each workspace can have a different version control setup. The High-performance Embedded Workshop allows you to store the version control settings independently so that you can import them into other workspaces. This greatly reduces the amount of time it takes to configure the same version control settings across several workspaces.

#### To export a version control setup

- 1. Select [Tools -> Version Control -> Configure]. The Version Control Setup dialog box opens.
- 2. Click the **Export** button. An **Export current Version Control configuration** dialog box opens.
- 3. Browse to the directory in which you would like to save the configuration.
- 4. Enter the name of the file and then click the **Save** button.

### To import a version control setup

- 1. Select [Tools -> Version Control -> Configure] menu option. The Version Control Setup dialog box opens.
- 2. Click the Import button. An Import a Version Control configuration dialog box opens.
- 3. Browse to the \*.HVC file that you would like to import.
- 4. Select the file and then click the **Import** button.



# 8. Custom Version Control System

If you have selected "Custom VCS" as the version control system in section 7.1, Select a Version Control System, the following definitions are necessary for connection of the High-performance Embedded Workshop and Custom VCS.

- Version control menu options and locations of the associated command executables (.EXE), command parameters, how to control the execution result of version control commands, etc.
- Locations of files for version control (directory mapping) and global variables
- Execution control of version control commands, user settings, and other general options

After defining this information, you can execute a command of the custom version control system by selecting a Highperformance Embedded Workshop menu option or toolbar button and view the result in the High-performance Embedded Workshop.

For installation and setting of the version control system, refer to the user's manual for respective version control systems.

For details on operations of the High-performance Embedded Workshop with the custom version control system, see section 8.11, Usage example of the Custom Version Control System.

# 8.1 Defining Version Control menu options

The custom version control system allows you to invoke a version control command, either by selecting an option from the [**Tools -> Version Control**] sub-menu, or by clicking a version control toolbar button. When either of these actions is performed, the associated commands are executed and the output is displayed in the **Version Control** tab of the output window.

## To execute a command by a version-control menu option or toolbar button

- 1. Select the items to which you want to apply the version control command, from the Projects tab of the workspace window. This may include a workspace, projects, folders and files. Right-click to invoke a pop-up menu. You can select a menu option you have defined from the **Version Control** submenu. A command associated with this menu option will be executed on the files contained in the workspace, project, or folder, or file itself selected in the workspace window.
- 2. For example, if you select the workspace icon, all of the files in all of the projects will be passed to the version control command (this will include any High-performance Embedded Workshop system files).
- 3. Select the required menu option from the [**Tools -> Version Control**] sub-menu or click the desired version control toolbar button.

The custom version control support allows you the highest degree of flexibility in specifying how a version control system is to be used. To configure it, select [**Tools -> Version Control -> Configure**]. The **Version Control Setup** dialog box opens.



Commands Projects General	
System menu options and toolbar buttons:	
Add to VCS	Modify
Remove from VCS	
Get from VCS	
Check out from VCS	
Check in to VCS	
Get VCS status	
User menu <u>o</u> ptions:	
	<u>A</u> dd
	<u>M</u> odify
	<u>R</u> emove
	Move <u>u</u> p
	Move <u>d</u> own

The **Commands** tab has two lists of menu options.

• System-defined menu options (System menu options and toolbar buttons)

**System menu options and toolbar buttons** lists the menu options always shown as the [**Tools -> Version Control**] sub-menus. These menu options also have an associated toolbar button on the **Version Control** toolbar. This contains the six menu options associated with the most frequently used version control commands.

• User-defined menu options (User menu options)

**User menu options** lists user-defined menu options. Menu options defined in this list will be added to the end of [**Tools -> Version Control**] sub-menus. If you wish to add menu options not included in the system menu options, they must be defined in this list. User-defined menu options are not shown on the version control toolbar.

## 8.1.1 System-defined menu options

There are six version control toolbar buttons. They provide you with a shortcut to the most commonly used version control commands. Initially, when you first create a workspace, these toolbar buttons are inactive because you have not yet associated any version control commands to them.

The toolbar buttons are equivalent to the six menu options on the [Tools -> Version Control] sub-menu. In other words, selecting [Tools -> Version Control -> Get from VCS] will have exactly the same effect as clicking the Get from VCS toolbar button. As the toolbar buttons themselves are fixed, the only operation that you can perform upon them is to define which commands should be executed when they are clicked.

In order to invoke commands from the toolbar or the system defined options of the [**Tools -> Version Control**] submenu, you must first define the associated commands that should be executed when they are activated. The names of the options and their intended action are listed.


Option	Description
Add to VCS	Add selected files to version control system.
Remove from VCS	Remove selected files from version control system.
Get from VCS	Get a read only local copy of the selected files from version control system.
Check out from VCS	Get a writable local copy of the selected files from version control system.
Check in to VCS	Put back, i.e. update, the selected files in version control system with the local copy.
Get VCS status	View the status of the selected files.

#### To define a command to be executed via the system menu or toolbar button

- 1. Select [Tools -> Version Control -> Configure]. The Version Control Setup dialog box opens.
- 2. Select a menu option in which you wish to define a command from **System menu options and toolbar buttons** list and click the **Modify** button. The **Define Commands** dialog box opens.
- 3. Click the **Add** button to define a command in the selected menu option. See section 8.2, Defining Version Control commands, for further information.
- 4. Close the **Define Commands** dialog box by clicking the OK button.
- 5. Close the Version Control Setup dialog box by clicking the OK button.

## 8.1.2 User-defined menu options

You can create as many user-defined menu options as you like, name them how you want and define their order in the menu.

If you have already created user-defined menu options, definitions of the commands can be modified or deleted. It is also possible to change the order of menu options being displayed.

User-defined menu options do not appear on the version control toolbar.

#### To create a new user-defined version control menu option

- 1. Select [Tools -> Version Control -> Configure]. The Version Control Setup dialog box opens.
- 2. Click the Add button. The Add Menu Option dialog box opens.
- 3. Enter the name of the menu option into the **Option** field.
- 4. Click the **Add** button. The **Add Command** dialog box opens. For details, see section 8.2, Defining Version Control commands.
- 5. Click the OK button to close the **Add Command** dialog box.
- 6. Click the OK button to close the Add Menu Option dialog box.
- 7. Click the OK button to close the **Version Control Setup** dialog box.



#### To modify a user-defined version control menu option

- 1. Select [Tools -> Version Control -> Configure]. The Version Control Setup dialog box opens.
- 2. Select the menu option to be modified from the **User menu options** list and then click the **Modify** button. The **Define Commands** dialog box opens.
- 3. Select the menu option to be modified from the **Commands** list and then click the **Modify** button. The **Modify Commands** dialog box opens. For details, see section 8.2, Defining Version Control commands.
- 4. Click the OK button to close the **Modify Commands** dialog box.
- 5. Click the OK button to close the **Define Commands** dialog box.
- 6. Click the OK button to close the Version Control Setup dialog box.

#### To remove a user-defined version control menu option

- 1. Select [Tools -> Version Control -> Configure]. The Version Control Setup dialog box opens.
- 2. Select the menu option to be removed from the [User menu options list and click the Remove button.
- 3. Close the Version Control Setup dialog box by clicking the OK button.

#### To change the ordering of user-defined version control menu options

- 1. Select [Tools -> Version Control -> Configure]. The Version Control Setup dialog box opens.
- 2. Select the menu option to be moved and then click the **Move up** and **Move down** buttons as necessary.
- 3. Close the Version Control Setup dialog box by clicking the OK button.

# 8.2 Defining Version Control commands

Version control commands are listed in the **Define Commands** dialog box. You can define as many commands as you want to and specify the order in which they execute. Existing commands can be modified or removed.

#### To define a new version control command

1. Click the Add button on the Define Commands dialog box. The Add Command dialog box opens.



Add Command	? ×
Command : Version control executable : V:\Vss\WINNT\Ss.exe Arguments : History \$(VCDIR)\\$(FILENAME) -Y\$(USERNAME), \$(P)	OK Cancel
Initial directory : \$(FILEDIR) Browse	
Executable return code :	

- 2. Enter the full path of the command into the **Version control executable** field, or browse to it graphically by clicking the **Browse** button.
- 3. Enter the arguments for the command into the Arguments field.
- 4. Enter into **Initial directory** the directory from which you would like to run the executable or browse to it graphically by clicking **Browse**. In most cases this should be set to the \$(FILEDIR) placeholder, which means that the command should be executed from the same directory as the file.
- 5. Set the Executable return code options as appropriate (see section 8.5, Executable return code).
- 6. Click the OK button to define the new command.

#### To modify a version control command

- 1. Select the command to be modified from the **Commands** list of the **Define Commands** dialog box.
- 2. Click the **Modify** button. The **Modify Command** dialog box will be displayed.
- 3. Modify the information as necessary and then click the OK button.

#### To remove a version control command

- 1. Select the command to be removed from the **Commands** list of the **Define Commands** dialog box.
- 2. Click the **Remove** button.

## To change the ordering of version control menu options

- 1. Select the menu option to be moved from the **Commands** list of the **Define Commands** dialog box.
- 2. Click the **Move up** and **Move down** buttons as necessary.



# 8.3 Specifying arguments

It is obvious that arguments must be specified correctly, otherwise the version control tool executed will not function as intended. However, it is also important, when using custom version control support, to specify the arguments in a flexible way, as a single version control command can be applied to more than one file. To facilitate this, the Arguments field has a placeholder button (see Reference 5, Placeholders, for an in depth discussion of placeholders), which when clicked on, invokes a pop-up menu of all available placeholders. An explanation of each version control placeholder and how their values are derived can be found in the table below.

Pop-up menu	Placeholder	Value And How It Is Determined
User login name	\$(USERNAME)	Current user login ('General' tab)
User login password	\$(PASSWORD)	Current user password ('General' tab)
Version control directory	\$(VCDIR)	'Virtual' version control mapping ("Projects" tab)
Comment	\$(COMMENT)	Comment specified before command execution
File path + name	\$(FULLFILE)	Full path and name of the file involved in the operation
Filename	\$(FILENAME)	Filename (including extension) of the file involved in the operation
File leaf	\$(FILELEAF)	Filename (excluding extension) of the file involved in the operation
File extension	\$(EXTENSION)	Extension of the file involved in the operation
File directory	\$(FILEDIR)	Directory of the file involved in the operation
Configuration directory	\$(CONFIGDIR)	Current configuration directory
Project directory	\$(PROJDIR)	Current project directory
Workspace directory	\$(WORKSPDIR)	Current workspace directory
Temp Directory	\$(TEMPDIR)	Temporary directory
Command directory	\$(EXEDIR)	Version control executable directory
Windows directory	\$(WINDIR)	Directory where Windows® is installed
Windows system directory	\$(WINSYSDIR)	Directory where Windows® system files exist
Workspace name	\$(WORKSPNAME)	Current workspace name
Project name	\$(PROJECTNAME)	Current project name
Configuration name	\$(CONFIGNAME)	Current configuration name

# 8.4 Specifying comments

If a version control command contains the placeholder '\$(COMMENT)', the High-performance Embedded Workshop will request that you enter the comment when the command is executed (via the **Please Enter Comment** dialog box).

You may specify a comment for each file or, if you would like to specify the same comment for all files, check the **Apply comment to all files** checkbox before clicking the OK button.

# 8.5 Executable return code

While each version control command executes, its output is redirected to the **Version Control** tab of the output window. When the command's execution is complete, its return code is obtained. When defining a command, you can determine whether this return code can be used to indicate that the command failed and that the remaining commands should not be executed (i.e. abort). The controls contained in the **Executable return code** group allow you to specify this behavior.



## High-performance Embedded Workshop V.4.09

If the return code of the commands can be used to indicate a failure then you should select the **Command has failed if the return code is** option and set the drop-down list and edit box as required. If the **Command has failed if the return code is** option is selected then the High-performance Embedded Workshop will check the return code of each command to determine whether a failure occurred. If this is the case, no further commands will be executed and any other processes which would follow the commands (e.g. build) will not be executed.

If the **Return code of tool is not meaningful** option is selected then the High-performance Embedded Workshop will not check the return code of each command. Consequently, all commands will execute regardless.

# 8.6 Specifying file locations

When referring to a file's location, be sure to use a placeholder, otherwise the command will only relate to a hardwired file. For example, let's imagine that a version control executable has been selected which uses a -GET command to obtain a read-only copy of a file. The **Arguments** field could be specified as:

-GET 'c:\vc\files\project\main.c'

However, when executed, this command can only ever -GET the file MAIN.C. To resolve this problem, Highperformance Embedded Workshop uses a system of placeholders and directory mappings. Directory mappings tell the High-performance Embedded Workshop which 'working' directories (i.e. where source files are being worked on) map to which 'controlled' directories (i.e. where the source files are stored in the version control system). Mappings between these two directory systems can be specified via the **Projects** tab of the **Version Control Setup** dialog box.

Once the mappings have been defined, you can use the **Version control directory** placeholder, \$(VCDIR), to represent the directory in which the project file is stored.

Projects		
Show details for project : Details : Version control directory mapping	roject1	
Source Directory	Version Control Directory	<u>A</u> dd
C:\work\Application\Project1 C:\work\Application\Src C:\work\Shared	x:\vc\sw\app\project1 x:\vc\sw\app\src x:\vc\sw\shared	<u>M</u> odify
	Þ	<u>R</u> emove
E <u>n</u> vironment :		
		A <u>d</u> d
		Modi <u>f</u> y
· ·		Remo <u>v</u> e

#### To define a new directory mapping

- 1. Select [Tools -> Version Control -> Configure]. The Version Control Setup dialog box opens.
- 2. Select the **Projects** tab.

- 3. Click the Add button, which is next to the Version Control Directory Mappings list. The Version Control Directory Mapping dialog box opens.
- 4. Specify a High-performance Embedded Workshop workspace directory or project directory in the **Source directory** field. Select one of the following operations:
  - Enter the directory name, **OR**
  - Click the placeholder button. Then select "Project directory" or "Workspace directory" from the menu, OR
  - Click the **Browse** button to open the **Browse to Source Directory** dialog box. Select the directory and click the **Select** button.
- 5. Enter the version control directory into the Version Control Directory field.
- 6. Click the OK button.

## To modify a directory mapping

- 1. Select [Tools -> Version Control -> Configure]. The Version Control Setup dialog box opens. Select the **Projects** tab.
- 2. Select the mapping to be modified from the Version Control Directory Mappings list and then click the Modify button. The Version Control Directory Mapping dialog box opens.
- 3. Make the necessary changes to the two directories and then click the OK button to confirm the new settings.

Version Control Directory Mapping		? ×
Source directory : C:\work\Shared	<u>B</u> rowse	ОК
Version control directory :		Cancel
x:\vc\vc\shared		

#### To remove a directory mapping

- 1. Select [Tools -> Version Control -> Configure]. The Version Control Setup dialog box opens. Select the **Projects** tab.
- 2. Select the mapping to be removed from the **Version Control Directory Mappings** list and then click the **Remove** button.



# 8.7 Specifying file locations example

Consider the scenario shown in the figure below. It shows three directories, which are mapped from a shared version control drive (X:\) to a local drive where the development is being done (C:\).



Now let's imagine that a version control executable has been selected which uses a -GET command to obtain a read-only copy of a file. In order to get all of the files in a project we need to use the following command:

# -GET '\$(VCDIR)\\$(FILENAME)'

When the High-performance Embedded Workshop executes the command for a given project file, it will replace \$(VCDIR) for the equivalent version control directory in the file mapping.

For example, suppose FILE1.C is located at c:\work\application\project1\FILE1.C. If the -GET command is applied to FILE1.C then:

- 1. 'x:\vc\sw\app\project1' is substituted for '\$(VCDIR)', as this is the version control directory mapping for 'c:\work\application\project1'.
- 2. 'FILE1.C' is substituted for '\$(FILENAME)'.



# 8.8 Specifying environment

Select the **Projects** tab of the **Version Control Setup** dialog box to view the current settings.

#### To add a new environment variable

1. Click the Add button beside the Environment list (the Environment Variable dialog box will be invoked).

Environment Variable	? ×	
Varia <u>b</u> le :	ОК	
I		
Value :	Cancel	
		_ Placeholder
		popup menu

- 2. Enter the variable name into the **Variable** field.
- 3. Enter the variable's value into the **Value** field.
- 4. Click the OK button to add the new variable to the Environment list.

#### To modify an environment variable

- 1. Select the variable that you want to modify from the **Environment** list.
- 2. Click the **Modify** button beside the list.
- 3. Make the required changes to the **Variable** and **Value** fields.
- 4. Click the OK button to add the modified variable back to the list.

#### To remove an environment variable

- 1. Select the variable that you want to remove from the **Environment** list.
- 2. Click the **Remove** button beside the list.

# 8.9 Controlling execution of a Version Control System

The **General** tab of the **Version Control Setup** dialog box allows you to control the way in which the version control system is executed. It also shows the full path to the current version control configuration file.



The execution of a version control system can be modified via the following three checkboxes:

#### • Prompt before executing command

If this checkbox is selected then, before any version control commands are executed, a dialog is displayed, which lists all of the files involved in the operation. Files may be deselected by clearing the associated checkbox. Clicking the OK button will apply the command to each of the selected files. Clicking the Cancel button will abort the operation.

#### • Run in DOS window

By default, the output of the version control commands is redirected to the **Version Control** tab of the output window. If you would rather run each command in a separate DOS window then set this checkbox. Select **Close DOS window on exit** checkbox if you wish the DOS window to close after execution has been completed.

#### • Use forward slash '/' as version control directory delimiter

By default, when the High-performance Embedded Workshop substitutes the \$(VCDIR) placeholder, it uses the backward slash character '\' to divide directories. However, if the version control system you are using uses a forward slash character (e.g. Visual SourceSafe) to divide directories then select the **Use forward slash '/' as version control directory delimiter** checkbox.

# 8.10 Specifying a user name and password

Most version control tools will require you to pass a username and password on the command line in order to keep files secure, and to keep a record of which files were changed by which users. The custom version control support provides two placeholders: **User login name**, \$(USERNAME), and **User login password**, \$(PASSWORD). When the command is executed, these placeholders will be replaced with the current settings in the **General** tab of the **Version Control Setup** dialog box.

In order to give the \$(USERNAME) and \$(PASSWORD) fields a value you will first need to login. If you have not logged in before a command is executed, which uses either of these placeholders, then you will be prompted to do so before the command can be executed.

#### To login (i.e. specify a username and password)

- 1. Select [Tools -> Version Control -> Configure]. The Version Control Setup dialog box opens.
- 2. Select the **General** tab.

User details :		
User :	JHARK	Log In
Password :	*****	( <u></u> )

3. Click the **Log In** button. The **Log in** dialog box opens.

User name : OK	
JHARK Cancel	
Password :	
*****	
Confirm password by retyping it below :	
****	

- 4. Enter your username into the User name field.
- 5. Enter your password into the **Password** field.
- 6. Re-type your password again into the **Confirm password by retyping it below** field.
- 7. Click the OK button to set the new username and password. If there is any inconsistency between the two versions of the password that you entered, you will be requested to type your password again.

# 8.11 Usage example of the Custom Version Control System

The custom version control system includes any version control system in which command-line commands should be executed via the MS-DOS command prompt. The usage examples in this section use the following version control systems to be connected with the High-performance Embedded Workshop.

- Connecting the High-performance Embedded Workshop with RCS
- Connecting the High-performance Embedded Workshop with CVS

# 8.11.1 Connecting the High-performance Embedded Workshop with RCS

In this example, the High-performance Embedded Workshop is to be connected with RCS.

# • RCS environment

The following types of version control system can be used:

- o GNU RCS
- o GNU diff

This procedure assumes that the version control system is located at "C:\RCS".



## Workspace

Create a High-performance Embedded Workshop workspace in the following folder.

C:\WorkSpace\rcs\_sample

Then create a folder for RCS.

 $C:\!\!\!\!WorkSpace\cs_sample\cs_sample\cs_sample$ 

The following pages describe the procedures to create a workspace, make settings for version control, check-in, compare, check-out, and view logs.

- Step 1: Create a workspace
- Step 2: Selecting Custom Version Control System
- Step 3: Making settings for Version Control
- Step 4: Using the Version Control facility (Check-in)
- Step 5: Using the Version Control facility (Compare)
- Step 6: Using the Version Control facility (Check-out)
- Step 7: Using the Version Control facility (View Logs)

#### 8.11.1.1 Step 1: Create a workspace

- 1. Select [File -> New Workspace] to open the New Project Workspace dialog box.
- 2. Create a workspace with the following conditions.
  - Workspace Name: rcs\_sample
  - Project Name: rcs\_sample
  - Directory: C:\WorkSpace\rcs\_sample
- 3. Create a folder for RCS.

C:\WorkSpace\rcs\_sample\rcs\_sample\RCS



# 8.11.1.2 Step 2: Selecting Custom Version Control System

1. Select [Tools -> Version Control -> Select]. The Select Version Control System dialog box opens.

? ×
OK
Cancel

- 2. Select Custom VCS.
- 3. Click the OK button.

# 8.11.1.3 Step 3: Making settings for Version Control

Select [Tools -> Version Control -> Configure] to open the Version Control Setup dialog box.

## • Setting environment variables

1. Select the **Projects** tab.

Version Control Setup	? ×
Commands Projects General	
Show details for project : rcs_sample   Details :  Version control directory mappings :	
Source Directory Version Control Directory	Add
	<u>M</u> odify
	<u>R</u> emove
Environment :	
LOGNAME=user path=%path%;C:\RCS TZ=JST-9	Modify
OK Cancel <u>E</u> xp	ort Import

- 2. Click the Add button on Environment.
- 3. The **Environment** dialog box appears.

? ×
ОК
Cancel

- 4. Enter "path" in **Variable** and "%path%;C:\RCS" in **Value**, respectively.
- 5. Click the OK button.
- 6. Click the Add button. Also enter the following environment variables in Environment.

LOGNAME=user

TZ=JST-9

## • Specifying actions

1. Select the **General** tab.

/ersion Control Setup
Commands Projects General
Execution :
🔽 Run in DOS Window 🔲 Close DOS window on exit
□ Use forward slash '/' as version control directory delimiter
User details :
User : None Log In
Configuration file: Current version control configuration file: C:\WorkSpace\rsc_sample\rsc_sample.hvc
OK Cancel <u>E</u> xport <u>I</u> mport

- 2. Deselect the **Prompt before executing command** checkbox. (Leave it selected, however, if you wish to see the file name before executing commands.)
- 3. Select the **Run in DOS Window** checkbox. (This is because the RCS commands must be input via the MS-DOS prompt.)

4. Deselect the **Close DOS Window on exit** checkbox. (This is because errors must be viewed in the MS-DOS prompt.)

# • Setting up commands

- 1. Select the **Commands** tab.
- 2. Click the Add button on User menu options.
- 3. The Add Menu Option dialog box appears.
- 4. Enter "Initialize" in **Option** and click the **Add** button.
- 5. The **Add Command** dialog box appears.

Add Command	? ×
Command : Version control executable : C:\RCS\CI.EXE Arguments : I \$(FILENAME) Initial directory : \$(PR0JDIR) Browse	OK Cancel
Executable return code :	

- 6. Enter "C:\RCS\CI.EXE" in **Version control executable**.
- 7. Enter "-1 \$(FILENAME)" in **Arguments**.
- 8. Enter "\$(PROJDIR)" in **Initial directory**.
- 9. Click the OK button.
- 10. Click the OK button in the Add Menu Option dialog box.

# Tip:

It is recommended that you only register some commands frequently used for version control. Other commands should be used via the MS-DOS command prompt.



Also set up the following tree commands in the same way.

Command	Option Name	Characters to be Input	
Check out	Version control executable	C:\RCS\CO.EXE	
	Argument	-I -f \$(FILENAME)	
	Initial directory	\$(PROJDIR)	
Diff	Version control executable	C:\RCS\RCSDIFF.EXE	
	Argument	\$(FILENAME)	
	Initial directory	\$(PROJDIR)	
Log	Version control executable	C:\RCS\RLOG.EXE	
	Argument	\$(FILENAME)	
	Initial directory	\$(PROJDIR)	

Preparation for version control is now completed.

## 8.11.1.4 Step 4: Using the Version Control facility (Check-in)

1. Enter a keyword at line 12 in the file "rcs\_sample.c" so that you will be able to see the changes made in the file.

//\$Id\$

- 2. Select the file "rcs\_sample.c" in the **Projects** tab of the workspace window and then select [**Tools** -> **Version Control** -> **Check in**].
- 3. The MS-DOS command prompt window opens.
- 4. Enter a comment for check-in after the prompt (">>") and press Enter.
- 5. Enter one period '.' and press Enter.

rcs\_sample.c,v <-- rcs\_sample.c

enter description, terminated with single '.' or end of file:

NOTE: This is NOT the log message!

>>.

initial revision: 1.1

done

C:\RCS>

- 6. Enter "exit" and press Enter. This closes the MS-DOS command prompt window.
- 7. A High-performance Embedded Workshop confirmation dialog box appears. Click **Yes**. The keyword now includes the file name, revision number, date, and user name.



## Tip:

If you do not wish to see the result of command execution in the MS-DOS command prompt window, select the **Close DOS Window on exit** checkbox in the **General** tab of the **Version Control Setup** dialog box.

# 8.11.1.5 Step 5: Using the Version Control facility (Compare)

- 1. To compare with the checked-in file, modify the file "rcs\_sample.c" and save it.
- 2. Select the file "rcs\_sample.c" in the **Projects** tab of the workspace window and select [**Tools -> Version Control -> Diff**].
- 3. The MS-DOS command prompt window appears and shows the result of comparison.
- 4. Enter "exit" and press the Enter key. This closes the MS-DOS command prompt window.

RCS file: rcs\_sample.c,v Working file: rcs\_sample.c head: 1.1 branch: locks: strict user: 1.1 access list: symbolic names: keyword substitution: kv total revisions: 1; selected revisions: 1 description: \_\_\_\_\_ revision 1.1 locked by: user; date: 2006/03/14 07:22:42; author: user; state: Exp; Initial revision \_\_\_\_\_

# 8.11.1.6 Step 6: Using the Version Control facility (Check-out)

Use the check-out command to undo the changes made in the file.

- 1. Select the file "rcs\_sample.c" in the **Projects** tab of the workspace window and select [**Tools** -> **Version Control** -> **Check out**].
- 2. The MS-DOS command prompt window appears and shows the result of check-out.



\_\_\_\_\_

- 3. Enter "exit" and press the Enter key. This closes the MS-DOS command prompt window.
- 4. A High-performance Embedded Workshop confirmation dialog box appears. Click **Yes**. The file has returned to its original state.

#### 8.11.1.7 Step 7: Using the Version Control facility (View Logs)

Use the log command to view the changes.

- 1. Select the file "rcs\_sample.c" in the **Projects** tab of the workspace window and select [**Tools** -> **Version Control** -> **Log**].
- 2. The MS-DOS command prompt window appears and shows the log information.
- 3. Enter "exit" and press the Enter key. This closes the MS-DOS command prompt window.

#### 8.11.2 Connecting the High-performance Embedded Workshop with CVS

In this example, the High-performance Embedded Workshop is to be connected with CVS.

#### • CVS tool environment

The following type of version control system can be used:

o CVS

This procedure assumes that the version control system is located at "C:\cvs-1-11-17".

#### Workspace

Create a High-performance Embedded Workshop workspace in the following folder.

C:\WorkSpace\sampleCVS

Then create a folder for CVS.

C:\src\sampleCVS

#### Note:

In this version control using the High-performance Embedded Workshop and CVS, only versions of source files can be controlled. It cannot control versions of any other files (such as workspace or project files).

The following pages describe the procedures to create a workspace, make settings for version control, create a repository, register a module, check-out, check the status, compare, register the result, and view logs.



- Step 1: Create a workspace
- Step 2: Selecting Custom Version Control System
- Step 3: Making settings for Version Control
- Step 4: Using the Version Control facility (Create a Repository)
- Step 5: Using the Version Control facility (Register a Module)
- Step 6: Using the Version Control facility (Check-out)
- Step 7: Using the Version Control facility (Modify the File)
- Step 8: Using the Version Control facility (Check the Status)
- Step 9: Using the Version Control facility (Compare)
- Step 10: Using the Version Control facility (Register the Result)
- Step 11: Using the Version Control facility (View Logs)

## 8.11.2.1 Step 1: Create a workspace

- 1. Select [File -> New Workspace] to open the New Project Workspace dialog box.
- 2. Create a workspace with the following conditions.
  - Workspace Name: sampleCVS
  - Project Name: sampleCVS
  - Directory: C:\WorkSpace\sampleCVS
- 3. Exit the High-performance Embedded Workshop.
- 4. Create a folder for CVS.

C:\src\sampleCVS

- 5. Move the file you wish to control from "C:\WorkSpace\sampleCVS\sampleCVS" to "C:\src\sampleCVS".
- 6. Start up the High-performance Embedded Workshop and open the created workspace "sampleCVS".
- 7. Select [**Project -> Remove Files**] to open the **Remove Project Files** dialog box.
- 8. Select the file you wish to control and click the **Remove** button.
- 9. Click OK button.
- 10. Select [Project -> Add Files] to open the Add files to project 'sampleCVS' dialog box.
- 11. Select the file you wish to control and click the Add button.



# 8.11.2.2 Step 2: Selecting Custom Version Control System

1. Select [Tools -> Version Control -> Select].

<u> Y</u> ×	(
OK	
Cancel	
	OK Cancel

- 2. The Select Version Control System dialog box opens.
- 3. Select Custom VCS.
- 4. Click the OK button.

# 8.11.2.3 Step 3: Making settings for Version Control

Select [Tools -> Version Control -> Configure] to open the Version Control Setup dialog box.

## • Setting environment variables

- 1. Select the **Projects** tab.
- 2. Click the **Add** button on **Environment**.
- 3. The **Environment** dialog box appears.
- 4. Enter "CVSROOT" in Variable and "C:\cvs-1-11-17\DATABASE" in Value, respectively.



Version Control Setup		? ×
Commands Projects General		
-	ampleCVS 💌	
Details : Ver <u>s</u> ion control directory mapp	pings :	
Source Directory	Version Control Directory	<u>A</u> dd
		<u>M</u> odify
		<u>R</u> emove
•	Þ	
E <u>n</u> vironment :		
CVSR00T=C:\cvs-1-11-17\	DATABASE	A <u>d</u> d
		Modi <u>fy.</u> .
		Remo <u>v</u> e
	Cancel <u>E</u> xp	ort

5. Click the OK button.

# • Setting up commands

- 1. Select the **Commands** tab.
- 2. Click the Add button on User menu options.
- 3. The Add Menu Option dialog box appears.
- 4. Enter "Initialize" in **Option** and click the **Add** button.
- 5. The Add Command dialog box appears.
- 6. Enter "C:\cvs-1-11-17\cvs.exe" in Version control executable.
- 7. Enter "init" in **Arguments**.
- 8. Enter "C:\src\sampleCVS" in **Initial directory**.



Add Command	? ×
Command : Version control executable : C:\cvs-1-11-17\cvs.exe Arguments : init	OK Cancel
Initial directory : C:\src\sampleCVS Browse Browse	
Executable return code :	

- 9. Click the OK button.
- 10. Click the OK button in the Add Menu Option dialog box.

Also set up the following six commands in the same way.

Command	Option Name	Characters to be Input
Register	Version control executable	C:\cvs-1-11-17\cvs.exe
	Argument	import -m "New Source" sampleCVS Renesas rel-20060228
	Initial directory	C:\src\sampleCVS
Check out	Version control executable	C:\cvs-1-11-17\cvs.exe
	Argument	checkout sampleCVS
	Initial directory	C:\src
Status	Version control executable	C:\cvs-1-11-17\cvs.exe
	Argument	status \$(FILENAME)
	Initial directory	C:\src\sampleCVS
Diff	Version control executable	C:\cvs-1-11-17\cvs.exe
	Argument	diff \$(FILENAME)
	Initial directory	C:\src\sampleCVS
Commit	Version control executable	C:\cvs-1-11-17\cvs.exe
	Argument	commit -m "modify" \$(FILENAME)
	Initial directory	C:\src\sampleCVS
Log	Version control executable	C:\cvs-1-11-17\cvs.exe
	Argument	log \$(FILENAME)
	Initial directory	C:\src\sampleCVS

Preparation for version control is now completed.



## 8.11.2.4 Step 4: Using the Version Control facility (Create a Repository)

- 1. Select [Tools -> Version Control -> Initialize].
- 2. Select the file in the **Initialize** dialog box and click the OK button.

Executing C:\cvs-1-11-17\cvs.exe init Command(s) completed successfully.

## 8.11.2.5 Step 5: Using the Version Control facility (Register a Module)

- 1. Select [Tools -> Version Control -> Register].
- 2. Select the file in the **Register** dialog box and click the OK button.

Executing C:\cvs-1-11-17\cvs.exe import -m "New Source" sampleCVS Renesas rel-20060228

N sampleCVS/dbsct.c

(Omit)

No conflicts created by this import

Command(s) completed successfully.

## 8.11.2.6 Step 6: Using the Version Control facility (Check-out)

- 1. Select [Tools -> Version Control -> Checkout].
- 2. Select the file in the **Checkout** dialog box and click the OK button.
- 3. The Version Control tab of the output window shows the following message.

Executing C:\cvs-1-11-17\cvs.exe checkout sampleCVS

U sampleCVS/dbsct.c

(Omit)

cvs.exe checkout: Updating sampleCVS

Command(s) completed successfully.

# 8.11.2.7 Step 7: Using the Version Control facility (Modify the File)

Use the High-performance Embedded Workshop editor to modify the file (and save it). In this example, the file "sampleCVS.c" is modified.



## 8.11.2.8 Step 8: Using the Version Control facility (Check the Status)

- 1. The status of the file "sampleCVS.c" should be checked.
- 2. Select [Tools -> Version Control -> Status].
- 3. Select the file in the **Status** dialog box and click the OK button.
- 4. The Version Control tab of the output window shows the following message.

Executing C:\cvs-1-11-17\cvs.exe status sampleCVS.c

File: sampleCVS.c Status: Locally Modified

Working revision: 1.1.1.1 Tue Feb 28 02:46:30 2006

Repository revision: 1.1.1.1 C:\cvs-1-11-17\DATABASE/sampleCVS/sampleCVS.c,v

Sticky Tag: (none)

Sticky Date: (none)

Sticky Options: (none)

Command(s) completed successfully.

The status of the file "sampleCVS.c" is shown as "Locally Modified".

#### 8.11.2.9 Step 9: Using the Version Control facility (Compare)

- 1. The changes made in the file "sampleCVS.c" should be checked.
- 2. Select [Tools -> Version Control -> Diff].
- 3. Select the file in the **Diff** dialog box and click the OK button.
- 4. The Version Control tab of the output window shows the following message.

Executing C:\cvs-1-11-17\cvs.exe diff sampleCVS.c

Index: sampleCVS.c

RCS file: C:\cvs-1-11-17\DATABASE/sampleCVS/sampleCVS.c,v

```
retrieving revision 1.1.1.1
```

```
diff -r1.1.1.1 sampleCVS.c
```

```
28c28,31
```

< printf("### Data Input ###\n");

----

```
>//loop
```

```
> while(1)
```



```
> {
> printf("### Data Input ###\n"); //loop
49a53,54
> //loop
> }
Command(s) completed successfully.
```

You can see the changes made in the file "sampleCVS.c".

# 8.11.2.10 Step 10: Using the Version Control facility (Register the Result)

- 1. Select [Tools -> Version Control -> Commit].
- 2. Select the file in the **Commit** dialog box and click the OK button.
- 3. The Version Control tab of the output window shows the following message.

Executing C:\cvs-1-11-17\cvs.exe commit -m "modify" sampleCVS.c

Checking in sampleCVS.c;

C:\cvs-1-11-17\DATABASE/sampleCVS/sampleCVS.c,v <-- sampleCVS.c

new revision: 1.2; previous revision: 1.1

done

Command(s) completed successfully.

# 8.11.2.11 Step 11: Using the Version Control facility (View Logs)

- 1. Select [Tools -> Version Control -> Log].
- 2. Select the file in the **Log** dialog box and click the OK button.
- 3. The Version Control tab of the output window shows the following message.

Executing C:\cvs-1-11-17\cvs.exe log sampleCVS.c RCS file: C:\cvs-1-11-17\DATABASE/sampleCVS/sampleCVS.c,v Working file: sampleCVS.c head: 1.2 branch: locks: strict access list: symbolic names:

rel-20060228: 1.1.1.1



Renesas: 1.1.1
keyword substitution: kv
total revisions: 3; selected revisions: 3
description:
revision 1.2
date: 2006/02/28 03:39:22; author: user; state: Exp; lines: +6 -1
modify
revision 1.1
date: 2006/02/28 02:46:30; author: user; state: Exp;
branches: 1.1.1;
Initial revision
revision 1.1.1.1
date: 2006/02/28 02:46:30; author: user; state: Exp; lines: +0 -0
New Source
New Source

Command(s) completed successfully.



# 9. Visual SourceSafe Version Control System

The High-performance Embedded Workshop provides specific support for the Microsoft® Visual SourceSafe version control system.

The Visual SourceSafe version control system associates a project in your workspace with a project inside a Microsoft® Visual SourceSafe database.

Select a menu option from the [Tools -> Version Control] submenu or click a Version Control toolbar button to quickly activate the commands most frequently used in the Microsoft® Visual SourceSafe version control system.

For installation and option settings of Microsoft® Visual SourceSafe, refer to the online help or user's manual for Microsoft® Visual SourceSafe. In this section, it is assumed that the login username and password have been set.

The outline of procedures for version control by the High-performance Embedded Workshop connected with Microsoft® Visual SourceSafe is described below. For other functions available, see section 9.2, Visual SourceSafe commands.

## 1. Attaching Microsoft® Visual SourceSafe to a workspace

Associate a project in a High-performance Embedded Workshop workspace with a project inside a Microsoft® Visual SourceSafe database.

- Select "Microsoft Visual SourceSafe" as the version control system to be connected with the Highperformance Embedded Workshop.
- Login Microsoft® Visual SourceSafe.
- Specify the name and location of the project to be created in Microsoft® Visual SourceSafe.

#### 2. Adding files

Select the files in a High-performance Embedded Workshop workspace that you would like to add to Microsoft® Visual SourceSafe.

#### 3. Checking out a file

Select the files that you would like to modify in the High-performance Embedded Workshop workspace and check them out.

#### 4. Checking in a file

Select the files modified in the High-performance Embedded Workshop workspace and check them in.

# 9.1 Attaching Visual SourceSafe to a workspace

The following sections describe how you can associate Microsoft® Visual SourceSafe with your current workspace.

First, you need to select Microsoft® Visual SourceSafe as the version control system to be connected with the Highperformance Embedded Workshop.



#### To attach Microsoft® Visual SourceSafe to a workspace

- 1. Select [Tools -> Version Control -> Select]. The Select Version Control System dialog box opens.
- 2. Select the "Microsoft Visual SourceSafe" entry from the Microsoft Version control systems list.

Select Version Control System	? ×
Version control system:	OK
Microsoft Visual SourceSafe	Cancel

3. Click OK. The **SourceSafe Login** dialog box opens.

SourceSafe L	ogin	? ×
<u>U</u> sername:	User01	Login
<u>P</u> assword:	*****	Cancel
<u>D</u> atabase pa	ath:	Browse

- 4. Enter your Visual SourceSafe username into the Username field and password into the Password field.
- 5. For the **Database path** field, specify the Visual SourceSafe database file (SRCSAFE.INI) into which you would like to add this project. Select either of the following operations:
  - Click the **Browse** button to open the **Select Visual SourceSafe database** dialog box. Select the Visual SourceSafe database file and click the **Select** button, **OR**

Select Visual SourceSafe database	? ×
Look in: 🔁 VSS 💽 🖛 🗈 💣 🎟	
ata	
File <u>n</u> ame: srcsafe.ini Selec	t
Files of type: Visual SourceSafe databases (srcsafe.ini) 💌 Cance	<u>، ا</u>

- Enter the full path of the Visual SourceSafe database file in the Database path field.
- 6. The Create SourceSafe Project For Workspace dialog box was displayed.

Create SourceSafe Project For Workspace	? ×
Project name:	ОК
Workspace_Name	Cancel
\$/Workspace_Name	
<b>.</b>	
1	

- 7. The **Project name** field shows the name of the High-performance Embedded Workshop workspace. The **Project name** field displays the name of the project (i.e. folder) to be created in the database. If necessary you can change this name to another.
- 8. The tree underneath the **Project name** field shows the structure of the database specified in Step 5. Select the folder into which you would like to create the folder specified in **Project name**.
- 9. Click the OK button in the Create SourceSafe Project For Workspace dialog box. The Create SourceSafe Project dialog box opens. The example below shows that the High-performance Embedded Workshop workspace "Workspace\_Name" is added as a project "Workspace\_Name" into Visual SourceSafe database. The Project name field shows "Poject\_Name" that is to be created next.

Create SourceSafe Project	? ×
Project name: Project_Name	OK Cancel
\$/Workspace_Name/Project_Name	

- 10. Click the OK button in the Create SourceSafe Project dialog box.
- 11. Repeat step 10 for each of the projects in the current workspace.



The structure image of projects and sub-projects in the Visual SourceSafe database, High-performance Embedded Workshop workspace and projects, and the workspace directory in the hard disk is shown below.

Projects and sub-projects in the Visual SourceSafe database	HEW workspace and projects	Workspace directory in the hard disk (Visual SourceSafe working folder)
⊡-∰ \$/ ⊡-∰ <mark>Workspace_Name</mark> Project_Name	⊡…~ Workspace_Name ⊡… @ Project_Name	□

The High-performance Embedded Workshop has now created the necessary projects within Visual SourceSafe, and set up the version control toolbar and menu for immediate access.

# 9.2 Visual SourceSafe commands

The following eight commands are available via the version control toolbar or menu:

Tools -> Version Control menu	Version Control Toolbar	Function
Add to VCS	<b>P</b>	Adding files to Visual SourceSafe
Remove from VCS	÷	Removing files from Visual SourceSafe
Get from VCS	₽	Getting read-only copy of files from Visual SourceSafe
Check out from VCS	€	Checking out writable copy of files from Visual SourceSafe
Check in to VCS	<u>Å</u>	Checking in writable copy of files into Visual SourceSafe
Get VCS status	<u>a</u>	Viewing the status of files in Visual SourceSafe
Undo Check Out	-	Undoing a check out operation in Visual SourceSafe
Show History	-	Viewing the history of files in Visual SourceSafe

# 9.2.1 Adding files to Visual SourceSafe

In the previous section, a mapping between the workspace directory on your hard disk (i.e. the working directory) and the project directory in Visual SourceSafe (i.e. the directory controlled by Visual SourceSafe) has been established. However, the project directory (and any subdirectories) on your hard disk may contain many source files, whereas the directory it is mapped to in Visual SourceSafe will be initially empty.

You need to add files to the Visual SourceSafe project.



## To add files to Visual SourceSafe

- Select the files that you would like to add to Visual SourceSafe, in the **Projects** tab of the workspace window. You may also select a file folder, project folder, workspace folder or combination thereof. When selecting the project or workspace folder, the High-performance Embedded Workshop system files will be added to the selected file list. For example, selecting the project folder will also add the project file (.HWP) to the **Files** list. If the project file is then checked out and the version is newer than when it was last loaded, you will be asked whether you want to reload the project.
- 2. Select one of the following operations:
  - Click the Add to VCS toolbar button (
    ), OR
  - Select [Tools -> Version Control -> Add to VCS], OR
  - Right-click to invoke a pop-up menu and select [Version Control -> Add to VCS].
- 3. The **Add to VCS** dialog box opens. Select the checkbox on the left to the name of the file that you wish to add and click the OK button.

Add to VCS		? ×
<u>F</u> iles:		ОК
✓Workspace_Name.hws	[C:\Worksp 🔺	
✓Project_Name.hwp	[C:\Worksp	Cancel
✓Project_Name.c	[C:\Worksp	
<b>⊘</b> typedefine.h	[C:\Worksp	
dbsct.c	[C:\Worksp	
•	ra	
✓ Keep checked out		
<u>C</u> omment:		
	<b></b>	
J	7	

When you add files to Visual SourceSafe, the local versions in your working directory will become read-only. If you select the **Keep checked out** checkbox in the **Add to VCS** dialog box before you check the files into Visual SourceSafe, these files can be writable even when they are added. To check that the **Add to VCS** operation was carried out as you expected, or to quickly review the status of all of the files in a project:

- 1. Select the project folder whose files you want to check, in the **Projects** tab of the workspace window.
- 2. Select one of the following operations:
  - Click the Get VCS status toolbar button ((a)), OR
  - Select [Tools -> Version Control -> Get VCS status], OR
  - Right-click the selected item to invoke a pop-up menu and select [Version Control -> Get VCS status].



#### High-performance Embedded Workshop V.4.09

3. The status of each file will be displayed in the **Version Control** tab of the output window, or the **File(s) Status** dialog box. For setting of location where the status is to be displayed, see section 9.3, Visual SourceSafe command options. The information shown includes whether the file is added to the project, if the file is checked out and, if it is checked out, and who did so.

# 9.2.2 Removing files from Visual SourceSafe

If you wish to take files out of Visual SourceSafe, the files must be removed.

## To remove files from Visual SourceSafe

- 1. Select the files that you would like to remove from Visual SourceSafe, in the **Projects** tab of the workspace window. You may also select a file folder, project folder, workspace folder or combination thereof.
- 2. Select one of the following operations:
  - Click the **Remove from VCS** toolbar button (**b**), **OR**
  - Select [Tools -> Version Control -> Remove from VCS], OR
  - Right-click to invoke a pop-up menu and select [Version Control -> Remove from VCS].

Remove from VCS		? ×
<u>F</u> iles:		ОК
✓Workspace_Name.hws	[C:\Worksp 🔺	
✓Project_Name.hwp	[C:\Worksp	Cancel
✓Project_Name.c	[C:\Worksp	
✓typedefine.h	[C:\Worksp	
<b>⊘</b> dbset.c	[C:\Worksp	
	·····	

3. The **Remove from VCS** dialog box opens. Select the checkbox on the left to the name of the file that you wish to remove and click the OK button.

# 9.2.3 Getting read-only copy of files from Visual SourceSafe

Visual SourceSafe protects your source files and ensures that only one user can have a writable copy of a controlled file at any one time. However, it is possible for any user to obtain a read-only copy of any file.



## To get read-only copy of files from Visual SourceSafe

- 1. Select the files that you would like to get from Visual SourceSafe, in the **Projects** tab of the workspace window. You may also select a file folder, project folder, workspace folder or combination thereof.
- 2. Select one of the following operations:
  - Click the **Get from VCS** toolbar button ( 🗘 ), **OR**
  - Select the [Tools -> Version Control -> Get from VCS] menu option, OR
  - Right-click to invoke a pop-up menu and select [Version Control -> Get from VCS].

Get from VCS		? ×
<u>F</u> iles:		ОК
✓Workspace_Name.hws	[C:\Worksp 🔺	
Project_Name.hwp	[C:\Worksp	Cancel
✓Project_Name.c	[C:\Worksp	
✓typedefine.h	[C:\Worksp	
<b>⊘</b> dbsct.c	[C:\Worksp	
	r~	
•	<u> </u>	

3. The **Get from VCS** dialog box opens. Select the checkbox on the left to the name of the file that you wish to obtain and click the OK button.

# 9.2.4 Checking out writable copy of files from Visual SourceSafe

Visual SourceSafe protects your source files and ensures that only one user can have a writable copy of a controlled file at any one time. The check out operation takes a writable copy of the file from Visual SourceSafe and places it on your local drive. This can only be done if another user has not already checked out (a blue mark next to file name) the files in question.

# To check out writable copy of files from Visual SourceSafe

- 1. Select the files that you would like to check out from Visual SourceSafe, in the **Projects** tab of the workspace window. You may also select a file folder, project folder, workspace folder or combination thereof.
- 2. Select one of the following operations:
  - Click the Check out from VCS toolbar button ( 😴 ), OR
  - Select [Tools -> Version Control -> Check out from VCS], OR
  - Right-click to invoke a pop-up menu and select [Version Control -> Check out from VCS].



## High-performance Embedded Workshop V.4.09

# 9. Visual SourceSafe Version Control System

Check out from VCS		? ×
Files: ♥Workspace_Name.hws ♥Project_Name.hwp ♥Project_Name.c ♥typedefine.h ♥dbsct.c	[C:\Worksp ▲ [C:\Worksp [C:\Worksp [C:\Worksp [C:\Worksp	OK Cancel
Comment:		

- 3. The **Check out from VCS** dialog box opens. Select the checkbox on the left to the name of the file that you wish to obtain and click the OK button.
- 4. When the operation is finished the file has a red mark next to its name. This means you as the current user of High-performance Embedded Workshop has checked it out.





# 9.2.5 Checking in writable copy of files into Visual SourceSafe

Visual SourceSafe protects your source files and ensures that only one user can have a writable copy of a controlled file at any one time. The check out operation takes a writable copy of the file from Visual SourceSafe and places it on your local drive. Once a file is checked out it is edited and then checked back in so that the edits can be made available to other users.

#### To check in edits made to files in Visual SourceSafe

- 1. Select the files that you would like to check back into Visual SourceSafe, in the **Projects** tab of the workspace window. You may also select a file folder, project folder, workspace folder or combination thereof.
- 2. Select one of the following operations:
  - Click the Check in to VCS toolbar button (<sup>2</sup>/<sub>2</sub>), OR
  - Select [Tools -> Version Control -> Check in to VCS], OR
  - Right-click to invoke a pop-up menu and select [Version Control -> Check in to VCS].

Check in to VCS		? ×
<u>F</u> iles:		ОК
✓Workspace_Name.hws	[C:\Worksp📥	
✓Project_Name.hwp	[C:\Worksp	Cancel
✓Project_Name.c	[C:\Worksp	
<b>⊘</b> typedefine.h	[C:\Worksp	
<b>⊘</b> dbsct.c	[C:\Worksp	
Keep checked out		
<u>C</u> omment:		
	<u>_</u>	

- 3. The **Check in to VCS** dialog box opens. Select the checkbox on the left to the name of the file that you wish to check back into Visual SourceSafe and click the OK button.
- 4. When the operation is finished the file has a red mark next to its name. This means you as the current user of High-performance Embedded Workshop has checked it out.
- 5. The red mark next to the file name is removed, which indicates that the file has been checked into Visual SourceSafe. If you select the **Keep checked out** checkbox in the **Check in to VCS** dialog box before you check the files into Visual SourceSafe, these files can be writable even when they are checked out.



# 9.2.6 Viewing the status of files in Visual SourceSafe

Although files appear in your High-performance Embedded Workshop project (in the **Projects** tab of the Workspace window), Visual SourceSafe is not necessarily controlling them. Some of the files that are being controlled by Visual SourceSafe will be checked in, and others will be checked out (i.e. being edited by a user). The **Get VCS status** command displays the current status of files.

The status of each file will be displayed in the **Version Control** tab of the output window, or the **File(s) Status** dialog box. For setting of location where the status is to be displayed, see section 9.3, Visual SourceSafe command options.

## To view the status of files in Visual SourceSafe

- 1. Select the files whose status you would like to view, in the **Projects** tab of the workspace window. You may also select a file folder, project folder, workspace folder or combination thereof.
- 2. Select one of the following operations:
  - Click the **Get VCS status** toolbar button (ⓐ), **OR**
  - Select [Tools -> Version Control -> Get VCS status], OR
  - Right-click to invoke a pop-up menu and select [Version Control -> Get VCS status].
- 3. The Version Control tab of the output window or the File(s) Status dialog box shows the status of files.

# 9.2.7 Undoing a check out command in Visual SourceSafe

Visual SourceSafe protects your source files and ensures that only one user can have a writable copy of a controlled file at any one time. The check out operation takes a writable copy of the file from Visual SourceSafe and places it on your local drive. Once a file is checked out it is edited and then checked back in so that the edits can be made available to other users. However, if the check out operation was carried out by mistake, or perhaps is no longer required, then the operation can be undone.

#### To undo a check out of files from Visual SourceSafe

- 1. Select the files upon which you would like to undo a previous check out operation, in the **Projects** tab of the workspace window. You may also select a file folder, project folder, workspace folder or combination thereof.
- 2. Select one of the following operations:
  - Select [Tools -> Version Control -> Undo Check Out], OR
  - Right-click to invoke a pop-up menu and select [Version Control -> Undo Check Out].



# 9. Visual SourceSafe Version Control System

Undo Check Out		? ×
<u>F</u> iles:		ОК
✓Workspace_Name.hws	[C:\Worksp 🔺	
✓Project_Name.hwp	[C:\Worksp	Cancel
✓Project_Name.c	[C:\Worksp	
<b>⊘</b> typedefine.h	[C:\Worksp	
<b>⊘</b> dbset.e	[C:\Worksp 🚽	
	Fo	

- 3. The **Undo Check Out** dialog box opens. Select the checkbox on the left to the name of the file that you would like to undo a previous check out operation and click the OK button.
- 4. If the file has been modified, the confirmation message box shown below appears. To continue the check-out operation, click the Cancel button. To load the unmodified file from the version control system, click the Yes button. If you click the No button, the contents of the local file will be different from those of the file in the version control system.



# 9.2.8 Viewing the history of files in Visual SourceSafe

Visual SourceSafe controls the edits to the files in its projects and allows you to view the complete history of these edits right back to the time that the file was first added to the project.

The status of each file will be displayed in the **Version Control** tab of the output window, or the **File(s) History** dialog box. For setting of location where the status is to be displayed, see section 9.3, Visual SourceSafe command options.

#### To view the revision history of files in Visual SourceSafe

- 1. Select the files whose history you would like to view, in the **Projects** tab of the workspace window. You may also select a file folder, project folder, workspace folder or combination thereof.
- 2. Select one of the following operations:
  - Select [Tools -> Version Control -> Show History], OR
  - Right-click to invoke a pop-up menu and select [Version Control -> Show History].
- 3. The **Version Control** tab of the output window or the **File(s) History** dialog box shows the revision history of files.


### 9.3 Visual SourceSafe command options

You can control the way in which the history and status commands are displayed by selecting [Tools -> Version Control -> Configure].

- 1. Select [Tools -> Version Control -> Configure]. The Version Control Setup dialog box opens.
- 2. Select the **General** tab.

Visual Source Safe:
Display dialog box for history
Display dialog box for file status

- 3. To display the results of a history command in a dialog box then check the **Display dialog box for history** checkbox or clear it if you would rather display the output in the **Version Control** tab of the Output window.
- 4. To display the results of a status command in a dialog box, check the **Display dialog box for file status** checkbox or clear it if you would rather display the output in the **Version Control** tab of the Output window.

# 9.4 Version Control setup

You can view or modify the general settings for the High-performance Embedded Workshop connected with Visual SourceSafe.

- 1. Select [Tools -> Version Control -> Configure]. The Version Control Setup dialog box opens.
- 2. Select the **General** tab.

-User details :		
User:	User01	Log In
Password:	xoxoxoxox	
Database:	C:\Program Files\Microsoft	t Visual Studio\VSS\srcsafe.ini
-Configuration file:		
Current version control configuration file:		
C:\Workspace_Name\Workspace_Name.hvc		

- 3. The User details group box shows the login user ID and password, and database file in Visual SourceSafe.
- 4. You can modify the user ID, password, or database in the **Log in** dialog box opened by clicking the **Log In** button.
- 5. The **Configuration file** group box shows the configuration file in the current version control system.



# 9.5 Specifying file locations

You can view the current mapping in the project. Select the project that you wish to view the mapping from the **Show mappings for project** drop-down list. The High-performance Embedded Workshop workspace and project directory, and Visual SourceSafe project directory are shown in the list. Mapping of projects can be added, modified, or removed.

For detail, see section 8.6, Specifying file locations.

Custom Commands Mappines Ge	eneral		
Show mappings for project: Project_Name			
Source Directory C:\Workspace\Workspace_Nam C:\Workspace\Workspace_Nam	\$/Workspace_Name/Project_Nar	dd dify move	
	Þ		

#### To define a new directory mapping

- 1. Select [Tools -> Version Control -> Configure]. The Version Control Setup dialog box opens.
- 2. Select the **Mapping** tab.
- 3. Click the Add button. The Version Control Directory Mapping dialog box opens.
- 4. Specify a High-performance Embedded Workshop workspace directory or project directory in the **Source directory** field. Select one of the following operations:
  - Enter the directory name, **OR**
  - Click the placeholder button. Then select "Project directory" or "Workspace directory" from the menu, OR
  - Click the **Browse** button to open the **Browse to Source Directory** dialog box. Select the directory and click the **Select** button.
- 5. Enter the name of the Visual SourceSafe project directory to be mapped to the source directory in the Version Control Directory Mapping field.
- 6. Click the OK button.

#### To modify a directory mapping

- 1. Select [Tools -> Version Control -> Configure]. The Version Control Setup dialog box opens.
- 2. Select the **Mapping** tab.

3. Select the mapping to be modified from the list and then click the **Modify** button. The **Version Control Directory Mapping** dialog box opens.

Version Control Directory Mapping	? ×
Source directory : C:\Workspace\Workspace_Name\Proje Browse Version control directory :	OK Cancel
\$/Workspace_Name	

- 4. To modify a mapping, make the same setting as required for adding a mapping in the Version Control Directory Mapping dialog box.
- 5. Click the OK button.

#### To remove a directory mapping

- 1. Select [Tools -> Version Control -> Configure]. The Version Control Setup dialog box opens.
- 2. Select the **Mapping** tab.
- 3. Select the mapping to be removed from the list and then click the **Remove** button.

# 9.6 Adding Visual SourceSafe commands

In the **Custom Commands** tab of the **Version Control Setup** dialog box, you can add a menu defining a Visual SourceSafe command not included in the version control submenu.

Selecting this menu executes the defined command. Output information on execution of this command will be shown in the **Version Control** tab of the Output window.

This section introduces the definition of Properties command as an example.



Custom Commands   Mappings   General	
<u>C</u> ommands:	
	<u>A</u> dd
	<u>M</u> odify
	<u>R</u> emove
	Move <u>up</u>
	Move <u>d</u> own
Command Execution:	
Prompt before executing command	
Run in DOS Window	
☑ Use forward-slash '/ as version control director	y delimiter

- Execution control of version control commands
  - 1. Select the **Prompt before executing command** checkbox.
  - 2. Deselect the **Run in DOS Window** checkbox. (This is because the RCS commands must be input via the MS-DOS prompt.)
  - 3. Select the **User forward-slash '/' as version control directory delimiter** checkbox. (Visual SourceSafe uses this format.)

For detail, see section 8.9, Controlling execution of a Version Control System.

#### • Setting up commands

- 1. Click the Add button. The Add Menu Option dialog box appears.
- 2. Enter "Properties" in **Option** and click the **Add** button. The **Add Command** dialog box appears.



Add Command	? ×
Command: Version control executable : C:\Program Files\Microsoft Visual Studio\VSS\win32\SS.EXE Browse Arguments : Properties =y\$(USERNAME),\$(PASSWORD) -i- \$(VCDIR)/\$(FILI)	OK Cancel
Initial directory : \$(FILEDIR) Environment : Add	
Modify Remo <u>v</u> e	
Command return code: © <u>R</u> eturn code of command is not meaningful © Command has <u>f</u> ailed if the return code is Not Equal To 🔽 0	

- 3. Enter the executable file (SS.EXE) for version control system in Version control executable.
- 4. Enter "Properties -y\$(USERNAME),\$(PASSWORD) -i- \$(VCDIR)/\$(FILENAME)" in **Arguments**.
- 5. Enter "\$(FILEDIR)" in **Initial directory**.
- 6. Click the OK button.
- 7. Click the OK button in the Add Menu Option dialog box.

For detail, see section 8.2, Defining Version Control commands.

#### • Command Execution

- 1. Select [Tools -> Version Control -> Properties].
- 2. Select the file in the Properties dialog box and click the OK button.
- 3. Output information on execution of this command will be shown in the **Version Control** tab of the Output window.



# **10. Sharing Projects by Network Facilities**

The High-performance Embedded Workshop is capable of sharing workspaces and projects across a network. This allows users to concurrently work on shared projects and see each other's changes as they happen. This system can be used in conjunction with version control. The major difference with using this system is that each user can modify and update the workspace and project without making all of the other users reload their project and potentially lose all their changes.

This system is implemented by making one of the machines attached to the network the server machine. All other client machines then use the service this machine is providing. So if one of the client machines adds a new file, the server machine is notified. The server then notifies all other clients the action has taken place. This procedure is shown below.



The network system allows users to be given access rights to files. This allows the project administrator to make sure the only people who can modify the project and source files are allowed to do so. This might allow the administrator to limit each user to only have write capabilities for their own area of the project, other areas would be read only. This could limit any potential conflict or damage one user could do to other areas of the project. These limitations can be set to a number of different levels. This is outlined later in this section.

# Notes:

- Sharing projects by network facilities is not possible under the Windows Vista® or Windows® 7 operating system.
- Certain operations are locked when other clients are carrying them out. This means that if one user is currently changing the toolchain options all of the other users can only see read only versions of this data.
- The performance of High-performance Embedded Workshop does suffer when using the network facilities. If working in a small team it might be more suitable to use the single user mode and version control.



# Procedures necessary before invoking the High-performance Embedded Workshop on a PC with Windows® XP Service Pack 2 or later

See the following section, 10.1, Sharing projects by network facilities in a PC with Windows® XP Service Pack 2 or later.

#### To setup a project file on the server machine

- 1. Create a project file for the network.
- As the directory of the New Project Workspace dialog, specify a network path that can be recognized by both the server machine and the client machine.
   Example: \\WorkGroup\Severl\ShareDirectory

#### Sharing workspaces

The directory specified by the above procedure (step 2) must be shared. The High-performance Embedded Workshop's network facilities share the source, object, and other files in the server machine. These files exist only in the server machine. From the client machine, the files are accessed via the network.

#### To open a workspace from the client machine

To open a workspace on the network from the client machine, search the network path and select the workspace in the Open Workspace dialog box.

# 10.1 Sharing projects by network facilities in a PC with Windows® XP Service Pack 2 or later

Windows® XP Service Pack 2 provides enhanced security measures.

In a PC with Windows® XP Service Pack 2 or later, you cannot share projects by network facilities if the Windows® operating system is in its default state. Before invoking the High-performance Embedded Workshop, you should take the procedures listed below to change the settings of the PC so that High-performance Embedded Workshop system will be accessed via network. When your server machine and client machine have Windows® XP Service Pack 2 or later, the settings are required for both machines.

#### • For a server machine with Windows® XP Service Pack 2 or later

#### Step1: Firewall

- 1. Select **Control Panel** from the Windows® start menu.
- 2. Click Network and Internet Connections in the Control Panel.
- 3. Click **Windows Firewall** in the **Network and Internet Connections**. The **Windows Firewall** dialog box is invoked.



- 4. Click the **Exceptions** tab.
- 5. Click the Add Program button to open the Add a Program dialog box.
- 6. Click the **Browse** button and select \System\Sec\HewServer\Hew3Server.exe under the High-performance Embedded Workshop installation directory.
- 7. Click the OK button and close the Add a Program dialog box.
- 8. Click the Add Port button to open the Add a Port dialog box.
- 9. Enter "DCOM" in Name and "135" in Port Number and select TCP option button.
- 10. Click the OK button and close the Add a Port dialog box.
- 11. Click the OK button in the Windows Firewall dialog box.
- 12. Close the Network and Internet Connections.
- 13. Close the **Control Panel**.

#### Step2: DCOM

- 1. Select **Run** from the Windows® start menu. The **Run** dialog box is invoked.
- 2. Enter "dcomcnfg" in the **Open** field and click the OK button. The **Component Services** window opens.
- 3. Expand the tree. If the **Window Security Alert** dialog box appears at expansion, click the **Keep Blocking** button.

Console Root -> Component Services -> Computers -> My Computer

- 4. Right-click on **My Computer** and select **Properties** from the pop-up menu. The **My Computer Properties** dialog box is invoked.
- 5. Click the **COM Security** tab.
- 6. Click the Edit Limits button in the Access Permissions area to open the Access Permission dialog box.
- 7. Select "ANONYMOUS LOGON" in the Group or user names list.
- 8. Select the Allow checkbox for Remote Activation in the Permission for ANONYMOUS LOGON list.
- 9. Click the OK button and close the Access Permission dialog box.
- 10. Click the **Edit Limits** button in the **Launch and Activation Permission** area to open the **Launch Permission** dialog box.
- 11. Select "Everyone" in the Group or user names list.
- 12. Select the **Allow** checkbox for **Local Launch** in the **Permission for Everyone** list. Select the **Allow** checkbox for **Remote Activation** in the **Permission for Everyone** list.
- 13. Click the OK button and close the Launch Permission dialog box.



- 14. Click the OK button on the My Computer Properties dialog box.
- 15. Close the **Component Services** window.

#### • For a client machine with Windows® XP Service Pack 2 or later

#### Firewall

- 1. Select Control Panel from the Windows® start menu.
- 2. Click Network and Internet Connections in the Control Panel.
- 3. Click **Windows Firewall** in the **Network and Internet Connections**. The **Windows Firewall** dialog box is invoked.
- 4. Click the **Exceptions** tab.
- 5. Click the Add Program button to open the Add a Program dialog box.
- 6. Click the **Browse** button and select **HEW2.exe** under the High-performance Embedded Workshop installation directory.
- 7. Click the OK button and close the **Add a Program** dialog box.
- 8. Click the Add Port button to open the Add a Port dialog box.
- 9. Enter "DCOM" in Name and "135" in Port Number and select TCP option button.
- 10. Click the OK button and close the Add a Port dialog box.
- 11. Click the OK button in the Windows Firewall dialog box.
- 12. Close the Network and Internet Connections.
- 13. Close the **Control Panel**.

# **10.2** Enabling network facilities to share projects

#### To use network facilities to share projects

- 1. Select [Setup -> Options]. The Options dialog box opens.
- 2. Select the **Network** tab.
- 3. Click the **Enable network data access** checkbox. This should add an administrator to the system without a password. The administrator is the only user that can add additional users to the system and change user access rights. The administrator has the highest level of access.
- 4. Before leaving the network dialog the administrator must set their password. It is not possible to leave this dialog box until this is completed. This is described in the following section.
- 5. Close the **Options** dialog box.



#### High-performance Embedded Workshop V.4.09

- 6. When the dialog box is closed you are asked if you want to save the workspace and then re-open it. This is because the workspace must be re-opened in the shared access mode. If the changes are not saved then they will be lost. Click OK to share projects by network facilities.
- 7. When you re-open the workspace and log into the system, a dialog is displayed showing your current access rights. For example if you are the admin user the level will be administrator. When this dialog is closed the High-performance Embedded Workshop server window is opened and the network facilities to share projects are enabled.

	Network	
Enable network data acces	2	
Network database access:		
User:	Log in	Access rights
Password:	Password	Select server

# 10.3 Setting the administrator user's password

#### To set the administrator users password

- 1. Continue from the previous sections (step 4).
- 2. Click the **Password** button. This should have been enabled when the network data access was enabled.
- 3. The Change password dialog box opens.

Change password	? ×
Username	OK
Admin Password:	Cancel
Confirm Password:	

- 4. The user name is read only in the top field. In this case it should be Admin.
- 5. Type the new password into both of the fields and click OK.
- 6. This should set the user and password on the **Network** tab of **Options** dialog box.

# 10.4 Adding new users to the system

The initial setting of the network database adds an administrator user and a guest user to the system. The following levels of access are possible in the High-performance Embedded Workshop system:

Administrator	Full access to every aspect of High-performance Embedded Workshop. The user can add and remove users from the projects and change access rights. The administration user can change the workspace and project files and also the source files.
Full read/write access	The workspace and project files can be modified, as can the source files. But it is not possible to change user access rights from this access level.
Read/write file access	Only the source files can be modified. All project settings can only be viewed not modified.
Read only	All source files and project files can only be viewed as read only. Nothing can be modified.

When any user opens a network-enabled project they must type in their user name and password. Until this is done no access can be granted. Once entered the user is given one of the levels of access as seen above.

#### To add a new user to the system

- 1. Log in with a user who has administrator access rights. The process for doing this is described above.
- 2. Select [Setup -> Options]. The Options dialog box opens.
- 3. Select the **Network** tab.
- 4. Click the Access rights button. The User access rights dialog is displayed.
- 5. Click the **Add** button. The **Add new user** dialog box opens. This allows you as the administration user to add a new log in name and password. Normally the password should be set to some default text or left blank. Then click OK.
- 6. Once OK is clicked the user is added with read only rights. To change the access level select the user you wish to modify and then click the required radio button. Then click OK to save the access rights changes.

#### To remove an existing user to the system

- 1. Log in with a user who has administrator access rights. The process for doing this is described above.
- 2. Select [Setup -> Options]. The Options dialog box opens.
- 3. Select the **Network** tab.
- 4. Click the Access rights button. The User access rights dialog is displayed.
- 5. Select the user you wish to remove in the users list.
- 6. Press the **Remove** button.
- 7. Then click OK to save the access rights changes.



User access rights		? ×
	User access rights Users Access level Admin Admin Guest R Access: Access: Administrator Full read/write access Read/write file access only Read only access	<u>A</u> dd <u>R</u> emove
	OK	Cancel

# 10.5 Changing your password

#### To change your password

- Log into the High-performance Embedded Workshop network database you are changing your password for. Select [Setup -> Options]. The Options dialog box opens.
- 2. Select the **Network** tab.
- 3. Click the **Password** button.
- 4. Enter your new password and confirm it in the second edit box.
- 5. Click OK.
- 6. Then click OK to save the password change.

# 10.6 Note on using the network facilities to share projects

When you share a networked project for the first time the High-performance Embedded Workshop automatically connects you to the correct network High-performance Embedded Workshop service. This is defined using machine name. If the workspace cannot be found using the machine name then the dialog below appears. Simply type or browse the location of the machine and click OK. If you want to be the server machine then leave the radio button on its default selection, use local machine.

	08
🔿 Remote	Cancel
Computer name:	
	<u>B</u> rowse



If you have previously been the server of a workspace then the following message will be displayed when you attempt connection to another machine. Clicking Yes then connects your machine to the new location.

High-perl	rformance Embedded Workshop	×
To share this workspace requires the workspace to be closed and re You have modified the workspace. Do you want to save the modifica		
	Yes <u>N</u> o Cancel	

#### Note:

If the network is running multiple High-performance Embedded Workshop workspaces with the network facilities to share projects enabled then a user can only access one of them at one time. The only instance when this is not the case is if the same machine is serving all of the network workspaces.



# 11. Comparing Files

The High-performance Embedded Workshop has an integrated difference view. You can perform a difference comparison with two files on your local drive or a local file and a file in the Microsoft® Visual SourceSafe system.

# **11.1** Opening the Difference window

#### To perform a difference comparison with two files on your local drive

- 1. Select one of the following operations:
  - Select a file in the Projects tab of the workspace window. Right-click on the selected file to open a pop-up menu. Select **Show Differences**, **OR**
  - Click the Compare Files toolbar button (P), OR
  - Select [View -> Differences]. Right-click within the window to open a pop-up menu. Select [Compare].
- 2. The **Compare Files** dialog box opens.

Compare Files	? ×
<u>R</u> eference file:	Compare
orkspace\DemoSH4\DemoSH4\dbsct.c 💌 <u>B</u> rowse	Cancel
• Compare with <u>fi</u> le on drive	Cancel
Target file:	<u>A</u> dvanced
Browse	
${f C}$ Compare with version control	

- 3. Ensure the **Compare with file on drive** radio button is enabled.
- 4. Enter the name of a file to be compared to in **Reference file**. If you have opened the **Compare Files** dialog box via the workspace window at step 1, the file name is already entered in **Reference file**. Select a previously used file from the drop-down list box or click the **Browse** button to browse a file.
- 5. Clicking the **Advanced** button displays the **Compare Files Advanced Options** dialog box. This allows you to perform the difference comparison without taking white space into account. Click OK when you are finished with this options dialog box.

Compare Files - Advanced Options	? ×
- Advanced:	ОК
🔽 Ignore blanks	Cancel

6. Click Compare.



#### To perform a difference comparison with a local file and a file in Visual SourceSafe system

- 1. The High-performance Embedded Workshop must be connected to the Visual SourceSafe system. Add the two files you wish to compare to the Visual SourceSafe system in advance.
- 2. Select one of the following operations:
  - Select a file in the Projects tab of the workspace window. Right-click on the selected file to open a pop-up menu. Select **Show Differences**, **OR**
  - Click the Compare Files toolbar button (P), OR
  - Select [View -> Differences]. Right-click within the window to open a pop-up menu. Select Compare.
- 3. The **Compare Files** dialog box opens.

Compare Files	? ×
Reference file: orkspace\DemoSH4\DemoSH4\dbsct.c 💌 Browse	
C Compare with <u>fi</u> le on drive	Cancel
Target file:	<u>A</u> dvanced
Browse.	
<ul> <li>Compare with version control</li> </ul>	

- 4. Click the **Compare with version control** radio button. This radio button is selectable if the High-performance Embedded Workshop has been connected to the Visual SourceSafe system.
- 5. Enter the name of a file to be compared to in **Reference file**. If you have opened the **Compare Files** dialog box via the workspace window at step 2, the file name is already entered in **Reference file**. Select a previously used file from the drop-down list box or click the **Browse** button to browse a file.
- 6. Clicking the **Advanced** button displays the **Compare Files Advanced Options** dialog box. This allows you to perform the difference comparison without taking white space into account. Click OK when you are finished with this options dialog.
- 7. Click Compare.

The **Difference** window is displayed.



### Window configuration

	Difference	×	
Toolbar—	🖽 🗛 🥁 🗛 🛨		
Header —	der— C:\Workspace\DemoSH4\DemoSH4\dbsct.c Source-Safe version		
	16 <b>a</b> 17	16 17	
	<pre>*#pragma section \$DSEC ** static const struct { ** _UBYTE *rom_s; /* Start ** _UBYTE *rom_e; /* End ad ** _UBYTE *ram_s; /* Start ** } DTBL[] = { b</pre>	<pre>** #pragma section \$DSEC ** static const struct { *** _UBYTE *rom_s; /* Stat *** _UBYTE *rom_e; /* End *** _UBYTE *rom_s; /* Stat ***********************************</pre>	
	<pre>24 {sectop("D"),secend("D") 23 }; c</pre>	<pre> } DTBL[] = { 24 {sectop("D"),secend("I 23 }; </pre>	
	Beference file	Taroet file	

- In the Difference window, the left and right panes respectively show the files to be compared.
- Their names are at the header of each pane.
- If you hover the mouse cursor on the boundary of the two panes, the cursor turns into a double-sided arrow. To adjust the widths of the panes, drag the mouse to a desirable position.

#### Options

Right-clicking displays a pop-up menu containing available options.

A basic operation is allocated to the toolbar.

The **Toolbar display** and **Customize toolbar** options are also included in the pop-up menu opened by right-clicking on the toolbar.

Pop-up Menu Option	I oolbar Button	Macro Recording	Function
Compare	2.	-	This opens a new compare window so that some new files can be compared and the differences displayed.
Export results to file	-	-	This opens a dialog which allows you to choose a file to export the current difference results to a textual format.
Ignore white space	$\bowtie$	-	The ignore white space option which is on the advanced options dialog can be toggled via this menu option.
Find	#4	-	Displays a standard find dialog. This uses the same find dialog as the High-performance Embedded Workshop editor.
Find previous	<b>H</b>	-	Finds the next previous string that meets the find requirements.
Find next	A	-	Finds the next string that meets the find requirements.

# Pop-up Menu Option Toolbar Button Macro Recording Function



Pop-up Menu Option	<b>Toolbar Button</b>	Macro Recording	Function
Previous difference	1	-	Automatically jumps the view to the next previous
			difference.
Next difference	₽	-	Automatically jumps the view to the next difference.
Refresh comparison Refreshes the view to manually run the diff		Refreshes the view to manually run the difference	
			comparison again. This can be used if either file has
			been modified since the last comparison.
Toolbar display	-	-	Shows or hides the toolbar
Customize toolbar	-	-	Customizes toolbar buttons.



# **12. Navigation Facilities**

The High-performance Embedded Workshop has a number of new integrated navigation facilities.

Workspace	×
INT_OFFSET	
SR_Init	Navigation entries
C++ Classes	Navigation category
📮 🚠 CFish ————	Class
GetClassName(void)	Class members
m_nLiveinWater	
🖻 👬 CGreatWhite	
CGreatWhite()	
CGreatWhite()	
Navigation	

The **Navigation** tab of the workspace window contains categories for all supported navigation types. In Highperformance Embedded Workshop the following navigation components are supported as standard:

Navigation Type (Category)	Function
C Defines	All #defines for C and C++ source files are displayed.
C Functions	All ANSI C standard functions are for C source files displayed.
C++ Class	All classes, functions and members are displayed for C++ source files.

By default, it is possible to display the navigation items grouped by a navigation category.

Underneath each category, the navigation items belonging to the active project are displayed in the alphabetical order.

#### To group the display of navigation items by a navigation category

- 1. If you right-click anywhere inside the **Navigation** tab, a pop-up menu will be invoked.
- 2. Un-check the Group By File menu option. This option is unchecked by default.

The Navigation view lets you see the navigation items on a file-by-file basis.

Each file in the active project is shown in the tree, and the navigation items belonging to each file are displayed below it in the alphabetical order.



#### To group the display of navigation items by a file

- 1. If you right-click anywhere inside the **Navigation** tab, a pop-up menu will be invoked.
- 2. Check Group By File.



It is possible to disable scanning for certain navigation categories if you do not require the information.

#### To switch off a navigation category

- 1. If you right-click anywhere inside the **Navigation** tab, a pop-up menu will be invoked.
- 2. Select **Select Categories**. The **Select Categories** dialog box is displayed.



Select Categories	<u>×</u>
Please select the categories to scan for: ♥C Defines ♥C Functions □C++ Classes	Cancel

- 3. Un-check any categories you are not interested in seeing definitions for.
- 4. Click **OK**.

#### To update the navigation view

- 1. If you right-click anywhere inside the **Navigation** tab, a pop-up menu will be invoked.
- 2. Select Refresh.

### Notes:

- The navigation items are displayed gradually as the files are scanned. This means it may take some time if there are many files to fully complete the Navigation view update.
- Files are re-scanned when they are saved. This means that navigation information will not be available for new classes and functions until the file or files are saved.
- **Group By File** and Group By Access cannot both be on at the same time. Switching one on will switch the other one off.

# 12.1 C function and #define navigation component

These components simply add the function and #define definitions to the navigation view.

Workspace X	C define navigation
🕞 💼 C Defines 🗛	_ category title
	– #define
INT_OFFSET	C function
SR_Init	navigation category
🚊 🦲 C Functions — — — — — — — — — — — — — — — — — — —	– title 🗍 👘
	<ul> <li>ANSLC function</li> </ul>
abort	
📃 🔤 main 🔽 🚺	
Navigation	



#### To jump to a definition

Select either of the following ways.

- Double-click a function or a #define definition on the **Navigation** tab.
- Right-click on a function or a #define definition on the **Navigation** tab. Select **Go to Definition** from the pop-up menu.

# 12.2 C++ navigation component

The C++ navigation component supports the following structures in the view for C++ source files. The basic structure of the information is shown below.



The C++ navigation view uses a number of icons to describe the type of function or variable the icon belongs too. These are listed in the table below:

Icon	Description	
·· 📀	Public member function	
୍ବର	Protected member function	
- 🔊	Private member function	
·· 🕙	Public member variable	
ଂଙ୍କ	Protected member variable	
· 🚳	Private member variable	



#### High-performance Embedded Workshop V.4.09

Double clicking on a navigation item by default jumps you to the associated navigation items declaration. This default behavior can be modified by selecting **Jump To Definition On Double-Click** from the pop-up menu (this option is unchecked by default). When this option is checked, double-clicking a navigation item jumps you to the associated navigation items definition.

#### To jump to the definition

Select either of the following ways.

- Right-click on navigation items on the **Navigation** tab to display a pop-up menu and check that **Jump To Definition On Double-Click** is checked. Double-click a navigation item on the **Navigation** tab.
- Right-click on navigation items on the Navigation tab to display a pop-up menu. Select Go to Definition.

#### To jump to the declaration

Select either of the following ways.

- Right-click on navigation items on the **Navigation** tab to display a pop-up menu and check that **Jump To Definition On Double-Click** is unchecked. Double-click a navigation item on the **Navigation** tab.
- Right-click on navigation items on the Navigation tab to display a pop-up menu. Select Go to Declaration.

#### To list the member variables and functions in the alphabetical order

- 1. Right-click on navigation items on the **Navigation** tab to display a pop-up menu.
- 2. Un-check Group By Access. This option is unchecked by default.

#### To group the display of public, private, and protected member variables and functions together

- 1. Right-click on navigation items on the **Navigation** tab to display a pop-up menu.
- 2. Check Group By Access. This option is unchecked by default.

Note that **Group By File** and **Group By Access** cannot both be on at the same time. Switching one on will switch the other one off.

Another useful facility is the capability of viewing the base or derived classes for a certain selection.

#### To view the Base or derived classes

1. Right-click on class on the **Navigation** tab to display a pop-up menu.



- 2. To see the derived classes for the selection click the **Show Derived Classes** menu option. To see the base classes for the selection click the **Show Base Classes** menu option.
- 3. Depending on the selection a dialog is displayed which shows the class structure selected in an expanded tree format.
- 4. Click **Close** to close this dialog once you have the information you require.

# 12.3 Jump to a definition from the editor

It is possible to select navigation items of #defines, C functions, or C++ classes from the source codes shown in the editor window and view the positions where these navigation items are defined.

Note, however, that these navigation items must be under the categories checked on the Navigation tab of the workspace window.

#### To jump to a definition

- 1. Right-click on a navigation item, which you wish to view where it is defined, within the **Source** field of the Editor window.
- 2. Select Go To Definition Of 'Navigation item' from the pop-up menu.



3. If two or more navigation items are found, the **Select Navigation Item** dialog box appears. File names and line numbers are listed in the dialog box. Double-click an item, or select an item and press the OK button.



S	elect Navigation Item	? ×
	Select the location of 'sort':	
	Filename	Line
	c:\workspace\demosh4\demosh4\demosh4.c	53
	c:\workspace\demosh4\demosh4\demosh4.c	57
	OK Cancel	

# 12.4 Drag and drop navigation items

It is possible 'drag and drop' navigation items of #defines, C functions, or C++ classes shown in the Navigation tab of the workspace window. The Category (C Defines, C Functions, and C++ Classes), Globals Folder, and File items will not be draggable.

#### Drag and drop of navigation items into a file that's currently open in the editor

This makes it easier for you to write code, as you will be able to drag the relevant items into the file, instead of having to type them.



- Multiple items can be dragged at the same time (these will appear on separate lines).
- The navigation item names will be inserted in alphabetical order.
- Note that if you drag a function, only the function's name will be dragged.



# 12.5 Smart edit capability

Another feature of the High-performance Embedded Workshop is its smart edit facility. This is enabled by default for all C++ source files. This feature allows the High-performance Embedded Workshop editor to access C++ navigation information and provide auto-completion help when using C functions, #defines, C++ classes, and member functions.

The High-performance Embedded Workshop editor accesses C functions, #defines, and C++ navigation information and shows the smart editor's list in a pop-up window.

#### To use the Smart edit facilities

- 1. Select [Setup -> Options]. The Options dialog box opens.
- 2. Select the **Editor** tab.
- 3. The Enable Smart-edit should be checked.
- 4. Click OK.

Enable sma<u>r</u>t-editor

With this option switched on if you are working on C or C++ file the smart-edit capability should be enabled.

#### Note:

When the navigation category **C Defines**, **C Functions**, or **C++ Classes** is unchecked in the Navigation tab of the workspace window, the High-performance Embedded Workshop's smart editor function is disabled.



#### High-performance Embedded Workshop V.4.09

During normal usage the following editor operations will make the smart edit facilities visible.

• If you are using an object and are trying to access the members using the '.' or '->'. If you do this a pop-up window will be displayed which may help you select the correct member more efficiently than typing. Whilst typing the pop-up window will keep track of the keys you have pressed to help your selection. If you press return then the currently selected member will be added. This pop-up window is also used when using the '::' method and it is displayed in figure below. Pressing CTRL+SPACE displays the pop-up window. This pop-up window disappears when a member is added.



• If you are trying to use C or C++ functions then the pop-up window in Figure below is displayed when the first open bracket is entered. This pop-up window allows you to see what functions are available for the current object. Selecting the function automatically enters the remaining parameters for you. Pressing CTRL+SHIFT+SPACE opens this pop-up window, which is visible until a closing bracket is entered.

class Sample *p_sam;
p_sam = new Sample; p_sam->change(
for( i=0; void change(long *a)
if(j < 0){ i = -i;
]]; }
a[i] = j; }

- It is possible to select a C function, #define, C++ class, or member function in the pop-up window and add it to the High-performance Embedded Workshop editor window by either of the following ways:
  - Right-click within the High-performance Embedded Workshop editor window to display a pop-up menu and select **List Members**. A pop-up window opens. Select an item you wish to add and press ENTER, **OR**
  - o Press CTRL+SPACE to display a pop-up window. Select an item you wish to add and press ENTER.



# 13. Map

Among the section settings of the linkage editor and the contents of the linkage list file output by the linkage editor, information on sections and symbols are respectively shown. There are the following features.

## Sections

- A tree view in the form of address section group section
- Assignment of sections can be changed by drag and drop
- You can add, modify, or delete addresses, section groups, and sections
- The source file for the address in the selected line can be opened in the editor window
- The contents can be printed out

#### Symbols

- Lists all symbol information or those for respective sections
- You can search for, filter, or sort (this function is not supported by the evaluation version) symbol information
- The source file for the address in the selected line can be opened in the editor window
- The contents can be printed out

#### • Supported toolchains

The toolchains included in the following compiler packages support the map function.

- o C/C++ Compiler Package for SuperH Family V.7.1.03 or later
- o C/C++ Compiler Package for H8SX, H8S, H8 Family V.5.0.05 or later
- o C Compiler Package for M16C Series and R8C Family V.5.42 Release 00 or later
- o C Compiler Package for M32C Series V.5.41 Release 00 or later
- o C Compiler Package for R32C Series V.1.01 Release 00 or later
- o C/C++ Compiler Package for RX Family V.1.00 Release 00 or later

#### • To view the information on sections and symbols

To view the information on sections and symbols, the following setting is required in advance:

1. Select [**Build -> xxxxx Standard Toolchain**] to open the build options dialog box.

- 2. Select List for Category in the Link/Library tab.
- 3. Select the Generate list file checkbox. This allows output of information on sections.
- 4. Click the **Enable all** button. This allows output of information on symbols.
- 5. Select [**Build -> Build**].

C/C++ Assembly Link/Library Standard I	.ibrary CPU 💶
Category : List	
☑ <u>G</u> enerate list file ┌ Contents : ────	
<ul> <li>✓ Show symbol</li> <li>✓ Show reference</li> <li>✓ Show section</li> <li>✓ Show cross reference</li> </ul>	<u>Enable all</u> Disable all
Options Link/Library : -noprelink -rom=D=R -nomessage -list=''\$(CONFIGDIR)\\$(PROJECTNAME).ma -show=symbol,reference,xreference -nooptim	

#### Note:

If a subcommand file is specified as shown below, the section setting information of the linkage editor will not be shown in the **Map Section Information** window.

C/C++ Assembly Link/Library Standard Library CPU
Category : Subcommand file
☑ Use external subcommand file
Subcommand file path :
\$(CONFIGDIR)\file1.sub
Options Link/Library :
-subcommand=\$(CONFIGDIR)\file1.sub



# 13.1 Managing section settings

You can add, modify, or delete information on sections on the GUI through the Map Section Information window. It is also easy to check the section settings after modification.

# 13.1.1 Opening the Map Section Information window

#### To open the Map Section Information window

- 1. Select one of the following operations to open the Select Map Window Type dialog box:
  - Select [View -> Map], OR
  - Click the **Map** toolbar button (**PPP**).
- 2. Select "Map Section Information" in the **Map** drop-down list.
- 3. Click OK.

### Window configuration

Map Section Information						×
Toolbar _	🖻 # # A 🗛 🛛 📾 🔀 🎒	<b>e</b>				
	🖃 🏧 Linker section setting 🔺	Name	St 🔺	End A	Size	0.
Address -	🖶 🛱 A000000	INTHandler	00000800	00000⊅57	00000558	
Section group	RSTHandler	VECTTBL	00000058	00000D83	0000002c	
group	🖨 🖧 00000800	IntPRG	00000084	00000D87	00000004	
	🗄 📲 INTHandler, VECTTBL	PResetPRG	00001000	0000106F	00000070	
	⊡…‱ 00001000	Р	00002000	000067BD	000047BE	- 1
	PResetPRG	С	000067c0	00006в43	00000384	- 1
		C\$BSEC	00006в44	00006B4B	00000008	- 1
		C\$DSEC	00006B4C	00006B57	0000000c	- 1
Sub-section	⊡ 📲 P,C,C\$BSEC,C\$DSEC,1	D	00006в58	00006в63	0000000c	- 1
group — Overlay	₽,C,C\$BSEC,C\$DSE	POverlayPRG	00002000	0000208F	00000090	Р
group	POverlayPRG	В	70000000	700005EB	000005EC	
	🖻 🖧 7000000	R	700005EC	700005F7	0000000c	- 1
		S	73FFFBFO	73ffffef	00000400	- 1
Section -	• B •	RSTHandler	A0000000	A000001B	0000001c	
1	·	1	1	1		_
	L	L				┯┛
		Section Name	Start Addres	End Addres	s Size Ov	/erlay
The map info	The map information and section settings of the linkage editor The section settings after a build					



- The left pane shows the map information and section settings of the linkage editor.
  - Section settings of the linkage editor are displayed.
  - While in the edit mode, each of subsection groups, overlay groups, and sections can be dragged and dropped.

User Operation	Appearance of the Mouse Pointer while an Item is Dragged	Position of the Dropped Item
Select a subsection group ( <sup>III</sup> ) or	At ITEM	Copied to the next of the item where dropped
section ( <sup> </sup> ) and drag it while holding the Ctrl key	'Normal select' pointer showing the item name and a '+' sign when placed over a destination where	
Select an overlay group ( <sup>111</sup> ) and drag it while holding the Ctrl key	the item can be copied to	Copied to the last of the items
Select a subsection group ( <sup>III</sup> ) or	Mr ITEM	Added before the item where dropped
section ( <sup> </sup> ) and drag it while holding the Shift key	'Normal select' pointer showing the item name and an upward-pointing arrow when placed over a destination where the item can be moved to	
Select a subsection group ( <sup>III</sup> ) or	👌 ITEM	Added to the next of the item where dropped
section ( 🎴 ) and drag it	- 'Normal select' pointer showing the item name	
Select an overlay group (	when placed over a destination where the item can be moved to	Added to the last of the items

- The right pane shows the section settings after a build.
  - Nothing is displayed in this pane if there is no linkage list file (.map) output by the linkage editor.
  - You can adjust the column width by dragging the mouse on a column header. Moreover, right-clicking the column header displays a pop-up menu. A tick mark right next to an entry indicates that this column is displayed. Clicking an entry will switch showing/hiding the column.
  - Clicking the column header sorts the listed items. When the window is first opened, the items are sorted by the value of **Start Address**.
  - The **Overlay** column shows the name of the primary section.

#### Window options

• Map information and section settings of the linkage editor (left pane)

Right-clicking displays a pop-up menu containing available options.

A basic operation is allocated to the toolbar.

The **Toolbar display** and **Customize toolbar** options are also included in the pop-up menu opened by right-clicking on the toolbar.



Pop-up Menu Option	<b>Toolbar Button</b>	Function
Section Edit Mode	1	Enters the edit mode.
Add Section Group	<b>.</b> 88	Adds the section group.
Add Section	°"	Adds the section.
Add Overlay Group *1	<b>8</b>	Adds the overlay group.
Section Auto Registration		Automatically adds missing sections.
Edit Selected Item	20	Edits the selected item.
Cut	ж	Cuts the section information.
Сору	E	Copies the section information.
Paste	<b>E</b>	Pastes the section information.
Delete	$\mathbf{x}$	Deletes the section information.
Move up	û.	Moves up the section information.
Move Down	🖑	Moves down the section information.
Change Primary Section *1		Changes primary section.
Memory Map Setting *2	-	Sets a memory map.
Memory Resource Auto Allocate *2		Automatically allocates the memory resource.
Print	<b>a</b>	Prints section setting tree view.
Toolbar display	-	Shows or hides the toolbar.
Customize toolbar	-	Customizes toolbar buttons.

#### Notes:

\*1. This menu option cannot be used when the linkage editor does not support overlay.

\*2. The following debuggers support this facility.

- The Simulator Debugger for the SuperH Family
- The Simulator Debugger for the H8SX, H8S, H8 Family
- The Simulator Debugger for the RX Family
- Section settings after a build (right pane)

Right-clicking displays a pop-up menu containing available options.

A basic operation is allocated to the toolbar.

The **Toolbar display** and **Customize toolbar** options are also included in the pop-up menu opened by right-clicking on the toolbar.

Pop-up Menu Option	<b>Toolbar Button</b>	Function
Show Unallocated Area		Shows the unallocated area.
Show No Size Section	0	Shows the no size section.
View source	E	Views the source code for the address in the selected line.
Print	9	Prints section list view.



#### 13.1.2 Entering/exiting the edit mode

To edit the section information shown in the left pane, the High-performance Embedded Workshop must enter the edit mode. By default, the edit mode is not selected.

#### To enter the edit mode

- 1. Right-click within the left pane to open a pop-up menu.
- 2. Select Section Edit Mode. This allows the High-performance Embedded Workshop to enter the edit mode.
- 3. There is a tick mark on **Section Edit Mode**.

While the High-performance Embedded Workshop is in the edit mode, the options listed in the table below are available.

Right-click on an item in the left pane. The following options with "♥ " in the table can be selected.

Pop-up Menu Option	-up Menu Option Toolbar				Item		
	Button	Root	Address	Section group	Subsection group	Overlay group	Section
		22					
Add Section Group	<b>.</b> ##	~	~	~	~	~	~
Add Section	°"		~	~	~	~	~
Add Overlay Group *1	37			~	~		✓ *2
Section Auto Registration	<b>1</b>	~	~	~	~	~	~
Edit Selected Item	24		~	~	~	~	~
Cut	Ж				~	~	~
Сору	Ē				~	~	~
Paste	Ē.		~	~	~	~	~
Delete	×		~	~	~	~	~
Move up	û.						~
Move Down	<b>4</b>						~
Change Primary Section *1	2					¥	
NT /							

Notes:

\*1. This menu option cannot be used when the linkage editor does not support overlay.

\*2. Will be supported by the toolchains included in the compiler packages listed below.

- C/C++ Compiler Package for SuperH Family V.9.00 Release 04 or later
- C/C++ Compiler Package for H8SX, H8S, H8 Family V.6.01 Release 02 or later
- C/C++ Compiler Package for RX Family V.1.00 Release 00 or later
- C/C++ Compiler Package for M16C Series and R8C Family V.6.00 Release 00 or later



#### To exit the edit mode

- 1. Right-click within the left pane to open a pop-up menu.
- 2. Select Section Edit Mode.
- 3. The confirmation dialog box shown below opens. To reflect the changes to the section settings of the linkage editor, select Yes.



4. The High-performance Embedded Workshop exits the edit mode. The tick mark on **Section Edit Mode** has disappeared.

#### 13.1.3 Adding a section group

In the edit mode, you can add a section group.

#### To add a section group

- 1. Right-click within the left pane to open a pop-up menu.
- 2. Select Add Section Group.
- 3. The Add Section Group dialog box appears.



- 4. Enter the start address of a section in Section Start Address.
- 5. Enter a section group name in **Section Group Name**. If you specify section names separated by a comma, each of them will be added as a section. In the example above, "Section1" and "Section2" are added as sections.



#### 13.1.4 Adding a section

In the edit mode, you can add a section under a section group, subsection group, or overlay group.

#### To add a section

- 1. Right-click on a section group, subsection group, or overlay group to open a pop-up menu.
- 2. Select Add Section.
- 3. The Add Section dialog box appears.

Add Section	? ×
Section <u>N</u> ame:	OK
Section3	Cancel

4. Enter a section name in **Section Name**.

In the tree of the left pane, the new section is added under the section group, subsection group, or overlay group.

#### 13.1.5 Adding an overlay group

In the edit mode, you can add an overlay group \*1 under a section group, subsection group, or section \*2.

#### To add an overlay group

- 1. Right-click on a section group, subsection group, or section to open a pop-up menu.
- 2. Select Add Overlay Group.
- 3. The Add Overlay Group dialog box appears.





4. Enter an overlay group name in **Overlay Group Name**.

In the tree of the left pane, the new overlay group is added under the section group, subsection group, or section.

#### Notes:

- \*1. This item cannot be used when the linkage editor does not support overlay.
- **\*2**. Will be supported by the toolchains included in the compiler packages listed below.
  - C/C++ Compiler Package for SuperH Family V.9.00 Release 04 or later
  - C/C++ Compiler Package for H8SX, H8S, H8 Family V.6.01 Release 02 or later
  - C/C++ Compiler Package for RX Family V.1.00 Release 00 or later
  - C/C++ Compiler Package for M16C Series and R8C Family V.6.00 Release 00 or later

#### 13.1.6 Automatically registering the unregistered section

In the edit mode, if any of the sections listed in the right pane is missing in the left pane, this section can automatically be added to the left pane.

#### To automatically add missing sections

- 1. Right-click within the left pane to open a pop-up menu.
- 2. Select Section Auto Registration.
- If some sections are missing in the left pane and the right pane includes a section group whose address value is smaller than the smallest address among the missing sections, all of these sections will be added to the last in the section group.



• If some sections are missing in the left pane and none of the addresses of section groups listed in the right pane is smaller than the smallest address among the missing sections, all of these sections will be added as a section group at the smallest address among the missing sections.

#### Note:

If an overlay section is missing, this will be added as a normal section.

#### 13.1.7 Editing a selected item

In the edit mode, you can also modify the address value and the name of a section group, subsection group, overlay group, or section.

#### To modify the selected item

- 1. Select either of the following operations:
  - Right-click on an item in the left pane to open a pop-up menu. Select Edit Selected Item, OR
  - Double-click to modify a value.
- 2. The **Edit Selected Item** dialog box appears.
- 3. Enter a string or integer value into an edit field.

Edit Selected Item	? ×
Section Start <u>A</u> ddress:	OK
A000000	Cancel
Section Group <u>N</u> ame:	
RSTHandler	

#### 13.1.8 Setting the primary section

In the edit mode, you can set a selected overlay group as the primary section.\*

#### To set the primary section

- 1. Right-click on an overlay group in the left pane to open a pop-up menu.
- 2. Select Change Primary Section.


In the tree of the left pane, the selected overlay group is set as the primary section.

#### Note:

\*. This function cannot be used when the linkage editor does not support overlay.

#### 13.1.9 Setting a memory map

While the High-performance Embedded Workshop is connected with a debugger, you can set a memory map.

#### To set a memory map

- 1. Right-click within the left pane to open a pop-up menu.
- 2. Select Memory Map Setting.
- 3. The memory map setting dialog box opens. The current mapping of the address space is shown as a list. Set a memory map or memory resource as necessary.
- 4. Click the OK button.

#### Note:

The following debuggers support this facility.

- The Simulator Debugger for the SuperH Family
- The Simulator Debugger for the H8SX, H8S, H8 Family
- The Simulator Debugger for the RX Family

# 13.1.10 Automatically allocating the memory resource

When there is a linkage list file (.map) output by the linkage editor, the memory resource can be automatically allocated according to the memory map and linkage map information. While the High-performance Embedded Workshop is connected with a debugger, you can allocate a memory resource.

#### To automatically allocate the memory resource

- 1. Right-click within the left pane to open a pop-up menu.
- 2. Select Memory Resource Auto Allocate. The dialog box shown below appears.





- 3. To continue processing, click OK.
- 4. When allocation is completed, the dialog box shown below appears.



#### Note:

The following debuggers support this facility.

- The Simulator Debugger for the SuperH Family
- The Simulator Debugger for the H8SX, H8S, H8 Family
- The Simulator Debugger for the RX Family

#### 13.1.11 Printing out the section settings tree

You can also print out the section settings tree in the left pane.

### To print out the section settings tree

- 1. Right-click within the right pane to open a pop-up menu.
- 2. Select **Print**.
- 3. The standard print formatting and selection dialog box opens. From here you can choose your printer and page setup options.



# 13.1.12 Viewing unallocated areas

You can view areas where no section is allocated. By default, this function is not selected.

#### To view unallocated areas

- 1. Right-click within the right pane to open a pop-up menu.
- 2. Select Show Unallocated Area.
- 3. There is a tick mark on **Show Unallocated Area**. The right pane shows all areas where no section is allocated, which is indicated by "Unallocated Area" in the Name column.

### 13.1.13 Viewing sections of size 0

You can view sections of size 0. By default, this function is not selected.

### To view sections of size 0

- 1. Right-click within the right pane to open a pop-up menu.
- 2. Select Show No Size Section.
- 3. The right pane shows all sections of size 0. At this time, the "End Address" column does not show the end addresses.

### 13.1.14 Viewing the source code for the address

The source file for the address in the selected line will be opened in the editor window.

#### To view the source code for the address

Select one of the following operations:

- Right-click on the section-list line in the right pane to open a pop-up menu. Then select View Source, OR
- Double-click on the section-list line in the right pane.

# 13.1.15 Printing out the section list

You can also print out the section list in the right pane.

### To print out the section list

- 1. Right-click within the right pane to open a pop-up menu.
- 2. Select Print.
- 3. The standard print formatting and selection dialog box opens. From here you can choose your printer and page setup options.

# 13.2 Viewing symbols

You can view symbols in each of the sections in the Map Symbol Information window.

Even if there are too many symbols, the search or filtering function can be used to view necessary information only.

# 13.2.1 Opening the Map Symbol Information window

#### To open the Map Symbol Information window

- 1. Select one of the following operations to open the Select Map Window Type dialog box:
  - Select [View -> Map], OR
  - Click the **Map** toolbar button (**PPP**).
- 2. Select "Map Symbol Information" in the **Map** drop-down list.
- 3. Click OK.

13. Map



### Window configuration

🖃 🗒 Link m	nap list 🔺	Name	St 🔺	End A	Size	At	Scope	Sec	Fil∈_
- 💑 [00	000800 -	INTHan	00000800		00000000	None	Global	INT	vhar
- 🖧 (00	000⊅58 -	ExpHan	00000800		00000000	None	Local	INT	vhar
🔏 (oo	000084 -	int_term	00000858		00000000	None	Local	INT	vhar
001 🖧	000084 -	_TLBmis	00000в00		00000000	None	Local	INT	vhar
- 🔏 [00	001000 -	_IRQHan	000000000		00000000	None	Local	INT	vhar
	002000 -	_RESET	00000D58		00000000	None	Global	VEC	vect
	0067c0 -	_INT_MASK	000000084		00000000	None	Global	INTTBL	vect
	006в44 -	_INT_Ve	000000084		00000000	None	Global		vect
	оосви <u>–</u> Ообв4с –	TBL_Reset			00000000	None	Global		intr
	ОО6B58 -	PowerO	00001000	00001049	0000004A	Entry	Global		rese
	000000	_Manual	0000104A	0000106F	00000026	Fu	Global		rese
	<b>_</b> _	main	00002000	000020 <b>E</b> 7	000000E8	Fu	Global	P	Demc.
	•								
				ا		┛┖──┯─	┛┖─┯──	لــــالــــــــــــــــــــــــــــــــ	
		Symbol Name		End Address		ymbol Attr		Section Na	

The map list of the linkage editor

The symbol information of the linkage editor

- The left pane shows the map list of the linkage editor.
  - Nothing is displayed in this pane if there is no linkage list file (.map) output by the linkage editor.
  - Selecting the root of the map list ("Linker map list") shows all symbol information in the right pane.
  - Selecting a section in the map list only shows symbol information of the selected section in the right pane.
- The right pane shows the symbol information of the linkage editor.
  - Clicking the column header sorts the listed items. When the window is first opened, the items are sorted by the value of **Start Address**.
  - The Attribute column shows the symbol attribute.

Entry:	Entry function
Function:	Function name
Data:	Variable name
None:	Undefined (label or assembly symbol)

- The **Scope** column shows the scope attribute.

Global:	Global symbol
Local:	Local symbol



### Window options

• Map list of the linkage editor (left pane)

Right-clicking displays a pop-up menu containing available options.

A basic operation is allocated to the toolbar.

The **Toolbar display** and **Customize toolbar** options are also included in the pop-up menu opened by right-clicking on the toolbar.

Pop-up Menu Option	<b>Toolbar Button</b>	Function
Print	<b>a</b>	Prints section tree view.
Toolbar display	-	Shows or hides the toolbar.
Customize toolbar	-	Customizes toolbar buttons.

• Symbol information of the linkage editor (right pane)

Right-clicking displays a pop-up menu containing available options.

A basic operation is allocated to the toolbar.

The **Toolbar display** and **Customize toolbar** options are also included in the pop-up menu opened by right-clicking on the toolbar.

Pop-up Menu Option	<b>Toolbar Button</b>	Function
Find	<b>#4</b>	Finds a symbol.
Find Next	<b>A</b>	Finds the next symbol that matches the search text.
Filter	<b>6</b> •	Filters the symbol information.
View source	R	Views the source code for the address in the selected line.
Print	9	Prints section list view.

# 13.2.2 Printing out the map list

You can also print out the map list in the left pane.

#### To print out the map list

- 1. Right-click within the left pane to open a pop-up menu.
- 2. Select **Print**.
- 3. The standard print formatting and selection dialog box opens. From here you can choose your printer and page setup options.



### 13.2.3 Finding symbols

You can search for the linkage editor symbols.

#### To search for symbols

- 1. Right-click within the right pane to open a pop-up menu.
- 2. Select Find. The Find Symbol dialog box appears.



- 3. Enter a symbol name you wish to find in **Name**, or select one from those you have previously searched for in the drop-down list box.
- 4. To distinguish uppercase and lowercase characters, select the Match case checkbox.
- 5. To use wild-cards (? or \*), select the **Wild card** checkbox.
  - ?: A character
  - \*: A string
- 6. Click OK.

When a symbol is found, a line that contains the symbol is highlighted in the right pane.

To find the next occurrence of the symbol, select Find Next.

# 13.2.4 Filtering the symbol information

You can view the filtered symbol information of the linkage editor.

# To filter the symbol information

- 1. Right-click within the right pane to open a pop-up menu.
- 2. Select Filter. The Filter Symbol dialog box appears.



Filter Symbol	? ×
Enable Filter	
Eilter Column: Attribute	•
Filter <u>I</u> tem:	
✓Function ✓Data	<u>S</u> et All
<ul> <li>✓ Data</li> <li>✓ Entry</li> <li>✓ None</li> </ul>	<u>C</u> lear All
OK	Cancel

- 3. Select the **Enable Filter** checkbox. By default, this checkbox is not selected.
- 4. Select a column in Filter Column.
- 5. Select a filtering condition in **Filter Item**.

Filter Column	Filter Item	Function
Attribute (symbol attribute)	Function	Filter by functions
	Data	Filter by data symbols
	Entry	Filter by positions to start execution
	None	Filter by other symbols (such as labels)
Scope (scope attribute)	Global	Filter by global symbols
	Normal	Filter by local symbols
Section (sections containing symbols)	Section name	Filter by section names
File (files containing symbols)	File name	Filter by file names

- 6. Clicking the **Set All** button selects the check boxes of all filtering conditions.
- 7. Clicking the **Clear All** button de-selects the check boxes of all filtering conditions.
- 8. Click OK.

Symbol information of the linkage editor shown in the right pane is filtered by the selected condition. For further filtering by combining a different condition, select another column and then select a filtering condition in the column.

To disable filtering, de-select the **Enable Filter** checkbox in the **Filter Symbol** dialog box.



# 13.2.5 Viewing the source code for the address

The source file for the address in the selected line will be opened in the editor window.

# To view the source code for the address

Select one of the following operations:

- Right-click on the symbol-list line in the right pane to open a pop-up menu. Then select View Source, OR
- Double-click on the symbol-list line in the right pane.

# **13.2.6 Printing out the symbol information**

You can also print out the symbol information in the right pane.

# To print out the symbol information

- 1. Right-click within the right pane to open a pop-up menu.
- 2. Select Print.
- 3. The standard print formatting and selection dialog box opens. From here you can choose your printer and page setup options.



# 14. Command Line

The High-performance Embedded Workshop Command Line Interpreter allows the user to control the debugger by sending text-based commands instead of the window menus and commands. It is especially useful if a series of predefined commands need to be sent to the debugger by calling them from a batch file and, optionally, recording the output in a log file.

### Notes:

# • Specifying a file in the command line

To specify a file in the command line, use a placeholder (excluding TCL). If you wish to specify a directory not included in the placeholder, specify an absolute path. After specifying the absolute path, this file will not be correctly found when it is in another host computer or environment where the path content is different. In such cases, specify the file again.

FILE\_LOAD ELF/DWARF2 \$(CONFIGDIR)\\demo.abs

### • Separating command parameters

You can use TCL commands while TCL is enabled. The use of the TCL command batch, in which parameters are separated by tabs, is also possible.

For	{set	i	0}	{\$i	<	2}	{incr	i}	{
puts	[memor	y_displa	ıy	300	10]				
}									

When TCL is disabled, TCL commands are not usable. Only white space can be used to separate parameters so using tabs instead will cause the following error.

>TCL

TCL Disabled

>memory\_display 300 10

Error: Invalid command

# 14.1 Opening the Command Line window

Choose [**View -> Command Line**] or click the **Command Line** toolbar button (**D**) to open the Command Line window. If available, the window title displays the current batch and log file names separated by colons.

The High-performance Embedded Workshop command and TCL commands can be input in this window. For information about the available commands, see Reference 3, Commands, and the on-line help.

The Command Line window has two tabs: Console and Debugger.



### Window configuration

### • Console tab

This pane allows the user to input text-based commands to control the target platform. The results of command execution shown on this pane are in synchronization with the lower pane of the **Debugger** tab.

	Command Line: C:\Workspace\Command.hdc: no log file: 🗵				
Toolbar —	📾 💟 💷 🔛 🖉 🏉 🗇				
Input	>#conmand				
commands	>reset				
and the	>step				
results of 🚽	Step complete at A0000002.				
command					
execution	>				
L	Console / Debugger /				

- The command can be executed by pressing the "Enter" key after the command is input at the prompt (>) on the last line in the window.
- Pressing the CTRL + UP ARROW or CTRL + DOWN ARROW keys on the last line in the window displays the previously executed command line.
- A series of command lines can be called from a batch file and the results can be output to a file. The contents of the selected batch file are shown in the upper pane of the Debugger tab.
- You can clear the information shown on this pane. The contents of the lower pane of the Debugger tab will also be cleared.

#### • **Debugger** tab

This pane allows the user to perform single stepping or set breakpoints in the selected batch file to control the target platform.

When a batch file has been loaded, placing the mouse cursor on the line dividing the panes turns the mouse cursor into a double-headed arrow. Click and drag the mouse cursor to a desired position to adjust the size of the panes.





• Upper pane (contents of the batch file)

The upper pane shows the contents of the selected batch file and allows debugging. However, you cannot modify data in this pane.

- The contents of the selected batch file are displayed.
- Single stepping of command lines can be performed in the selected batch file.
- Execution can start from the top of the batch file.
- A breakpoint (•) can be set in the Breakpoint column on the left to the command line.
- A yellow arrow ( $\clubsuit$ ) in the Breakpoint column indicates the command line where execution is to start.
- The line where execution is to start (i.e. the yellow arrow (♥)) can be returned to the top of the batch file.
- The results of command execution can be output to a file.
- o Lower pane (results of command execution)

The lower pane shows the results of command execution. This information is in synchronization with the contents of the Console tabbed pane. However, you cannot modify data in this pane.

• You can clear the information shown in this pane. The contents of the Console tabbed pane will also be cleared.



# Options

• Console tab

Right-clicking displays a pop-up menu containing available options.

A basic operation is allocated to the toolbar.

The **Toolbar display** and **Customize toolbar** options are also included in the pop-up menu opened by right-clicking on the toolbar.

Pop-up Menu Option		Toolbar Button	Macro Recording	Function
Set Batch File		<b>EXT</b>	-	Specifies a batch file.
Play		<b>1</b>	-	Plays the current batch file.
Stop		<u>st@</u>	-	Stops execution of the current batch file.
Set Log File		<b>6</b>	-	Specifies a log file.
Logging			-	Starts or stops output to log file.
Browse		2	-	Enters a full path to the file.
Placeholder	Configuration directory	Configuration Directory 💌	-	Pastes the \$(CONFIGDIR) placeholder.
	Configuration name		-	Pastes the \$(CONFIGNAME) placeholder.
	Project directory	-	-	Pastes the \$(PROJDIR) placeholder
	Project name	-	-	Pastes the \$(PROJECTNAME) placeholder.
	Workspace directory		-	Pastes the \$(WORKSPDIR) placeholder.
	Workspace name	-	-	Pastes the \$(WORKSPNAME) placeholder.
	HEW Installation directory		-	Pastes the \$(HEWDIR) placeholder.
Select All		<b>2</b>	-	Selects (i.e. highlights) the entire contents of the active window.
Сору		<b>W</b> B	-	Places a copy of the highlighted text into the Windows® clipboard.
Cut		¥	-	Removes the selected text block and copy them onto the Windows® clipboard.
Paste		2	-	Copies the contents of the Windows® clipboard into the active window at the position of the insertion cursor.
Clear Windo	W	Ø	-	Clears the contents of the Command Line window.
Undo		ഗ	-	Undoes the last operation.
-			-	Once an opening bracket "[" or "{" is input and the nesting follows, this button is visible. (This button selection is impossible.)
Toolbar disp	blay	-	-	Shows or hides the toolbar.
Customize to	oolbar	-	-	Customizes toolbar buttons.



### • **Debugger** tab

• Upper pane (contents of the batch file)

Right-clicking displays a pop-up menu containing available options.

A basic operation is allocated to the toolbar.

The **Toolbar display** and **Customize toolbar** options are also included in the pop-up menu opened by right-clicking on the toolbar.

Pop-up Menu Option	Toolbar Button	Macro Recording	Function
Set Batch File	BAT	-	Specifies a batch file.
Play	<b>1</b>	-	Plays the current batch file.
Stop	STOP	-	Stops execution of the current batch file.
Reset	-	-	Resets the status of a batch file.
Step	0+	-	Single-steps in a batch file.
Insert/Remove Breakpoint	-	-	Toggles a break point on the current line of a batch file.
Enable/Disable Breakpoint	-	-	Enables or disables the current breakpoint on the current line of a batch file.
Clear All Breakpoints	-	-	Removes all breakpoints of a batch file.
Set Log File	<b>5</b> 3	-	Specifies a log file.
Logging		-	Starts or stops output to log file.
Toolbar display	-	-	Shows or hides the toolbar.
Customize toolbar	-	-	Customizes toolbar buttons.

• Lower pane (results of command execution)

Right-clicking displays a pop-up menu containing available options.

A basic operation is allocated to the toolbar.

The **Toolbar display** and **Customize toolbar** options are also included in the pop-up menu opened by right-clicking on the toolbar.

Pop-up Menu Option	<b>Toolbar Button</b>	Macro Recording	Function
Clear Window	61	-	Clears the contents of the Command Line window.

# 14.2 Specifying a batch file

It is useful to use a batch file when a series of predefined command lines need to be executed. Create a batch file by a text editor and write necessary command lines. The default extension of a batch file is .hdc.

Choose **Set Batch File** from the pop-up menu to open the Set Batch File dialog box, in which the name of a batch file (\*.hdc) can be specified. Clicking the OK button displays the specified batch file name as the window title. Clicking the Cancel button closes the dialog box without modifying the setting.



Set Batch File	? ×
Batch File:	<u>P</u> lay
Browse	<u>о</u> к
	<u>C</u> ancel

After a batch file has been specified, the upper pane of the **Debugger** tabbed pane in the Command Line window shows the contents of the batch file. When saving the session, the batch file name will also be saved.

#### Notes:

- The batch file is automatically closed if any of the GUI features or command listed below is run. The information shown in the upper pane of the **Debugger** tabbed pane is cleared.
  - Closing a workspace (CLOSE\_WORKSPACE)
  - o Creating a new workspace
  - Inserting a project into the workspace
  - o Adding a session
- If you modify the contents of the batch file after it has been specified, the information shown in the upper pane of the **Debugger** tabbed pane in the Command Line window will be updated. The status of the batch file will be reset, with all breakpoints deleted.

If you modify the contents after the batch file has been specified, the High-performance Embedded Workshop shows a message telling that the batch file will be loaded again.

If you do not wish to open this confirmation dialog box, select the Don't ask this question again checkbox.

#### To open the confirmation dialog box again

- 1. Select [Setup -> Options]. The Options dialog box opens
- 2. Select the Confirmation tab.
- 3. Select the Auto-reload Batch File checkbox. This checkbox is selected by default.
- 4. Click OK.



# 14.3 Executing a batch file

Click the **Play** button in the **Set Batch File** dialog box or choose **Play** from the pop-up menu to execute the batch file. The **Play** menu is displayed in gray while the file is running and can be used when the batch file execution stops and control returns to the user.

The message box shown below appears when a user attempts to close the Command Line window during execution of a command file.

ecxcmdline			×
8	The batc	h file will be abo	rted!
	DK )	Cancel	

To terminate the execution of the batch file and close the Command Line window, click OK. To continue the execution, click Cancel. This does not close the Command Line window.

# 14.4 Stopping command execution

Choose **Stop** from the pop-up menu to stop command execution. The **Stop** menu becomes valid during command execution.

# 14.5 Specifying a log file

Choose **Set Log File** from the pop-up menu to open the **Open Log File** dialog box, in which a log file to store the command execution results can be specified.

Open Log File		? ×
Log File:	Browse	<u> </u>

Enter the name of a log file (\*.log). The logging option is automatically set and the name of the file is shown on the window title bar.

Opening a previous log file will ask the user if they wish to append or overwrite the current log.

# 14.6 Starting or stopping logging

Choose **Logging** from the pop-up menu to toggle logging to file on and off. When logging is active, the button becomes effective. Note that the contents of the log file cannot be viewed until logging is completed, or temporarily disabled by clearing the check box. Re-enabling logging will append to the log file.



# 14.7 Entering a full path to the file

It is recommended that the full path to a file is specified as a file name in the Command Line window because the current directory can be moved. However, care must be taken to enter the correct full path to a file when it is entered from the keyboard. To save this trouble, a full path can be easily specified by browsing through files.

Choose **Browse** from the pop-up menu to open the Browse dialog box. Select a file and click **Open** to paste the full path to the selected file to the cursor location. This option can only be used when the cursor is located on the last line.

# 14.8 Pasting a placeholder

Select a placeholder from the **Placeholder** submenu in the pop-up menu to paste the selected placeholder to the cursor location. This function is only available when the cursor is located on the last line.

Placeholder sub-menu	Placeholder
Configuration directory	\$(CONFIGDIR)
Configuration name	\$(CONFIGNAME)
Project directory	\$(PROJDIR)
Project name	\$(PROJECTNAME)
Workspace directory	\$(WORKSPDIR)
Workspace name	\$(WORKSPNAME)
HEW Installation directory	\$(HEWDIR)

# 14.9 Selecting all the window contents

Choose Select All from the pop-up menu to select all contents in the Command Line window.

# 14.10 Copying the selection onto the clipboard

Choose **Copy** from the pop-up menu to copy the selected text block onto the Windows® clipboard. Only available if the test block is selected.

# 14.11 Cutting out the selection to the clipboard

Choose **Cut** from the pop-up menu to remove the selected text block and copy them onto the Windows® clipboard. This option is available only when you have selected the text block currently being input at the latest cursor position.

# 14.12 Pasting the contents of the clipboard

Choose **Paste** from the pop-up menu to insert the content of the Windows® clipboard at the current cursor position. This option can only be used when the cursor is at the last line.



# 14.13 Clearing the contents of the Command Line window

Selecting **Clear window** from the pop-up menu of the Command Line window clears all of the information shown in the Console tabbed pane and in the upper pane of the Debugger tabbed pane of the Command Line window.

# 14.14 Undoing the last operation

Choose **Undo** from the pop-up menu to undo the last operation at the latest cursor position (on the text currently being input).

# 14.15 Checking brace matching

While the **Brackets Opened Indicator** toolbar button () is displayed, users can see if brackets match. The button cannot be operated.

Once an opening bracket "[" or "{" is input and the nesting follows, the **Brackets Opened Indicator** toolbar button is visible ( $\square$ ) until "]" or "}" is input as the closing bracket.

Example of a TCL Function Input in the State Command Line Window		Toolbar Button is Visible	
>set bit 1	None	3	
1	None		
>set value 1	None		
1	None		
>set r 1	None		
1	None		
>if {\$bit & \$value} {	Opening brace is input		
> set r 0\$r	None	2	
>} else {	Opening and closing braces are input	<u>a</u>	
> set r 1\$r	None	2	
>}	Closing brace is input	2	
>	None		
01			

# 14.16 Resetting the status of a batch file

The status of a batch file can be reset on the **Debugger** tabbed pane of the Command Line window. This feature is not available when no batch file has been specified.

A reset involves the following operations:

- The yellow arrow returns to the top of the batch file.
- The command interactive mode is exit.



• If a closing parenthesis is missing, it will automatically be entered.

#### To reset the status of a batch file

- 1. Open the **Debugger** tabbed pane of the Command Line window.
- 2. Right-click on the upper pane to open the pop-up menu.
- 3. Select Reset.

#### Notes:

- Information such as the name of the batch file and breakpoints in the batch file is saved in the session. If you change the session on the **Debugger** tabbed pane of the Command Line window, the pane now shows the contents of the batch file specified for the current session. The status of the batch file is also automatically reset even in cases where the batch file includes any of the following commands.
  - Setting the current project (CHANGE\_PROJECT)
  - o Selecting a session (CHANGE\_SESSION)
- If you change the configuration, the status of the batch file is also automatically reset even in cases where the batch file includes the following command.
  - Selecting a build configuration (CHANGE\_CONFIGURATION)

# 14.17 Single-stepping in a batch file

Single-stepping of command lines starting from the top of the batch file is possible on the **Debugger** tabbed pane of the Command Line window. This feature is not available when no batch file has been specified.





#### To single-step in a batch file

- 1. Open the **Debugger** tabbed pane of the Command Line window.
- 2. Perform one of the following operations in the upper pane.
  - Click on the **Step** toolbar button ( **1**).
  - Right-click within the pane to open the pop-up menu and select Step.
  - With a focus in the pane, press the "Ctrl+F10" keys.

The yellow arrow ( $\Rightarrow$ ) moves to the next command line. If some errors have occurred, however, the arrow does not move to the next line.

#### Note:

If the batch file includes the commands listed below, debugging of the batch file on the **Debugger** tabbed pane of the Command Line window cannot proceed. Instead, use the **Console** tabbed pane of the Command Line window to execute the batch file.

• CLOSE\_WORKSPACE

When a command is run, the batch file will be closed.

• CHANGE\_PROJECT or CHANGE\_SESSION

When a command is run, the contents of the batch file specified for the current session will be shown in the pane.

• CHANGE\_CONFIGURATION

When a command is run, the status of the batch file will automatically be reset.

# 14.18 Setting a breakpoint in a batch file

Breakpoints can be set on command lines in a batch file on the **Debugger** tabbed pane of the Command Line window. This feature is not available when no batch file has been specified.

#### To set a breakpoint in a batch file

- 1. Open the **Debugger** tabbed pane of the Command Line window.
- 2. Perform one of the following operations in the upper pane.
  - Double-click on the Breakpoint column for the line where the execution should stop.
  - Place the cursor on the line where the execution should stop. Then right-click to open the pop-up menu and select **Insert/remove breakpoint**.

A breakpoint icon (•) appears on the Breakpoint column. This indicates that a breakpoint has been set.



- 3. To switch enabling and disabling of the breakpoint that has been set, right-click and select **Enable/disable breakpoint** from the pop-up menu.
- 4. To delete the breakpoint, right-click and select **Insert/remove breakpoint**, or double-click on the column.
- 5. To delete all breakpoints, right-click and select Clear all breakpoints from the pop-up menu.

Execute the batch file with the **Debugger** tabbed pane open in the Command Line window. At a break, the execution stops before the command line where the breakpoint has been set. If you select **Step** or **Play** after the break, the execution restarts at the command line indicated by the yellow arrow ( $\clubsuit$ ). Up to 256 breakpoints can be set. When saving the session, all breakpoints that have been set are also saved.

### Notes:

- No break occurs if the batch file has been executed with the **Console** tabbed pane open in the Command Line window.
- If you modify the contents of the batch file after it has been specified, the information shown in the upper pane of the **Debugger** tabbed pane in the Command Line window will be updated. The status of the batch file will be reset, with all breakpoints deleted.



# 15. Macro-Recording Support Facility

The macro-recording support facility allows you to record operations, which are associated with the High-performance Embedded Workshop system applications **\*1**, build **\*2**, and debugging **\*3**, as High-performance Embedded Workshop command-line commands or to execute these recorded commands.

Files to record the operations (High-performance Embedded Workshop macro files) are command line batch files that have "hdc" as the extension and can be modified. These files are stored in the "Macros" folder within the High-performance Embedded Workshop management information folder under the application data folder for each user profile. By default, the file Default.hdc is created.

Note, however, that High-performance Embedded Workshop is not capable of recording all operations in the High-performance Embedded Workshop system corresponding to the High-performance Embedded Workshop command-line commands. For the operations that can be recorded, a macro record icon ( $\bigcirc$ ) is shown in the "Macro Recording" column of the menu list. This indicates that this operation can be recorded into a High-performance Embedded Workshop macro file.

# Notes:

- \*1. Changing a project, session, or configuration
- **\*2**. Compilation and build.
- **\*3**. Downloading a module, changing a memory value or register value, setting/deleting a software breakpoint, and running a program

The macro-recording support facility is available in the Tools menu and on the Macros toolbar.

The procedure from selection of Record Macro (•) to that of **Stop Macro** consists one macro and one High-performance Embedded Workshop macro file can record multiple macros (a set of High-performance Embedded Workshop command-line commands). A macro includes multiple High-performance Embedded Workshop command-line commands.

# 15.1 Macro menu and toolbar

The macro recorder has both a menu and toolbar integrated into the High-performance Embedded Workshop.

The macro menu is available on the **Tools** menu. It has the following standard menu options:

Macros
<u>Record Macro</u>
Play Macro
Stop Macro

The **Macros** menu option opens the Macro dialog box. This allows management of the current macro files and functions. The other menus options are the same as those seen on the **Macros** toolbar.

#### High-performance Embedded Workshop V.4.09

It is also possible to associate macro functions with a menu, when you do this additional menu options are added underneath the standard macro menu options. This is shown below.



Items that also have an associated toolbar have their icons displayed alongside the menu options.

The standard buttons on the Macros toolbar are shown below.



- The **Record Macro** button starts the macro recording process. After this button is clicked debugging and Highperformance Embedded Workshop management operations are recorded. This operation is the same as the **Record macro** menu on the Tools menu.
- The **Play Macro** button starts a macro function playback. If more than one macro is defined, the select macro function dialog is displayed to ask you to select specific macro for playback.
- The **Pause Macro** button is only enabled when recording or playing back a macro function. This pauses the current operation so that you can return to it later.
- The **Stop Macro** button is only enabled when recording or playing back a macro function. This halts the current macro operation that is being carried out.

In the same way you can associate macros with a menu you can associate them with a toolbar. When you associate the macro you can decide to use a user defined toolbar icon or the default High-performance Embedded Workshop macro function icon. Once added the associated macros appear to the right of the standard macro buttons. Clicking these buttons then automatically launches the associated macro function.





# 15.2 Using the Macro dialog box

The **Macro** dialog box allows you to access and manager the macro files currently in use in the High-performance Embedded Workshop system. The **Macro** dialog box is shown below.

Macro	? ×
Current macro file:	
Default	<u>N</u> ew
Macro <u>f</u> unctions:	Import
	<u>R</u> ecord
	Play
	<u>E</u> dit
	A <u>s</u> sign
	Re <u>m</u> ove
Show all available macro functions	
Initial directory: \$(FILEDIR) Browse	Close

The **Current macro file** drop-down list holds all of the macro files currently in use in the High-performance Embedded Workshop system. These files are stored in the "Macros" folder within the High-performance Embedded Workshop management information folder under the application data folder for each user profile. Any files that are copied into this directory are automatically added to the **Current macro file** drop-down list.

The macros functions in the **Current macro file** selection are shown in the **Macro functions** list box. Changing the **Current macro file** selection will automatically update this list. If you check the **Show all available macro functions** check box then all macros currently defined are displayed in the list. When selected this option ignores the value of the **Current macro file** drop-down list box.

It can be useful to use multiple macro files to logically hold related areas. For example you might like to create a different macro file for use with different debugger targets.

To execute macro files that have relative paths, specify the initial directory in **Initial directory**. Change the directory in the same way as using the change-directory command and execute macro files in the initial directory.

#### To create a new blank macro file

- 1. Select [Tools -> Macros]. The Macro dialog box opens.
- 2. Click the New button.
- 3. Enter the new macro file name.
- 4. Click OK. The new file name is added to the Current macro file drop-down list.



#### To import an existing macro file

- 1. Select [Tools -> Macros]. The Macro dialog box opens.
- 2. Click the **Import** button.
- 3. Browse to the existing macro file.
- 4. Click **Select**. The new file name is added to the **Current macro file** drop-down list.
- 5. When you select the macro file name in the list the available macros are displayed.

It is possible to manually edit any of the macro files. The files are text based High-performance Embedded Workshop command batch files.

#### To edit an existing macro file

- 1. Select [Tools -> Macros]. The Macro dialog box opens.
- 2. Select the macro file, the function exists in.
- 3. Select the macro in the function list you wish to edit.
- 4. Click the **Edit** button.
- 5. The Macros dialog box is dismissed and the file is opened in the editor.

You can also drag and drop a macro file into the High-performance Embedded Workshop window to open this file (in the same way as opening an ordinary text file).

#### To remove the currently selected macro from the existing macro file

- 1. Select [Tools -> Macros]. The Macro dialog box opens.
- 2. Select the macro file, the function exists in.
- 3. Select the macro in the function list you wish to remove.
- 4. Click the **Remove** button.

Pressing the Close button closes the Macro dialog box. It is not possible to cancel operations on this dialog box.



# **15.3** Importing a macro file with existing macros

If you want to import an existing High-performance Embedded Workshop macro file from another machine which already has a number of macros this is easily achieved by the following operations.

One method for importing macro files is from the macro dialog. The instructions are shown below:

### To import an existing macro file into High-performance Embedded Workshop

- 1. Select [Tools -> Macros]. The Macro dialog box opens.
- 2. Click the **Import** button. A standard file browser is opened for you to select the existing macro file you wish to import.
- 3. Select the file and click Select. The file is now imported. It automatically becomes the default macro file in the Current macro file drop-down list. The file is also copied into the "Macros" folder within the High-performance Embedded Workshop management information folder under the application data folder for each user profile.
- 4. Then click **Close** on the **Macro** dialog box and the macro file is ready for use.

Another method is to simply copy the new macro file into the **Macros** folder in the High-performance Embedded Workshop root directory. High-performance Embedded Workshop will automatically detect the new file and add its information to the High-performance Embedded Workshop macros dialog box. The new file will not be the default file for use so you will then have to enter the macro dialog box to switch the default macro file to the newly copied one.

# 15.4 Recording a macro

There are a number of ways to record a macro file in the High-performance Embedded Workshop, these options are discussed below.

#### To record a macro from the toolbar or menu

- Click the **Record Macro** button (●) or menu option on the **Macros** toolbar or the **Tools** menu. To indicate that a recording operation is taking place the mouse cursor is modified to include the record icon (२०). When the macro is being recorded the Pause Macro (□) and Stop Macro (□) buttons and menus are enabled. When the macro is being recorded the pause and stop buttons (२०) and menus are enabled.
- 2. If you are using a debugger that does not support the macro facility a warning is displayed that states the following "Warning: The target does not support macros. Recording will be limited. Do you wish to continue?". If you click yes the record operation will continue.
- 3. Recording will continue until you click **Stop Macro**. For details, see section 15.5, Functions that can be recorded into macro files.
- 4. When **Stop Macro** is clicked a dialog is displayed that allows you to enter the macro name.



5. Clicking OK saves the name and adds it to the current macro file. Clicking cancel dismisses the dialog box and loses the macro recording.

Another method is to launch the recording process via the Macro dialog box.

#### To record a macro from the macro dialog box

- 1. Select [Tools -> Macros]. The Macro dialog box opens.
- 2. Click the **Record** button. The Macro dialog box is dismissed and the recording operation begins.

# 15.5 Functions that can be recorded into macro files

High-performance Embedded Workshop is not capable of recording all operations in the High-performance Embedded Workshop system into macro files as scripts. For the operations that can be recorded, "Record Macro" in the menu list shows a macro record icon. This indicates that the operation can be recorded into a macro file.

- Recordable functions (common to all High-performance Embedded Workshop products)
- Recordable functions (dependent on the debugger)

# 15.5.1 Recordable functions (common to all High-performance Embedded Workshop products)

Recordable functions common to all High-performance Embedded Workshop products included in tool packages are listed in the table below.

#### • Handling menu options, shortcut keys, and toolbar buttons

While a macro is being recorded, if you handle a menu option, shortcut key, or toolbar button of those listed below or make any setting in a dialog box opened, these operations will be recorded into a macro file.

Menu	Menu Option	Shortcut Key	Toolbar Button
File	Open Workspace	-	-
	Save Workspace	-	-
	Close Workspace	-	-
	New Session	-	-
	Import Session	-	-
	Save Session	-	-
	Refresh Session	-	-
	Download A New Module *2	-	-
	Recent Workspaces	-	-
	Recent Downloaded Modules	-	-



Menu	Menu Option		Shortcut Key	Toolbar Button
Edit	Toggle Breakpoint	t	F9	
	Enable/Disable Br	eakpoint	CTRL+F9	1
Project	Set Current Project	ct	-	-
	Insert Project		-	-
	Edit Project Config	guration *3	-	-
Build * <b>1</b>	Build File		CTRL+F7	٢
	Build		F7	
	Build All		-	initia
	Build Multiple		-	-
	Clean Current Pro	ject	-	-
	Clean All Project		-	-
	Build Configuration	ns * <b>4</b>	-	SimDebug_SH-3 💽
Debug	Debug Sessions *	5	-	SimSessionSH-3
	Reset CPU		-	∃ <b>T</b>
	Go		F5	≣↓
	Reset Go		SHIFT+F5	≣Q
	Free Go *3		-	≣ <b>Ţ</b>
	Go to Cursor		-	
	Set PC to Cursor		-	I <sub>PC</sub>
	Run		-	-
	Step In		F11	{ <del>\</del> }
	Step Over		F10	8
	Step Out		SHIFT+F11	P
	Step		-	-
	Step Mode	Auto	-	-
		Assembly	-	-
		Source	-	-
	Halt Program		-	<b>STOB</b>
	Initialize		-	-
	Connect *3		-	
	Disconnect *3		-	
	Save Memory		-	-
	Verify Memory *3		-	-
	Download	<file download="" module="" name="" of="" the=""></file>	-	-
	Modules	All Download Modules	-	-
	Unload Modules	<file download="" module="" name="" of="" the=""></file>	-	-
		All Downloaded Modules	-	
Setup	Radix	Hex	-	<u>16</u>
		Decimal	-	<u>10</u>
		Oct	-	8
		Bin	-	2



#### Notes:

- \*1. This menu is not displayed while a debug-only project "Debugger only xxxxxx" created by High-performance Embedded Workshop V.4.01 or later is in use.
- \*2. Only options **Offset**, **File format**, **Filename**, **Access size**, and **Perform memory verify during download** can be recorded; **Download debug information only** and **Download automatically on target connection** are not recordable.
- **\*3**. Support for this function depends on the debugger.
- \*4. Selections made from the **Current configuration** drop-down list can be recorded.
- **\*5**. Selections made from the **Current session** drop-down list can be recorded.

### • Windows

While a macro is being recorded, operations made in the windows listed below will be recorded into a macro file. For information on the types of recordable operations, see the descriptions of the respective windows.

Window Name	Opened by	
Projects tab of the Workspace window [View -> Workspace]		
Editor	Double-clicking on a file in the Projects tab of the workspace window	
Disassembly	[View -> Disassembly]	
Register	[View -> CPU -> Registers]	
Memory	[View -> CPU -> Memory]	
IO	[View -> CPU -> IO]	

# 15.5.1.1 Projects tab of Workspace window

While a macro is being recorded, the following operations will be recorded into a macro file.

Target		Operation	Function
♂ Workspace	Clean All Projects pop-up menu option *1	Click on the menu option	Deletes intermediate and output files from configurations in all projects in this workspace.
Project	Build pop-up menu option *1	-	Builds out of date project files.
	Build All pop-up menu option *1	_	Builds project files, regardless of whether the project files are out of date.
	Clean Current Project pop- up menu option *1	_	Deletes intermediate and output files from the current configuration in this project.
	Set as Current Project pop up menu option	-	Sets this project as the current project.



Target		Operation	Function
Project file	Build <file name=""> pop-up menu option *1</file>		Builds a file.
Download	Download all module pop- up menu option	-	Loads all object (program) files.
modules folder	Download A New Module pop-up menu option	Make setting in the dialog box opened by clicking on the menu option <b>*2</b>	Loads an object (program) file.
	-	Double-click on a download module	
Download module	<b>Download module</b> pop-up menu option	Click on the menu option	
	Download module (debug data only) pop-up menu option	_	
	Unload module pop-up menu option	_	Unloads an object (program) file from memory.
	Download A New Module pop-up menu option	Make setting in the dialog box opened by clicking on the menu option <b>*2</b>	Loads an object (program) file.

Notes:

\*1. Available only when there is a toolchain installed.

\*2. Only options **Offset**, **File format**, **Filename**, **Access size**, and **Perform memory verify during download** can be recorded; **Download debug information only** and **Download automatically on target connection** are not recordable.

# 15.5.1.2 Editor window

While a macro is being recorded, the following operations will be recorded into a macro file.

Display Mode	Target	Operation	Function
Source mode	<b>Build</b> " <file name="">" pop-up menu option</file>	Click on the menu option	Builds the selected file.
	Toggle Breakpoint pop-up menu option	-	Sets or clears a software breakpoint.
	Enable/Disable Breakpoint pop-up menu option	-	Enables or disables the current software breakpoint.
	Go To Cursor pop-up menu option	_	Runs program until the PC reaches cursor.
	Set PC Here pop-up menu option	-	Sets PC to the address at cursor.
	S/W Breakpoints column	Double-click on the column	Sets or clears a software breakpoint.
Mixed/disassembly	Go To Cursor pop-up menu option	Click on the	Sets or clears a software breakpoint.
mode	Set PC Here pop-up menu option	menu option	Enables or disables the current software breakpoint.
	Toggle Breakpoint pop-up menu option	-	Runs program until the PC reaches cursor.
	Enable/Disable Breakpoint pop-up menu option	-	Sets PC to the address at cursor.



Display Mode	Target	Operation	Function
	S/W Breakpoints column	Double-click on	Sets or clears a software breakpoint.
		the column	

### 15.5.1.3 Disassembly window

While a macro is being recorded, the following operations will be recorded into a macro file.

Display Mode	Target	Operation	Function
Source mode	Toggle Breakpoint pop-up menu	Click on the	Sets or clears a software breakpoint.
	option	menu option	
	Enable/Disable Breakpoint pop-up		Enables or disables the current software
	menu option		breakpoint.
	Go To Cursor pop-up menu option		Runs program until the PC reaches cursor.
	Set PC Here pop-up menu option	_	Sets PC to the address at cursor.
	S/W Breakpoints column	Double-click on	Sets or clears a software breakpoint.
		the column	
Mixed/disassembly	Go To Cursor pop-up menu option	Click on the	Sets or clears a software breakpoint.
mode	node Set PC Here pop-up menu option menu option		Enables or disables the current software breakpoint.
	Toggle Breakpoint pop-up menu option	-	Runs program until the PC reaches cursor.
	Enable/Disable Breakpoint pop-up menu option	-	Sets PC to the address at cursor.
	S/W Breakpoints column	Double-click on	Sets or clears a software breakpoint.
		the column	

# 15.5.1.4 Register window

While a macro is being recorded, the following operations will be recorded into a macro file.

Target	Operation	Function
Edit pop-up menu option	Make setting in the dialog box opened by clicking on the menu option *1	Changes a
Flag register *2	Click on the flag register	register's content.
Value	In-place edit	_
	Make setting in a dialog box opened by double-clicking a value *1	

#### Notes:

\*1. Only option Value can be recorded; Radix and Set As are not recordable.

\*2. Support for this function depends on the debugger.



# 15.5.1.5 Memory window

While a macro is being recorded, the following operations will be recorded into a macro file.

Target	Operation	Function	
Set Pop-up menu option/toolbar button	Make setting in the dialog box	Modifies memory contents.	
Fill Pop-up menu option/toolbar button	opened by clicking on the menu	Fills a block of memory.	
Move Pop-up menu option/toolbar button	option or toolbar button	Moves a block of memory.	
Compare Pop-up menu option/toolbar button *	-	Compares memory contents.	
Save Pop-up menu option/toolbar button		Saves memory to a file.	
Load Pop-up menu option/toolbar button	-	Loads a memory area contents from a file.	
Value	In-place edit	Modifies memory contents.	
	Make setting in the dialog box opened by double-clicking a value	-	

# Note:

\*. Support for this function depends on the debugger.

### 15.5.1.6 IO window

While a macro is being recorded, the following operations will be recorded into a macro file.

Target	Operation	Function
Value	In-place edit	Modifies the I/O register contents.
	Make setting in the dialog box opened by double-clicking a value	

# 15.5.2 Recordable functions (dependent on the debugger)

Recordable functions dependent on the debugger included in tool packages are listed in the table below.

- E10A-USB Emulator Software (H8SX Device Group)
- H8S/Tiny E8a Emulator Debugger
- H8 Tiny/Super Low Power E8a Emulator Debugger
- H8 Tiny/Super Low Power E8 Emulator Debugger All debugging functions can be recorded into a macro file.
- M32C/80 E8 Emulator Debugger
- M16C/Tiny, M16C/62P E8 Emulator Debugger
- R8C/Tiny E8 Emulator Debugger

While a macro is being recorded, operations made in the windows listed below will be recorded into a macro file. For information on the types of recordable operations, see the descriptions of the window.

Window Name	Opened by
Editor	Double-clicking on a file in the Projects tab of the workspace window
Disassembly	[View -> Disassembly]
Label	[View -> Symbol -> Label]



#### High-performance Embedded Workshop V.4.09

Window Name	Opened by
Watch	[View -> Symbol -> Watch]
Locals	[View -> Symbol -> Locals]
Event	[View -> Code -> Eventpoints]

- E10A-USB Emulator Software (other than the H8SX device group)
- E10T-USB Emulator Software
- M32C E8a Emulator Debugger
- M16C E8a Emulator Debugger
- R8C E8a Emulator Debugger
- R32C E8a Emulator Debugger
- 740 E8a Emulator Debugger
- E6000H Emulator Software

### • E6000 Emulator Software

While a macro is being recorded, operations made in the windows listed below will be recorded into a macro file. For information on the types of recordable operations, see the descriptions of the window.

Opened by		
Double-clicking on a file in the Projects tab of the workspace window		
[View -> Disassembly]		
[View -> Symbol -> Label]		
[View -> Symbol -> Watch]		
[View -> Symbol -> Locals]		

#### • E1/E20 Emulator Software

While a macro is being recorded, if you handle a menu option, shortcut key, or toolbar button of those listed below, these operations will be recorded into a macro file.

Menu	Menu Option		Shortcut Key	Toolbar Button
Debug	RTOS Debug	Go To Cursor	-	t is t
		Step In	Alt+F11	<b>}</b> }
		Step Over	Alt+F10	<b>₽</b>
		Step Out	Shift+Alf+F11	P

While a macro is being recorded, operations made in the window or dialog box listed below will be recorded into a macro file. For information on the types of recordable operations, see the descriptions of the window or dialog box.

Window/Dialog Box Name	Opened by
Editor	Double-clicking on a file in the Projects tab of the workspace window
Disassembly	[View -> Disassembly]
Label	[View -> Symbol -> Label]
Watch	[View -> Symbol -> Watch]
Locals	[View -> Symbol -> Locals]
On-Chip Break *	[View -> Event -> On-chip Break]
Trace Conditions *	[View -> Event -> Trace Conditions]
Performance *	[View -> Event -> Performance Analysis Conditions]
Notes	

#### Note:

\*. Support for this function depends on the debugger.



### • E100 Emulator Software

While a macro is being recorded, if you handle a menu option, shortcut key, or toolbar button of those listed below, these operations will be recorded into a macro file.

Menu	Menu Option		Shortcut Key	<b>Toolbar Button</b>
Debug	RTOS Debug	Go To Cursor	-	
		Step In	Alt+F11	<u>}</u> }
		Step Over	Alt+F10	₽ ₽
		Step Out	Shift+Alf+F11	<b>P</b>

While a macro is being recorded, operations made in the window or dialog box listed below will be recorded into a macro file. For information on the types of recordable operations, see the descriptions of the window or dialog box.

Window/Dialog Box Name Opened by		
Editor	Double-clicking on a file in the Projects tab of the workspace window	
Disassembly	[View -> Disassembly]	
Label	[View -> Symbol -> Label]	
Watch	[View -> Symbol -> Watch]	
Locals	[View -> Symbol -> Locals]	
Code Coverage	[View -> Code -> Code Coverage]	
Data Coverage	[View -> Code -> Data Coverage]	
Hardware Break	[View -> Event -> Hardware Break]	
Trace conditions	[View -> Event -> Trace Conditions]	
Performance Analysis Conditions	[View -> Event -> Performance Analysis Conditions]	
Trigger Output Conditions	[View -> Event -> Trigger Output Conditions]	
Realtime Profile	[View -> Performance -> Realtime Profile]	
Configuration properties	[Setup -> Emulator -> System]	
Device Setting	[Setup -> Emulator -> Device Setting]	
Start/Stop Function Setting *	[Setup -> Emulator -> Start Stop Function Setting]	
Nota		

Note:

\*. Support for this function depends on the debugger.

# • E200F Emulator Software

While a macro is being recorded, operations made in the window or dialog box listed below will be recorded into a macro file. For information on the types of recordable operations, see the descriptions of the window or dialog box.

Window/Dialog Box Name	Opened by	
Editor	Double-clicking on a file in the Projects tab of the workspace window	
Disassembly	[View -> Disassembly]	
Label	[View -> Symbol -> Label]	
Watch	[View -> Symbol -> Watch]	
Locals	[View -> Symbol -> Locals]	
Trace	[View -> Code -> Trace]	
Event	[View -> Code -> Eventpoints]	
Realtime Profile	[View -> Performance -> Realtime Profile]	



Window/Dialog Box Name Open	ned by
Window/Dialog Box Name Open	led by

Configuration

[Setup -> Emulator -> System]

- M32C FoUSB/UART Debugger
- M16C R8C FoUSB/UART Debugger
- M32C Compact Emulator Debugger
- M16C R8C Compact Emulator Debugger
- 740 Compact Emulator Debugger
- M32C PC7501 Emulator Debugger
- M16C R8C PC7501 Emulator Debugger
- M32C PC4701 Emulator Debugger
- M16C PC4701 Emulator Debugger

### • 740 PC4701 Emulator Debugger

While a macro is being recorded, operations made in the windows listed below will be recorded into a macro file. For information on the types of recordable operations, see the descriptions of the window.

Window Name	Opened by	
ASMWatch	[View -> Symbol -> ASMWatch]	
CWatch	[View -> Symbol -> CWatch]	
S/W Break Points	[View -> Break -> S/W Break Points]	
Address Interrupt Break Points	[View -> Break -> Address Interrupt Break Points]	

### • H8/300H Tiny Compact Emulator Debugger

While a macro is being recorded, operations made in the windows listed below will be recorded into a macro file. For information on the types of recordable operations, see the descriptions of the window.

Window Name	Opened by	
ASMWatch	[View -> Symbol -> ASMWatch]	
CWatch	[View -> Symbol -> CWatch]	

- E30A Emulator Debugger
- M32C Simulator Debugger
- M16C R8C Simulator Debugger
- R32C Simulator Debugger
- 740 Simulator Debugger

While a macro is being recorded, operations made in the windows listed below will be recorded into a macro file. For information on the types of recordable operations, see the descriptions of the window.

Window Name	Opened by
ASMWatch	[View -> Symbol -> ASMWatch]
CWatch	[View -> Symbol -> CWatch]
S/W Break Points	[View -> Break -> S/W Break Points]

# • Simulator Debugger for SuperH Family

• Simulator Debugger for RX Family

# • Simulator Debugger for H8SX, H8S, H8 Family

While a macro is being recorded, operations made in the windows listed below will be recorded into a macro file. For information on the types of recordable operations, see the descriptions of the window.



### High-performance Embedded Workshop V.4.09

Opened by	
Double-clicking on a file in the Projects tab of the workspace window	
[View -> Disassembly]	
[View -> Code -> Coverage]	
[View -> Code -> Trace]	
[View -> Code -> Eventpoints]	
[View -> CPU -> Simulated I/O]	
[View -> Code -> Trace] [View -> Code -> Eventpoints]	

Support of this function depends on the debugger in use. For details, refer to the user's manual, help files, or the release notes for the emulator or the simulator debugger.

#### 15.5.2.1 Editor window

While a macro is being recorded, the following operation will be recorded into a macro file.

Display Mode	Target	Operation	Function
Source mode	Instant Watch pop-up menu	Make setting in the dialog box opened by	Adds a symbol.
	option	clicking on the menu option	

#### • E8 emulator software

Target	Operation	Function
Event Breakpoints column	Double-click on the column	Sets or clears a hardware breakpoint.

# 15.5.2.2 Disassembly window

While a macro is being recorded, the following operation will be recorded into a macro file.

Display Mode	Target	Operation	Function
Source mode	Instant Watch pop-up menu	Make setting in the dialog box opened by	Adds a symbol.
	option	clicking on the menu option	

### • E8 emulator software

Target	Operation	Function
Event Breakpoints column	Double-click on the column	Sets or clears a hardware breakpoint.


#### 15.5.2.3 Label window

While a macro is being recorded, the following operations will be recorded into a macro file.

Target	Operation	Function	
Add pop-up menu option/toolbar button	Make setting in the dialog box opened by clicking on the menu option or toolbar button	Defines a symbol.	
Delete pop-up menu option/toolbar button	Click on the menu option or toolbar button	Deletes a symbol.	
Delete All pop-up menu option/toolbar button	_	Deletes all symbols.	
Load pop-up menu option/toolbar button	Make setting in the dialog box opened by clicking on the menu option or toolbar button	Defines symbols.	
BP column	Double-click on the column	Sets or clears a software breakpoint.	

#### 15.5.2.4 Watch window

While a macro is being recorded, the following operations will be recorded into a macro file.

Target	Operation	Function
Enable Auto Update pop-up menu option/toolbar	Click on the menu option or toolbar	Sets real-time check of a symbol
button	button	to enable.
Enable Auto Update All pop-up menu		Sets real-time check of all
option/toolbar button	_	symbols to enable.
Disable Auto Update pop-up menu option/toolbar		Sets real-time check of a symbol
button	_	to disable.
Disable Auto Update All pop-up menu		Sets real-time check of all
option/toolbar button		symbols to disable.
Record Update Value -> Start Recording pop-up		Starts to record the updated
menu option/toolbar button	opened by clicking on the menu	values.
	option or toolbar button	
Record Update Value -> Stop Recording pop-up	Click on the menu option or toolbar	Stops recording the updated
menu option/toolbar button	button	values.
Add Watch pop-up menu option/toolbar button *	Make setting in the dialog box	Adds a symbol.
Edit Name pop-up menu option/toolbar button	opened by clicking on the menu	Deletes and adds a symbol.
Edit Value pop-up menu option/toolbar button	option or toolbar button	Edits the value of a symbol.
Delete pop-up menu option/toolbar button	Click on the menu option or toolbar	Deletes a symbol.
Delete All pop-up menu option/toolbar button	button	Deletes all watch items.
Radix -> Hexadecimal pop-up menu		Sets radix for hexadecimal for
option/toolbar button	_	this value.
Radix -> Decimal pop-up menu option/toolbar	_	Sets radix for decimal for this
button	_	value.
Radix -> Octal pop-up menu option/toolbar button		Sets radix for octal for this value.
Radix -> Binary pop-up menu option/toolbar	-	Sets radix for binary for this
button		value.
'+' or '-' sign	Click on the sign	Expands or collapses a symbol.
Watched variable name	Make setting in the dialog box	Deletes and adds a symbol.
	opened by double-clicking a value	
Value	In-place edit	Edits the value of a symbol.



Target	Operation	Function	
	Make setting in the dialog box		
	opened by double-clicking a value		

Note:

\*. Scope is not recordable.

#### 15.5.2.5 Local window

While a macro is being recorded, the following operations will be recorded into a macro file.

Target	Operation	Function
Edit pop-up menu	Make setting in the dialog box opened by clicking on the menu option	Changes a local item's
option/toolbar button	or toolbar button	value.
Value	Make setting in the dialog box opened by double-clicking on a value	_
	In-place edit	

### 15.5.2.6 Event window

While a macro is being recorded, the following operations will be recorded into a macro file.

Target	Operation	Function
Add pop-up menu option/toolbar button	Make setting in the dialog box	Adds the break point or break condition.
Edit pop-up menu option/toolbar button	opened by clicking on the menu option or toolbar button	Edits the break point or break condition.
Enable pop-up menu option/toolbar button	Click on the menu option or toolbar button	Enables the selected break points or break conditions.
<b>Disable</b> pop-up menu option/toolbar button	-	Disables the selected break points or break conditions.
<b>Delete</b> pop-up menu option/toolbar button	-	Removes the selected break points or break conditions.
<b>Delete All</b> pop-up menu option/toolbar button	-	Removes all break points or break conditions.

#### 15.5.2.7 Code Coverage window

While a macro is being recorded, the following operation will be recorded into a macro file.

Target	Operation		Function	
Hardware Settings pop-up	Make setting in the dialog box opened by	٠	Allocates code coverage memory.	
menu option/toolbar button	clicking on the menu option or toolbar button	٠	De-allocates code coverage memory.	



#### 15.5.2.8 Data Coverage window

While a macro is being recorded, the following operation will be recorded into a macro file.

Target	Operation	Fι	Inction
Hardware Settings pop-up	Make setting in the dialog box opened by	٠	Allocates data coverage memory.
menu option/toolbar button	clicking on the menu option or toolbar button	٠	De-allocates data coverage memory.

#### 15.5.2.9 On-Chip Break dialog box

While a macro is being recorded, the following operation will be recorded into a macro file.

Target	Operation	Function
Apply button	Click on the button	Changes the on-chip break conditions.

### 15.5.2.10 Hardware Break dialog box

While a macro is being recorded, the following operation will be recorded into a macro file.

Target	Operation	Function
Apply button	Click on the button	Changes the hardware break conditions.
Detail button for Access Protect Violation	Make setting in the	Sets the access violation area.
Detail button for Read from an uninitialized memory	dialog box opened by	
Detail button for Stack access violation *	clicking on the button	
Detail button for Task stack access violation	-	

Note:

\*. Support for this function depends on the debugger.

#### 15.5.2.11 Trace Conditions dialog box

While a macro is being recorded, the following operation will be recorded into a macro file.

• E1/E20 Emulator Software

Target	Operation	Function
Apply button	Click on the button	Changes the trace conditions.

### • E100 Emulator Software

Target	Operation	Function
Apply button	Click on the button	Changes the trace conditions.
Detail button for Access Protect Violation	Make setting in the	Sets the access violation area.
Detail button for Read from an uninitialized memory	dialog box opened by	
Detail button for Stack access violation *	clicking on the button	
Detail button for Task stack access violation	-	



#### Note:

\*. Support for this function depends on the debugger.

#### 15.5.2.12 Performance dialog box

While a macro is being recorded, the following operation will be recorded into a macro file.

Target	Operation	Function
Apply button	Click on the button	Changes the performance conditions.

#### 15.5.2.13 Performance Conditions dialog box

While a macro is being recorded, the following operation will be recorded into a macro file.

Target	Operation	Function
Apply button	Click on the button	Changes the performance conditions.

### 15.5.2.14 Trigger Output Conditions dialog box

While a macro is being recorded, the following operation will be recorded into a macro file.

Target	Operation	Function
Apply button	Click on the button	Changes the trigger output conditions.

#### 15.5.2.15 Realtime Profile window

While a macro is being recorded, the following operations will be recorded into a macro file.

• E200F Emulator Software

Target	Operation	Function
Clear Data pop-up menu	Click on the menu option or toolbar button	Clears all measurement data of real-
option/toolbar button		time profile.
Save To File pop-up menu	Make setting in the dialog box opened by clicking	Saves real-time profile
option/toolbar button	on the menu option or toolbar button	measurement results to a file.

#### • E100 Emulator Software

Target	Operation	Function
Set pop-up menu option/toolbar button	Make setting in the dialog box opened by clicking on the menu option or toolbar button	<ul> <li>Allocates real-time profile memory.</li> <li>De-allocates real time profile memory.</li> <li>Sets real-time profile measurement mode.</li> <li>Enables or disables each task ID individually.</li> </ul>
Properties pop-up menu option/toolbar button		Set the unit of the measurement of a realtime profile.



Target	Operation	Function
Save To File pop-up menu option/toolbar		Saves real-time profile measurement results to a
button		file.
Clear Data pop-up menu option/toolbar button Click on the menu		Clears all measurement data of real-time profile.
option or toolbar button		

### 15.5.2.16 Configuration properties dialog box

While a macro is being recorded, the following operation will be recorded into a macro file.

Target	Operation	Function
OK button	Click on the button	Selects the operating clock.
		Sets up the debugger.
		Selects a switching function.
		<ul> <li>Allocates emulation memory. *</li> </ul>
		<ul> <li>Sets on overwriting of the internal flash memory. *</li> </ul>
		Sets the warning of exceptional events.
		• Selects whether or not to maintain Ram monitor condition. *
		Set up the work area. *

Note:

\*. Support for this function depends on the debugger.

#### 15.5.2.17 Device Setting dialog box

While a macro is being recorded, the following operation will be recorded into a macro file.

Target	Operation	Function
OK button	Click on the button	<ul> <li>Selects a device and operation mode.</li> <li>Sets the external bus area information. *</li> <li>Allocates emulation memory. *</li> </ul>

Note:

\*. Support for this function depends on the debugger.

#### 15.5.2.18 Start/Stop Function Setting dialog box

While a macro is being recorded, the following operation will be recorded into a macro file.

Target	Operation	Function
OK button	Click on the button	Specifies the address of a work area and an executing routine.



#### 15.5.2.19 ASMWatch window

While a macro is being recorded, the following operations will be recorded into a macro file.

Target	Operation	Function
<b>Set</b> pop-up menu option/toolbar button	Make setting in the dialog box opened by clicking on the menu option or toolbar button	Sets new data to selected watch point.
Value	Make setting in the dialog box opened by double-clicking a value	
	In-place edit	

#### 15.5.2.20 CWatch window

While a macro is being recorded, the following operations will be recorded into a macro file.

Target	Operation	Function
<b>Set</b> pop-up menu option/toolbar button	Make setting in the dialog box opened by clicking on the menu option or toolbar button	Sets new data to selected C watch point.
Value	Make setting in the dialog box opened by double-clicking on a value	
	In-place edit	

### 15.5.2.21 S/W Breakpoints window

While a macro is being recorded, the following operations will be recorded into a macro file.

Target	Operation	Function
Add button	Click on the button	Adds the break point.
Delete button		Removes the selected break point.
Delete All button	_	Removes all break points.
Enable button	_	Enables the selected break point.
All Enable button	_	Enables all break points.
Disable button		Disables the selected break point.
All Disable button	_	Disables all break points.
Selected item in the	Double-click on the item	Enables or disables the selected software breakpoint.
S/W Break Point list		

### 15.5.2.22 Address Interrupt Break Points dialog box

While a macro is being recorded, the following operations will be recorded into a macro file.

Target	Operation	Function
Add button	Click on the button	Adds the break point.
Delete button		Removes the selected break point.
Delete All button		Removes all break points.
Enable button		Enables the selected break point.
All Enable button		Enables all break points.
Disable button		Disables the selected break point.
All Disable button		Disables all break points.



Target	Operation	Function
Selected item in the Address	Double-click on the item	Sets or clears a software breakpoint.
Interrupt Break Point list		

#### 15.5.2.23 Trace window

While a macro is being recorded, the following operation will be recorded into a macro file.

Target	Operation	Function
Acquisition pop-up menu	Make setting in the dialog box opened by clicking on the	Configures trace acquisition
option/toolbar button	menu option or toolbar button	parameters.

#### 15.5.2.24 Configuration dialog box

While a macro is being recorded, the following operation will be recorded into a macro file.

Target	Operation	Function
OK button	Click on the button	Sets the emulator operation conditions.

#### 15.5.2.25 Coverage window

While a macro is being recorded, the following operations will be recorded into a macro file.

Target	Operation	Function	
Enable All pop-up menu option/toolbar button	Click on the menu option or	Enables all the coverage functions.	
Clear All pop-up menu option/toolbar button toolbar button		Clears the coverage data for all functions.	
Add Range pop-up menu option/toolbar button	Make setting in the dialog box	Adds a new coverage range.	
Edit Range pop-up menu option/toolbar button	opened by clicking on the menu option or toolbar button	Edits selected coverage range.	
Enable pop-up menu option/toolbar button	Click on the menu option or	Enables or disables coverage.	
Clear Data pop-up menu option/toolbar button	toolbar button	Clears the coverage data.	
Save Data pop-up menu option/toolbar button	Make setting in the dialog box	Saves the coverage data.	
Load Data pop-up menu option/toolbar button	opened by clicking on the menu option or toolbar button	Loads the coverage data from file.	

#### 15.5.2.26 Simulated I/O window

While a macro is being recorded, the following operation will be recorded into a macro file.

Target	Operation	Function
Erase All pop-up menu	Click on the menu option	n Clears the contents of the Simulated I/O window.
option/toolbar button	or toolbar button	



### 15.6 Playing a macro

There are a number of ways to play a macro file in the High-performance Embedded Workshop, these options are discussed below.

#### To play a macro from the toolbar or menu

- 1. Click the **Play Macro** button ( $\triangleright$ ) or menu option on the **Macros** toolbar or the Tools menu.
- 2. If there is only one macro defined it is automatically played. If multiple macro functions exist in the current default macro file a dialog is displayed that asks you to choose which macro you wish to execute. When playing the mouse cursor is modified to include the play icon  $(\mathbf{k} \mathbf{P})$ .
- 3. Click OK and the macro is executed. When the macro is being executed the pause ( $\mathfrak{M}$ ) and stop ( $\Box$ ) buttons and menus are enabled. When pausing the mouse cursor is modified to include the pause icon ( $\Im_{\mathfrak{M}}^{\mathfrak{p}}$ ).

It is also easy to run a macro from the macro toolbar, menu or via a keyboard shortcut if you have set up macro function assignments in the Macro dialog box. For information on this area read the "Assigning a macro" information.

Another method is to launch the macro function via the Macro dialog box.

#### To play a macro from the macro dialog box

- 1. Select [Tools -> Macros]. The Macro dialog box opens.
- 2. Select the macro file which contains the function you want to play.
- 3. Select the macro function in the function list that you want to play.
- 4. Click the **Play** button.
- 5. The Macro dialog box is dismissed and the macro function executed.

### 15.7 Editing a macro

It is also possible to edit macros in the following way.

#### To edit a macro

- 1. Select [Tools -> Macros]. The Macro dialog box opens.
- 2. Select the macro file which contains the function you want to edit.
- 3. Select the macro function in the Macro functions that you want to edit.
- 4. Click the **Edit** button.
- 5. The Macro dialog box is dismissed and the file opened in the High-performance Embedded Workshop editor.



#### High-performance Embedded Workshop V.4.09

The Macros are stored in the "Macros" folder within the High-performance Embedded Workshop management information folder under the application data folder for each user profile. These files are just text based TCL files that can also be manually edited in the High-performance Embedded Workshop editor window. Any changes made will automatically be picked up by the High-performance Embedded Workshop next time the macro is executed.

## 15.8 Assigning a macro

It is possible to assign a macro to either a custom menu option, toolbar or keyboard shortcut. This allows you to quickly access macros which you often use. It is possible to assign a single macro to a **Tools** menu, **Macros** toolbar and keyboard shortcut all at the same time if you want. A custom menu and a toolbar button are shown in alphabetical order.

#### To assign a macro to a toolbar button

- 1. Select [Tools -> Macros]. The Macro dialog box opens.
- 2. Select the macro file which contains the function you want to assign.
- 3. Select the macro function in the Macro functions list that you want to assign.
- 4. Click the Assign button. The Macro assign dialog box is displayed.
- 5. Select the Assign to Toolbar check box.
- 6. You can now modify the macro description in the **Description** field. This will be used as the toolbar tool tip if you assign the macro to a toolbar button.
- 7. Now you can decide whether you will use the default macro toolbar button (D) or specify your own toolbar button image from an existing bitmap file. (\*.bmp)
- 8. To specify your own image simply click the browse button to open a standard file browser. This will allow you to manually locate the file on your machine.
- 9. Click OK and the toolbar is added to the **Macros** toolbar.

#### To assign a macro to a menu

- 1. Select [Tools -> Macros]. The Macro dialog box opens.
- 2. Select the macro file which contains the function you want to assign.
- 3. Select the macro function in the **Macro functions** list that you want to assign.
- 4. Click the Assign button. The Macro assign dialog box is displayed.
- 5. Select the Assign to Menu Name check box.
- 6. You can now modify the macro description in the **Description** field. This will be used as the menu tool tip if you assign the macro to menu.
- 7. Enter the menu name as you want it to appear on the tools menu.



8. Click OK and the menu is added to the **Tools** menu.

#### To assign a macro to a keyboard shortcut

- 1. Select [Tools -> Macros]. The Macro dialog box opens.
- 2. Select the macro file which contains the function you want to assign.
- 3. Select the macro function in the Macro functions list that you want to assign.
- 4. Click the Assign button. The Macro assign dialog box is displayed.
- 5. Select the Assign to Keyboard Shortcut check box.
- 6. Select the keyboard shortcut you wish to assign to the macro in the drop-down list.
- 7. Click OK and the shortcut is now available for use.

### 15.9 Configuring the Macro tab of the Output window

Shows the current records of macros. You can view information such as High-performance Embedded Workshop command-line commands recorded into a High-performance Embedded Workshop macro file from execution of [Tools - > Macro Recording] to [Tools -> Stop Macro]. It is also possible to view this information while recording.



Right-clicking displays a pop-up menu containing available options.

A basic operation is allocated to the toolbar.

The **Toolbar display** and **Customize toolbar** options are also included in the pop-up menu opened by right-clicking on the toolbar.

Pop-up Menu Option	<b>Toolbar Button</b>	Function
Clear Window	0	Clears the contents of the window.
Save		Saves the contents of the window into a text file.
Сору		Copies the selected contents onto the Windows® clipboard.
Toolbar display	-	Shows or hides the toolbar.
Customize toolbar	-	Customizes toolbar buttons.



# 16. Test Support Facility

The High-performance Embedded Workshop provides regression testing facilities for your application by the test support facility. Regression test is re-execution of a previous test after modification of the program to verify the result. The test tool allows High-performance Embedded Workshop macro and batch files to be executed and the system can then be compared to see if it matches the original captured data.

It is possible to create many tests and scripts and automate the testing procedure. Then once the testing has been completed the results are displayed in an easy to understand format in the test browser window. This can significantly reduce the work needed to execute many regression tests.

The test tool uses the TCL command line batch files as its method of executing the features and functions of Highperformance Embedded Workshop. These functions can be created easily using the macro recorder feature or by manually editing them yourself.

If you need to automate the execution of the actual tests the test tool also supports command line operation. Many commands are available that allow you to open and close suites, run and compare test data.

The first step towards using the test tool is to create a test suite. This is available on the main **Test** menu. Then you must edit the test suite to create some actual tests. Then you can execute the tests using the "Run tests" menu option.

Note, however, that High-performance Embedded Workshop is not capable of acquiring test-image information of all functions in the High-performance Embedded Workshop system. For the items from which test-image information can be acquired, see section 16.6, Functions that can be saved as test-image data into test-image files.

### 16.1 Creating a test suite

A test suite is a set of tests. A test suite has a very similar concept to a workspace. The test suite can contain many tests which in turn can contain many macros to be executed.





Test suites are independent to the current High-performance Embedded Workshop workspace. This allows your test suite to work with multiple workspaces and projects to test different situations in one test suite. The "open\_workspace" and "change\_project" commands all work with the test suite feature so allow you to control which workspace is currently in use.

#### To create a test suite

1. Select [**Test -> Create New Test Suite**]. The **Create New Test Suite** dialog box opens.

Create New Test Suite	? ×
Test suite name:	ОК
	Cancel
Directory for test suite:	
C:\Workspace\DemoSH4	<u>B</u> rowse

- 2. Enter the test suite name.
- 3. Initially the workspace directory is shown in **Directory for test suite**. This can be modified as required.
- 4. Clicking OK. The test suite is then created. This then enables a number of other options on the Test menu.

Once the test suite is created the Test tab of the workspace window has the test suite added to it. This tab allows quick navigation around your test suite and fast access to the tests.

A file is located in the destination location with the filename ".HTS" (High-performance Embedded Workshop test suite).

### 16.2 Opening and closing test suites

Once a test suite is created it is saved to a file with the extension ".HTS" (High-performance Embedded Workshop test suite). To open this file again after you have created it previously you can use the following operation.

#### To open a test suite

- 1. Select [Test -> Open Test Suite]. The Open Test Suite dialog box opens.
- 2. Select the test suite file. Then click Select. The test suite is loaded. This then enables a number of other items on the **Test** menu and opens the test suite contents in the Test tab of the workspace window.

When test suites are opened the filename of the suite is added to the **Recently used test suite** list on the **File** menu. This allows access to open your test suites after the initial **Open test Suite** menu operation.



#### To close a test suite

Select [**Test -> Close Test Suite**]. Once clicked the current test suite is closed and all items are removed from the workspace tab.

There are TCL command line commands for these operations that can be used via the command line window. These commands are named "open\_test\_suite" and "close\_test\_suite".

You can also close a test suite from the workspace window pop-up menu.

### 16.3 Editing a test suite

Once your test suite has been created the next step is to add some tests. This operation allows you to access the **Modify Test Suite** dialog box and will allow you to add tests which will then be executable.

#### To edit a test suite

1. Select [Test -> Edit Test Suite]. The Modify Test Suite dialog box is displayed.

odify Test Suite	:	?
<u>T</u> est cases:		ОК
Test Name	Description	Cancel
		<u>A</u> dd
		<u>M</u> odify
		<u>R</u> emove

The **Modify Test Suite** dialog box displays the currently defined test cases. Each test case has a name and can have a detailed description that you have defined to explain the tests purpose.

Clicking the Add button on this dialog box displays the Add New Test dialog box, this is shown below.



dd New Test		? ×
Test_name: Test_Demo		ОК
T <u>e</u> st description: Demonstration		Cancel
Tests to execute:		
Macro Name Demo	File Path Default	Add <u>Fi</u> emove Move to <u>t</u> op Move <u>up</u> Move <u>d</u> own Move to <u>b</u> ottom
Test jmage file: C:\Workspace\Dem	oSH4\Test_Suite_Demo_Test_Demo.hif	Bro <u>w</u> se

Selecting the test and clicking modify allows you to edit currently defined details for the selected test. These details are also shown in the same dialog as that used for "Add new test". If you click **Remove** the test is removed.

The **Add New Test** dialog box allows you to configure tests for execution later. This is discussed in "Adding tests to the test suite".

You can also edit a test suite from the workspace window pop-up menu.

### 16.4 Adding tests to the test suite

Once your test suite has been created the next step is to add some tests. This operation allows you to access the **Modify Test Suite** dialog box and will allow you to add tests which will then be executable.

#### To add tests to a test suite

- 1. Select [Test -> Edit Test Suite]. The Modify Test Suite is then displayed.
- 2. Click Add. The Add New Test dialog box is displayed.



dd New Test		? ×
Test <u>n</u> ame:		ок
Test_Demo		Cancel
Test description:		
Demonstration		
Tests to execute:		
Macro Name	File Path	<u>Add</u>
Demo	Default	<u>R</u> emove
		Move to <u>t</u> op
		Move <u>u</u> p
		Move <u>d</u> own
		Move to <u>b</u> ottom
Test jmage file:		
C:\Workspace\Der	noSH4\Test_Suite_Demo_Test_Demo.hif 🔰 🕨	Bro <u>w</u> se

#### To add a new test you should setup the following data

- 1. Enter the test name. There can be no spaces in this name.
- 2. Enter the test description. This should describe the test in a verbose way so you will understand it at a later date.
- 3. To setup the actual tests to execute you should click the Add button.
- 4. The Add New Test Script dialog box is displayed (See below).
- 5. The **Script type** box allows you to select a High-performance Embedded Workshop macro ("HEW macro") or a High-performance Embedded Workshop command line batch file ("TCL command line batch file").
- 6. Selecting **HEW Macro** in the **Script type** drop-down list shows all of the registered macro in the **Select macros to add to test** list.
- 7. You can select multiple macro files to be executed for each test. This is achieved by selecting the checkbox next to the macro names you wish to use.
- 8. Selecting "TCL command line batch file" in the drop-down list changes the **Add Test Script** dialog box so you can define the file to execute when the test is executed.
- 9. Clicking OK stores the results and adds it to the Add New Test dialog box.
- Finally you must set the Debug system for test image comparison file. This file stores the comparison data for the system to be compared to after the tests have executed. This file can be created by using [Test -> Create Test Image File].
- 11. Clicking OK adds the test to the **Modify Test Suite** dialog box.



High-performance Embedded Workshop V.4.09

Add Test Script		? ×
Script type: HEW Macro Select macros to add		OK Cancel
Test Name	File Path	
Download	Default Default	

### 16.5 Creating a test image file

The test image data is what is used for any comparison in the test system. Only items that are saved in the test file can be compared.

For example you may only be interested in comparing data that is located in a small area memory not the memory for the entire device. The more items that you compare for each test will slow down the comparison and this can make a big difference if you are executing many tests.

#### To create test image data to be saved into a test image file

1. Select the [Test -> Create New Test Image File]. The Create New Test Image dialog box is then displayed.

Create New Test Image	? ×
Available components:	OK Cancel
CPU ID-SimSessionSH-4 Memory-SimSessionSH-4 Register-SimSessionSH-4	<u>C</u> heck All
□ 💭 Simulated I/O-SimSessionSH-4 □ 💭 Status-SimSessionSH-4 □ 🔊 Output	<u>S</u> ettings
Build Output Window     Debug Output Window-SimSessionSH-4      Symbol      Locals-SimSessionSH-4	Import
Watch-SimSessionSH-4	



- 2. Clicking the **Check All** button selects all checkboxes, while clicking the **Uncheck All** button deselects all checkboxes.
- 3. Click the **Import** button to open the **Import the Test Image File** dialog box. Browse to the HIF file location. The settings of an existing test image file are imported.
- 4. Each component is listed in the dialog box. Check the checkbox next to the component name to save data for that component.



- 5. Select the component in the dialog and then click the **Settings** button. A dialog box will be displayed that is dependent on the selected component. This will allow you to customize the data that is saved to the file. For details, see section 16.6, Functions that can be saved as test-image data into test-image files.
- 6. Clicking OK will dismiss the dialog and store the changes that you have set up.
- 7. The Save Test Image File dialog box opens.
- 8. Save the data into the High-performance Embedded Workshop test-image file selected in the **Add New Test** dialog box or a new High-performance Embedded Workshop test-image file. Only test-image data of the selected test items (with tick marks in checkboxes) will be saved into the file.

You can also edit a test image file from the workspace window pop-up menu.

### 16.6 Functions that can be saved as test-image data into test-image files

Create New Test Image	? ×
Available components:	OK Cancel
CPU IO-SimSessionSH-4 Memory-SimSessionSH-4 Register-SimSessionSH-4 Register-SimSessionSH-4	<u>C</u> heck All
Simulated I/O-SimSessionSH-4 □ □ □ Status-SimSessionSH-4 □ □ □ □ Output	<u>S</u> ettings
Build Output Window	Import
Symbol Cocals-SimSessionSH-4 Watch-SimSessionSH-4	



#### High-performance Embedded Workshop V.4.09

High-performance Embedded Workshop is not capable of saving all High-performance Embedded Workshop functions as test-image data. For details on the functions that can be saved into test-image files, see the topics below. These topics also include information about failed test items (FAIL) shown after test or at comparison of test-image files.

- Functions that can be saved into test-image files (common to all High-performance Embedded Workshop products)
- Functions that can be saved into test-image files (dependent on the debugger)

In some cases it is necessary to make detailed setting specific to each of the test items before saving test-image data. If you double-click on a test item in the Create New Test Image dialog box, a further dialog box for detailed setting opens. When a test item has any detailed setting, it is possible to customize test-image data of this test item before saving. On the other hand, when a test item does not allow detailed setting and the checkbox for this test item is selected, all test-image data of this item will be saved into the file.

A save file icon (**[]**) in the "Saving into Test-Image File" column of an item listed in the **View** menu indicates that this data can be saved into a test-image file.

### 16.6.1 Functions that can be saved into test-image files (common to all High-performance Embedded Workshop products)

Among the functions common to all High-performance Embedded Workshop products, test-image data of the test items listed below can be saved into test-image files.

The following table shows the test items in the Create New Test Image dialog box and the corresponding windows from which the data will be saved into test-image files.

For more information on the test-image data to be saved into test-image files, how to make detailed setting, and test results, see the descriptions of test items for the corresponding windows.

Create New Test Image Dialog Box		Window (Tab) Name	Opened by	
Test Group Name	Test Item Name			
Output	Build Output Window	Build tab of the Output window	[View -> Output]	
	Debug Output Window	Debug tab of the Output window		
CPU	Register *	Register	[View -> CPU -> Registers]	
	IO *	IO	[View -> CPU -> IO]	
	Status *	Status	[View -> CPU -> Status]	
	Memory *	Memory	[View -> CPU -> Memory]	
Code	StackTrace *	StackTrace	[View -> Code -> Stack Trace]	

Note:

\*. Test items cannot be selected when the High-performance Embedded Workshop is not connected to any target.



#### 16.6.1.1 Output-Build/Debug (Output window)

The following table shows information on the test-image data to be saved into test-image files and test results (not matched).

Test group name	Output			
Test item name	Build Output Window			
Test-image data to be	All informa	ation in the Build tab of the output window.		
saved into test-image files	·			
Test result details	Original	Driginal Content of the test-image file		
	New	Data in the current High-performance Embedded Workshop system or in another test-image file to be compared with		
	Example	Original:		
		New:		
		Building - New - Debug		

Test group name	Output	Output		
Test item name	Debug Out	Debug Output Window		
Test-image data to be saved into test-image file	All information in the Debug tab of the output window.			
Test result details	Original	Content of the test-image file		
	New	Data in the current High-performance Embedded Workshop system or in another test-image file to be compared with		
	Example	<b>Original:</b> Connected <b>New</b> : Step Normal End		

No detailed setting specific to the test item can be made.

#### 16.6.1.2 CPU-Register (Register window)

The following table shows information on the test-image data to be saved into test-image files and test results (not matched).

Test group name	CPU			
Test item name	Register			
Test-image data to be saved into test-image files	All information in the Register window.			
Test result details	Failed at register	Name of the unmatched register		
	Src	Content of the test-image file		
	Dest	Data in the current High-performance Embedded Workshop system or in another test-image file to be compared with		
	Example	Failed at register R11, Src = 0x00000000, Dest = 0x00000fff		

No detailed setting specific to the test item can be made.



#### 16.6.1.3 CPU-IO (IO window)

The following table shows information on the test-image data to be saved into test-image files and test results (not matched).

Test group name	CPU			
Test item name	10			
Test-image data to be saved into test-image files	Values set as detailed information and the range of data in the <b>IO</b> window acquired with this setting. No I/O register is specified by default.			
Test result details	Failed at	Name of the unmatched I/O register		
	Src	Content of the test-image file		
	Dest	Data in the current High-performance Embedded Workshop system or in another test-image file to be compared with		
	Example	Failed at IPRC, register value is different: Src = 0000, Dest = FFFF		

#### To make detailed setting

dit Test IO regist			? >
Compared 10 registe	rs: Address		<b></b>
	FF000000		
	FF000004		
П ТТВ	FF000008		
TEA	FF00000C		
MMUCR	FF000010		
CCR	FF00001C		
🗖 TRA	FF000020		
	FF000024		<u> </u>
Current <u>S</u> etting:			
			A.
🔲 Ignore address		OK	Cancel

- 1. Selecting the checkbox for an I/O register in the **Compared I/O registers** list allows this I/O register to be saved into the test-image file. These check boxes are not selected by default. The selected I/O register is shown in **Current Setting**.
- 2. If you do not wish to compare the address of the selected register, select the **Ignore address** checkbox. This checkbox is not selected by default.
- 3. Click OK.



#### 16.6.1.4 CPU-Status (Status window)

The following table shows information on the test-image data to be saved into test-image files and test results (not matched).

Test group name	CPU			
Test item name	Status			
Test-image data to be	Values set a	as detailed information and all data in the <b>Status</b> window. No sheet is specified		
saved into test-image files	by default.			
Test result details	Sheet	Name of the unmatched sheet		
	Line	Unmatched line number		
	Src	Src Content of the test-image file		
	Dest	Data in the current High-performance Embedded Workshop system or in another test-image file to be compared with		
	Example	Status differs.		
		Sheet = Platform		
		<b>Line</b> = 6		
		Src = Execute From, Pipeline Reset		
		Dest = Execute From, EX Stage		

#### To make detailed setting

Edit Test Status	? ×
Sheet: Memory Platform	OK.
Events	Cancel

- 1. To specify the sheet to be compared in the test-image file, select the **Memory**, **Platform**, or **Events** checkbox. These checkboxes are not selected by default.
- 2. Click OK.



#### 16.6.1.5 CPU-Memory (Memory window)

The following table shows information on the test-image data to be saved into test-image files and test results (not matched).

Test group name	CPU				
Test item name	Memory	Memory			
Test-image data to be	Values set	Values set as detailed information and the range of data in the Memory window acquired			
saved into test-image files	with this se	with this setting. No memory range is specified by default.			
Test result details	Failed at	Unmatched address			
	address				
	Src	Content of the test-image file			
	Dest	Data in the current High-performance Embedded Workshop system or in another test-image file to be compared with			
	Example	Failed at address 0x70000014, Src = 0x00002f5a, Dest = 0x00002704			

#### To make detailed setting

Edit Test Memory Rai	nges		? ×
Memory ranges: Start address 🔸	End address	Size	OK Cancel
			<u>A</u> dd
			<u>M</u> odify <u>R</u> emove

- To add a memory range
  - 1. Click on the **Add** button on the **Edit Test Memory Ranges** dialog box to open the **Add Memory Range** dialog box.



Add Memory Range	? ×
Start address:	OK
	Cancel
End address:	
Access size:	

- 2. Specify Start address, End address, and Access size to set a memory range to be saved into a test-image file.
- 3. Click OK.
- To modify a memory range
  - 1. Click on the **Modify** button on the **Edit Test Memory Ranges** dialog box. The memory range selected in the **Memory ranges** list will be modified. Only one variable is selectable in the list. The **Modify Memory Range** dialog box opens.

Modify Memory Range		? ×
Start address:		ОК
	▼ 🕖	Cancel
End address: 000000FF	- 🕫	
<u>A</u> ccess size:		
1		

- 2. Specify **Start address**, **End address**, and **Access size** to set a memory range to be saved into a test-image file.
- 3. Click OK.
- To remove a memory range
  - 1. Click on the **Remove** button on the **Edit Test Memory Ranges** dialog box. The memory range selected in the **Memory ranges** list will be deleted. Only one trace range is selectable in the list.

Then click OK.



#### 16.6.1.6 Code-StackTrace (StackTrace window)

The following table shows information on the test-image data to be saved into test-image files and test results (not matched).

Test group name	Code		
Test item name	StackTrace		
Test-image data to be	Values set as detailed information and data within nested function calls in the StackTrace		
saved into test-image files	window. The	e default value in <b>Nest level</b> is 10.	
Test result details	Src Content of the test-image file		
	Dest	Data in the current HEW system or in another test-image file to be compared with	
	Example	Value differs. <b>Src</b> = F, PowerON_Reset_PC(), { 0000080E } <b>Dest</b> = F, PowerON_Reset_PC(), { 0000081C }	

#### To make detailed setting

Edit Test StackTrace	? ×
<u>N</u> est level (1-64): 10 🗮	OK I
Symbol: <u>P</u> arameter <u>L</u> ocal Variable	Cancel
Ignore address	

- 1. The range of stack trace information to be saved into the test-image file is determined by the number of nest levels specified in **Nest level**.
- 2. To compare parameters or local variables, select checkbox **Parameter** or **Local Variable**. These checkbox are not selected by default.
- 3. If you do not wish to compare the address of the selected register, select the **Ignore address** checkbox. This checkbox is not selected by default.
- 4. Click OK.



Create New Test Image Dialog Box Window Name Opened by

#### 16.6.2 Functions that can be saved into test-image files (dependent on the debugger)

Among the High-performance Embedded Workshop functions dependent on the debugger, test-image data of the test items listed below can be saved into test-image files.

The following table shows the test items in the Create New Test Image dialog box and the corresponding windows from which the data will be saved into test-image files.

Supported Debugger

For more information on the test-image data to be saved into test-image files, how to make detailed setting, and test results, see the descriptions of test items for the corresponding windows.

Test Group Name	Test Item Name	-		
Symbol	Watch	Watch	[View -> Symbol -> Watch]	<ul><li>E1/E20 Emulator Software</li><li>E10A-USB Emulator Software</li></ul>
	Locals	Locals	[View -> Symbol - > Locals]	<ul> <li>E10T-USB Emulator Software</li> <li>E8a Emulator Software</li> <li>E8 Emulator Software</li> <li>E100 Emulator Software</li> <li>E200F Emulator Software</li> <li>E6000H Emulator Software</li> <li>E6000 Emulator Software</li> <li>Simulator Debugger for SuperH Family</li> <li>Simulator Debugger for H8SX, H8S, H8 Family</li> </ul>
	ASMWatch	ASMWatch	[View -> Symbol -> ASMWatch]	<ul><li>E30A Emulator Debugger</li><li>M32C FoUSB/UART Debugger</li></ul>
	CWatch	CWatch	[View -> Symbol -> CWatch]	<ul> <li>M16C R8C FoUSB/UART Debugger</li> <li>M16C R8C Compact Emulator Debugger</li> <li>H8/300H Tiny Compact Emulator Debugger</li> <li>740 Compact Emulator Debugger</li> <li>M32C PC7501 Emulator Debugger</li> <li>M16C R8C PC7501 Emulator Debugger</li> <li>M32C PC4701 Emulator Debugger</li> <li>M16C PC4701 Emulator Debugger</li> <li>740 PC4701 Emulator Debugger</li> <li>M32C Simulator Debugger</li> <li>M16C R8C Simulator Debugger</li> <li>R32C Simulator Debugger</li> <li>740 Simulator Debugger</li> </ul>
CPU	Simulated I/O	Simulated I/O	[View -> CPU -> Simulated I/O]	Simulator Debugger for RX Family
Code	Coverage	Coverage	[View -> Code -> Coverage]	<ul> <li>Simulator Debugger for H8SX, H8S, H8 Family</li> </ul>

Feb 01, 2011



Create New Test Image Dialog Box	Window Name	Opened by	Supported Debugger
Test Group Name Test Item Name	-		
Trace	Trace	[View -> Code -> Trace]	<ul> <li>E10A-USB Emulator Software</li> <li>E10T-USB Emulator Software</li> <li>E8a Emulator Software</li> <li>E8 Emulator Software</li> <li>E200F Emulator Software</li> <li>E6000H Emulator Software</li> </ul>
			<ul> <li>E6000 Emulator Software</li> <li>Simulator Debugger for SuperH Family</li> <li>Simulator Debugger for RX Family</li> <li>Simulator Debugger for H8SX, H8S, H8 Family</li> </ul>

Test items cannot be selected when the High-performance Embedded Workshop is not connected to a target.

Support of this function depends on the debugger in use. For details, refer to the user's manual, help information, and release notes for the emulator or simulator debugger.

#### 16.6.2.1 Symbol-Watch (Watch window)

The following table shows information on the test-image data to be saved into test-image files and test results (not matched).

Test group name	Symbol		
Test item name	Watch		
Test-image data to be	Values set as detail	iled information and the range of data in the Watch window acquired	
saved into test-image files	with this setting. Check boxes for all symbols are blank by default.		
Test result details	Failed at symbol	Name of the unmatched symbol	
	Src	Content of the test-image file	
	Dest	Data in the current High-performance Embedded Workshop system or in another test-image file to be compared with	
	Example	Failed at symbol a[6], Src = (long)H'00002704{70000018}, Dest = (long)H'00000daa{70000018}	

#### To make detailed setting

If you double-click on a test item in the Create New Test Image dialog box, a further dialog box for detailed setting opens.

The scope for all watch items to be tested must be **Current Scope**. If any other scope has been selected, the items are not shown in the **Symbols** list.



Edit Test Watch		? ×
Watch Sheet:		
Watch1	•	OK
<u>Symbols:</u>		Cancel
[_]a[6]		
Current Setting:		
Watch1 : a[6]	A F	
Ignore address		

- 1. The **Watch Sheet** drop-down list contains "Watch1", "Watch2", "Watch3", and "Watch4". Symbols shown in the **Symbols** list depend on the selection made in the **Watch Sheet** drop-down list.
- 2. Each of the symbols in the **Symbols** list has a check box. By default, the check boxes are not selected. Select a watch sheet and check the boxes for symbols as required. The selected symbol is shown in **Current Setting**.
- 3. Select the **Ignore address** check box to disable comparison of the addresses of watch points. The check box is not selected by default.

Edit Test Watch	? ×
Watch Sheet:	
Watch1	OK
<u>S</u> ymbols:	Cancel
<b>✓</b> a[6]	
<u>C</u> urrent Setting:	
Watch1 : a[6]	
Ignore address	

4. Click OK.



#### 16.6.2.2 Symbol-Locals (Locals window)

The following table shows information on the test-image data to be saved into test-image files and test results (not matched).

Test group name	Symbol		
Test item name	Locals		
Test-image data to be	Values set as detai	iled information and the range of data in the Locals window acquired	
saved into test-image files	with this setting. Checkboxes for all symbols are blank by default.		
Test result details	Failed at symbol	Name of the unmatched symbol	
	Src	Content of the test-image file	
	Dest	Data in the current High-performance Embedded Workshop system or in another test-image file to be compared with	
	Example	Failed at symbol i, Src = (int)H'00000001{R14}, Dest = (int)H'00000000{R14}	

#### To make detailed setting

If you double-click on a test item in the Create New Test Image dialog box, a further dialog box for detailed setting opens.

Edit Test Locals	? ×
<u>Symbols:</u>	
min	OK
□max □i	Cancel
<u>C</u> urrent Setting:	
A.	

1. To specify a symbol to be saved into the test-image file, select the checkbox for that symbol. The selected symbol is shown in **Current Setting**.





2. Click OK.

#### 16.6.2.3 Symbol-ASMWatch (ASMWatch window)

The following table shows information on the test-image data to be saved into test-image files and test results (not matched).

Test group name	Symbol		
Test item name	ASMWatch		
Test-image data to be saved into test-image files	Values set as detailed information and the range of data in the <b>ASMWatch</b> window acquired with this setting. Checkboxes for all watch points are blank by default.		
Test result details	Failed at xxx , data value is different:	Name of the unmatched watch point	
	Src Content of the test-image file		
	Dest	Data in the current High-performance Embedded Workshop system or in another test-image file to be compared with	
	Example	Failed at 414, data value is different: Src = 0000, Dest = 0001	

#### To make detailed setting



High-performance Embedded Workshop V.4.09

Edit Test ASM	₩atch Points		? ×			
Compared <u>w</u> atcł	Compared <u>w</u> atch					
Expression	Address	Size	OK			
			Cancel			
			<u>A</u> dd			
			Add <u>B</u> it			
			<u>D</u> elete			
☐ <u>I</u> gnore addre	ss for lables and	symbols.				

- To add a watch point
  - 1. Click on the Add button on the Edit Test ASM Watch Points dialog box to open the Add dialog box.

A	dd		X
	Address:		
	Size:	Byte 💌	
	Radix:	● Hex O Dec O Bin	
		OK Cancel	_

- 2. Specify Address, Size, and Radix to set a watch point to be saved into a test-image file.
- 3. Click OK.
- To add a bit-level watch point
  - 1. Click on the Add Bit button on the Edit Test ASM Watch Points dialog box to open the BitAdd dialog box.



BitAdd			×
BitSymbol:			•
Address:		•	<u></u>
Size:	Byte		•
Bit:			
40	(	Cancel	

- 2. Specify BitSymbol, Address, Size, and Bit to set a bit-level watch point to be saved into the test-image file.
- 3. Click OK.
- To delete a watch point
  - 1. Click on the **Delete** button on the **Edit Test ASM Watch Points** dialog box. The watch point selected in the **Compared watch** list will be deleted. Only one point is selectable in the list.
- To disable comparison of the addresses of watch points
  - 1. Select the **Ignore address for labels and symbols** check box on the **Edit Test ASM Watch Points** dialog box. The check box is not selected by default.

E	Edit Test ASM Watch Points 🛛 📪 🔀				
	Compared <u>w</u> atch				
	Expression	Address	Size	OK	
	msize	000468	Word	Cancel	
				<u>A</u> dd	
				Add <u>B</u> it	
				<u>D</u> elete	
	Ignore address for lables and symbols.				

Then click OK.



#### 16.6.2.4 Symbol-CWatch (CWatch window)

The following table shows information on the test-image data to be saved into test-image files and test results (not matched).

Test group name	Symbol		
Test item name	CWatch		
Test-image data to be saved into test-image files	Values set as detailed information and the range of data in the <b>CWatch</b> window a <b>files</b> with this setting. Checkboxes for all variables are blank by default.		
Test result details	Failed at xxx , data value is different:	Name of the unmatched variable	
	Src	Content of the test-image file	
	Dest	Data in the current High-performance Embedded Workshop system or in another test-image file to be compared with	
	Example	Failed at (a)[0] ([Global]), data value is different: Src = 1783, Dest = 0	

#### To make detailed setting

E	dit Test C Variables		? ×
	Compared		
	Name	Scope	OK
			Cancel
			<u>A</u> dd
			<u>D</u> elete
	•	•	

- To add a variable
  - 1. Click on the **Add** button on the **Edit Test C Variables** dialog box to open the **Add Variables** dialog box, in which variables can be added in units of scopes.



Add Variat	? ×	
<u>S</u> cope:	[Global]	
Name		
🗖 +(sign	ed long (10)) a	
🗖 +(strue	ot Sample ) st	
	OK C	ancel

- 2. The Scope drop-down list contains Global, Local, and file names. Variables shown in the Name list depend on the selection made in the Scope drop-down list. The Name list shows global variables when Global has been selected, and local variables that can be viewed in the current scope (at the position of the program counter) when Local has been selected. When you have selected a file name, the Name list shows the file local variables included in the compilation unit to which the selected file belongs.
- 3. Each of the variables in the **Name** list has a check box. By default, the check boxes are not selected. A '+' sign shown next to a variable in the **Name** list indicates that the variable can be expanded by double-clicking. However, double-clicking on a '-' sign does not collapse the expanded variable. Even if the check box for a variable with '+' is selected, the boxes for its elements are still blank.
- 4. Select a scope and check the boxes for variables as required.
- 5. Click OK.
- To delete a variable
  - 1. Click on the **Delete** button on the **Edit Test C Variables** dialog box. The variable selected in the **Compared** list will be deleted. Only one variable is selectable in the list.



### High-performance Embedded Workshop V.4.09

Edit Test C Va	? ×	
Compared		
Name	Scope	ОК
<b>(a)(0)</b> (st).s0	(Global) [Global]	Cancel
		<u>A</u> dd
		<u>D</u> elete
I		

Then click OK.

#### 16.6.2.5 CPU-Simulated I/O (Simulated I/O window)

The following table shows information on the test-image data to be saved into test-image files and test results (not matched).

Test group name	CPU		
Test item name	Simulated I/O		
•		etailed information and the range of data in the <b>Simulated I/O</b> window s setting. No I/O simulation range is specified by default.	
Test result details	Failed at Line	Unmatched line number	
	Src	Content of the test-image file	
	Dest	Data in the current High-performance Embedded Workshop system or in another test-image file to be compared with	
	Example	Failed at Line 1: Src = ### Data Input ### Dest = a[1]=21468	

#### To make detailed setting



Edit Test Simulated I/O ? 🔀			
Simulated 1/	D:		
Options	Start Line	End Line	OK
			Cancel
			<u>A</u> dd
			<u>M</u> odify
			<u>R</u> emove
•			

- To add a simulated I/O range
  - 1. Click on the **Add** button on the **Edit Test Simulated I/O** dialog box to open the **Add Simulated I/O** dialog box.

Add Simulated I/O	? ×
Options:	Cancel

- 2. Specify All or Line No to set a I/O simulation range to be saved into a test-image file.
- 3. Click OK.
- To modify a simulated I/O range
  - Click on the Modify button on the Edit Test Simulated I/O dialog box. The simulated I/O range selected in the Simulated I/O list will be modified. Only one variable is selectable in the list. The Modify Simulated I/O dialog box opens.

Modify Simulated 1/0	? ×
Ogtions:	Cancel



- 2. Specify All or Line No to set the I/O simulation range to be saved into a test-image file.
- 3. Click OK.
- To remove a simulated I/O range
  - 1. Click the **Remove** button on the **Edit Test Simulated I/O** dialog box. The simulated I/O range selected in the **Simulated I/O** list will be removed. Only one simulated I/O range is selectable in the list.

Then click OK.

#### 16.6.2.6 Code-Coverage (Coverage window)

The following table shows information on the test-image data to be saved into test-image files and test results (not matched).

Test group name	Code		
Test item name	Coverage		
Test-image data to be saved into test-image files	Values set as detailed information and the range of data in the <b>Coverage</b> window acquired with this setting. No test range is specified by default.		
Test result details	Range	Coverage range	
	Instruction	Unmatched test option	
Execution mismatch at Unmatched address			
		Unmatched address	
	Image	Content of the test-image file	
	System	Data in the current High-performance Embedded Workshop system or in another test-image file to be compared with	
Example         Range DemoSH4.c:           Instruction Execution Time mismatch at 0x20E4		Range DemoSH4.c:	
		Instruction Execution Time mismatch at 0x20E4 Image: 1 System: 0	

#### To make detailed setting


Edit Test Coverage		? ×
Available Ranges	Ranges to be tested	
		Test Options
		☑ Address
	>>>	☑ Execution Path
		Execution <u>T</u> imes
		🗖 <u>P</u> ass
		Cancel

- 1. The Available Ranges list shows the coverage ranges that are currently available.
- 2. If you select coverage ranges in the **Available Ranges** list and click the ">>" button, these coverage ranges then appear in the **Ranges to be tested** list.
- 3. The Address checkbox among test options is always ON because addresses must be saved whenever coverage ranges are saved into test-image files (users cannot control this selection). To set other test options Execution Path, Execution Times, and Pass, select the check boxes for these options. The Execution Path checkbox is selected by default. Other check boxes are not selected by default.
- 4. Click OK.

## 16.6.2.7 Code-Trace (Trace window)

The following table shows information on the test-image data to be saved into test-image files and test results (not matched).

Test group name	Code		
Test item name	Trace		
Test-image data to be saved into test-image files	Values set as detailed information and the range of data in the <b>Trace</b> window acquired with this setting. No trace range is specified by default.		
Test result details	Comparing PTR	Compared trace range	
	Trace type	Trace type	
	Trace data is not matching. PTR	Unmatched trace range	
	Src	Content of the test-image file	
	Dest	Data in the current High-performance Embedded Workshop system or in another test-image file to be compared with	
	Example	Comparing PTR -3 to -1. Trace type : Trace. Trace data is not matching. PTR:-3 to PTR:-1 Trace data at the beginning of difference. PTR:-3 Src : 0000001879 0000107C F->DEMMW MOV.L @R15+, R1 R1<- 00000010 Dest: 0000001887 00000818 FD <e @r5="" jsr="" pc<-00001000<="" th=""></e>	



#### To make detailed setting (with one trace type)

If you double-click on a test item in the Create New Test Image dialog box, a further dialog box for detailed setting opens.

Edit Test Trace Ranges 🛛 🔋 🗙			
<u>T</u> race ranges:			
Start PTR	EndPTR		OK
			Cancel
			<u>A</u> dd
			<u>M</u> odify
			<u>R</u> emove
1			

- To add a trace range
  - 1. Click on the Add button on the Edit Test Trace Ranges dialog box to open the Add Trace Range dialog box.

Add Trace Range	? ×
Start PTR	ОК
<u>E</u> nd PTR	
	Cancel

- 2. Specify **Start PTR**, and **End PTR** to set a trace range to be saved into a test-image file.
- 3. Click OK.
- To modify a trace range
  - 1. Click on the **Modify** button on the **Edit Test Trace Ranges** dialog box. The trace range selected in the **Trace ranges** list will be modified. Only one trace range is selectable in the list.
  - 2. The Modify Trace Range dialog box opens.



Modify Trace Range	? ×
Start PTR	ОК
End PTR -2	Cancel

- 2. Specify Start PTR, and End PTR to set a trace range to be saved into a test-image file.
- 3. Click OK.
- To remove a trace range
  - 1. Click on the **Remove** button on the **Edit Test Trace Ranges** dialog box. The trace range selected in the **Trace** ranges list will be deleted. Only one trace range is selectable in the list.

Then click OK.

## To make detailed setting (with two or more trace types)

If you double-click on a test item in the Create New Test Image dialog box, a further dialog box for detailed setting opens.

Ε	dit Test Trace Range	25		? ×
	Trace ranges:			
	Trace type	Start PTR	End PTR	OK
				Cancel
				<u>A</u> dd
				Modify
				<u>R</u> emove
	•			

- To add a trace range
  - 1. Click on the Add button in the Edit Test Trace Ranges dialog box to open the Add Trace Range dialog box.



Add Trace Range	? ×
Irace type	
Internal/AUD/Usermemory_trace	
Start PTR	
	ОК
End PTR	
	Cancel

- 2. Specify **Trace type**, **Start PTR**, and **End PTR** to set a trace range to be saved into a test-image file.
- 3. Click OK.
- To change a trace range
  - Click on the Modify button on the Edit Test Trace Ranges dialog box. The trace range selected in the Trace ranges list will be modified. Only one trace range is selectable in the list. The Modify Trace Range dialog box opens.

Modify Trace Range	? ×
<u>I</u> race type	
Internal/AUD/Usermemory_trace	-
Start PTR	
-10	OK
End PTR	
-2	Cancel

- 2. Specify Trace type, Start PTR, and End PTR to set a trace range to be saved into a test-image file.
- 3. Click OK.
- To remove a trace range
  - 1. Click on the **Remove** button on the **Edit Test Trace Ranges** dialog box. The trace range selected in the **Trace** ranges list will be deleted. Only one trace range is selectable in the list.

Then click OK.



# 16.7 Comparing a test image file

#### To compare a test image file

1. Select [**Test -> Compare Test Image File**]. The **Compare** dialog box opens.

Compare	? ×
<u>I</u> est image file: <u>Compare test image file to:</u> © Current <u>s</u> ystem © Test image <u>fi</u> le:	OK Cancel
Browse	

- 2. Enter the test file location that you want to compare in **Test image file** field. This is a "\*.HIF" (Highperformance Embedded Workshop test image file) file and contains details about the test image data.
- 3. Then select what you want to compare the file you have selected with. You can choose either the Current system or another previously saved test image file. The **Current system** option is useful if you have manually executed a script and want to check the current test image data with some you saved previously.
- 4. Click OK. The results will be loaded into the test browser.

It is possible to compare test image file separately to test execution. This allows you to compare manually the current system or two test image files at a later date to when the tests were actually executed.

You can also compare test image file from the workspace window pop-up menu.

# 16.8 Running tests

This allows you to select tests and automate their execution.

### To run tests

1. Select [**Test -> Run Tests**]. The **Run Tests** dialog box opens.



High-performance Embedded Workshop V.4.09

R	un Tests - Test_S	uite_Demo	? ×
	Test <u>c</u> ases:		OK
	Test Name	Description	Cancel
	Tes_Demo	Demonstration	Bun
			Move up
			Move <u>d</u> own
	l <u>A</u> ction after test exec	ution:	
	Compare system aga	ainst saved test image file 📃 💌	
	Automatically say           Stop at first failure		

- 2. All tests currently defined in the High-performance Embedded Workshop test system are listed in the **Test cases** list.
- 3. Clicking the check box selects the test for execution in this test run.
- 4. It is also possible to modify the order a test is executed in by selecting the test and clicking the **Move up** and **Move down** buttons.
- 5. When one or more tests have been selected the **Run** button is enabled. Clicking this **Run** button starts the test run. Information will be shown in Test tab of the output window as the tests are executed. Once completed the test browser will be displayed and will show the test results for all tests executed.

The Action after test case execution drop-down list allows you two options.

- The "Compare system against saved test image file" is the normal operation and allows the current High-performance Embedded Workshop system to be compared with the test image file (\*.HIF) file that was attached to the relevant test case. These results will then be added to the test browser and will give pass, fail and failure details information to you.
- The "Refresh test image file" option simply runs all of the test cases and updates the test image file (\*.HIF) file with the system image after each test.

The **Automatically save test results** check box allows you to automatically save the results of each test execution to a text file in the same directory as the test suite. The filename used is made up from the current test suite and the date of time of the test execution.

The **Stop at first failure** will stop the automated test execution when the first test failure is met. This avoids running many tests when perhaps the first test causes all others to fail anyway.

The **Test time out** is for use when you may be executing user code. In some cases if there is a bug the code many execute and never finish during the tests execution. If any one test takes longer than the number of seconds in this box the test is terminated and is flagged as a failure.



You can also run test from the workspace window pop-up menu.

## 16.9 Using the test browser

The test browser displays the results of the last test execution.

The test browser can be easily opened by selecting [Test -> Test Results Browser].

×					I
T.	Test Name	Descriptio	n	Result	
	Tes_Demo	Demonstr	ation	FAIL	l
	<ul> <li>✓</li> <li>Component</li> </ul>		Result	 Details	
	CPU - Memory-SimSes	sionSH-4	FAIL	Failed in memory range: 0x70000000 - 0x70000027	
	Symbol - Watch-SimSe		FAIL	Symbol data is not matching.	
	CPU - 10-SimSessionSl	H-4	SUCCESS		
	•			Þ	l

The top pane of the test browser lists all of the tests that were executed during the last test run. A green icon to the left of the test name indicates the test was successful. A red icon indicates the test failed.

Selecting a test in the top pane shows the test details in the pane below. The bottom pane of the test browser lists all of the components that were checked. A green icon to the left of the component name indicates the test was successful. A red icon indicates the test failed. If a test fails the details are shown in this window.

Double clicking on the component displays more information on why the test data comparison failed. This can also be viewed by selecting the test and then clicking the details button on the toolbar.

The results of the tests can be exported to file. The two options are **Export** and **Export** All, **Export** only saves the results of the currently selected test. **Export** All saves the results of all the tests executed.

The results can be saved as text or as a comma delimited file for importing into other tools for analysis. You can also display **Test Browser** from the workspace window pop-up menu.

• Tests (top pane)

Right-clicking displays a pop-up menu containing available options.

Pop-up Menu Option	<b>Toolbar Button</b>	Function
Export		Exports the result.
Export All	ø	Exports all results.
Clear	-	Clears all results.



## • Test items (bottom pane)

Right-clicking displays a pop-up menu containing available options.

Pop-up Menu Option	<b>Toolbar Button</b>	Function
Details	<b>1</b>	Displays the details of the result.

# 16.10 Configuring the Test tab of the Workspace window

The **Test** tab was created to allow fast access to the tests in your test suite.

When the test suite is opened the test suite is added to the Test tab of the workspace window and the tests in the test suite are added underneath them. This is shown below.



Right-clicking on either the test suite name or the test gives fast access to the test facilities features.

Right-clicking on the test suite icon ((1)) allows you the following options:

Pop-up Menu Option	Function
Edit Test Suite	Edits the current test suite. Allows you to add and remove tests to the High-performance Embedded Workshop test system.
Close Test Suite	Closes the current test suite.
Compare Test Image File	Compares test image file allows you to compare a test image file with the current High- performance Embedded Workshop system or with another test image file already on your disk. The results are then displayed in the test browser.
Run Tests	Runs tests allows you to run multiple tests that you have defined in the test suite and see the results of the comparisons in the test browser. There are various options to configure the test run execution. You can also use this option to refresh the test image files automatically if you make a minor modification and they need to be updated.
Test Results Browser	This menu displays the test results for one or more test executions. It shows the pass and fail results and the detailed reason why the test failed.
Properties	The Test Suite Properties dialog box is displayed. It displays the test suite name, test suite description, test suite location and the last modified date.



Right-clicking on a test icon (

Pop-up Menu Option	Function
Run Test Case	Executes the selected test case.
Edit Test Case	Modifies the selected test case.
Remove Test Case	Deletes the selected test case.
Create New Test Image File	Creates a test-image file for the selected test case.
Compare Test Image File	Compares the test-image file with data in the current system.
Edit Test Image File	Modifies the test-image file.
Save Test Image File	Saves the current setting in the High-performance Embedded Workshop system into the test image file.
Save Test Image File As	Saves the current setting in the High-performance Embedded Workshop system as another test image file.
Properties	The Test properties dialog box opens. It displays the test name, test description, test location and the last modified date.

# 16.11 Configuring the Test tab of the Output window

The Test tab of the output window displays the results and progress of the current test execution. The test execution progress shows the current test being executed and the number of tests remaining. If errors occur then these are displayed in this window.

**Starting tests, test suite** <Test Suite Name>, <Compare Conditions>, <Save Conditions>, continue on failure, timeout=10

Running test <Test Name>, (X of Y) Running batch file/macro <Macro Name>, (X of Y)

Classification Item		Run Tests dialog box		Example (Figure below)	
Starting tests,	<test name="" suite=""></test>	-	-	'TestSuite'	
test suite	<compare conditions=""></compare>	Action after test execution drop-	Compare system against saved test image file	comparing results	
		down list box	Refresh test image file	refreshing results	
	<save conditions=""></save>	Automatically save	🔽 (Default)	saving results	
		test results		not saving results	
	<error condition=""></error>	Stop at first failure	$\checkmark$	stop on failure	
			🗖 (Default)	continue on failure	
	<time out=""></time>	Test timeout	Optional (Default = 10)	timeout=10	
Running test	<test name=""></test>	-	-	'Test_01'	
				'Test_02'	
	(X of Y)	-	-	(1 of 2)	
	X is the current test			(2 of 2)	
	number and Y is the tota	l			
	number of tests.				
Running batch	<macro name=""></macro>	-	-	'Macro_01'	
file/macro				'Macro_02'	



Classification	Item	Run Tests dialog box	Example (Figure below)
	(X of Y) -	-	(1 of 2)
	X is the current test		(2 of 2)
	number and Y is the total		
	number of tests.		

9	01 01 AL AT   \$1 \$1 02   🗈 🖬   ?
S	Starting tests, test suite 'TestSuite', comparing
R	Running test 'Test_01', (1 of 2)
	Running batch file/macro 'Macro_01', (1 of 2) Batch file/macro finished successfully
R	Running batch file/macro 'Macro_02', (2 of 2) Batch file/macro finished successfully
R	Running test 'Test_02', (2 of 2)
	Running batch file/macro 'Macro_01', (1 of 2) Batch file/macro finished successfully
R	Running batch file/macro 'Macro_02', (2 of 2) Ratch file/macro finished successfully
A	all tests completed
	Test /

Right-clicking displays a pop-up menu containing available options.

A basic operation is allocated to the toolbar.

The **Toolbar display** and **Customize toolbar** options are also included in the pop-up menu opened by right-clicking on the toolbar.

Pop-up Menu Option	<b>Toolbar Button</b>	Function
Clear Window	a	Clears the contents of the window.
Save		Saves the contents of the window into a text file.
Сору		Copies the selected contents onto the Windows® clipboard.
Toolbar display	-	Shows or hides the toolbar.
Customize toolbar	-	Customizes toolbar buttons.



# 17. Debugging Facility

This chapter describes the debugging operations and their related windows and dialog boxes.

See Chapter 1, Overview, for the fundamental concepts of the High-performance Embedded Workshop (High-performance Embedded Workshop).

For details on the functions available with the debugger in use, refer to the user's manual or help of the emulator or simulator.

# 17.1 Preparations for debugging

This section describes the preparations for debugging your program. You will learn how to select and configure a debugger with which to debug, how to load the user program, and what the debugger sessions are.

## 17.1.1 Compiling for debug

In order to be able to debug your program at C/C++ source level, your C/C++ program must be compiled and linked with the **Debug** option enabled. When this option is enabled, the compiler puts all the information necessary for debugging your C/C++ code into the absolute file or management information file, which are usually called **Debug Object Files**. When you create your project the initial setup will normally be configured for debug.

## Notes:

- Make sure you have the debug option enabled on your compiler and linker when you generate an object file for debugging.
- If your debug object file does not contain any debugging information (for example, the S-Record format), then you can still load it into the debugger, but you will only be able to debug at the assembly language level.

## 17.1.2 Selecting a debugger

Selecting the debugger is very dependent on the installation of the High-performance Embedded Workshop. If the Highperformance Embedded Workshop has a toolchain installed then the application project generator will be able to set up both the toolchain and the debugger targets simultaneously. This allows the options for targets and toolchain to be matched closely so that no inconsistencies occur. If there is no toolchain installed you will only be able to select debugonly project types. By default, High-performance Embedded Workshop will display a debug-only project generation type for each CPU family in the **New Project Workspace** dialog box.

#### Note:

The project generator in the High-performance Embedded Workshop automatically generates sample source files for the microcomputer in use (for some project types, however, no files are generated). These files contain sample code and this does not indicate that the operations of all programs based on that code are guaranteed. Please read the hardware manual for the microcomputer to check if the generated source code is applicable and modify the code as required.





The **New Project Workspace** dialog box allows you to select a project type for generation, which matches your CPU target.

Project Type	Description
Application	Project for generating an execution program that includes the initial routine file written in the C/C++ language.
Assembly Application	Project for generating an execution program that includes the initial routine file written in the assembly language.
Demonstration	Project for generating a demonstration program written in the C language.
C source startup Application	Project for generating a startup program written in the C language.
Empty Application	Project for only setting the toolchain environment (no generation file).
Import Makefile	A project to create an executable program by importing an existing makefile.
Library	Project for generating a library file (no generation file).
Debugger only - xxxxxx	A debug-only project (no generation file).

In the example below, the SH-4 simulator/debugger is assumed. For details on the functions available with the debugger in use, refer to the user's manual or help of the debugger.



New Project-1/9-Select Target CPL	J.Toolchain version 🛛 🔋 🗙
	Toolchain version : 9.2.0.0
	Which CPU do you want to use for this project? CPU Series:
	SH2A-FPU SH2-DSP SH-3 SH3-DSP SH-4
	CPU Type:
	SH7750         Image: Constraint of the second
	If there is no CPU type to be selected, select the "CPU Type" that a similar to hardware specification or select "Other".
< Back	Next > Finish Cancel

1. Select the CPU and Toolchain version in Step 1. The CPU types (**CPU Type**) are classified according to the CPU series (**CPU Series**). Select the CPU corresponding to the program to be developed because the generation file differs according to the **CPU Series** and **CPU Type** settings. If there is no corresponding CPU, select a CPU with similar hardware specifications or **Other**.

The following buttons at the bottom of the dialog box are the same as those in the New Project wizard dialog box.

Next>	Moves to the next display.
<back< td=""><td>Returns to the previous display.</td></back<>	Returns to the previous display.
Finish	Opens the <b>Summary</b> dialog box (selections followed by this button are default).
Cancel	Returns to the New Project Workspace dialog box.

To move to Step 2, click the **Next>** button in Step 1.



New Project-2/9-Option Setting	? ×
	Specify global options. Division : CPU Endian : Big FPU : Mix Round to : Zero
	<ul> <li>Denormalized number allower as a result</li> <li>Position independent code (PIC)</li> <li>Treat double as float</li> <li>Bit field's members are allocated from the low</li> <li>Pack struct, union and class</li> <li>Use try, throw and catch of C++</li> <li>Enable/disable runtime information</li> </ul>
< Back	Next > Finish Cancel

Specify the options common to all project files in Step 2. The specifiable items depend on the CPU selected in Step 1.

To move to Step 3, click the **Next>** button in Step 2.



What kind of initialization routine would you like to create?   Use I/O Library Number of I/O Streams: 20	would you like to create?	New Project-3/9-Setting the Conte	ents of Files to be Generated	? ×
Generate Hardware Setup Function			What kind of initialization routine would you like to create? Use I/O Library Number of I/O Streams: 20 20 20 20 20 20 20 20	? ×

3. Specify the generation file in Step 3.

Use I/O Library:		Checking enables use of standard I/O libraries.		
	Number of I/O Streams:	Specifies the number of I/O streams that can be used simultaneously.		
Use	Heap Memory:	Checking enables use of the heap area management function sbrk().		
	Heap Size:	Specifies the unit of the size of the heap area to be managed.		
	Generate main() Function:	Selects generation of a model main function. Generates a main function file (Project name).c(cpp).		
I/O	Register Definition Files:	Checking generates an I/O register definition file (iodefine.h) written in the C language.		
	Generate Hardware Setup Function:	Selects generation of a model I/O register initial setting program. Generates a hardware setting file (hwsetup.c(cpp) or hwsetup.src).		

#### Note:

To include a main function that has already been made, select **None** in **Generate main() Function** and after making the project, add the file containing the main function to the project. Note that if the name of the function to be included is different, the function calling section in resetprg.c must be modified. Be sure to refer to the hardware manual of the CPU for actual values of the sample file contents, such as the vector table definition and I/O register definition, which are generated by the project generator.

To move to Step 4, click the **Next>** button in Step 3.



New Project-4/9-Setting the Standard Library
Image: Section of the section of th

4. Specify the configuration of the standard libraries used by the C/C++ compiler in Step 4.

The functions defined in the checked items and the runtime functions are included.

Enable all:	Selects all standard library functions.
Disable all:	Does not select all standard library functions. Note that only the minimum required
	functions, runtime and new, are selected.

To move to Step 5, click the **Next>** button in Step 4.



New Project-5/9-Setting the Stack Area		? ×
	What are the stack settings?	
	Stack Pointer Address: (power-on reset) H'73FFFFFC Stack Size: H'400	
< Back Ne	xt> Finish Can	cel

5. Specify the stack area in Step 5. This is done by setting the initial value of the stack pointer and the stack size. The initial value of the stack areas depends on the CPU selected in Step 1.

## Note:

The stack area is defined by stacksct.h which is generated by the High-performance Embedded Workshop. If stacksct.h has been modified by an editor, it cannot be modified from [**Project -> Edit Project Configuration**] in the High-performance Embedded Workshop.

To move to Step 6, click the **Next>** button in Step 5.



## 17. Debugging Facility

New Project-6/9-Setting the Vector	? ×
	What supporting files would you like to create? Vector Definition Files Vector Handlers: Handler Vector PowerON_Reset H'000 Power On Rese Manual_Reset H'020 Manual Reset I
< Back	Next > Finish Cancel

6. Specify the vector in Step 6.

Vector Definition Files:	Checking generates a vector definition file and a vector table setting function definition file.		
Vector Handlers:	Handler:	Displays the handler program name of the reset vector. To modify the handler program, after selecting the handler program name by clicking on it, enter the new handler program name. Note that if the handler program is modified, a reset program (resetprg.c) is not generated.	
	Vector:	Displays a description of the vector.	

## Note:

Since the generated reset program, interrupt functions, reset vector handlers, and interrupt source register definitions are samples, be sure to refer to the CPU hardware manual.

To move to Step 7, click the **Next>** button in Step 6.



Image: Simulator (Little endiar)         Image	New Project-7/9-Setting the Target System for Debugging	×
	SH-4 Functional Simulator         SH-4 Functional Simulator(Little endian)         SH-4 Simulator(Little endian)         SH-4 Simulator(Little endian)         SH-4 With BSC Simulator(Little endian)         SH-4(SH7750R) Functional Simulator(Little endian)         SH-4(SH7750R) Simulator(Little endian)         Image type :         SH-4         Image type :         All CPUs	

7. Specify the debugger targets in Step 7.

Targets:	Sets the debugger targets. Select (by checking) the debugger targets. No selection or a selection of more than one target is possible.
Target type:	Specifies the type of the targets displayed in Targets.
Target CPU:	Specifies the CPU of the targets displayed in Targets.

#### Note:

The endian type selected in step 2 will be applied to the compiler settings. This is separate from the endian type of the debugger target selected in step 7.

To move to Step 8, click the **Next>** button in Step 7.



New Project-8/9-Setting the Debugger	r Options ? 🗙
	Target name : SH-4 Simulator
	Core : <single core=""></single>
	Configuration name : SimDebug_SH-4
	Detail options :
	Item         Setting           Simulator I/O         disable           Simulator I/O addr.         0x4           Bus mode         0
	☐ Initial session
< Back	Next > Finish Cancel

8. Set the options for the debugger targets selected in Step 8.

Core:	Specifies the target core.		
Configuration name:	By default, the High-performance Embedded Workshop generates two configurations: <b>Release</b> and <b>Debug</b> . If a debugger target is selected, a configuration for the selected target is also generated (an abbreviation including the target name). This configuration name can be changed in <b>Configuration name</b> .		
Detail options:	ptions: Sets the debugger target options. To modify an option, select If the selected item cannot be modified, <b>Modify</b> remains gray every selected.		
	Simulator I/O:	System call for standard I/O or file I/O from the user program is enabled ( <b>Enable</b> ) or disabled ( <b>Disable</b> ).	
	Simulator I/O addr.:	Address for above system call.	
	Bus mode:	Currently cannot be used by the simulator/debugger.	
	Initial session:	If <b>Initial session</b> is checked, this session becomes an initial session.	

To move to Step 9, click the **Next>** button in Step 8.



New Project-9/9-Changing the File Nam	nes to be C	reated	? ×
	The following generated: File Name	g source Ext	files will be Description
	dbsct typedefine sbrk iodefine intprg vecttbl vect env vhandler resetprg Tutorial sbrk stacksct	c h src src inc src c h h	Setting of B,R Section Aliases of Integer Type Program of sbrk Definition of I/O Regis Interrupt Program Initialize of Vector Tab Definition of Vector Define Interruput Ever Reset/Interrupt Handle Reset Program Main Program Header file of sbrk file Setting of Stack area
and the second	•		Þ
< Back	Next >	Fini	sh Cancel

9. The files to be generated by the High-performance Embedded Workshop based on the settings made so far are displayed as a list in Step 9.

File Name:	File name
	To change a file name, after selecting the file name by clicking on it, enter the new file
	name.
Extension:	File extension
Description:	Description of the file

Clicking the Finish button in Step 9 displays the Summary dialog box.



Summary	? ×	
Project Summary:		
CPU SERIES : SH-4 CPU TYPE : SH7750	efine.h c ne.h	
Click OK to generate the project or Cancel to abort.		
OK	Cancel	

The project generator displays information on the project to be generated in the Summary dialog box. After confirming the display contents, click the OK button. Clicking Cancel returns to the New Project wizard dialog box. Checking Generate Readme.txt as a summary file in the project directory will save the project information displayed in the Summary dialog box as a text file named Readme.txt in the project directory.

## 17.1.2.1 To create a new project having information from makefile

High-performance Embedded Workshop can analyze GNU make format and Hmake format (High-performance Embedded Workshop generated) and create a workspace which has file information from makefile.

Open the **New Project Workspace** dialog box and select the **Import Makefile** as project type. After supplying some fields (e.g.: Workspace Name) and pressing OK button, **New Project - Import Makefile** dialog box opens.





When a makefile is selected by **Makefile path**, **Source files** shows source files in the makefile. To view the source files in the makefile again, click **Start**.

To apply toolchain options such as the compiler, select the **Import options** check box.

If you want to remove a file from the project, you can remove it by selecting the file and pressing **Remove** button. And if you want to add a file to the project, you can add it by pressing **Add** button.

Selecting the **Show file path** check box shows the full path of the file.

## 17.1.3 Editing project configuration

If you are using the SuperH or H8SX, H8S, and H8 family toolchains then it is possible to configure the simulator again using the project generator. This feature is not enabled for the demonstration project type.

- 1. Select [Project -> Edit Project Configuration]. The Edit Project Configuration dialog box opens.
- 2. Click on the **Target** tab.
- 3. Select the target you wish to use and then click OK.



## 17.1.4 Configuring the debugger

Before you can load a program into your debugger you must set it up to match your application's system. The items that must be set-up are typically device type, operating mode, clock speed and the memory map. It is particularly important to set-up the memory map, as you must have memory in the debugger, into which your user code will be loaded.

In the High-performance Embedded Workshop, the project generation process will have completed much of this work. However if you are using a different configuration of board from the standard types then some customization will be essential.

## 17.1.4.1 Setup

To set-up the debugger configuration choose [**Setup -> Simulator**] or [**Setup -> Emulator**]. Under this sub menu will be the menus which can be used to configure your debug platform.

In the case of the SuperH family Simulator the available menus are the **System** and **Memory resource**. These options both allow the simulator to be customized and setup to your requirements.

You will be presented with a set-up dialog specific to the debugger that you chose in the Debug Settings dialog.

For a detailed description of the features available in your debugger, please refer to the separate Debugger User Manual.

## 17.1.4.2 Memory mapping

For the debugger to correctly represent your user system, the memory map must be set up. It needs to know which areas in the device's address space are RAM, ROM, on-chip registers or areas where there is no memory.

When you select the device type and mode in the project generator, the High-performance Embedded Workshop will automatically set up the map for that device and the mode in which the processor is operating. For example, in a device with internal ROM and RAM, the areas where these are located in the device's memory map will be set by default.

If you are using a device that does not have internal memory, or a device with external memory instead of (or in addition to) the internal memory, then you must tell the debugger that you have memory there.

## Tip:

If you are trying to debug code with an emulator and need some memory available that does not exist either on-chip or externally (in your hardware), then you can map some *emulation memory* from the emulator to the address space for your application to use.

The details will be specific to the debugger that you chose in the new project.

Additional information about the memory mapping can be viewed in the **System Status** view's **Memory** pane. The **Device Configuration** area shows how the memory in the device's address space.

#### Note:

Due to page length limitations in some emulators, the range addresses may not exactly match the entered addresses.



## 17.1.4.3 Editing debug settings

Normal operation of the High-performance Embedded Workshop debugger means that your target and download modules will be automatically configured in the project generation process. However in some cases it may be necessary for you to manually configure your debug session. This is often the case when using old toolchains and project generators that do not support the latest High-performance Embedded Workshop interfaces.

To check your debug session setup display the Debug Settings dialog box. Select one of the following operations

- Select [Debug -> Debug Settings], OR
- Right-click on the download module or within the **Download modules** folder on the Projects tab of the workspace window and select **Debug Settings** from the pop-up menu.

Debug Settings		? X
Debug Settings SimSessionSH-4  Project_Name	Target       Options         Iarget:       SH-4 Simulator         Core:       Single Core Target         Debug format:       Image:         Elif/Dwarf2       Image:         Download modules:       Image:         Filename       Offset Address         \$(CONFIGDIR)\\$(PR0       00000000         Elif/Dwarf2       Image:	Add Modify Remove
	ОК	Cancel

**Target**, **Core**, **Debug format**, or **Download modules** can be selected on the **Target** tab of the **Debug Settings** dialog box. Note that **Core** can be specified by using Synchronized Debugging facility.

#### To change the target the following operations is necessary

- 1. Select the project that needs to be changed in the tree on the left of the dialog. It defaults to the current project.
- 2. Select the session which is to be modified in the drop list above the tree.
- 3. Change the target using the target drop list control. This removes any target specific setup options that have been previously been set.



Moreover, the **Options** tab of the **Debug Settings** dialog box provides the following options.

- Configuring the automatic command line batch file execution
- Not executing a batch file when a module containing only debug information is downloaded
- Downloading modules after build
- Removing breakpoints on download
- Disabling memory access until after target connection command file execution
- Limiting disassembly memory access
- Not performing automatic target connection
- Resetting CPU after download module
- Disabling memory access by GUI when target is executing

#### (1) Configuring the automatic command line batch file execution

The High-performance Embedded Workshop debugger is tightly integrated with the TCL command line facilities. This means that it is possible to write batch files for the High-performance Embedded Workshop debugger which can be executed automatically at certain times. The **Command batch file load timing** list displays the order in which the files will be executed. It is possible to Add, Modify, Remove, Up and Down the files in this list.

#### To configure the automatic command line batch file execution

- 1. Select [**Debug -> Debug Settings**]. The Debug Settings dialog box opens.
- 2. Select the **Options** tab.
- 3. Select the **Command batch file load timing**. This can be the following values, "At target connection", "Before download of modules", "After download of modules", and "After reset".

Command batch file load timing:	
At target connection	
At target connection Before download of modules	<u>A</u> dd
After download of modules After reset	<u>M</u> odify
	<u>R</u> emove
	Up
	D <u>o</u> wn

4. Then click Add. The debugger will then display the add Command Line File dialog box.



Command Line File	? ×
Eilename: Browse	OK.
Initial Directory: Browse	Cancel
Apply to all timings	

- 5. Enter the command-batch file name in the **Filename** field. If you wish to insert a placeholder into the **Filename** field, click the placeholder button () and select the placeholder from the pop-up menu. To browse a file, click the Browse button.
- 6. In the **Initial Directory** field, enter the name of the directory where you wish to execute the command batch file. By this setting, you can also use command batch files that have relative paths. If you wish to insert a placeholder into the **Initial Directory** field, click the placeholder button ()) and select the placeholder from the pop-up menu. To browse a directory, click the Browse button.
- 7. Selecting the **Apply to all timings** check box adds the batch file to every timing. ("At target connection", "Before download of modules", "After download of modules", and "After reset".)
- 8. Click OK to add the batch file.
- 9. Once added it can moved into the correct place in the order by using the **Up** and **Down** buttons. This is only valid if you are adding multiple command line batch files.
- 10. Click OK.

#### Notes:

- If you use the FILE\_LOAD or FILE\_LOAD\_ALL command to download a module, the command batch file specified for "Before download of modules" or "After download of modules" is not executed.
- If you use the RESET command to reset the CPU, the RESET command written in the command batch file specified for "After reset" is not executed.

#### (2) Not executing a batch file when a module containing only debug information is downloaded

If this checkbox is selected, a batch file will not be executed when **Download** (**Debug Data Only**) is selected in the popup menu displayed by right-clicking the module in the workspace window even in cases where **Before download of modules** or **After download of modules** has been selected for Command batch file load timing and a command batch file has been specified.

#### To not execute a batch file when a program containing only debug information is downloaded

- 1. Select [Debug -> Debug Settings]. The Debug Settings dialog box opens.
- 2. Select the **Options** tab.
- 3. Select the **Disable batch file execution when downloading debug information**. By default, this checkbox is selected.



4. Click OK.

#### (3) Downloading modules after build

If this checkbox is selected, the user program will be automatically downloaded after a build.

#### To download modules after build

- 1. Select [Debug -> Debug Settings]. The Debug Settings dialog box opens.
- 2. Select the **Options** tab.
- 3. Select the **Download modules after build**. By default, this checkbox is selected.
- 4. Click OK.

#### (4) **Removing breakpoints on download**

If this checkbox is selected, the breakpoints will be automatically removed after downloading the user program.

#### To remove breakpoints on download

- 1. Select [**Debug -> Debug Settings**]. The **Debug Settings** dialog box opens.
- 2. Select the **Options** tab.
- 3. Select the **Remove breakpoints on download**. Default for this checkbox depends on the target.
- 4. Click OK.

#### (5) Disabling memory access until after target connection command file execution

When this checkbox is selected, if a command batch file to be automatically executed at connection to the target has been specified, the memory in the target will not be accessed until execution of the command batch file is finished.

#### To disable memory access until after target connection command file execution

- 1. Select [**Debug -> Debug Settings**]. The **Debug Settings** dialog box opens.
- 2. Select the **Options** tab.
- 3. Select the **Disable memory access until after target connection command file execution**. This option is useful when initial setting of memory is necessary at connection to the target. By default, this checkbox is not selected.
- 4. Click OK.



#### (6) Limiting disassembly memory access

Disables reading data from memory outside the displayed range in disassembly mode.

#### To limit disassembly memory access

- 1. Select [Debug -> Debug Settings]. The Debug Settings dialog box opens.
- 2. Select the **Options** tab.
- 3. Select the Limit disassembly memory access. Default for this checkbox depends on the target.
- 4. Click OK.

#### (7) Not performing automatic target connection

If this checkbox is selected, the target is not connected until you select [**Debug->Connect**]. Support for this function depends on the debugger.

#### To not perform automatic target connection

- 1. Select [Debug -> Debug Settings]. The Debug Settings dialog box opens.
- 2. Select the **Options** tab.
- 3. Select the **Do not perform automatic target connection**. This checkbox is only enabled if the feature is supported by the selected target. Selecting a new target will reset this option to the default setting for the target. Default for this checkbox depends on the target.
- 4. Click OK.

#### (8) Resetting CPU after download module

If this checkbox is selected, the target will be automatically reset after downloading the user program. Support for this function depends on the debugger.

#### To reset CPU after download module

- 1. Select [Debug -> Debug Settings]. The Debug Settings dialog box opens.
- 2. Select the **Options** tab.
- 3. Select the **Reset CPU after download module**. This checkbox is only enabled if the feature is supported by the selected target. Selecting a new target will reset this option to the default setting for the target. Default for this checkbox depends on the target.
- 4. Click OK.

#### (9) Disabling memory access by GUI when target is executing

By setting this option, the user can limit memory accesses from the High-performance Embedded Workshop components during execution. This prevents the target being overloaded providing memory data and degrading execution. Support for this function depends on the debugger.

#### To disable memory access by GUI when target is executing

- 1. Select [Debug -> Debug Settings]. The Debug Settings dialog box opens.
- 2. Select the **Options** tab.
- 3. Select the **Disable memory access by GUI when target is executing**. This checkbox is only enabled if the feature is supported by the selected target. Selecting a new target will reset this option to the default setting for the target. Default for this checkbox depends on the target.
- 4. Click OK.

It does not completely prevent memory access.

• Operations in which memory accesses are prevented

In the following operations, handling of memory is disabled or no data will actually be read from or written to memory in the target if attempted. Values to be shown are dummy values ('FF').

- Viewing tooltip watch information in the Editor or Disassembly window in the source mode
- Viewing or modifying the disassembly code in the Editor or Disassembly window in the mixed or disassembly mode
- Viewing or modifying the memory values in the Memory, IO, or MR window \*
- Operations in which memory accesses are not prevented

Memory accesses are not prevented in the following operations.

- Executing commands in the Command Line window or command files
- o Viewing or modifying the memory values in the Watch window with real-time update enabled \*
- Viewing the memory values in the Monitor, Image, or Waveform window with real-time update enabled \*
- Viewing or modifying the memory values in the CWatch or ASMWatch window with real-time RAM monitor function \*
- Viewing the memory values in the RAM Monitor or MR window with real-time RAM monitor function \*
- Viewing or modifying the memory values in the CWatch or ASMWatch window \*
- o Viewing the memory values in the GUI I/O, Image, or Waveform window \*

#### Note:

\*. Support for this function depends on the debugger.

## 17.1.5 Downloading modules

## 17.1.5.1 Setting the downloading a program

Once you have made sure that there is memory in your system in which to download your code, you can then proceed to download a program to debug. The initial selection of download module is automatic with regard to an application generator, as it is the output from the linker. However, with regard to the debug-only project generator, it is possible for you to browse to the module that you wish to download. This is outlined in the section 17.1.2, Selecting a debugger.

Debug Settings		? ×
SimSessionSH-4	\$(CONFIGDIR)\\$(PRO 00000000 Elf/Dwarf2	Add Addify Hemove Down
	OK	Cancel

It is also possible to manually choose download modules after the project creation. This is achieved via the **Debug Settings** dialog box. This dialog box allows you to control the debug settings throughout your workspace. The tree on the left of the dialog contains all of the current projects. Selecting a session in this tree will then show you the settings for that project and the session selection in the session drop-down list. In this list box, it is possible to select multiple sessions or all sessions. If you select multiple configurations you can choose to modify the settings for one or more configurations at once. The **Debug Settings** dialog box displays the following debug options:

- Current debug target for the current project and configuration selection.
- Download modules for the current project and configuration selection.



Select the debug format in the **Debug format** list. You can only debug modules that match the format specified here. Even if there is a mixture of file formats in the **Download modules** list below only the one chosen here can be viewed.

The **Download modules** list displays the order in which the files will be downloaded to the target. It is possible to Add, Modify, Remove, Up and Down modules in this list.

## To add a new download module

- 1. Select one of the following operations to open the **Debug Settings** dialog box:
  - Select [Debug -> Debug Settings], OR
  - Right-click on a module or within the Download modules folder on the Projects tab of the workspace window to display a pop-up menu. Select **Debug Settings**.
- 2. In the project tree, select the project and configurations to which you want to add a download module.
- 3. Click the **Add** button. The **Download Module** dialog box is displayed. All fields must be setup for the download module to be configured correctly.

Download M	odule	? ×
<u>O</u> ffset:		OK
<u>F</u> ile format:	Elf/Dwarf2	Cancel
File <u>n</u> ame:	\$(CONFIGDIR)\\$(PROJECTNAME).abs	<u>B</u> rowse
<u>A</u> ccess size:	1	
Download	I debug information only	
Perform memory verify during download		
Download automatically on <u>target</u> connection		

- 4. The Offset field specifies the memory address offset the module will be loaded at. It defaults to 0. After the module has been added, the offset value is shown on the right of the module name in the [Debug -> Build] submenu, the Download modules folder on the Projects tab of the workspace window, and the [File -> Recently Download Module] submenu.
- 5. The **File format** drop-down list box contains a list of supported object format. The file format of the download module is selected here. So this does not have to match the debug format on the **Debug Settings** dialog box. However, the Stack Trace window will be supported for instance depending on the format specified in the **Debug format** field.
- 6. The **Filenames** field can be specified with placeholders or as an absolute setting. It is recommended to use placeholders.
- 7. The **Access size** field specifies the access size when the memory is accessed. If the memory needs to be accessed by a specific access size, specify the access size by using this option.



- 8. The **Download debug information only** checkbox will download the debug information only without downloading the code.
- 9. The **Perform memory verify during download** checkbox can be used to do additional checks when downloading the module to ensure it was correctly downloaded to the target device.
- 10. The **Download automatically on target connection** checkbox can be used to automatically download the module when the target is being connected.
- 11. When you click the OK button, the debug download module is added to the bottom of the list.

For automatically downloading an existing module to the target when adding it to the Download modules list, see section 17.1.5.2, Downloading modules.

#### To change the setting of a download module

- 1. Select one of the following operations to open the **Download Module** dialog box:
  - Open the **Debug Settings** dialog box and select a module, for which you wish to modify the setting, in the **Download modules** list and click the **Modify** button, **OR**
  - Right-click on the download module or within the Download modules folder on the Projects tab of the workspace window and select **Properties** from the pop-up menu.
- 2. Change the setting and click the OK button.

#### To remove download modules

- 1. Open the **Debug Settings** dialog box.
- 2. Select a module, which you wish to remove, in the **Download modules** list and click the **Remove** button.

#### To remove selected download modules using the Projects tab of the Workspace window

- 1. Select the download modules that you want to remove in the Projects tab of the Workspace window. Multiple download modules can be selected by holding down the SHIFT or CTRL key.
- 2. Select one of the following operations:
  - Select **Remove** from the pop-up menu opened by right-clicking, **OR**
  - Press **Delete** key.
- 3. A confirmation dialog box opens for you to select whether or not to remove the selected download modules from the project. To delete the selected download modules, select **Yes**. Otherwise select **No** or **Cancel**.

If you do not wish to open this confirmation dialog box, select the **Don't ask this question again** checkbox. To open this dialog box again, select [**Setup -> Options**] to open the Options dialog box. Select the **Remove download module(s) from project** checkbox on the Confirmation tab. By default, this checkbox is selected.



#### To change the order of a module to be downloaded to the target

- 1. Open the **Debug Settings** dialog box.
- 2. Select a module in the **Download modules** list and click the **Up** or **Down** button.

Any changes made in the Debug Settings dialog box are only changed when you click OK.

#### 17.1.5.2 Downloading modules

Download the object program to be debugged.

#### To download modules

Select one of the following operations:

- Select the module, which you wish to download, from the [Debug -> Download Modules] submenu, OR
- Double-click the module in the Download modules folder on the Projects tab of the workspace window, OR
- Right-click on the module in the Download modules folder on the Projects tab of the workspace window to display a pop-up menu. Select **Download** or **Download** (**Debug Data Only**).

The High-performance Embedded Workshop shows the most recently downloaded modules on the submenu of [File -> Recently Download Modules]. This is useful when you wish to download a module that you recently used.

#### To download all modules

Select one of the following operations:

- Select [Debug -> Download Modules -> All Download Modules], OR
- Right-click within the Download folder on the Projects tab of the workspace window to display a pop-up menu. Select **Download all modules**.

Select the [**Debug -> Debug Settings**] menu option. The Debug Settings dialog box opens. The files will be downloaded to the target in order of the **Download modules** list.

An existing module, which is not yet added to the Download modules list, can be automatically downloaded to the target when it is added to the Download modules list.



#### To automatically download an existing module to the target when adding it to the Download modules list

- 1. Select one of the following operations to open the **Download Module** dialog box:
  - Select [File -> Download A New Module], OR
  - Right-click on a module or within the Download modules folder on the Projects tab of the workspace window to display a pop-up menu. Select **Download A New Module**.

All fields must be setup for the download module to be configured correctly.

Download M	odule	? ×
<u>O</u> ffset:		OK
<u>F</u> ile format:	Elf/Dwarf2	Cancel
File <u>n</u> ame:	\$(CONFIGDIR)\\$(PROJECTNAME).abs	<u>B</u> rowse
<u>A</u> ccess size:	1	
Download	debug information only	
Perform memory verify during download		
Download automatically on target connection		

- 2. The Offset field specifies the memory address offset the module will be loaded at. It defaults to 0. After the module has been added, the offset value is shown on the right of the module name in the [Debug -> Build] submenu, the Download modules folder on the Projects tab of the workspace window, and the [File -> Recently Downloaded Modules] submenu.
- 3. The **File format** drop-down list box contains a list of supported object format. The file format of the download module is selected here. So this does not have to match the debug format on the **Debug Settings** dialog box. However, the Stack Trace window will be supported for instance depending on the format specified in the **Debug format** field.
- 4. The **Filename** field can be specified with placeholders or as an absolute setting. It is recommended to use placeholders.
- 5. The **Access size** field specifies the access size when the memory is accessed. If the memory needs to be accessed by a specific access size, specify the access size by using this option.
- 6. The **Download debug information only** checkbox will download the debug information only without downloading the code.
- 7. The **Perform memory verify during download** checkbox can be used to do additional checks when downloading the module to ensure it was correctly downloaded to the target device.
- 8. The **Download automatically on target connection** checkbox can be used to automatically download the module when the target is being connected.
- 9. Click the **OK** button.



Select [**Debug -> Debug Settings**]. The **Debug Settings** dialog box will be invoked. The module is added to the bottom of the **Download modules** list on the **Target** tab.

If a module previously downloaded is modified outside the High-performance Embedded Workshop, a confirmation dialog box appears asking if you wish to download this module again. To download the module again, select **Yes**. Otherwise select **No** or **Cancel**.



If you do not wish to open this confirmation dialog box, select the Don't ask this question again checkbox.

#### To open the confirmation dialog box again

- 1. Select [Setup -> Options]. The Options dialog box will be displayed.
- 2. Select the **Confirmation** tab.
- 3. Select the **Reload out-of-date download modules** checkbox. This checkbox is selected by default.
- 4. Click OK.

## 17.1.5.3 Check for changed source files before download

The High-performance Embedded Workshop can check to see if any of the source files have changed before the download module is downloaded in the current project. If files have been modified then a confirmation is launched which asks the user if they wish to rebuild the code before the download takes place. To rebuild the code before the download again, select **Yes**. To not rebuild the code before the download again, select **No**.



If you do not wish to open this confirmation dialog box, select the **Don't ask this question again** checkbox.


### To open the confirmation dialog box again

- 1. Select [Setup -> Options]. The Options dialog box will be displayed.
- 2. Select the **Confirmation** tab.
- 3. Select the Build out-of-date download modules checkbox. This checkbox is selected by default.
- 4. Click OK.

## 17.1.5.4 Showing a source tree on download (debug-only project)

While a debug-only project (i.e., "Debugger only - xxxxx") is in use, a tree of source files for a download module are automatically shown in the **Projects** tab of the workspace window once the module has been downloaded.

The files for a download module are retrieved when the module is downloaded.

If the files are not full paths and are relative or short file names, they will automatically be searched for relative to the download module directory.

Files that cannot be found after the initial search will be shown to you in the Locate Files dialog box.

However, you can choose not to open the Locate Files dialog box at downloading.

#### To prevent opening of the Locate Files dialog box at downloading

- 1. Select [Setup -> Options]. The Options dialog box will be displayed.
- 2. Select the **Confirmation** tab.
- 3. Select the **Do not show Debug only workspace browse dialog on download** checkbox. This checkbox is not selected by default.
- 4. Click **OK**.



## (1) Locating files

Files that cannot be found or do not exist after the initial search will be shown to the user in the **Locate Files** dialog box (in the **Files to find** list).

Locate Files	? ×
Last searched directory: C:\Workspace\\DebugOnly\Debug_Emulator	
Files to find: Search status: 0 / 13 found/unfound	
_h_c_lib.h dbsct.c machine.h resetprg.cpp sbrk.c sbrk.h	▲ ▼
Browse:	
Workspace     DebugOnly     DebugOnly     Debug_Emulator     Source     OmenoSH4     Ovrkspace_Name     Compact Disc (D:)	
Look in subdirectories Do not show on download Search OK C	Cancel

#### To locate files

- 1. The **Last searched directory** holds the last directory that was searched for these module files.
- 2. The **Search status** field shows the number of files that were found against the number that have not this is only relevant to the files in the dialog and do not account for any previous searches.
- 3. The **Files to find** list holds a list of all the files that could not be found, they will appear in short filename format, and will be searched for in this manner.
- 4. The Browse section allows the user to select a directory in which to search the remaining files for.
- 5. The Look in subdirectories check box will look for the files in all sub-directories for the chosen directory.
- 6. The **Do not show on download** check box will not bring up this dialog box on download if there are files that can not be located. This checkbox is also present in the **Confirmation** tab of the **Options** dialog box, and can be turned off and on here.
- 7. Pressing **Search** will look for the files in the **Browse** list in the chosen directory, files that have been found will be removed from the **Files to find** list.

Any searched directories in the **Browse** section will be shown by a folder and magnifying glass icon (()). Once the user has selected **Search** this button will be changed to a **STOP** button allowing the user to cancel a search. Any files that had been found before the search was cancelled will return to the unlocated state.



- 8. Clicking **OK**, then the files that were searched for and found will be located to that directory, accept all searches, and close the dialog box.
- 9. Clicking **Cancel** completely cancel all searches that have been done, and close the dialog without changing anything.

If you had done some searches and they clicked on cancel a confirmation box will appear as below;

Confirmation Request	? ×
) Are you sure you want to cancel all file relocations and discard o	changes?
Don't ask this question again	
Yes <u>N</u> o	

If you select **Yes** this will invoke the Locate Files dialog box and close the dialog box. If you select **No** the dialog box will remain open allowing you to select OK. If the confirmation is selected not to be shown again, the default behavior will be **Yes**.

#### To open the confirmation dialog box again

- 1. Select [Setup -> Options]. The Options dialog box will be displayed.
- 2. Select the **Confirmation** tab.
- 3. Select the **Show dialog to search on OK for directory browse dialog** checkbox. This checkbox is selected by default.
- 4. Click OK.

If you select the OK button after selecting a directory, and not selecting search, a confirmation box will be shown, as below.



If you select **Yes** a search will be made, and the Locate Files dialog box will remain open, allowing you to stop the search if needed.

Selecting **No** will invoke the message box and close the Locate Files dialog box.

If the confirmation is ticked not to be shown again, the default behavior will be No.

#### To open the confirmation dialog box again

- 1. Select [Setup -> Options]. The Options dialog box will be displayed.
- 2. Select the **Confirmation** tab.
- 3. Select the **Show message when canceling for directory browse dialog** checkbox. This checkbox is selected by default.
- 4. Click **OK**.

#### (2) Showing a source tree

Files retrieved from the download module will be shown in the workspace window. The files will be shown under the download module.

Files will be grouped according to file type and shown under their relevant folder. If no file type for that file exists they will be grouped under their extension folder. When you double-click on a file, the external editor is launched with the file. If you wish to open the file in the editor window, see section 2.5, File extensions and file groups.

Files that could be located will be shown under the download module, shown in a normal icon ().



Files that could not be located will be shown under the download module but will have a red icon ().



To distinguish these files, you can view the full path of the file.

#### To view the full paths of all files

- 1. Right-click anywhere inside the **Projects** tab of the **Workspace** window.
- 2. Select the Configure View menu option. The Configure View dialog box will be displayed.
- 3. Select the **Show file paths** check box.
- 4. Click OK.



If you wish to relocate files in the download module to a different directory this is possible via pop-up menu.

#### (3) Relocating files to a different directory

If you wish to relocate the whole module i.e. want to relocate all files in the module to a different directory this is possible via the download module pop-up menu.

#### To relocate all files in the module to a different directory

1. Right-click on a download module in the **Projects** tab of the workspace window.



2. Select the **Relocate Module** menu option. Selecting this will bring up a standard Windows® directory open dialog box.

Please select new location of module files		? ×
Look jn: 🔁 Debug_Emulator	🔽 🗧 🗈 💣 🎟 -	
1		
	<u></u> pe	n
	Cano	

3. Select a directory, and then click the **Open** button.

Any files that could not be found in this directory will invoke the **Locate Files** dialog box, allowing you to search for the unlocated files.

You can also relocate files, via the pop-up menu option shown for files.

## To relocate a file to a different directory

- 1. Right-click on a file in the **Projects** tab of the workspace window.
- 2. Select the **Relocate file(s)** menu option. Selecting this will bring up a standard Windows® file open dialog box.



High-performance Embedded Workshop V.4.09

Please selec	t new location of file		? ×
Look jn: 🔁	Debug_Emulator	- + 1	💣 🎟 -
File <u>n</u> ame:	stdlib		<u>O</u> pen
Files of <u>type</u> :	(stdlib.h)	•	Cancel

- 3. This dialog will only allow you to select a file of the same name.
- 4. Select a file, and then click the **Open** button.

You can select multiple files to be relocated, this will bring up the **Locate Files** dialog box with the files to be relocated in the files to find section.

## 17.1.5.5 Displaying main functions on download

The High-performance Embedded Workshop automatically displays the source file including main functions in source mode within the High-performance Embedded Workshop editor after downloading modules.

#### To display main functions on download

- 1. Select [Setup -> Options] to open the Options dialog box.
- 2. Select Debug tab.
- 3. Select the **Display main function on download** checkbox. By default, this checkbox is not selected.
- 4. Click **OK**.
- 5. Download a module. For details on download, see section 17.1.5.2, Downloading modules.

## 17.1.5.6 Unloading of modules

It is possible to manually unload downloaded modules.

When a module is unloaded, its symbols are erased from the High-performance Embedded Workshop debugging system, but the memory contents of the target remains unmodified. After a module has been unloaded, it cannot be debugged unless it is reloaded.



#### To unload modules

Select one of the following operations:

- Select the module, which you wish to unload, from the [Debug -> Unload Modules] submenu, OR
- Right-click on the module in the **Download modules** folder on the Projects tab of the workspace window to display a pop-up menu. Select **Unload**. It is possible to select two or more modules.

#### To unload all modules

Select [Debug -> Unload Modules -> All Downloaded Modules].

Select [**Debug -> Debug Settings**]. The **Debug Settings** dialog box opens. This will unload the modules from the target in the order specified in **Download modules** list on the **Target** tab.

## 17.1.6 Debugger sessions

The High-performance Embedded Workshop allows you to store all of your builder options into a configuration. This means that you can "freeze" all of the options and give them a name. In a similar way, High-performance Embedded Workshop allows the user to store his debugger options in a session. Later on, you can select the session and all of the debugger options will be restored. These sessions allow the user to specify target, download modules and debug options. This means that potentially each session can be targeted at a different debugger.

This facility can allow you to have many different sessions, each with different debugger options defined. For example, it is possible to have each session using the same target but with slight variations in the session options. This can mean it is very easy for the user to switch session and modify such things as register values, or target settings such as clock speed. The figure below shows this principle. The five sessions share the same target but the sessions can be slightly different, with regard to the options defined. This means that both sessions can share the same download module and avoid an unnecessary rebuild. This is because sessions are not directly related to the build configuration data.

DefaultSession 💌
DefaultSession
SimSession_System_01
SimSession_System_02
SimSession_System_03
SimSessionSH-4

Each session's data is stored in a separate file to the High-performance Embedded Workshop project. You can then manipulate the data to share or modify as is required in the project.

#### 17.1.6.1 Selecting a session

#### To select a session

1. Select [Debug -> Debug Sessions]. The Debug Sessions dialog box opens.

- 2. Select the session that you want to use from the **Current session** drop-down list box.
- 3. Click the **OK** button.

DefaultSession	•
DefaultSession	
SimSessionSH-4	

You can also select a different debug session by selecting it from the Current Session drop-down list on the **Standard** toolbar.

## 17.1.6.2 Adding a session

You can now create a session with a target attached and setup ready for use. This session can be given a name and the target chosen.

#### To create a new session, with a target attached and setup

1. Select [File -> New Session]. The New Session dialog box opens.

New Session	? ×
Session name:	ОК
I	Cancel
Create a <u>n</u> ew session on:	
SH-1 Simulator	
Using internal core:	
Single Core Target 💌	
Session generator to use:	
Renesas SH Project Generator	

- 2. Enter the name of a new session into the **Session name** field. This can be up to 32 characters in length and contain letters, numbers, and the underscore character. Especially, do not use a minus sign, or a space.
- 3. Select the target you wish to use in the **Create a new session on** list.
- 4. Select the core in the **Using internal core** list. This item is only available when a synchronized debugging facility has been used.
- 5. Select the generator in the **Session generator to use** list. This should default to the correct selection. However sometimes there may be multiple generators that support the same target.
- 6. Click the **OK** button. This should launch the generation process, the process depends on the generator that was selected. At this point an additional dialog box may be displayed for target setup options.
- 7. When finished, a new session is added to the current project. It should be available in the sessions drop-down list box on the main toolbar.



You can create a new empty session in the project directory. The session will use the session name as its new file name. If the file name already exists then an error is displayed.

## To add a new empty session

- 1. Select [Debug -> Debug Sessions]. The Debug Sessions dialog box opens.
- 2. Click the Add button. The Add Session dialog box opens.
- 3. Click the Add New Session radio button.
- 4. Enter a name for the session.
- 5. Click the **OK** button.

⊙ <u>A</u> dd a new session				
NewSession				

You can import session data from another file and create a new session file in the project directory. All information is an exact copy of the file the data was imported from.

#### To import an existing session into a new session file

- 1. Select [Debug -> Debug Sessions]. The Debug Sessions dialog box opens.
- 2. Click the Add button. The Add Session dialog box opens.
- 3. Click the **Use an existing session file** radio button.
- 4. Enter a name for the session.
- 5. Browse to an existing session file location, which you would like to import into the current project.
- 6. Click the **OK** button.

Use an existing session file <u>N</u> ame: Tutorial			
Session file path:	Browse		
Den and maintain link to session file			
Make session file link read only			

This operation can also be achieved by using the [File -> Import session].

#### To import an existing session using [File -> Import Session]

1. Select [File -> Import Session]. The Session Name dialog box opens.



Session Name	? ×
Session name:	OK
J	Cancel

- 2. Enter the new session name.
- 3. Select the session file you wish to import into the new session.
- 4. Click **OK**. A new session is added with the same settings as the file you browsed to but with the new name.

## 17.1.6.3 Importing a link to a session

You can add a new session to the High-performance Embedded Workshop system but link to the session file in its location rather than importing or copying the file to the project directory. This is useful when sharing debugger information with other users in a network environment.

#### To import a link to an existing session file

- 1. Select [**Debug -> Debug Sessions**]. The **Debug Sessions** dialog box opens.
- 2. Click the Add button. The Add new session dialog box opens.
- 3. Click the **Use an existing session file** radio button.
- 4. Enter a name for the session.
- 5. Browse to an existing session file location, which you would like to import into the current project.
- 6. Click the **Open and maintain link to session file** checkbox. This means the session will not be imported into the project directory but instead the High-performance Embedded Workshop will link to the session location. This file location was entered in step 5 and it will save all of the session data in this location.
- 7. Click the OK button to close the Debug Sessions dialog box.

It is possible to make the link to session file read-only. This is useful if you are sharing debugger-setting files and you do not want data to be modified accidentally.

#### To import a link to an existing session file and make it read-only

- 1. Select [Debug -> Debug Sessions]. The Debug Sessions dialog box opens.
- 2. Click the Add button. The Add new session dialog box opens.
- 3. Click the Use an existing session file radio button.
- 4. Enter a name for the session.



- 5. Browse to an existing session file location, which you would like to import into the current project.
- 6. Click the **Open and maintain link to session file** checkbox.
- 7. Click the **Make session file link read only** checkbox. This means that the High-performance Embedded Workshop will be unable to save changes to this session and will only be able to read the data when the session is opened.

Use an existing session file			
<u>N</u> ame: Tutorial			
Session file path:	<u>B</u> rowse		
☑ Open and maintain link to session file			
☑ <u>M</u> ake session file link read only			

## 17.1.6.4 Removing a session

#### To remove a session

- 1. Select [**Debug -> Debug Sessions**]. The **Debug Sessions** dialog box opens.
- 2. Select the session you would like to remove.
- 3. Click the **Remove** button.
- 4. Click the OK button to close the Debug Sessions dialog box.

It is not possible to remove the current session.

#### 17.1.6.5 Making a session read-only

#### To make a session read only

E Read only

- 1. Select [**Debug -> Debug Sessions**]. The **Debug Sessions** dialog box opens.
- 2. Select the session you would like to view the properties for.
- 3. Click the **Properties** button. The properties dialog is displayed.
- 4. Click the **Read only** checkbox. This makes the link read only. This is useful if you are sharing debugger-setting files and you do not want data to be modified accidentally.
- 5. Click the OK button to close the Debug Sessions dialog box.



## 17.1.6.6 Saving session information

To save a session

Select [File -> Save Session].

If you have the **Prompt before saving session** checkbox checked, a dialog box is displayed which asks you whether you wish to save the information. Clicking No loses the changes you made in the session. This checkbox is located in the **[Setup -> Options]** dialog box on the **Workspace** tab.

## To save a session with a different name

- 1. Select [Debug -> Debug Sessions]. The Debug Sessions dialog box opens.
- 2. Select the session you would like to save.
- 3. Click the **Save as** button. The **Save Session** dialog box opens.
- 4. Browse to the new file location.
- 5. If you only want to export the session file to another location then leave the **Maintain link** checkbox unchecked. If you would like High-performance Embedded Workshop to use this location instead of the current session location then check the **Maintain link** checkbox.
- 6. Click the OK button.

#### To save a session with a different name using [File -> Save Sessions As]

1. Select [File -> Save Sessions As]. The Session Name dialog box opens.



- 2. Enter the new session name.
- 3. Click the OK button.

## 17.1.6.7 Reloading session information

## To reload a session

Select [File -> Refresh Session].



Clicking this will lose any changes to your session currently and the reload the current session into High-performance Embedded Workshop.

# 17.2 Viewing a program

This section describes how to look at your program as source code and assembly language mnemonics.

The source code will be shown in the source mode in two types of windows. The main features are:

## **Editor Window**

- A source file opened in the editor window can be edited.
- When debugging and stepping through source code, each source file (which contains the current PC address) will be opened in a separate window as you step through code from one file to another. When stepping into an area of disassembly that has no associated source-code line, the source-mode will switch to disassembly mode.

#### **Disassembly Window**

- Any source file opened in the Disassembly window is read-only and cannot be edited. The Disassembly window popup menu will contain a menu-item to open the current file for editing.
- When debugging and stepping through source code, the Disassembly window will automatically switch to the relevant source file (which contains the current PC address) making this simple for you when debugging. Whilst the Disassembly window is open there will be no standard editor windows opened, the source files will be opened within the Disassembly window view as necessary. When stepping into an area of disassembly that has no associated source-code line, the source-mode will switch to disassembly mode.
- You can use disassembly source-mode as primary debug view. If this facility is available, the following actions will be performed:
  - If the Disassembly window is open and in Disassembly mode, then when a break occurs at a known source address, the Disassembly window will switch to source-mode.
  - o If the Disassembly window is open, then when a break occurs, no other editor windows are opened.
  - When a break occurs, the window order will not be changed automatically.
  - The position of the source file in the editor window is fixed.

If you wish to continue step in Disassembly mode of the Disassembly window, select the [**Debug -> Step mode -> Assembly**].

#### Note:

After a break occurs, the High-performance Embedded Workshop displays the location of the program counter (PC). In some cases, for example if a project has been moved from its original path, the source files may not be automatically found. In this case the High-performance Embedded Workshop will open a source file browser dialog to allow you to manually locate the file - this path will then be used to update any other source files in this debug project.



## 17.2.1 Opening the Editor window

To view a source file's code, double-click on its icon in the file tree, or right-click on the source file and click the **Open** option on the pop-up menu. The High-performance Embedded Workshop opens the file in the editor.

## Toolbar

View Source View mixed mode View Disassembly					
I	Line	Source Ad	C.,	S.,	Source
	24 25 26	00002000			void main(void) { long min, max;

The editor in version 4.00 onwards has been enhanced to include an integrated disassembly view. This integrated view has a toolbar which allows the switching of mode. When each mode is available it is possible to click the button and change to the new view.

Three different modes are possible these are listed below:

Mode	Function		
Source mode	This mode is the standard High-performance Embedded Workshop editor. It allows source file editing		
	and keywords are highlighted correctly if you are viewing source files.		
	This view allows line numbers, addresses, breakpoints, bookmarks, and source codes to be viewed.		
Mixed mode	The mixed mode facility in a source file is different to the disassembly window mixed mode. Instead		
	of showing the continuous disassembly it shows the disassembly that is related to each line of source		
code. This view cannot be edited and is only available when the module is downloaded.			
showing both source and disassembly) to be viewed.			
Disassembly mode	The disassembly mode shows the true continuous disassembly code in address order. This is the		
	same as clicking [View -> Disassembly].		
	This view is only available when a target is attached to the session.		
	This view allows breakpoints, address, object codes, labels, and disassembled codes to be viewed.		

### Notes:

It is not possible to switch from the source view to the mixed display under the following conditions.

- 1. The target is not connected to the current session.
- 2. No download modules have been downloaded.
- 3. No debug information is available for the current project.
- 4. The currently displayed file has been edited and the changes not saved.



If an address included in the address range of a source file is then shifted to be in the range of another source file in **View Disassembly mode**, selecting **View Source mode** or **View mixed mode** displays the Confirmation Request dialog box shown below.

To open the new source file for the address, click Yes.

To view the source code in the previous mode, click No.

If you do not wish to select View Source mode or View mixed mode, click Cancel.

Confirmation Request	? X				
Your current disassembly address does not correspond to the original file. Do you wish to open the file which corresponds to the current disassembly address?					
Don't ask this question again					
Yes <u>N</u>	Lo Cancel				

If you do not wish to open this confirmation dialog box, select the **Don't ask this question** again checkbox.

#### To open the confirmation dialog box again

- 1. Select [Setup -> Options]. The Options dialog box opens.
- 2. Select the **Confirmation** tab.
- 3. Select the **Switch from disassembly to new source** checkbox. This checkbox is selected by default.
- 4. Click OK.



## 17.2.1.1 Viewing the code

To view a source file's code, click the **View Source** button.

## Window configuration

Minutering descents	🚸 D e	emoSH4.c		
View mixed mode View Source View Disassembly		51 SZ		
Column —	Line	Source Address	S/W Breakpoints	Source
header	22	00002000	\$	void main(void) 🗧
	23			{
	24			long min, max;
	25			long j;
	26			int i;
	27	00002006		printf("### Data
	28			
	29	0000201A	٠	for( i=0; i<10; 💌
	•			► /h
Line N		column S7 urce Address colur	 W Breakpoints colu	imn Source

- Clicking another toolbar button switches the display mode. To view disassembly codes in mixed mode, click the "View mixed mode" button. To view disassembly codes in disassembly mode, click the "View Disassembly" button.
- The columns listed in the table below are on the left of the "Source" field.

Column Name	Description
Line	Displays the line number for the source file.
Source Address	After your program has been downloaded, the editor window displays the addresses for the current source file.
S/W Breakpoints	Display the PC location ( $\diamondsuit$ ), breakpoints ( $\blacklozenge$ ) and bookmark ( $\Box$ ). Setting PC breakpoint by double-click.
The "Source" field includ	as and as highlighting the syntax

• The "Source" field includes codes highlighting the syntax.

## Options

Right-clicking within the "Source" field opens a pop-up menu containing available options.

Pop-up Menu Option	Macro Recording	Function	
Build File " <file name="">"</file>	•	Builds the selected files.	
Open ' <file name="">'</file>	-	Opens a file shown in the editor window.	
Cut	-	Removes highlighted text and places it on the Windows® clipboard.	
Сору	-	Places a copy of the highlighted text into the Windows® clipboard.	
Paste	-	Copies the contents of the Windows® clipboard into the active window at the position of the insertion cursor.	
Add File To Project	-	Adds the file to a project.	



Pop-up Menu Option		Macro Recording	Function		
Go To Definition Of ' <navigation< td=""><td>-</td><td>Shows the positions where navigation items of #defines, C</td></navigation<>		-	Shows the positions where navigation items of #defines, C		
Item>'			functions, or C++ classes are defined.		
List Membe	rs	-	While the smart editor function is enabled, selecting this option		
			displays an alphabetical list of C #defines, C functions, or C++		
			classes as a pop-up window. Double-clicking on an item copies this		
			character string and pastes it to the current cursor position.		
Find		-	Finds text in the current file.		
Replace		-	Replaces text in the current file.		
Goto Line		-	Jumps to a line in a file.		
Match Brac	es	-	Finds a matching brace.		
Bookmarks	Toggle Bookmark	-	Sets a bookmark at the current line or clears a bookmark at the		
			current line.		
	Next Bookmark	-	Jumps to the next bookmark in the current file from the current line.		
	Previous Bookmark	-	Jumps to the previous bookmark in the current file from the current $% \left( {{{\bf{n}}_{{\rm{n}}}}} \right)$		
			line.		
	Clear All Bookmarks	; -	Clears all bookmarks in the current file.		
Templates	Define Templates	-	Defines a template.		
	Insert Template	-	Inserts a template.		
Toggle Brea	akpoint	•	Sets or clears a software breakpoint at the line showing the address		
Enable/Disa	able Breakpoint	•	Enables or disables the current software breakpoint.		
Define Colu	ımn Format	-	Sets the status of editor and disassembly columns.		
Columns	Column name	-	Sets the status of editor and disassembly columns.		
Turn Heade	er On/Off	-	Shows or hides the column header.		
Instant Wat	ch *	-	Launches the Instant Watch dialog box with the name extracted		
			from the view at the current text cursor (not mouse cursor) position.		
Add Watch	*	-	Adds the selected variable to the Watch window.		
Go To Curs	or	•	Starts executing the user program at the current PC and continues		
			until the PC equals the address indicated by the current text cursor		
			(not mouse cursor) position.		
Set PC Here		•	Changes the value of the Program Counter (PC) to the address at		
			the row of the text cursor (not mouse cursor).		
Display PC		-	Opens the source file or disassembly at the address of the PC.		
View Disass	sembly	-	Opens a Disassembly window at the address mating the current		
			source line.		
Properties		-	Displays file properties.		

Note:

\*. Support for this function depends on the debugger.

## 17.2.1.2 Viewing assembly-language code

The **Disassembly** mode shows at the same address of the cursor line as the **Source** mode. You can view assemblylanguage codes only when the debugger is connected



## Window configuration

View mixed mode —	🚸 DemoSH4.c - Disa	ssembly			
View Source- View Disassembly-					
Column —	S/W Breakpoints - ASM	Disassembly Address	Obj code	Label	Disassembly
header	\$	00002000	2FE6	_main	MOV.L
		00002002	4F22		STS.L 📃
	•	00002004	7FF0		add 🗾
	•				► //.
-	Disassen	hbly Address column	l	.abel colu	umn
S/W B	reakpoints - ASM column	Ob	oj code colur	mn	Disassembly

- Clicking another toolbar button switches the display mode. To view disassembly codes in mixed mode, click the "View mixed mode" button. To view source codes in source mode, click the "View Source" button.
- The columns listed in the table below are on the left of the "Disassembly" field.

Column Name	Description
S/W Breakpoints - ASM	Display the PC location ( $\Rightarrow$ ), breakpoints ( $\spadesuit$ ). Setting PC breakpoint by double-click.
Disassembly Address	Display the disassembly address. Open the <b>Set Address</b> dialog box by double-click. Enter the address to jump to.
Obj code	Display the object codes.
Label	Display the Labels. This column is not available if no module has been downloaded.
Devela aliabia a the "Discourse	which field (in discoundly mode) are us the Assembler dislocker. Enter

• Double-clicking the "Disassembly" field (in disassembly mode) opens the **Assembler** dialog box. Enter assembly-language codes.

## Options

Right-clicking within the "Disassembly" field opens a pop-up menu containing available options.

Pop-up Menu Option	Macro Recording	Function	
Refresh	-	Acquires the latest disassembly information to update the contents of the window.	
Lock Refresh	-	It is possible to lock a memory range in the disassembly (in disassembly mode) so that it does not refresh.	
View Source	-	Launches editor at location in source.	
Set Address	-	Enters a new start address.	
Go To Cursor	•	Commences to execute the user program starting from the current PC address. The program will continue to run until the PC reaches the address indicated by the text cursor (not the mouse cursor) or another break condition is satisfied.	
Display PC	-	Displays the disassembled codes at the PC location.	
Set PC Here	•	Changes the value of the PC to the address indicated by the text cursor (not the mouse cursor).	
Edit	-	Modifies the instruction at that address.	



Pop-up Menu Option	Macro Recording	Function
Find in Range	-	Searches the range for the specified text string.
Сору	-	Places a copy of the highlighted text into the Windows® clipboard.
Define Column Format	-	Sets the status of editor and disassembly columns.
Turn Header On/Off	-	Shows or hides the column header.
Save Disassembly Text	-	Saves the specific range.
Print	-	Prints the specific range.
Toggle Breakpoint	•	Sets or clears a software breakpoint at the line showing the address.
Enable/Disable Breakpoint	•	Enables or disables the current software breakpoint.

## 17.2.1.3 Refreshing the disassembly view

Even if some external operation changes the memory contents shown in the **Disassembly** view, the High-performance Embedded Workshop cannot detect the change. For example, if you use the external flash utility to program the range of memory being displayed, the **Disassembly** view will not be reflected.

Right-click on the Disassembly filed (in disassembly mode) and select Refresh.

This acquires the latest information of the data being displayed and updates the contents of the view.

## 17.2.1.4 Disassembly lock refresh

It is possible to lock a memory range in the disassembly view so that it does not refresh. This function is called the disassembly "lock refresh".

Right-click on the Disassembly field (in disassembly mode) and select **Lock Refresh**. The **Lock Refresh** dialog box will be displayed.

Lock Refresh		? ×
✓ Locked		ОК
<u>S</u> tart Address:	00002000 💌 ፳	Cancel
End Address:	0000201A 💌 🗾	

Check the Locked check box. The controls should now enable.

Select the start and end address that should be locked and cached so that no updates are displayed. Click **OK**. The view refreshes to only show the locked area.

#### 17.2.1.5 Looking at a specific address

When you are looking at your program in a Disassembly view, you may want to look at another area of your program's code. Rather than scrolling through a lot of code in the program, you can go directly to a specific address.



Right-click on the Disassembly field (in disassembly mode) and select the **Set Address** option. The **Set Address** dialog box will be displayed.



Enter the address or label name in the edit box and either click the **OK** button or press the Enter key. The Disassembly view updates to show the code at the new address. When an overloaded function or a class name is entered, the **Select Function** dialog box opens for you to select a function (Support for this function depends on the debugger.).

## 17.2.1.6 Modifying assembly-language code

You can modify the assembly-language code in the disassembly view (in disassembly mode) by double-clicking on the instruction that you wish to change. The **Assembler** dialog box will be displayed.

Assembler	? ×
Address Code 00002408 D15F	(COK
<u>M</u> nemonic:	_ Cancel
MOV.L @(H'017C:8,PC),R1	

The address, machine code and disassembled instructions are shown. Type the new instruction or edit the old instruction in the **Mnemonic** field. Pressing Enter will assemble the instruction into memory and move on to the next instruction. Clicking the **OK** button will assemble the instruction into memory and close the dialog box. Clicking the **Cancel** button or pressing ESC will close the dialog box.

#### Note:

The assembly-language display is disassembled from the actual machine code in the debugger's memory. If the memory contents are changed the dialog box (and Disassembly view) will show the new assembly-language code, but the source view will be unchanged. This is true even if the source file contains assembler.

#### 17.2.1.7 Disassembly find in range

The disassembly find in range can be used to find a certain text string in the disassembly view between two addresses.

Right-click on the Disassembly field (in disassembly mode) and select **Find in Range**. The **Find in Range** dialog box opens.



Find in Range		? ×
<u>F</u> ind:	<b>•</b>	OK
<u>S</u> tart Address:	<b>_</b>	Cancel
End Address:	<b>I</b>	

Enter your search string, the start and end address that should be searched. Click **OK**. The view then selects the first instance of that string in the range.

## Note:

Subsequent find operations will find strings only in the paged disassembly area not the complete range.

## 17.2.1.8 Saving disassembly text

The contents of the disassembly view can be saved by using the **Save Disassembly Text** menu from the **Disassembly** field (in disassembly mode) pop-up menu.

When **Save Disassembly Text** is selected the **Save Disassembly Text** dialog box is displayed that asks you the range of addresses to save.

Save Disasse	mbly Text		? ×
<u>S</u> tart Address:	00002000	<b>-</b>	ОК
End Address:	0000201A	- 🦻	Cancel
<u>F</u> ilename:			Browse

Both a start and end address should be supplied. You also need to specify the full filename to save the information to. If needed you can browse to the file to use. Click **OK**.

## 17.2.1.9 Printing the disassembly view

The disassembly view can be printed by using [File -> Print] or the Print toolbar button ( ) when it is in focus or by using the menu Print on the Disassembly field (in disassembly mode) pop-up menu.

When **Print** is selected the **Disassembly Print Range** dialog box is displayed that asks you the range of addresses to print.



Disassembly Print Range		? ×
Start Address: 00002000	▼ 🔊	OK
End Address: 0000201A	- 2	Cancel

Both a start and end address should be supplied.

Clicking **OK** on this dialog box then passes the print selection to the standard print formatting and selection dialog box. From here you can choose your printer and page setup options.

## 17.2.2 Opening the Disassembly window

If you wish to view code at assembly-language level, then select one of the following operations:

- Choose [View -> Disassembly], OR
- Press CTRL+D, **OR**
- Click on the View Disassembly toolbar button (

The **Disassembly** window opens at the current program counter (PC) location.

## Toolbar

\	View Source View mixed mode View Disassembly						
ĺ							
I	C	S.,	Disassemb	ОЫ	Label	Disassembly	
1		\$	00002000	2FE6	_main	MOV.L	R14,0
1			00002002	4F22		STS.L	PR,0-
1			00002004	7FF0		ADD	#H'FC

The disassembly window in version 4.02 onwards has been enhanced to include a source view.

This integrated view has a toolbar which allows the switching of mode. When each mode is available it is possible to click the button and change to the new view.



Mode	Function		
Source mode	The source file relating to the current PC address will be opened if available.		
	It allows keywords are highlighted correctly if you are viewing source files.		
	This view is read-only and cannot be edited.		
	This view is only available when the module is downloaded.		
	This view allows line numbers, address, breakpoints, and source codes to be viewed.		
Mixed mode	The mixed mode facility in the disassembly window is different to a source file mixed mode. Instead		
	of showing the continuous disassembly it shows the disassembly that is related to each line of		
	source code.		
	This view can be edited the assembly-language code and is only available when the module is		
	downloaded.		
	This view allows breakpoints, address, object codes, labels, and disassembled codes to be viewed.		
Disassembly mode	The disassembly mode shows the true continuous disassembly code in address order.		
	This view can be the assembly-language code and is only available when a target is attached to the		
	session.		
	This view allows breakpoints, address, object codes, labels, and disassembled codes to be viewed.		

Three different modes are possible these are listed below:

## 17.2.2.1 Viewing the code

When switching to the source mode from within the disassembly window, the source file relating to the current PC address will be opened if available. Otherwise either the most recently open disassembly window source file will be reopened where possible or a prompt will be given to select a source file to be opened.

Any text-based files may be opened in the disassembly window and not just project source files. Any source file opened in the disassembly window is read-only and cannot be edited; it is only intended for the viewing and debugging of source files.

This view is only available when the module is downloaded.

## Window configuration





## High-performance Embedded Workshop V.4.09

- Clicking another toolbar button switches the display mode. To view disassembly codes in mixed mode, click the "View mixed mode" button. To view disassembly codes in disassembly mode, click the "View Disassembly" button.
- The columns listed in the table below are on the left of the "Source" field.

Column Name	Description
Line	Displays the line number for the source file.
Source Address	After your program has been downloaded, the source view displays the addresses for the current source file.
S/W Breakpoints	Display the PC location ( $\clubsuit$ ), and breakpoints ( $\textcircled{\bullet}$ ). Setting PC breakpoint by double-click.

• The "Source" field includes codes highlighting the syntax.

## Options

Right-clicking within the "Source" field opens a pop-up menu containing available options.

Pop-up Menu Option	Macro Recording	Function	
Open Source File For Editing	-	Opens an editor window containing the file for editing.	
Open File in Source Mode	-	Opens a file shown in the disassembly window.	
Сору	-	Places a copy of the highlighted text into the Windows® clipboard.	
Go To Definition Of	-	Shows the positions where navigation items of #defines, C functions,	
<pre>'<navigation item="">'</navigation></pre>		or C++ classes are defined.	
Find	-	Finds text in the current file.	
Goto Line	-	Jumps to a line in a file.	
Toggle Breakpoint	•	Sets or clears a software breakpoint at the line showing the address.	
Enable/Disable Breakpoint	•	Enables or disables the current software breakpoint.	
Turn Header On/Off		Shows or hides the column header.	
Define Column Format	-	Sets the status of editor and disassembly columns.	
Columns Column name	-	Sets the status of editor and disassembly columns.	
Instant Watch *	-	Launches the Instant Watch dialog box with the name extracted from	
		the view at the current text cursor (not mouse cursor) position.	
Add Watch *	-	Adds the selected variable to the Watch window.	
Set Address	-	Enters a new start address.	
Go To Cursor	•	Starts executing the user program at the current PC and continues until	
		the PC equals the address indicated by the current text cursor (not	
		mouse cursor) position.	
Set PC Here	•	Changes the value of the PC to the address at the row of the text	
		cursor (not mouse cursor).	
Display PC	-	Opens the source file or disassembly at the address of the PC.	
Properties	-	Displays file properties.	
Note:			

Note:

Support for this function depends on the debugger.



## 17.2.2.2 Opening a source file for editing

The Disassembly source-mode allows only to view and debug source files, and no editing is permitted from within this view.

To edit a source file that is currently open in the Disassembly source-mode view, the pop-up menu option **Open Source File for Editing** must be selected. This will open an editor window containing the file for editing.

Alternatively you may open the file for editing manually via the Workspace Window or by the main menu [File -> **Open**].

## 17.2.2.3 Opening a source file in the Disassembly window

Opening a source file is possible by selecting **Open File in Source Mode** from the pop-up menu in Disassembly window. This menu option is available on the pop-up menu in all view-modes within the Disassembly window.

Upon selecting this menu option a file-browse dialog will be displayed prompting for the source file to be opened. Any text-based file may be selected and not just source files.

## 17.2.2.4 Using disassembly source-mode as primary debug view

It is possible to use disassembly source-mode as primary debug view.

#### To use disassembly source-mode as primary debug view

- 1. Select [Setup -> Options]. The Options dialog box opens.
- 2. Select the **Debug** tab.
- 3. Select the **Use disassembly source-mode as primary debug view** checkbox. By default, this checkbox is not selected.
- 4. Click OK.

If this option is set, the following actions will be performed:

- If the Disassembly window is open and in Disassembly mode, then when a break occurs at a known source address, the Disassembly window will switch to source-mode.
- If the Disassembly window is open, then when a break occurs, no other editor windows are opened.
- When a break occurs, the window order will not be changed automatically.
- The position of the source file in the editor window is fixed.

If you wish to continue step in Disassembly mode of the Disassembly window, select [**Debug -> Step mode -> Assembly**].



## 17.2.2.5 Viewing assembly-language code

The **Disassembly** mode shows at the current PC location. You can view assembly-language codes only when the debugger is connected.

## Window configuration

View mixed mode—	🚸 Disassembly				<u>- 🗆 ×</u>
View Source- View Disassembly-					
Column —	S/W Breakpoints - ASM	Disassembly Address	Obj code	Label	Disassembly
header	⇒ ●	00002000 00002002 00002004	2FE6 4F22 7FF0	_main	MOV.L STS.L ADD
SAVB	Disassen Disassen	hbly Address column	bi code colur	Label colu	umn Disassembly

- Clicking another toolbar button switches the display mode. To view disassembly codes in mixed mode, click the "View mixed mode" button. To view source codes in source mode, click the "View Source" button.
- The columns listed in the table below are on the left of the "Disassembly" field.

Column Name	Description
S/W Breakpoints - ASM	Display the PC location ( $\Rightarrow$ ), breakpoints ( $\spadesuit$ ). Setting PC breakpoint by double-click.
Disassembly Address	Display the disassembly address. Open the <b>Set Address</b> dialog box by double-click. Enter the address to jump to.
Obj code	Display the object codes.
Label	Display the Labels. This column is not available if no module has been downloaded.

• Double-clicking the "Disassembly" field opens the Assembler dialog box. Enter assembly-language codes.

## Options

Right-clicking within the "Disassembly" field opens a pop-up menu containing available options.

- -

Pop-up Menu Option	Macro Recording	Function	
Open File in Source Mode	-	Opens a file shown in the disassembly window.	
Refresh	-	Acquires the latest disassembly information to update the contents of window.	
Lock Refresh	-	It is possible to lock a memory range in the disassembly so that it does not refresh.	
View Source	-	Launches editor at location in source.	
Set Address	-	Enters a new start address.	



Pop-up Menu Option	Macro Recording	Function	
Go To Cursor	•	Commences to execute the user program starting from the current PC address. The program will continue to run until the PC reaches the address indicated by the text cursor (not the mouse cursor) or another break condition is satisfied.	
Display PC	-	Displays the disassembled codes at the PC location.	
Set PC Here	•	Changes the value of the PC to the address indicated by the text cura (not the mouse cursor).	
Edit	-	Modifies the instruction at that address.	
Find in Range	-	Searches the range for the specified text string.	
Сору	-	Places a copy of the highlighted text into the Windows® clipboard.	
Define Column Format	-	Sets the status of editor and disassembly columns.	
Turn Header On/Off	-	Shows or hides the column header.	
Save Disassembly Text	-	Saves the specific range.	
Print	-	Prints the specific range.	
Toggle Breakpoint	•	Sets or clears a software breakpoint at the line showing the address.	
Enable/Disable Breakpoint	•	Enables or disables the current software breakpoint.	

## 17.2.2.6 Refreshing the disassembly view

Even if some external operation changes the memory contents shown in the **Disassembly** view, the High-performance Embedded Workshop cannot detect the change. For example, if you use the external flash utility to program the range of memory being displayed, the **Disassembly** view will not be reflected.

Right-click on the Disassembly filed (in disassembly mode or mixed mode) and select Refresh.

This acquires the latest information of the data being displayed and updates the contents of the view.

## 17.2.2.7 Disassembly lock refresh

It is possible to lock a memory range in the disassembly view so that it does not refresh. This function is called the disassembly "lock refresh".

Right-click on the Disassembly field (in disassembly mode or mixed mode) and select **Lock Refresh**. The **Lock Refresh** dialog box will be displayed.

Lock Refresh			? ×
Cocked			OK
<u>S</u> tart Address:	00002000 💌 🧶	<u>F</u>	Cancel
End Address:	0000201A 💌	<u>7</u>	

Check the **Locked** check box. The controls should now enable.

Select the start and end address that should be locked and cached so that no updates are displayed. Click **OK**. The view refreshes to only show the locked area.



## 17.2.2.8 Looking at a specific address

When you are looking at your program in a Disassembly window, you may want to look at another area of your program's code. Rather than scrolling through a lot of code in the program, you can go directly to a specific address.

Right-click on the Source field or the Disassembly field and select the **Set Address** option. The **Set Address** dialog box will be displayed.

Set Address		? ×
Address:		OK
0000201A	<u> </u>	Cancel

Enter the address or label name in the edit box and either click the **OK** button or press the Enter key. The Disassembly view updates to show the code at the new address. When an overloaded function or a class name is entered, the **Select Function** dialog box opens for you to select a function (Support for this function depends on the debugger.).

When the Disassembly window is in source-mode, and an address is entered which is source-related but not in the currently open source file, the related source file will automatically be opened in the source-mode. If an address is specified which is not source-related, Disassembly view will switch mode to disassembly mode in order to view the specified address.

#### 17.2.2.9 Modifying assembly-language code

You can modify the assembly-language code in the disassembly view (in disassembly mode or mixed mode) by doubleclicking on the instruction that you wish to change. The **Assembler** dialog box will be displayed.

Assembler	? ×
Address Code 00002408 D15F	(OK)
<u>M</u> nemonic:	Cancel
MOV.L @(H'017C:8,PC),R1	

The address, machine code and disassembled instructions are shown. Type the new instruction or edit the old instruction in the **Mnemonic** field. Pressing Enter will assemble the instruction into memory and move on to the next instruction. Clicking the **OK** button will assemble the instruction into memory and close the dialog box. Clicking the **Cancel** button or pressing ESC will close the dialog box.

#### Note:

The assembly-language display is disassembled from the actual machine code in the debugger's memory. If the memory contents are changed the dialog box (and Disassembly view) will show the new assembly-language code, but the source view will be unchanged. This is true even if the source file contains assembler.



## 17.2.2.10 Disassembly find in range

The disassembly find in range can be used to find a certain text string in the disassembly view between two addresses.

Right-click on the Disassembly field (in disassembly mode or mixed mode) and select **Find in Range**. The **Find in Range** dialog box opens.

Find in Range		? ×
<u>F</u> ind:	×	OK
<u>S</u> tart Address:	<b>_</b>	Cancel
End Address:	<b>I</b>	

Enter your search string, the start and end address that should be searched. Click **OK**. The view then selects the first instance of that string in the range.

#### Note:

Subsequent find operations will find strings only in the paged disassembly area not the complete range.

#### 17.2.2.11 Saving disassembly text

The contents of the disassembly view can be saved by using the **Save Disassembly Text** menu from the **Disassembly** field (in disassembly mode or mixed mode) pop-up menu.

When **Save Disassembly Text** is selected the **Save Disassembly Text** dialog box is displayed that asks you the range of addresses to save.

Save Disasse	mbly Text			? ×
<u>S</u> tart Address:	00002000	•	<u></u>	OK
End Address:	0000201A	•	<u></u>	Cancel
<u>F</u> ilename:			<u>B</u> row:	se

Both a start and end address should be supplied. You also need to specify the full filename to save the information to. If needed you can browse to the file to use. Click **OK**.

#### 17.2.2.12 Printing the disassembly view

The disassembly view can be printed by using [File -> Print] or the Print toolbar button ( ) when it is in focus or by using the menu Print on the Disassembly field (in disassembly mode or mixed mode) pop-up menu.

When **Print** is selected the **Disassembly Print Range** dialog box is displayed that asks you the range of addresses to print.

Disassembly Print Range		? ×
Start Address: 00002000	▼ 🔊	ОК
End Address: 0000201A	• 🖉	Cancel

Both a start and end address should be supplied.

Clicking **OK** on this dialog box then passes the print selection to the standard print formatting and selection dialog box. From here you can choose your printer and page setup options.

## 17.2.3 Looking at the current PC position

To automatically display the program counter (PC) click the **Display PC** toolbar button ( $\succeq_{PC}$ ), or select [**Debug** -> **Display PC**]. This will open the editor or disassembly at the current PC.

## 17.2.4 Highlighting the line at the PC

In the Editor and Disassembly windows, an icon in the **S/W Breakpoints** column indicates the PC location. The source or assembly-language code at the PC can be highlighted.

## To highlight the line at the PC location

- 1. Select [Setup -> Options]. The Options dialog box opens.
- 2. Select the **Debug** tab.
- 3. Select the Enable line highlight for PC position checkbox. By default, this checkbox is selected.
- 4. Click OK.

The color of the line can be customized in the **Format Views** dialog box. The text (foreground) is in black and the background color is yellow by default.

#### To change the colors of the line at the PC

- 1. Select [Setup -> Format Views]. The Format Views dialog box opens.
- 2. Select an item, for which you wish to change the color, from the left-hand tree in the dialog box and expand it.
  - If you are opening the editor window or disassembly window in source mode, select **Source** and expand it.
  - If you are opening the editor window or disassembly window in mixed mode or disassembly mode, select **Disassembly** and expand it.



- 3. Select the **PC Line Highlight** category.
- 4. Change the selection in the **Foreground** and **Background** lists of the **Color** tab.
- 5. Click OK.

# 17.3 Operating memory

This section describes how to look at memory areas in the CPU's address space. How to look at a memory area in different formats, how to fill and move a memory block, and how to load and verify a memory area with a disk file are described.

## 17.3.1 Opening the Memory window

The Memory window displays the contents of contiguous memory areas.

To open the **Memory** window, choose [**View -> CPU -> Memory**], or click the **Memory** toolbar button (**F**).

You can specify the display start address and the scroll range at opening. The **Display Address** dialog box opens. Enter the **Display Address**, **Scroll Start Address** and **Scroll End Address**.



Click the OK button or press the Enter key, and the dialog box closes and the **Memory** window opens. The display can be scrolled within the range of the entered display scroll start and end addresses.

#### Window configuration

	Memory [0	00000]																	
olbar —		m ===	<u>16</u> <u>10</u> ± <u>10</u>	8	2	dbc	為	あ	٥.	Ξ.	æ	£.	.d	.16	.32	2			
olumn	Address	Label	Register +0	+1	+2	+3	+4	+5	+6	+7	+8	+9	+A	+B	+C	+D	+E	+ F	ASCII
	000000		[FB][3B][1 00	00	00	00	00	00	00	00	00	00	00	00	00	00	00	00	
	000010		00	00	00	00	00	00	00	00	00	00	00	00	00	00	00	00	
	000020		00	00	00	00	00	00	00	00	00	00	00	00	00	00	00	00	
	000030		00	00	00	00	00	00	00	00	00	00	00	00	00	00	00	00	
	000040		00	00	00	00	00	00	00	00	00	00	00	00	00	00	00	00	
	000050	]	00	00	00	00	00	00	00	00	00	00	00	00	00	00	00	00	<u> </u>
																			►
																			<u> </u>
Ac	dress colum	in   Ri abel colu	egister column							D	ata c	olumn	1						Code colu



- The Label and Register columns are hidden by default.
- The **Label** column shows the name of the label allocated to the first address of the memory data displayed on this row.
- The **Register** column shows the name of the register allocated to the first address of the memory data displayed on this row.
- '+n' in the column header of the **Data** column means the offset value from the first address of the row.
- The column header of the **Code** column shows the code name.
- Double-clicking the **Address** or **Label** column opens a dialog box, which allows you to change first address to be displayed..
- Double-clicking on the **Data** or **Code** column opens a dialog box, which allows you to change the memory data at the selected address. Changing of the value can be recorded in a macro (Macro Recording).
- Values in the Address, Data, and Code columns can be changed by in-place editing. (Macro Recording)

#### Options

Right-clicking displays a pop-up menu containing available options.

A basic operation is allocated to the toolbar.

The Toolbar display and Customize toolbar options are also included in the pop-up menu opened by right-clicking on the toolbar.

Toolbar Macro	Function
Button Recording	9
Ø 😐	Sets data at specified address.
🏊 😐	Fills specified memory block with data.
🚔 🔶	Moves specified memory block to.
🐙 😐	Compares the contents of two memory blocks.
<u>s</u> -	Tests an area of memory.
-	Saves memory contents in a text file.
魚	Finds a value in memory.
<b>#</b> -	Finds a next value in memory.
-	Specifies the display start address.
=	Specifies scroll range.
	Starts address to value of the register.
<b>H</b> -	Keeps tracking of the stack pointer position.
<b>-</b>	Changes the program display position immediately after
	downloading.
<del>с</del> ,	Refreshes the Memory window.
	Disables refresh of the Memory window.
<b>e</b> + -	Regularly refreshes the Memory window.
<b>36</b> -	Specifies the refresh interval.
	Button         Recording           ↓         ●           ↓



# High-performance Embedded Workshop V.4.09

Pop-up Me	enu Option		Toolbar	Macro	Function						
				Recording							
	1byte			-	Data is displayed in 1 byte unit.						
Length	2bytes			-	Data is displayed in 2-bytes units.						
	4bytes			-	Data is displayed in 4-bytes units.						
	8bytes		***	-	Data is displayed in 8-bytes units.						
Radix	Hex		<u>16</u>	-	Data is displayed in hexadecimal.						
	Dec		10	-	Data is displayed in decimal.						
	Signed De	с	± <u>10</u>	-	Data is displayed in signed decimal.						
	Oct		8	-	Data is displayed in octal.						
	Bin		2	-	Data is displayed in binary.						
Code	ASCII		dbc	-	Displays memory data as ASCII characters (default).						
	SJIS			-	Not supported.						
	JIS		-	-							
	UNICODE	UTF-8	-	-	-						
Pop-up Me Data Length Radix Code Code		UTF-16	-	-	-						
	EUC		-	-	-						
	Float	Float	1.	-	Displays memory data as single-precision floating-point values.						
		Double	.d	-	Displays memory data as double-precision floating-point values.						
Ċ	Complex	Float Complex	†. ₩1	-	Displays complex numbers as single-precision floating-poin values.						
		Double Complex	.d †.dI	-	Displays complex numbers as double-precision floating- point values.						
		Float Imaginary	IL	-	Displays imaginary numbers as single-precision floating- point values.						
		Double Imaginary	.dI	-	Displays imaginary numbers as double-precision floating- point values.						
	Fixed	16bit Fixed	.16	-	Displays memory data as 16 bit fixed.						
		32bit Fixed	.32	-	Displays memory data as 32 bit fixed.						
		24bit Accum	-	-	Displays memory data as 24 bit accumulate.						
		40bit Accum	-	-	Displays memory data as 40 bit accumulate.						
Layout	Label		ы	-	Switches display or non-display of Label area.						
	Register		R	-	Switches display or non-display of Register area.						
	Code		a	-	Switches display or non-display of Code area.						
Column				-	Changes the number of digits displayed.						
-	Enable		0	-	Switches display or non-display of measurement result.						
				•	Saves memory contents in a file.						
			<i>è</i>	•	Loads a memory area contents from a file.						
			-	-	Splits up the window display.						
Toolbar dis	play		-	-	Shows or hides the toolbar.						
Customize	toolbar		-	-	Customizes toolbar buttons.						



## Notes:

- \*1. Support for this function depends on the debugger.
- **\*2**. Available only when the debugger supports this function.

## 17.3.2 Setting data at a desired address in the Memory window

To set data at a desired address in the Memory window, follow the procedure below.

• In-place edit in the Data column.

+0	+1	+2	+3
00	00	00	00
00	00	00	00
00	00	00	00
00	00	00	00

• In-place edit in the Code column.

A	S	0	Ι	Ι													ĺ
I																	j
ŀ	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•		1
ŀ	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•		
Ŀ	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•		

- To change the contents of memory, open the **Set** dialog box by selecting one of the following operations:
  - o Double-click the Data column you want to change, **OR**
  - Double-click the Code column you want to change, **OR**
  - Select the data you want to change and choose **Set** from the pop-up menu.

Set					? ×
ł	Address:	0000000	)	•	
[	Data:	00			
Г	Data Type				
	⊙ Va	lue	🔿 String		
	Length:	1byte			-
	Code:	ASCII			~
Γ	Verify				
	(	ОК	Cano	el	



Enter the value (value or character) to be set in the **Data** field. Select the **Verify** check box. Support for verify function depends on the debugger.

#### When setting the value

Click the Value button in the Data Type group. Specify the data length in the Length field.

#### When setting the character

Click the **String** button in the **Data Type** group.

## 17.3.3 Selecting a memory range

If the memory address range is in the **Memory** view, you can select the range by clicking on the first memory unit (depending on your **Memory** view display choice) and dragging the mouse to the last unit. The selected range is highlighted.

+0	+1	+2	+3	+4	+5	+6	+7	+8	+9	+A	+B	+C	+D	+E	+F
00	00	00	00	00	00	00	00	00	00	00	00	00	00	00	00
00	00	00	00	00	00	00	00	00	00	00	00	00	00	00	00
00	00	00	00	00	00	00	00	00	00	00	00	00	00	00	00
00	00	00	00	00	00	00	00	00	00	00	00	00	00	00	00
00	00	00	00	00	00	00	00	00	00	00	00	00	00	00	00
00	00	00	00	00	00	00	00	00	00	00	00	00	00	00	00
00	00	00	00	00	00	00	00	00	00	00	00	00	00	00	00
00	00	00	00	00	00	00	00	00	00	00	00	00	00	00	00
00	00	00	00	00	00	00	00	00	00	00	00	00	00	00	00

#### 17.3.4 Filling an area of memory with constant data

You can set the contents of a range of memory addresses to a value using the memory fill feature. Select an address range to fill in the Memory window by dragging the mouse. Choose **Fill** from the pop-up menu of the memory window. The **Fill** dialog box opens.

Fill	? ×
Start Address: 00000000 💌	<u>F</u>
End Address: 0000007F	
Data:	
Data Type • Value C String	
Length: 1byte	
Gode: ASCII 💌	
🗖 Verify	
OK Cancel	


Enter the data (value or character) to be filled in the **Data** field.

Select the **Verify** check box. Support for verify function depends on the debugger.

If you did not drag the address range to be filled, you must enter the start/end address. The end address can also be prefixed by a plus (+); the end address will become the (start address) + (entered value).

### When specifying the value

Click the Value button in the Data Type group. Specify the data length in the Length field.

### When setting the character

Click the **String** button in the **Data Type** group.

When the display data length is two bytes, two bytes' worth of a character can be specified.

Please use the Set dialog box to specify the character string. (Select menu Set)

# 17.3.5 Copying an area of memory

You can copy an area of memory using the memory copy feature. Select a copy-source address range in the Memory window by dragging the mouse. Choose **Move** from the pop-up menu of the memory window. The **Move** dialog box opens.

Move		? ×
Start Address:	0000000	• 🗾
End Address:	0000007F	•
Move Address:		•
Length:	1byte	-
🗖 Verify		
OK	Cancel	

Enter the copy destination start address in the Move Address field.

Select the Verify check box. Support for verify function depends on the debugger.

If you did not drag the copy-source address range, you must enter the start/end address. The end address can also be prefixed by a plus (+); the end address will become the (start address) + (entered value).

### **Drag and Drop**

Type of dropped data	Operation
Selected range on the Memory Window's Data column	Copy the contents of a selected range of data to an area starting from the dropped position.



# 17.3.6 Comparing the memory contents

The contents of two memory blocks can be compared. Select a source address range in the Memory window by dragging the mouse. Choose **Compare** from the pop-up menu of the memory window. The **Compare Memory** dialog box opens.

Compare Memory		? ×
Start Address:	000000	<u></u>
End Address:	0000FF 💌	<u></u>
Compare Address:		<u></u>
DataLength:	1byte	-
ОК	Cancel	

Enter the start address of the destination memory area in the **Compare Address** field and the data length in the **Data Length** field. If you did not drag the comparison-source address range, you must enter the start/end address. The end address can also be prefixed by a plus (+); the end address will become the (start address) + (entered value).

If there is a mismatch, the address where it was found is displayed in a message box.

When the contents of the two memory areas match at comparison, the message "Comparison successful." appears.

Support for this function depends on the debugger.

### 17.3.7 Testing an area of memory

You can test an area of memory in the address space using the **Memory Test** feature. Select an address range to test in the Memory window by dragging the mouse. Choose the **Test** option from the pop-up menu of the memory window. The **Test Memory** dialog box opens.

Test Memory		? ×
Start Address:	000000	- 🔊
End Address:	0000FF	•
OK	Cancel	

If you did not drag the address range to be tested, you must enter the start/end address. The end address can also be prefixed by a plus (+); the end address will become the (start address) + (entered value).

Support for this function depends on the debugger.

### Note:

The exact test is target dependent. However, in all cases the current contents of the memory will be overwritten - YOUR **PROGRAM OR DATA WILL BE ERASED**.



# 17.3.8 Saving memory contents in a text file

You can save an area of memory in the address space to a text file using the **Save Memory Contents** feature. Select an address range to save in the Memory window by dragging the mouse. Choose **Save Memory contents** from the pop-up menu of the memory window. The **Save memory contents** dialog box opens.

Select the output range in **Memory Save Area**, data format in **Data Length**, number of digits in **Column**, radix in **Radix**. It is possible to select showing/hiding of the Label column and Register column by **Output the label column** and **Output the register column**, respectively.

If you did not drag the address range to be saved, you must enter the output range.

Save memory cont	ents			? ×
Save in: 🔁 SH4d	lemo	•	( <del>-</del> 🔁 (	* 🎟 -
🔁 Debug				
SimDebug_SH-4				
🗒 Readme.txt				
File <u>n</u> ame:				<u>S</u> ave
Save as <u>t</u> ype: Text	Files (*.txt)		•	Cancel
Memory Save Area:	00000000 💌 👳		000000FF	- 🗾
Data Length:	1byte 💌	Column:	16	
Radix:	Hex 💌	Code:	ASCII	-
	Output the label control	olumn.	🔽 Output t	he register column

# 17.3.9 Finding a value in memory

You can find a value in memory using the **Search Memory** feature. Select an address range to search in the Memory window by dragging the mouse. Choose **Search** from the pop-up menu of the memory window. The **Search memory** dialog box opens.

Search Memory		? ×
Start Address:	000000	<u></u>
End Address:	0000FF	
Data:	1F5B	
Search Data As:	2byte	•
🛛 🔽 Search Equal Data	a 🛛 🦵 Search Backwa	ard
ОК	Cancel	

Enter a value you want to find in **Data** and select the data format in **Search Data As**.

If "pattern" is selected as the **Search Data As**, a byte string of up to 256 bytes or less (512 characters or less) can be searched for. Specify the data for each byte with two characters.



### High-performance Embedded Workshop V.4.09

Search conditions other than pattern search are data match/mismatch and search direction. Note that only data match and forward direction can be selected with pattern search.

If you did not drag the address range to be found, you must enter the start/end address.

The end address can also be prefixed by a plus sign (+), which will use the entered value as a range.

If the data is found, the data which has been found is highlighted in the Memory window. If **Search Next** is selected from the pop-up menu in the state where data has been found, the search will continue from the next address.

Support for this function depends on the debugger.

### 17.3.10 Changing the display address

Use the scroll bar, Page Up/Page Down key and Up/Down key to change the display position.

To change the display address directly, follow the procedure below.

1. In-place edit in the Address column.

Address	L٤
000000	-
000010	
000020	
000030	I

- 2. To change the display address, open the **Display Address** dialog box by selecting one of the following operations:
  - Double-click the Address column you want to change, **OR**
  - Choose **Address** from the pop-up menu.



Specify the displaying address in the Address field.

### **Drag and Drop**

Manipulation	Operation
Select address on Memory Window's Address area and	Changes the window's display start address to that address.
drop it into another Memory Window's Address column	
Select variable name (text string) and drop it into Memory	Changes the window's display start address to that address.
Window's Address column	



### 17.3.11 Changing the scroll area

Select Scroll Area from the pop-up menu of the memory window. The Scroll Area dialog box opens.

Scroll Area		? ×
Start Address:	000000	• 🔊
End Address:	OFFFFF	• 🗾
ОК	Cancel	

Specify the scroll range to be displayed. By default, the scroll range is set to 0 to the maximum address of MCU.

### 17.3.12 Starting address to value of the register

Select the register from the followings in the pop-up menu **Register** of the memory window.

Support for this function depends on the debugger.

### 17.3.13 Tracking the stack pointer position

The memory window has a function that alters the display address while tracking the stack pointer position (By default, the display does not track the stack pointer position.). To track the stack pointer position, choose **Followed Stack Pointer** from the pop-up menu of the memory window. The **Followed Stack Pointer** dialog box opens.

Followed Stack Pointer 🛛 🔋 🗙				
Stack Pointer:	lot Following			
OK	Cancel			

Select the stack pointer to track from the drop-down list box. The Memory Window has its display addresses automatically altered by tracking the selected stack pointer position. Selecting "Not Following" in the **Followed Stack Pointer** dialog box causes the debugger to stop tracking the stack pointer position.

Support for this function depends on the debugger.

# 17.3.14 Changing the program display position immediately after downloading

To specify the source file position, select **Set Start Up Symbol** option from pop-up menu of the memory window. The **Set start up symbol** dialog box opens.





Input start up symbol to drop-down list box.

### 17.3.15 Refreshing the Memory window

The **Memory** window contents can be forcibly refreshed. Selecting **Refresh** from the pop-up menu of the **Memory** window.

### 17.3.16 Disabling refresh of the Memory window

Automatic refresh of the **Memory** window, which is performed when user program execution stops and in other cases, can be disabled.

Select **Lock Refresh** from the pop-up menu of the **Memory** window. While **Lock Refresh** is active, the contents of the Memory window are grayed-out.

Auto Refresh in the Memory window cannot be used with Lock Refresh.

# 17.3.17 Regularly refreshing the Memory window

If you select **Auto Refresh** from the pop-up menu of the **Memory** window, the **Memory** window will regularly be refreshed while the user program is running.

The actual refresh interval is shown on the leftmost column header in the Memory window.

Support for this function depends on the debugger.

	Memory []	Watch_1]					×
		11 16	<u>10 ±10</u>	8 2 4	bc み あ	あま あ	.f.d
Refresh	[[ 3328ms]		+4	+8	+C	ASCII	
interval	00005400	0000598E	000054E6	0000498A	000039A5		I9.
	00005410	0000255E	000020C3	000020AF	000012C8	<u>×</u> ^ .	
	00005420	000010D0	0000020B	00000000	00005400		T.
	00005430	D98EB9F0	00000000	00000001	00000000	•••••	
	00005440	000014D4	00000000	00005000	00000000		°
	00005450	D32E561F	7B36C775	F1E4CCF9	175BE4A2	V.{6.u	···· L·· 🗾
	•						►

To specify the refresh interval, select **Refresh Interval** from the pop-up menu. This item is only selectable when it is supported by the debugger.

Lock Refresh in the Memory window cannot be used with Auto Refresh.



# 17.3.18 Specifying the refresh interval

You can specify the interval to refresh the **Memory** window while the user program is running.

Support for this function depends on the debugger.

### To specify the refresh interval

- 1. Right-click within the window to open a pop-up menu.
- 2. Select Refresh Interval. The Auto Refresh Interval dialog box opens.

Auto Refresh Interva	? ×
Interval (300 - 9000mse	ec : 10msec unit):
1500	msec
ОК	Cancel

3. Specify the refresh interval in 10-ms units. The default value and valid range vary depending on the debugger.

The refresh interval can be specified for each **Memory** window.

After **Auto Refresh** is activated, the actual refresh interval is shown on the leftmost column header in the **Memory** window during execution of the user program.

The actual refresh interval may be longer than the specified value depending on the state of execution.

# 17.3.19 Changing the data length

Select the data length from the followings in the pop-up menu Data Length of the memory window.

Either the following can be specified.

- 1byte Displays in 1-byte units (default).
- 2bytes Displays in 2-bytes units.
- 4bytes Displays in 4-bytes units.
- 8bytes Displays in 8-bytes units.

# 17.3.20 Changing the radix

Select the data radix from the followings in the pop-up menu Radix of the memory window.

Either the following can be specified.



Hex	Displays in hexadecimal (default).
Dec	Displays in decimal.
Signed Dec	Displays in signed decimal.
Oct	Displays in octal.
Bin	Displays in binary.

# 17.3.21 Changing the code

Select the code from the followings in the pop-up menu **Code** of the memory window.

Either the following can be specified.

ASCII		Displays memory data as ASCII characters (default).
SJIS		Not supported.
JIS		Not supported.
UNICODE	UTF-8	Not supported.
	UTF-16	Not supported.
EUC		Not supported.
Float	Float	Displays memory data as single-precision floating-point values.
	Double	Displays memory data as double-precision floating-point values.
Complex	Float Complex	Displays complex numbers as single-precision floating-point values.
	Double Complex	Displays complex numbers as double-precision floating-point values.
	Float Imaginary	Displays imaginary numbers as single-precision floating-point values.
	Double Imaginary	Displays imaginary numbers as double-precision floating-point values.
Fixed	16bit Fixed	Displays memory data as 16bit fixed.
	32bit Fixed	Displays memory data as 32bit fixed.
	24bit Accum	Displays memory data as 24bit accumulate.
	40bit Accum	Displays memory data as 40bit accumulate.

# 17.3.22 Setting the layout

Select the layout from the followings in the pop-up menu Layout of the memory window.

The followings can be selected:

Label	Switches display or non-display of the Label column.
Register	Switches display or non-display of the Register column.
Code	Switches display or non-display of the Code column.

When the label, register or code is shown, the option is checked.

# When the Label, Register and Code columns are hidden:

Address	+0	+1	+2	+3	+4	+5	+6	+7	+8	+9	+A	+B	+C	+D	+E	+F
00000000	00	00	00	00	00	00	00	00	00	00	00	00	00	00	00	00
00000010	00	00	00	00	00	00	00	00	00	00	00	00	00	00	00	00
00000020	00	00	00	00	00	00	00	00	00	00	00	00	00	00	00	00



# 17.3.23 Changing the number of digits displayed

Choose Column from the pop-up menu of the memory window. The Set Column dialog box opens.

Set Column	? X
1E	OK
	Cancel

Specify the number of digits in which you want data to be displayed (1-256).

### 17.3.24 Switching display or non-display of measurement result

In the memory window, a display of coverage measurement is set to "Disable" by default. To enable the display, select [**Coverage -> Enable**] from the pop-up menu of the memory window. In the Memory window, the background of the executed lines is displayed in sky blue, and the background of the unexecuted lines is displayed in gray.

During coverage measurement, the default foreground and background colors of the executed codes are black and light blue, respectively, while those of the codes not executed are black and gray. The colors in both cases can be customized in the **Format Views** dialog box.

For detail, see section 17.3.29, Changing text colors.

Ī	+0	+1	+2	+3	+4	+5	+6	+7	+8	+9	+A	+B	+C	+D	+E	+F	ASCII
1	00	00	00	00	00	00	00	00	00	00	00	00	00	00	00	00	
	00	00	00	00	00	00	00	00	00	00	00	00	00	00	00	00	
	00	00	00	00	00	00	00	00	00	00	00	00	00	00	00	00	
	00	00	00	00	00	00	00	00	00	00	00	00	00	00	00	00	

Support for this function depends on the debugger.

### 17.3.25 Saving an area of memory

You can save an area of memory in the address space to a disk file using the **Save Memory** feature. Select an address range to save in the **Memory** window by dragging the mouse. Choose **Save** from the pop-up menu of the memory window. The **Save Memory As** dialog box opens

(This operation can also be achieved by selecting [Debug -> Save Memory].)



High-performance Embedded Workshop V.4.09

Save Memory As	? ×
<u>F</u> ile format: Binary	Save
File <u>n</u> ame:	
<u>S</u> tart address: 00001000	
End address: 000010FF	
Access size:	

If you did not drag the address range to be saved, you must enter the start/end address. The end address can also be prefixed by a plus sign (+), which will use the entered value as a range.

Enter the file format in **File format**, file name in **Filename**, and access size in **Access size**. The **File name** drop-down list contains the last four filenames used for saving memory. Alternatively, click the "..." button to launch a standard **Save As** dialog box. The access size for saving data can be selected from the **Access size** drop-down list. When the data is saved in memory with little endian, the order of data depends on the access size.

When the file save is complete a confirmation message box will be displayed.

### 17.3.26 Loading a memory area from a file

A file can be loaded to the debugger's memory. Choose **Load** from the pop-up menu of the memory window. The **Load Program** dialog box opens.

Load Program	? ×
Eile format: Binary	Open Cancel
File <u>n</u> ame:	
<u>O</u> ffset address: 00000000	J
Access size:	nload

Enter the file format in **File format**, file name in **Filename**, offset address in **Offset address**, and access size in **Access** size. To verify memory, check **Perform memory verify during download**. If the load address value is to be changed, enter the offset value in the Offset field, otherwise enter 0.

# 17.3.27 Splitting up the window display

To vertically divide the Memory window display into two, select **Split** from the pop-up menu and move the split-up bar.

To cancel the split-up display, select **Split** from the pop-up menu again.

### 17.3.28 Verifying a memory area

A memory area in the address space can be verified using the memory verify function. Choose [**Debug -> Verify Memory**]. The **Verify Memory** dialog box opens.

Verify Memory	? ×
Eile format: Binary	Verify
File <u>n</u> ame:	Cancel
Offset address:	J
Access size:	

Enter the file format in **File format**, file name in **Filename**, offset address in **Offset address**, and access size in **Access** size.

The message "File verified OK" appears when verification is successful. If the verification failed, "Verify failed" appears.

Support for this function depends on the debugger.

# 17.3.29 Changing text colors

The color and font of the data or codes shown in the Memory window can be customized in the **Format Views** dialog box (in the same manner as the color and font in other windows). Now it is also possible to customize the color of the changed values or accessed memory, which are shown in the data or code field of the Memory window, in the **Format Views** dialog box.

### To change the look of the Memory window

- 1. Select [Setup -> Format Views]. The Format Views dialog box opens.
- 2. Select the Memory item in the tree and expand it.
- 3. Select the category of objects to be customized.



Category	Objects to be Customized	Foreground color (default)	Background color (default)
Normal	Text shown in the windows	Black	White
Accessed *	During coverage measurement: Executed codes	Black	Sky blue
Not Accessed *	During coverage measurement: Codes not executed	Black	Gray
Unknown *	During coverage measurement: The outside in coverage area	Black	White
No Memory *	The outside in memory area	Gray	Gray
Modified	Changed values	Red	White

Note:

\*. Support for this function depends on the debugger.

- 4. Modify the **Foreground** and **Background** color selection on the **Color** tabbed page as desired.
- 5. Click OK.

# 17.4 Displaying memory contents as an Image

The memory contents can be displayed as an image in the **Image** window.

Support for this function depends on the debugger.

# 17.4.1 Opening the Image window

Choose [View -> Graphic -> Image] or click the Image toolbar button (



High-performance Embedded Workshop V.4.09

je Properties	
- Color Information:	
Mode:	Bit/Pixel: 24bit
C RGB	
🔿 BGR	<u>Sampling:</u> 4:4:4
C YCbCr	Eormat: chunky
Buffer Information:	
Data <u>A</u> ddress: 00000	0000 💌 🗾
Palette Address: 00000	0000 🔽 💌
_ Width/Height Size(Pixel)	D:
Width:	00000000 Byte
Height:	0 0000000 Byte
Enable <u>R</u> edraw Contir     Image Data:     Frame Count:	Buffer Full Size(Hex):
	Dyte
View Information:	Position:
✓IEW Mode. ● Full Size	X Position:
C Part Size	⊻ Position: 0
Start Position:	Width/Height Size(Pixel):
О Тор	Width: 0
<ul> <li>Bottom</li> </ul>	Height: 0
	<u> </u>

The Image Properties dialog box is used to specify the display method of the Image window.

The following items are to be specified:

Color	Specifies the color information of the image to be displayed.							
Information	Mode	Specifies the format.						
		MONOCHROME	Black and white.					
		RGB	R (red), G (green), and B (blue)					
		BGR	B (blue), G (green), and R (red)					
		YCbCr	Y (luminance), Cb (color difference for blue), and Cr (color					
			difference for red)					
	Bit/Pixel	Specifies bits/pixel selected).	according to the selected mode (valid when RGB or BGR is					



# Sampling

Specifies the format of sampling (valid when YCbCr is selected). An example is shown below.

		7 m example 13 sho	wii below.			
		Sampling Ratio	Original Data	Sampling Data	Size of an Image (4*4 pixels)	
		4:4:4	11,12,13,14	11,12,13,14	16 bytes	
			21,22,23,24	21,22,23,24		
			31,32,33,34	31,32,33,34		
			41,42,43,44	41,42,43,44		
		4:2:2		11,11,13,13	8 bytes	
				21,21,23,23		
				31,31,33,33		
				41,41,43,43		
		4:1:1		11,11,13,13	4 bytes	
				11,11,13,13		
				31,31,33,33		
				31,31,33,33		
		4:2:0		11,11,11,11	4 bytes	
				21,21,21,21		
				31,31,31,31		
				41,41,41,41		
		Format Chunky(invalid wh planar planar2	nen 4:2:0 is sele	Y,Y,Y,,Y,	Cb,Cr,Y,Cb,Cr,,Y,Cb,Cr Cb,Cb,Cb,,Cb,Cr.Cr.Cr,,Cr Cb,Cr,Cb,Cr,Cb,Cr,,Cb,Cr	
Buffer	Specifies the area to	-	d the address of			
Information	Data Address	Specifies the first address in memory of the area for display as image data (in hexadecimal notation).				
	Palette Address	Specifies the first address in memory of the color-palette data (in hexadecimal notation; valid when "8Bit" has been selected for RGB or BGR).				
	Width/Height Size	Specifies the width and height of the image.				
		Width (Pixel)	-	width of the image ated as decimal nu	e (unless a prefix is included, mbers).	
		Height (Pixel)	•	height of the imag ated as decimal nu	e (unless a prefix is included, mbers).	
		Buffer Size (Hex)	Displays the	size of the buffer r	equired for image display as ght (in hexadecimal notation).	
	Enable Redraw Continuously	Specifies the redraw continuously function has been enabled.				
	Image Data	Specifies the numb	Specifies the number of frames to be redrawn image continuously.			
	č	Frame Count		number of frames	•	
		Buffer Full Size	-		equired for images display as	
		(Hex)	1.0		ight, and the number of frames	
		()		1		



(in hexadecimal notation).

View Mode	Specifies the entire/	part to be displayed in the image.
	Full Size	The image is displayed on the whole screen.
	Part Size	The image is displayed on part of the screen.
Start Position	Specifies where the	display of data is to start.
	Тор	Display of data starts at the upper-left position.
	Bottom	Display of data starts at the lower-left position.
Position	Specifies the start p	osition of the image where part of the image is to be displayed.
	(Valid when "Part S	Size" is selected)
	X Position	Specifies the X-coordinate of the start position (unless a prefix
		is included, values are treated as decimal numbers).
	Y Position	Specifies the Y-coordinate of the start position (unless a prefix
		is included, values are treated as decimal numbers).
Width/Height Size	Specifies the height	and width of an image to be displayed on part of the screen.
	Width (Pixel)	Specifies the width of the display (unless a prefix is included,
		values are treated as decimal numbers).
	Height (Pixel)	Specifies the height of the display (unless a prefix is included,
		values are treated as decimal numbers).

View Specifies whether display is on all or part of the screen (in hexadecimal notation; valid when "8Bit" has Information been selected for RGB or BGR).

After the settings have been made in the Image Properties dialog box, clicking the OK button opens the Image window.

Even after the **Image** window is displayed, the display contents can be modified by opening this dialog box by choosing **Properties** from the pop-up menu.

The Image window shows the memory contents as an image.



• Double-clicking within the window displays information on the pixel on which the mouse pointer is located in the **Pixel Information** dialog box.



# Options

Right-clicking displays a pop-up menu containing available options.

A basic operation is allocated to the toolbar.

The **Toolbar display** and **Customize toolbar** options are also included in the pop-up menu opened by right-clicking on the toolbar.

Pop-up Menu Option		<b>Toolbar Button</b>	Macro Recording	Function
Auto	Nonrefresh	Nonrefresh 💌	-	Disables refresh of the Image window.
Refresh	Stop		-	Automatically refreshes the Image window when user
				program execution stops.
	Real time		-	Regularly refreshes the Image window.
Refresh N	Now	\$	-	Refreshes the Image window.
Update In	nterval *	<b>1</b>	-	Specifies the refresh interval.
Next Imag	ge	*	-	Redraws the next image.
Previous	Image	+	-	Redraws the previous image.
Top Imag	е	₩	-	Redraws the top image.
Last Imag	je	≯	-	Redraws the last image.
Redraw	0.5s	1.0s 💌	-	The image is redrawn continuously every 0.5 seconds.
Interval	1.0s		-	The image is redrawn continuously every 1.0 seconds.
	2.0s		-	The image is redrawn continuously every 2.0 seconds.
	4.0s	_	-	The image is redrawn continuously every 4.0 seconds.
Redraw C	Continuously		-	Starts redrawing continuously.
Stop			-	Stops redrawing continuously.
Properties	S	<b>6</b>	-	Opens the Image Properties dialog box.
Toolbar display		-	-	Shows or hides the toolbar.
Customiz	e toolbar	-	-	Customizes toolbar buttons.
<b>N</b> T /				

Note:

\*. Available only when the debugger supports this function.

# 17.4.2 Regularly refreshing the Image window

Selecting [Auto Refresh -> Nonrefresh] from the pop-up menu will not refresh the window.

Selecting [Auto Refresh -> Stop] from the pop-up menu will allow the window contents to be automatically refreshed when user program execution stops.

Selecting [Auto Refresh -> Real time] from the pop-up menu will allow the window contents to be refreshed while the user program is running.

To specify the refresh interval, select **Update Interval** from the pop-up menu. This item is only selectable when it is supported by the debugger.



# 17.4.3 Refreshing the Image window

Selecting **Refresh Now** from the pop-up menu immediately refreshes the window contents.

### 17.4.4 Specifying the refresh interval

You can specify the interval to refresh the Image window while the user program is running.

Support for this function depends on the debugger.

### To specify the refresh interval

- 1. Right-click within the window to open a pop-up menu.
- 2. Select **Update Interval**. The **Update Interval** dialog box opens.

Update Interval 🛛 ? 🗙				
Interval (10 - 10000msec : 10msec unit):				
1000	msec			
<u>0</u> K	<u>C</u> ancel			

3. Specify the refresh interval in 10-ms units. The refresh interval is specifiable in the range from 10 to 10000 ms and the default value is 1000 ms.

The specified refresh interval is applied to all Image windows.

The actual refresh interval may be longer than the specified value depending on the state of execution.

# 17.4.5 Viewing Images as Consecutive Frames

The **Image** window can also show images as consecutive frames. This allows you to view multiple images easily because it is not necessary to set addresses for all images.

After images have been loaded into consecutive addresses in memory as equal-sized frames, you can view the images by switching the frames in order.

### To enable the function to show images as consecutive frames

- 1. Take either of the following ways to open the Image Properties dialog box.
  - When the **Image** window is currently open, right-click on the window and select **Properties** from the pop-up menu.



- When no Image window is open, select [View -> Graphic -> Image] or click on the Image toolbar button

   (
   ).
- 2. Select the Enable Redraw Continuously checkbox in the Image Properties dialog box.
- 3. In the **Frame Count** edit box, enter the number of frames (2 or more) that you wish to view.
- 4. Click on the **OK** button.

After this function has been enabled, you can view the images by either manually or automatically switching the frames.

# 17.4.5.1 Manually switching the frames

While the function to show images as consecutive frames is enabled, the pop-up menu options and toolbar buttons listed in the table below are available for manually switching the frames.

Pop-up Menu Option	<b>Toolbar Button</b>	Function
Next Image	+	Redraws the next image.
Previous Image	+	Redraws the previous image.
Top Image	₩	Redraws the top image.
Last Image	→	Redraws the last image.

Whether the pop-up menu options and toolbar buttons are active or not depends on which frame is currently displayed.

### To manually switch the frames

Clicking on **Next Image** goes to the next frame. If you wish to view three consecutive frames, for example, the operation will be as follows.

Image		×
Nonrefresh		
۲	Next Image	



1. Click on the **Next Image** toolbar button to view the second frame.

Image		×
Nonrefresh		
	Next Image	

2. Click on the **Next Image** toolbar button to view the third frame.

Image	×
Nonrefresh 💽 🔯 🔶 🗲 🛧 🕂 1.0s 💌 🕨 🔳	

# 17.4.5.2 Regularly switching the frames

While the function to show images as consecutive frames is enabled, the pop-up menu and toolbar items listed in the table below are available for regularly switching the frames.

Pop-up Menu Option Toolbar Button			Function	
Redraw Interval * 0.5s		1.0s 💌	The image is redrawn continuously every 0.5 seconds.	
	1.0s		The image is redrawn continuously every 1.0 seconds.	
	2.0s		The image is redrawn continuously every 2.0 seconds.	
	4.0s		The image is redrawn continuously every 4.0 seconds.	
Redraw Continuously			Starts redrawing continuously.	
Stop 🗖			Stops redrawing continuously.	

Note:

\*. Images may not be updated in the defined interval depending on the debugger in use and the image size.

Whether the pop-up menu and toolbar items are active or not depends on which frame is currently displayed.



### To regularly switch the frames

While **Redraw Continuously** is enabled, the frames are regularly switched. If you wish to view three consecutive frames, for example, the operation will be as follows.

1. On the first frame, select an update interval from the **Redraw Interval** drop-down list box and click on the **Redraw Continuously** toolbar button.



2. When the specified time has elapsed, the second frame is automatically displayed.

Image						×
Nonrefresh	- ka	₫ + 4	- 14)	1.0s 🔻	▶ ■	1
302						

3. When the specified time has elapsed, the third frame is automatically displayed. The automatic switching of frames has ended.





# 17.4.6 Displaying the pixel information

Double-clicking within the window displays information on the pixel on which the mouse pointer is located in the **Pixel Information** dialog box.

Pixel Information				? ×
Color mode:	N	Ionochror	ne	<u> </u>
Pixel:		Black		
Position X:	52	Y:	11	
Buffer Size Width:	80	Height:	80	
Image Size Width:	80	Height:	80	

This dialog box displays pixel information on the cursor location.

Color Mode	Displays the format of the image.		
Pixel	Displays color information of the cursor location. (Displayed in decimal)		
Position	Displays the cursor location in X and Y coordinate. (Displayed in decimal)		
	Х	Displays the X-coordinate of the cursor location.	
	Y	Displays the Y-coordinate of the cursor location.	
Buffer Size	Displays the buffe	er size. (Displayed in decimal)	
	Width	Displays the buffer width.	
	Height	Displays the buffer height.	
Image Size	Displays the width	n and height of the display. (Displayed in decimal)	
	Width	Displays the width.	
	Height	Displays the height.	

# 17.5 Displaying memory contents as Waveforms

Memory contents can be displayed as wave forms in the **Waveform** window.

Support for this function depends on the debugger.

# 17.5.1 Opening the Waveform window

Choose [View -> Graphic -> Waveform] or click the Waveform toolbar button ( to open the Waveform Properties dialog box.



# High-performance Embedded Workshop V.4.09

# 17. Debugging Facility

Waveform Properties		?	×
Data <u>A</u> ddress: 0000		<u>ρ</u> κ	]
Data <u>S</u> ize: 8bit		Ca <u>n</u> cel	1
- <u>C</u> hannel: Mono	C Stereo		-
Buffer Size(Hex): 0000	0000	•	

Specifies the waveform format. The following items can be specified.

Data Address	Specifies the start address of data in memory. (Displayed in hexadecimal)
Data Size	Selects 8bit or 16bit.
Channel	Specifies Mono or Stereo.
Buffer Size (Hex)	Specifies the buffer size of data. (Displayed in hexadecimal)

After the settings have been made in the **Waveform Properties** dialog box, clicking the OK button opens the **Waveform** window.

Even after the **Waveform** window is displayed, the display contents can be modified by opening this dialog box by choosing **Properties** from the pop-up menu.

Displays the memory contents as waveforms. The X-coordinate shows the number of sampling data and the Y-coordinate shows the sampling value.



• If you double-click the coordinate where you wish to view the sampling information, the **Sample Information** dialog box appears.



# Options

Right-clicking displays a pop-up menu containing available options.

Pop-up Menu Option Macro Reco		Macro Recording	ng Function		
Auto Refresh Nonrefre		-	Disables refresh of the Waveform window.		
	Stop	-	Automatically refreshes the Waveform window when user		
			program execution stops.		
	Real time	-	Regularly refreshes the Waveform window.		
Refresh Now		-	Refreshes the Waveform window.		
Update Interval *		-	Specifies the refresh interval.		
Zoom In		-	Zoom-in display.		
Zoom Out		-	Zoom-out display.		
Reset Zoom		-	Resets the zoom display.		
Zoom Magnification X2		-	The zoom magnification is 2.		
	X4	-	The zoom magnification is 4.		
	X8	-	The zoom magnification is 8.		
Scale 128		-	The size of the X-coordinate is 128 pixels.		
	256	-	The size of the X-coordinate is 256 pixels.		
	512	-	The size of the X-coordinate is 512 pixels.		
Clear Cursor		-	Hides the cursor display.		
Sample Information		-	Displays the sampling information of the cursor location.		
Properties		-	Opens the Waveform Properties dialog box.		
Note:					

\*. Available only when the debugger supports this function.

# 17.5.2 Regularly refreshing the Waveform window

Selecting [Auto Refresh -> Nonrefresh] from the pop-up menu will not refresh the window.

Selecting [Auto Refresh -> Stop] from the pop-up menu will allow the window contents to be automatically refreshed when user program execution stops.

Selecting [Auto Refresh -> Real time] from the pop-up menu will allow the window contents to be refreshed while the user program is running.

To specify the refresh interval, select **Update Interval** from the pop-up menu. This item is only selectable when it is supported by the debugger.

# 17.5.3 Refreshing the Waveform window

Selecting Refresh Now from the pop-up menu immediately refreshes the window contents.



# 17.5.4 Specifying the refresh interval

You can specify the interval to refresh the **Waveform** window while the user program is running.

Support for this function depends on the debugger.

### To specify the refresh interval

- 1. Right-click within the window to open a pop-up menu.
- 2. Select **Update Interval**. The **Update Interval** dialog box opens.

Update Interval	? ×
Interval (10 - 10000mse	ec : 10msec unit):
1000	msec
<u> </u>	Cancel

3. Specify the refresh interval in 10-ms units. The refresh interval is specifiable in the range from 10 to 10000 ms and the default value is 1000 ms.

The specified refresh interval is applied to all **Waveform** windows.

The actual refresh interval may be longer than the specified value depending on the state of execution.

# 17.5.5 Zoom-in display

Selecting **Zoom In** from the pop-up menu displays the waveforms with the horizontal coordinate enlarged.

# 17.5.6 Zoom-out display

Selecting Zoom Out from the pop-up menu displays the waveforms with the horizontal coordinate reduced.

### 17.5.7 Resetting the zoom display

Selecting **Reset Zoom** from the pop-up menu displays the waveforms in its original size.

### 17.5.8 Setting the zoom magnification

In the Zoom Magnification submenu of the pop-up menu, the zoom magnification can be selected from 2, 4, or 8.



# 17.5.9 Setting the horizontal scale

In the Scale submenu of the pop-up menu, the size of the X-coordinate can be selected from 128, 256, or 512 pixels.

# 17.5.10 Non-display of cursor

Selecting **Clear Cursor** from the pop-up menu hides the cursor display.

# 17.5.11 Displaying the sampling information

When anywhere within the graph is clicked, a cursor (green vertical line) appears. The cursor can be moved by pressing the left or right arrow key. Right-click in the window to display a pop-up menu. Selecting **Sample Information** from the pop-up menu displays the **Sample Information** dialog box.

If you double-click the coordinate where you wish to view the sampling information, the **Sample Information** dialog box appears.

Sample Infor	mation	? ×
Data Size:	8bit	<u>0</u> K
Channel:	Mono	
X: Value	00000014	
Yaide Y:	FF	

Displays the sampling information of the cursor location in the **Waveform** window. The following information is displayed.

Data Size	Displays	8bit or 16bit.
Channel	Displays	the data channel.
Value	Х	Displays the X-coordinate of cursor location.
	Y	Displays the Y-coordinate of cursor location (displays Y-coordinate for both the upper and
		lower plots when Stereo is selected).

# 17.6 Looking at I/O memory

As well as a CPU and ROM/RAM, a micro-controller also contains on-chip peripheral modules. The exact number and type of peripheral modules differ between devices but typical modules are DMA controllers, serial communications interfaces, A/D converters, integrated timer units, a bus state controller and a watchdog timer. Accessing registers, which are mapped to the micro-controller's address space, programs the on-chip peripherals.

Since the setting up and use of these on-chip peripheral registers is usually very important in an embedded microcontroller application, it is useful to be able to look clearly at the contents of these registers. The Memory view only allows you to look at data in memory as byte, word, long word, single-precision floating-point, double-precision floating-



point, or ASCII values, so the High-performance Embedded Workshop also provides an **IO** window to ease the inspection and setting up of these registers.

# 17.6.1 Opening the IO window

To open the **IO** window, choose [**View -> CPU -> IO**] or click the **View IO** toolbar button (**ID**). Modules that match the on-chip peripheral modules organize the I/O register information.

The IO window has two tabs: All Register and Selected Register.

When the **IO** window is first opened, only a list of module names is shown on the **All Register** tabbed pane.

### Window configuration



Tab	Description
All Register	Shows all I/O registers.
Selected Register	Shows selected I/O registers. This page is blank by default.

- I/O registers can be expended.
- If the value of an I/O register is changed, the value is displayed in red.
- Double-clicking on the line of an I/O register opens a dialog box, in which you can change the value. Changing of the value can be recorded in a macro (Macro Recording).
- The values of I/O registers and their bits can be changed in the **Value** column by in-place editing (Macro Recording).



# Options

Right-clicking displays a pop-up menu containing available options.

A basic operation is allocated to the toolbar.

The **Toolbar display** and **Customize toolbar** options are also included in the pop-up menu opened by right-clicking on the toolbar.

Pop-up Menu Option	<b>Toolbar Button</b>	Macro Recording	Function
Refresh	<u>0</u>	-	Refreshes the IO window.
Lock Refresh	×	-	Disables refresh of the IO window.
Load IO File	<b>F</b>	-	Manually loads an I/O file.
Add to Selected Register tab *		-	Adds the selected I/O register(s) to the <b>Select</b> <b>Register</b> tabbed pane.
Select IO Register	R	-	Selects the I/O register(s) to be shown on the <b>Select Register</b> tabbed pane.
Print	<b>e</b>	-	Prints the current data of the window.
Save To File		-	Saves the current data of the window to a text file.
Find	ĝiĝ.	-	Finds an I/O register.
Find Next	Ĥ	-	Finds the next I/O register to match.
Toolbar display	-	-	Shows or hides the toolbar.
Customize toolbar	-	-	Customizes toolbar buttons.
NT 4			

Note:

\*. This menu option is selectable only when the **All Register** tabbed pane is open.

# 17.6.2 Expanding an I/O register display

To display the names, addresses and values of the I/O registers, double-click on the module name or select the module name, by clicking on it or using the cursor keys, and press the cursor right key. The module display will expand to show the individual registers of that peripheral module and their names, addresses and values. Double-clicking (or pressing the cursor left key) again on the module name will close the I/O register display.

# Note:

If you are using an emulator-based debugger, reading data from an I/O register can sometimes affect the operation of your program. For example, reading a data register can cancel a pending interrupt. Data is only read from I/O modules that have been expanded in the **IO** window (so that the register values are displayed). Therefore, as long as I/O modules are collapsed when they no longer need to be displayed, this will not cause a problem. Also, note that having a Memory window (or Disassembly window) open on the I/O area can have the same effect.

While Lock Refresh of the IO window is active, no data will be read out even if an I/O register display is expanded.



# 17.6.3 Modifying the values of I/O registers

The values of I/O registers can be modified in the **IO** window.

# To modify the value of an I/O register

Select either of the following ways.

• In-place edit in the Value column.

10			×
🔄 🔀 🖪 🛤 🔐 🖬			
Name	Address	Value	Access
🖃 🚍 Memory Management Unit			
🖻 🗁 🤤 РТЕН	FF000000	00000001	
ASID		1	
All Register Selected Register			

• Double-click on the line of the I/O register or bit to open a dialog box in which the values can be changed. Enter a value in **New value**.

Edit ASID at FF000000	? ×
<u>C</u> urrent value: 1	OK
New value:	Cancel

# 17.6.4 Refreshing the IO window

The IO window contents can be forcibly refreshed. Selecting Refresh from the pop-up menu of the IO window.

### 17.6.5 Disabling refresh of the IO window

Automatic refresh of the **IO** window, which is performed when user program execution stops and in other cases, can be disabled.

Select Lock Refresh from the pop-up menu of the IO window.

The contents of the **IO** window are grayed-out but the text color of the I/O register for which the value has been changed remains.

While Lock Refresh is active, you cannot modify the I/O register settings expanded on the All Register and Selected Register tabbed panes of the IO window.



# 17.6.6 Selecting the I/O register(s) to view

You can select the I/O register(s) to be shown on the Select Register tabbed pane of the IO window.

### To select the I/O register(s) to view

- All Register tab
  - When adding a single I/O register to the **Select Register** tabbed pane
    - 1. Select an I/O register and select **Add to Selected Register tab** from the pop-up menu opened by rightclicking within the window.
  - o When adding two or more I/O registers to the Select Register tabbed pane at one time

 Select I/O Register from the pop-up menu opened by right-clicking within the window. The Select IO Register dialog box opens. The figure given below is an example of the Select I/O Register dialog box for the SH-4 simulator debugger. Each of the items in the Available Registers list has a checkbox that is unselected by default. Clicking on '+' expands the collapsed item while clicking on '-' collapses the expanded item. If you select the checkbox for an item with '+' or '-', the checkboxes for all of its elements are also selected. Similarly, if you deselect the checkbox for an item, all of its elements are also deselected. Clicking the Check All button selects all checkboxes, while clicking the Uncheck All button deselects all checkboxes.

Select IO Register	? ×
Available Begisters:	<u>Q</u> K <u>C</u> ancel Check <u>A</u> ll <u>U</u> ncheck All

- 2. Select the checkboxes for the I/O register(s) you wish to view.
- 3. Clicking OK closes the **Select IO Register** dialog box.



# • Select Register tab

- Select I/O Register from the pop-up menu opened by right-clicking within the window. The Select IO Register dialog box opens. Each of the items in the Available Registers list has a checkbox that is unselected by default.
- 2. Select the checkboxes for the I/O register(s) you wish to view.
- 3. Clicking OK closes the **Select IO Register** dialog box.

The selected I/O registers are shown on the Select IO Register tabbed pane of the IO window.

10			×
🔄 🔀 民 👭 🔐 📧			
Name	Address	Value	Access
🖃 🚍 Memory Management Unit			
🗄 🔄 РТЕН	FF000000	00000000	
ASID		0	
All Register Selected Register			

### Note:

This function cannot be used during execution of the user program.

### 17.6.7 Loading an I/O file

I/O files can be manually loaded to the IO window.

### To load an I/O file

- 1. Right-click in the **IO** window to display a pop-up menu.
- 2. Select Load IO File. The Set I/O File dialog box opens.
- 3. Specify an I/O file you wish to load in the I/O File field. To insert a placeholder in the I/O File field, click the Placeholder button () and select a placeholder from the pop-up menu. To browse a file, click the Browse button.
- 4. If the **Save path name of the I/O file** checkbox is not selected, the path name in the **I/O File** field will not be saved when you save the session. The next time this session is selected, the contents to be loaded to the **IO** window depend on the debugger. When the I/O file has been automatically downloaded, ensure that this checkbox is not selected. The I/O file dependent on the selected CPU is chosen by the debugger and automatically loaded.

If the **Save path name of the I/O file** checkbox is selected, the path name in the **I/O File** field will be saved when you save the session. The next time this session is selected, the most recently loaded I/O file will be



loaded to the **IO** window. This checkbox is not selected by default.

5. Click OK.

If the register names in the loaded I/O file and those shown on the **Select I/O Register** tabbed pane of the **IO** window do not match, an  $\bigotimes$  icon appears on the items as shown below.

10			×
🔄 🔀 🖪 🛤 🔐 📧			
Name	Address	Value	Access
🖃 🔀 Memory Management Unit			
🖻 🔀 РТЕН			
ASID			
All Register Selected Register	1		

See Reference 6, I/O File Format, for more information about an I/O file format.

### 17.6.8 Printing the currently displayed contents

The contents currently displayed on the selected pane of the **IO** window can be printed in a text file. Select **Print** from the pop-up menu.

### 17.6.9 Saving the currently displayed contents

The contents currently displayed on the selected pane of the **IO** window can be saved in a text file. Select **Save to File** from the pop-up menu.

### 17.6.10 Finding an I/O register

You can search for an I/O register shown in the IO window.

### To find an I/O register

- 1. Right-click within the window to open a pop-up menu.
- 2. Select Find. The Find dialog box opens.



# 17. Debugging Facility

# High-performance Embedded Workshop V.4.09

Find	? ×
C <u>o</u> lumn	
Name	<u>0</u> K
Find: ASID	<u>C</u> ancel
Match whole word	

3. Select the column in which you wish to search for an I/O register.

Address	Address of the I/O register
Name	Name of the I/O register

- 4. In Find, enter the string to be found in the selected column. The characters are not case-sensitive.
- 5. To find the I/O register that completely matches the string entered in Find, select the **Match whole word** checkbox. When this option is not selected, all I/O registers that partially match the entered string will be found.
- 6. Clicking OK starts the search from the first line. When an I/O register is found, this line is highlighted.

To find the next I/O register to match, select Find Next in the pop-up menu.

### 17.6.11 Finding the next

After an I/O register in the **IO** window has been found, you can also use **Find Next** to find the next I/O register that meets the requirement.

# 17.7 Looking at registers

If you are debugging at assembly-language level using the **Disassembly** view, you will probably find it useful to see the contents of the CPU's general registers. You can do this by using the **Register** window.

### 17.7.1 Opening the Register window

The **Register** window displays the register data and flag data. You can change a register/flag value from the window.

To open the **Register** window, choose [**View -> CPU -> Registers**], or click the **Registers** toolbar button (**E1**). The **Register** window opens showing all of the CPU's general registers and values.



# Window configuration

Screenshot of the SH-4 debugger

	Register			×
Column header	Name	Value	Radix	
	RO	00000010	Hex	
	RI	000000 <b>F</b> 0	Hex	
	R2	00040001	Hex	•

- Double-clicking the register display line opens a dialog box, which allows you to change a register value. (Macro Recording)
- The register's contents can be changed by using in-place editing. (Macro Recording)
- You can change a flag value by clicking the button corresponding to the flag. (Macro Recording)
- The right-click menu allows you to change the display radix point and the register bank (Change of the register bank can be selected only when the debugger supports this function.).

# Options

Right-clicking displays a pop-up menu containing available options.

Pop-up	Menu Option	Macro Recording	Function
Radix	Hex	-	Displays in hexadecimal.
	Dec	-	Displays in decimal.
	Oct	-	Displays in octal.
	Bin	-	Displays in binary.
Bank0 *		-	Displays registers of bank 0.
Bank1 *		-	Displays registers of bank 1.
Layout	Radix	-	Switches display or non-display of radix.
	FLAGs	-	Switches display or non-display of flags display area.
	Settings	-	Chooses a register to be displayed.
Edit		•	Changes a register's contents.
Refresh		-	Refreshes the Register window.
Lock Ref	resh	-	Disables refresh of the Register window.
Split		-	Splits up the window display.
Save To	File	-	Saves register contents in a text file.
NT 4			

Note:

\*. Support for this function depends on the debugger.



# 17.7.2 Changing the register display radix

You can change the display radix by register.

To do this, click the mouse right button on the register to be changed and select the display radix from the pop-up menu which is opened.

The followings can be selected.

Hex	Display in hexadecimal.
Dec	Display in decimal.
Oct	Display in octal.
Bin	Display in binary.

# 17.7.3 Switching Register Bank

Immediately after opening the Register window, the register data for the bank corresponding to the value of flag is displayed.

To switch the bank, you can also use the pop-up menu which is displayed by clicking the mouse right button on the register display area in the Register window, or change the value of flag.

(If you change the value of flag, the register bank also changes in response to the value.)

To reference the register data of Bank1, select **Bank1** from the pop-up menu with the Register window active.

The register specific to Bank1 is displayed in the gray background.

# Screenshot of the M16C family debugger

1 BANK -	Register Wind	× wot
Name	Value	Radix 🔺
RO	0000	Hex
R1	0000	Hex
R2	0000	Hex
R3	0000	Hex
AO	0000	Hex
A1	0000	Hex
FB	0000	Hex
USP	0000	Hex
ISP	0000	Hex
PC	0F0000	Hex 🔻
•		
IPL U I	OBSZD	с
0 0 0	0 0 0 0 0	0

To reference the register data of Bank0, select **Bank0** from the pop-up menu with the Register window active. (Through the operation of option **Bank0** and **Bank1**, the value of flag does not change.)

Support for this function depends on the debugger.



# 17.7.4 Setting the layout

To set the layout of the Register Window, choose **Layout** from the register Window pop-up menu. The followings can be selected:

Radix	Switch display or non-display of radix.
FLAGs	Switch display or non-display of flags display area.

When the radix or flag is shown, the option is checked.

# When the radix and flags are displayed

Register			×
Name	Value		Radix 🔺
RO	00000010		Hex
Rl	000000F0		Hex
R2	00040001		Hex 💌
MD RB BL FI	d m q imask	.I3 IMASK	.12
1 1 1 1	. 1 1 1	1	
IMASK.I1 I	MASK.IO S 1	FR SZ PF	DN
1	1 01	. 0 0 0	1
CAUSE.E CA	USE.V CAUSE	.z CAUSE	.0 CAUSE.U
0	0 0	0	0
CAUSE.I EN	ABLE.V ENAF	BLE.Z ENA	BLE.0
0	0	0	0
ENABLE.U E	NABLE.I FLA	AG.V FLAG	.Z FLAG.O
0	0	0 0	0
FLAG.U FLA	.G.I RM1 RMC	)	
0 (	) 0 1		

# 17.7.5 Choosing a register to be displayed

To choose a register to be displayed in the Register window, choose **Settings** from the register pop-up menu. This dialog box is shown in following figure.



Settings			? ×
<u>R</u> egisters: ☑R0	<b>₽</b> R10	MACH	Select all
<ul> <li>✓R1</li> <li>✓R2</li> <li>✓R3</li> <li>✓R4</li> <li>✓R5</li> </ul>	<ul> <li>✓R11</li> <li>✓R12</li> <li>✓R13</li> <li>✓R14</li> <li>✓R15</li> </ul>	IMACL IPR ISSR ISPC IRO_BANK	Unselect all
▼R6 ▼R7 ▼R8 ▼R9	♥PC ♥SR ♥GBR ♥VBR	♥R1_BANK ♥R2_BANK ♥R3_BANK ♥R4_BANK	<u>C</u> ancel
Flags:		<u> </u>	
✓MD ✓RB ✓BL	☑Q ☑IMASK.I3 ☑IMASK.I2	IS I IS IS IS IS IS IS IS IS IS IS IS IS	
IFD IM	☑IMASK.I1☑IMASK.I0	⊠SZ ⊻PR	
•		<u>•</u>	

# 17.7.6 Modifying register contents

To change register contents, follow the procedure below.

Enter a value in the **Value** field of the register you want to change.

Register			x
Name	Value	Radix	<u> </u>
PC	A0000000	Hex	
SR	MDRBBL	Bin	•

# To change a register's contents, open the Set Value dialog by selecting one of the following operations

- Double-click the register you want to change, **OR**
- Select the register you want to change and choose the **Edit** option from the pop-up menu.

PC - Set V	/alue	? ×
Value :	00001044	
Radix :	Hex	•
<u>S</u> et As:	Whole Register	•
	OK Cancel	

You can enter a number or C/C++ expression in the **Value** field.


#### High-performance Embedded Workshop V.4.09

You can choose the radix from the **Radix** drop-down list box.

You can choose whether to modify the entire contents of the register, a masked area, floating bits or flag bits, by selecting an option from the **Set As** drop-down list box (the contents of this list depends on the CPU and selected register).

When you have entered the new number or expression, click the **OK** button or press Enter. The dialog box closes and the new value is written into the register.

#### 17.7.7 Setting the flag value

#### When the flag itself is displayed

Click the button of the flag to be changed. Every time you click the button the flag status (1/0) is switched. If a flag is composed of multiple bits, a dialog is opened, where you can enter a value to be changed.

#### Screenshot of the SH-4 debugger

Register					×
FLAG.Z	FLAG.O	FLAG.U	FLAG.I	RM1	RMO
0	0	0	0	0	1

#### When the flag is displayed in the register

Double-click the FLG line. A dialog opens. Enter the value to be changed.

#### Screenshot of the SH-4 debugger (FLG line (FPSCR))

Register			×
Name	Value		
FPSCR	DN	RMO	
DBR	0000000		-
•			

#### 17.7.8 Splitting up the window display

To vertically divide the Register window display into two, select **Split** from the pop-up menu and move the split-up bar.

To cancel the split-up display, select Split from the pop-up menu again.



#### 17.7.9 Saving register contents

To save register contents in a text file, choose **Save To File** from the register pop-up menu. The **Save As** dialog box opens. Specify the file name.

#### 17.7.10 Refreshing the Register window

The **Register** window contents can be forcibly refreshed. Selecting **Refresh** from the pop-up menu of the **Register** window.

#### 17.7.11 Disabling refresh of the Register window

Automatic refresh of the **Register** window contents, which is performed when user program execution stops and in other cases, can be disabled.

Select Lock Refresh from the pop-up menu of the Register window.

The contents of the Register window are grayed-out.

When two windows are open and this function is enabled in either of the windows, values of registers and flags can be compared.

#### 17.7.12 Using register contents

It can be useful to be able to use the value contained in a CPU register when you are entering a value elsewhere in the High-performance Embedded Workshop, for example when displaying a specified address in the **Disassembly** or **Memory** views. You can do this by specifying the register name prefixed by the '#' or '%' character, e.g.: '#PC' or '%PC'.

The supported prefix depends on the debugger.

#### 17.7.13 Changing text colors

The text color and font of the register or flag values shown in the **Register** window can be customized in the **Format Views** dialog box (in the same manner as the color and font in other windows).

Now it is also possible to customize the color of the changed register or flag values in the **Format Views** dialog box. The default foreground and background colors of the changed values are red and white, respectively. The color of the background will be the same as in cases where the text category is selected.

#### To change the look of the Register window

- 1. Select [Setup -> Format Views]. The Format Views dialog box opens.
- 2. Select the **Register** item in the tree and expand it.



- 3. Select the **Modified** category.
- 4. Modify the **Foreground** color selection on the **Color** tabbed page as desired.
- 5. Click **OK**.

# 17.8 Specifying the radix

If no prefix (e.g. "0x") has been added to a numerical value entered in a dialog box, window, or command line, specify the radix for the value.

#### Notes:

- If the dialog box or window has a field for selecting the radix (e.g. a button or column), the radix selected in this field will be applied.
- The radix in some dialog boxes, windows, and commands may be fixed.

To specify the radix, select a submenu from [Setup -> Radix] or click on a button on the Debug toolbar.

[Setup -> Radix] Submenu	Debug Toolbar	Description
Hex	<u>16</u>	Sets the radix to hexadecimal. (default)
Decimal	10	Sets the radix to decimal.
Octal	8	Sets the radix to octal.
Binary	2	Sets the radix to binary.

# 17.9 Resetting the target MCU

To reset the target MCU, click the **Reset CPU** toolbar button (1), or select [**Debug -> Reset CPU**].

Resetting the target MCU initializes the on-chip I/O registers and makes the program counter jump to the address set in the reset vector.

# 17.10 Setting PC to the address at cursor

To changes the value of the PC to the address at the row of the text cursor, click the **Set PC to cursor** toolbar button  $(\mathbf{I}_{PC})$ , or select [**Debug -> Set PC to Cursor**].

# 17.11 Initializing the debugger

#### Select [**Debug -> Initialize**].

It will close down any open child windows and shut down the link to the debugger. If this is successful, an attempt to reestablish the link to the debugger will be made.



# 17.12 Connecting/disconnecting the debugger

#### To connect the debugger

Select one of the following operations:

- Click the **Connect** toolbar button (**P**), **OR**
- Select [**Debug->Connect**].

#### To disconnect the debugger

Select one of the following operations:

- Select [**Debug->Disconnect**].

Support for this function depends on the debugger.

# 17.13 Executing your program

This section describes how you can execute your program's code. You will learn how to do this by either running your program continuously or stepping single or multiple instructions at a time.

#### 17.13.1 Continuing run

When your program is stopped, the High-performance Embedded Workshop will display a yellow arrow ( $\Rightarrow$ ) in the gutter of the line in the source and disassembly views that correspond to the CPU's current Program Counter (PC) address value. This will be the next instruction to be executed if you perform a step or continue running.

#### To continue running from the current PC address

- Click the **Go** toolbar button ( $\blacksquare$ ), **OR**
- Choose [**Debug -> Go**].

To continue running from a specified address which is not the stop address, change the PC value in one of the following ways, and click the **Go** toolbar button ( $[] \downarrow$ ) or choose [**Debug** -> **Go**].

- Change the PC value in the **Register** window.
- Place the text cursor (not the mouse cursor) to a target line in the editor or disassembly window, and choose **Set PC Here** from the pop-up menu.



#### 17.13.2 Running from reset

To reset your user system and run your program from the Reset Vector address, click the **Reset Go** toolbar button (E) or choose [**Debug->Reset Go**]. The program will run until it hits a breakpoint or a break condition is met. You can stop the program manually at any time by clicking the **Halt** toolbar button () or by choosing [**Debug -> Halt Program**].

#### Note:

The program will start running from whatever address is stored in the Reset Vector location. Therefore it is important to make sure that this location contains the address of your startup code.

#### 17.13.3 Running program, ignoring any breakpoints

Sometimes when you are debugging your application you may need to start running continuously but have breakpoints configured. Instead of disabling all the breakpoints, you can quickly execute the code using the **Free Go** feature.

#### To run from the current PC address, but ignore any (both software and hardware) breakpoints

- Click the **Free Go** toolbar button (**E**), **OR**
- Choose [**Debug -> Free Go**].

#### Note:

Support for this function depends on the debugger.

#### 17.13.4 Running to cursor

Sometimes as you are going through your application you may only want to run a small section of code, that would require many single steps to execute. In this case it would be useful to be able to run to a particular point. You can do this by using the **Go To Cursor** feature.

#### How to use the Go To Cursor feature

- 1. Make sure that an editor or disassembly view is open, showing the address at which you wish to stop.
- 2. Position the cursor on the line containing the address at which you wish to stop.
- 3. Choose Go To Cursor from the pop-up menu.

The debugger will run your code from the current PC value until it reaches the address indicated by the cursor's position.



#### Notes:

- If your program never executes the code at this address, the program will not stop. If this happens, you can stop code execution by pressing ESC, choosing [Debug -> Halt Program], or clicking the Halt toolbar button (1990).
- The **Go To Cursor** feature requires a temporary breakpoint if you have already used all those available then the feature will not work.

#### 17.13.5 Running from a specified address

The **Run Program** dialog box allows the user to run the program from any address. Choose [**Debug -> Run**] to open the **Run Program** dialog box.

Run Program	? ×
Program counter:	<u>G</u> o
	<u>R</u> eset go
Temporary PC breakpoints:	
	Cancel

The following execution conditions can be specified in this dialog box:

Program counterInstruction address to start execution. The initial value is the current PC value.Temporary PC BreakpointsA temporary PC breakpoint. When execution started by this dialog box stops, this<br/>breakpoint is cleared.

Clicking the **Go** button starts execution according to the settings. Clicking the **Reset Go** button starts execution from the reset vector. Clicking the Cancel button closes this dialog box without executing instructions.

#### Note:

The **Temporary PC Breakpoints** feature requires a temporary breakpoint - if you have already used all those available then the feature will not work.

#### 17.13.6 Continuing execution to a main function at a reset

The High-performance Embedded Workshop has a function to continue execution of the program at a reset until it reaches the top of a main function. Then the High-performance Embedded Workshop sets a temporary software breakpoint at the main function. When execution of the program is stopped, this breakpoint is deleted.

#### To continue execution of the program to the top of a main function at a reset

1. Select [Setup -> Options] to open the Options dialog box.



- 2. Select the **Debug** tab.
- 3. Select the Run to main function on reset checkbox. By default, this box is not selected.
- 4. Reset the target microcomputer. Select one of the following operations:
  - Click **Reset CPU** toolbar button (**b**), **OR**
  - Select [**Debug -> Reset CPU**].

#### Note:

The **Run to main function on reset** feature requires a temporary breakpoint - if you have already used all those available then the feature will not work.

#### 17.13.7 Single step

When you are debugging your code it is very useful to be able to step a single line or instruction at a time and examine the effect of that instruction on the system. In the source view, a step operation will step a single source line. In the disassembly view, a step operation will step a single assembly-language instruction. If the instruction calls another function or subroutine, you have the option to either step into or step over the function. If the instruction does not perform a call, then either option will cause the debugger to execute the instruction and stop at the next instruction.

If you choose to step into the function, the debugger will execute the call and stop at the first line or instruction of the function.

#### To step into the function

- Click the **Step In** toolbar button (**?**), **OR**
- Select [Debug -> Step In].

If you choose to step over the function the debugger will execute the call and all of the code in the function (and any function calls that function may make) and stop at the next line or instruction of the calling function.

#### To step over the function

- Click the Step Over toolbar button ( **}**), OR
- Select [**Debug -> Step Over**].

During debugging, there are occasions when you may have entered a function, finished stepping through the instructions that you want to examine and would like to return to the calling function without tediously stepping through all the



#### High-performance Embedded Workshop V.4.09

remaining code in the function. Alternatively you may have stepped into a function by accident, when you meant to step over it and so want to return to the calling function without stepping all the way through the current function. You can do this with the **Step Out** feature.

#### To step out of the current function

- Click the **Step Out** toolbar button ({}, **OR**
- Select [Debug -> Step Out].

#### To choose the step mode to use while stepping

Select [**Debug -> Step Mode**].

Sub-menu	Function	
Auto (default)	Automatically chooses the step mode	
Assembly	Steps through assembly instructions	
Source	Steps through source code	

While performing **Step In** in source mode within the High-performance Embedded Workshop editor, you may wish to keep the source mode display of the library functions (e.g. printf) containing no debugging information. By default, the execution jumps to the **Disassembly** window and steps every assembly-language instruction.

You can select not to actually step in addresses where no debugging information exists even when the **Step In** execution reaches such addresses.

#### To select not to step into addresses where no debugging information exists

- 1. Select [Setup -> Options]. The Options dialog box opens.
- 2. Select the **Debug** tab.
- 3. Select the **Only step in when debug information is available** checkbox. By default, this checkbox is not selected.

#### Note:

Support for this function depends on the debugger.



## 17.13.8 Multiple steps

Sometimes you may find it useful to step through several instructions at a time. You can do this by using the **Step Program** dialog box. The dialog box also provides an automated step with a selectable delay between steps. Open it by choosing [**Debug -> Step**]. The **Step Program** dialog box opens.

Step Program	? ×
Steps:	OK
Delay: 2.5 seconds	Cancel
I Step over calls I Source level step	

Steps	Number of steps to be executed. By default, this value is 1.	
Delay	Delay between steps when the program is automatically stepped.	
	"No Refresh" (Prevents the update of the windows) or 0 to 3 seconds can be selected in 0.5	
	second units. By default, this value is 2.5 seconds.	
Step over calls	Selecting this box steps over function calls. By default, this checkbox is not selected.	
Source level step	Selecting this box steps the program at the source level. By default, this checkbox is selected.	

Click the OK button or press Enter to start stepping.

# 17.14 Stopping your program

This section describes how you can halt execution of your application's code. This section describes how to do this directly by using the **Halt** toolbar button (1990) and by setting breakpoints at specific locations in your code.

#### 17.14.1 Stopping the program by the Halt toolbar button

When your program is running, the **Halt** toolbar button is enabled (100), and when the program has stopped it is disabled (100).

#### To stop the program

- Click on the Halt toolbar button (1990), OR
- Choose [Debug -> Halt Program].

When the program has been stopped, a information including the cause of a stop is displayed in the **Debug** pane of the **Output** window.



#### 17.14.2 Standard breakpoints (PC breakpoints)

When you are trying to debug your program, you will want to be able to stop the program running when it reaches specific points in your code. You can do this by setting a program counter (PC) breakpoint on the line or instruction at which to want the execution to stop. The following instructions will show you how to quickly set and clear simple PC breakpoints.

#### To set a PC breakpoint in the Editor window

- 1. Make sure that the source or disassembly view is open at the place at which you want to set a PC breakpoint.
- 2. Select the **Toggle Breakpoint** pop-up menu option, or press F9, at the line showing the address at which you want the program to stop. You will see a red circle (•) appear in the gutter to indicate that a PC breakpoint has been set.
- 3. It is possible to enable or disable the current breakpoint by selecting [Enable/Disable Breakpoint] from the pop-up menu.
- 4. Now when you run your program and it reaches the address at which you set the PC breakpoint, execution halts with the message 'PC Breakpoint' displayed in the **Debug** pane, and the source or disassembly view is updated with the PC breakpoint line highlighted.

#### Additional information:

When there are two or more instances of a template function, breakpoints can even be set at the addresses corresponding to a single line of source code.

```
template <typename T> T1 tempfunc(T t1)
{
    g_iVal += t1; -> A breakpoint set here is also applied to tempfunc(c) and tempfunc(s).
    return g_iVal;
}
void main(void)
    char c;
    short s;
    ...
    tempfunc(c);
    tempfunc(s);
}
```

When you set a breakpoint, a dialog box appears to ask if you wish to apply this breakpoint to all addresses. To apply the breakpoint to all addresses, select Yes. Otherwise select No.

If you do not wish to open this confirmation dialog box, select the Don't ask this question again checkbox.

#### To open the confirmation dialog box again

- 1. Select [Setup -> Options]. The Options dialog box will be displayed.
- 2. Select the **Confirmation** tab.

- 3. Select the **Set multiple breakpoints** checkbox. This checkbox is selected by default.
- 4. Click OK.

#### Note:

The line or instruction at which you set a PC breakpoint is not actually executed - the program stops just before it is about to execute it. If you choose to **Go** or **Step** after stopping at the PC breakpoint, then the highlighted line will be the next instruction to be executed.

#### To change the PC breakpoint setting by using the Breakpoints dialog box

The breakpoint dialog can be displayed by selecting [Edit -> Source Breakpoints]. It allows you to view the current breakpoints set in the workspace and view the code associated with each one. From this dialog it is also possible to remove one or all breakpoints.

#### To toggle PC breakpoints

It is possible to toggle PC Breakpoints either by double-clicking in the breakpoint (BP) column of the line at which the PC breakpoint is set, or by placing the caret on the line and using the F9 key. The display will cycle through the available standard breakpoint types - a red circle () will be shown in the gutter.

# 17.15 Viewing the current status

To check the configuration and status of the debugger in the **Status** window, choose [**View -> CPU -> Status**], or click the **View Status** button (

The **Status** window has three tabs.

#### Window configuration

Status				×
Item			Status	
Memory	Mode		None	
Target	Device	Configuration	00000000-07FFFFFF	EXT
			20000000-27FFFFFF	EXT
			40000000-47FFFFF	EXT
			C0000000-C0000FFF	EXT
			FFFF8000-FFFFBFFF	RAM
			FFFFFE00-FFFFFFFF	I/O
System	Memory	Resources	00000000-00007FFF	Read/
			06000000-06000FFF	Read/
			07FFF000-07FFFFFF	Read/
			FFFFFE00-FFFFFFFF	Read/
Program	n Name		Memory Loaded Area	a 💽
<b>Ⅰ</b>				•
<b>▲</b> ► \ M	emory 🖉 P	latform \ Events /		



Tab	Description
Memory	Contains information about the current memory status including the memory mapping resources and
	the areas used by the currently loaded object file.
Platform	Contains information about the current status of the debugger, typically including CPU series and
	mode, run status and timing information.
Events	Contains information about the current event (breakpoint) status, including resource information.

# 17.16 Viewing the function call history

The Stack Trace window shows the function call history.

#### 17.16.1 Opening the Stack Trace window

To open the Stack Trace window, choose [View -> Code -> Stack Trace] or click the Stack Trace toolbar button (2010).

#### Window configuration

Stack Trace	
Kind Name	Value
<pre>F func3(short *) P param_3 L local_3 F func2(short *) P param_2 L local_2 F func1(short *) P param_1 L local_1</pre>	<pre>{ 0x00000094 }     0x00003ffa { 0x00003fd8 } (short*)     D'3 ( 0x00003fd4 ) (unsigned long) { 0x00000072 }     0x000003ffa { 0x00003fe4 } (short*)     D'2 { 0x00003fe0 } (unsigned long) { 0x000003e }     0x000003ffa { 0x00003ff0 } (short*)     D'1 { 0x00003fec } (unsigned long)</pre>
F main() L start	{ 0x00000012 } D'103 { 0x00003ffa } (short)

The following items are displayed.

Kind	Indicates the type of the symbol.	
	F: Function	
	P: Function parameter *	
	L: Local variable *	
Name	Indicates the symbol name.	
Value	Indicates the value, address, and type of the symbol.	

#### Note:

\*. Support for this function depends on the debugger.



#### Options

Right-clicking displays a pop-up menu containing available options.

Pop-up Menu Option	Macro Recording	Function
Go to Source	-	Go to the associated source line.
View Setting	-	Specifying the Stack Trace window settings.
Сору	-	Places a copy of the highlighted text into the Windows® clipboard.

#### 17.16.2 Viewing the source program

Select a function and choose **Go to Source** from the pop-up menu to display, then the source program corresponding to the function, which has been selected by opening the editor window, is displayed.

#### 17.16.3 Specifying the view

Choose **View Setting** from the pop-up menu to open the **Stack Trace Setting** dialog box, which allows the user to specify the Stack Trace window settings.

Stack Trace Setting	? ×
<u>N</u> est level (1-64)	
Display symbol <u>P</u> arameter <u>L</u> ocal Variable	
Diaplay Radix © <u>H</u> exadecimal	
O <u>D</u> ecimal O <u>O</u> ctal O <u>B</u> inary	
	<u>C</u> ancel

Nest level	Specifies the level of function call nesting to be displayed in the <b>Stack Trace</b> window.
Display symbol *	Specifies the symbol types to be displayed in addition to functions.
Display Radix *	Specifies the radix for displays in the Stack Trace window.

#### Note:

\*. Support for this function depends on the debugger.



#### 17.16.4 Selecting an encoding format

If the values of variables are characters or strings and they should be displayed as UTF-8 code, select [Setup -> Options] to open the Options dialog box. Click on the Debug tab and select UTF-8 for Encoding Format.

The Watch window now shows the values of variables as UTF-8 code. If Local Code Page (default) has been selected for Encoding Format, the values are displayed as ASCII code.

#### Note:

The default font for display in the Watch window is Courier New, which is not supported by UTF-8. Select another font compatible with UTF-8 in the Format Views dialog box in advance.

# 17.17 Using an external debugger

The High-performance Embedded Workshop can launch an external debugger tool. If you want to use another debugger then you must add it to the **Tools** menu.

The **Debugger** tab of the **Setup Customize** dialog box is where the external debugger related information is configured. You may wish to use an older version of the debugger if certain targets are not currently supported in the new environment. Invoke it by selecting [**Setup -> Customize**] and then selecting the **Debugger** tab.

The first choice to make is which debug tool you would like to use.

Once this has been selected the external debugger must be configured.

Hitachi Debugger Interface (version 4.x or greater)	Configuring the Hitachi Debugging Interface to integrate with High-performance Embedded Workshop
Renesas PD debugger	Configuring the PD debugger to integrate with High- performance Embedded Workshop
Other external debugger	Configuring an external debugger to integrate with High- performance Embedded Workshop
Non selected	Not use the external debugger

Click the Launch External Debugger toolbar button (*P*) to invoke the debugger with the specified session file.

After a build, if the download module has been updated, the High-performance Embedded Workshop will switch back to the debugger to enable immediate debugging. Whilst using an external debugger, double-clicking in any editor window will switch back to the High-performance Embedded Workshop with the source file open at the line that was double-clicked.

# 17.17.1 Configuring the Hitachi Debugging Interface to integrate with High-performance Embedded Workshop

The following details the information required to setup the Hitachi Debugging Interface to integrate with Highperformance Embedded Workshop and launch from the external debugger option in High-performance Embedded Workshop.



External debugger selected:	
Hitachi Debugger Interface (version 4.x or greater) 🔽	
Debugger settings	
HDI location:	
Session file: Browse	
Download module:	

#### To configuring the Hitachi Debugging Interface to integrate with High-performance Embedded Workshop

- 1. Firstly, the location of the debugger executable must be specified. This must be version Hitachi Debugging Interface 4.0 or greater, otherwise its behavior is not guaranteed. This may have been configured by the installation program or a project generation utility.
- 2. The second item of data is the session file. This tells the debugger which session to load when it is launched.
- 3. Finally, the location of the download module is required. This allows the High-performance Embedded Workshop to automatically switch to the debugger when the download module changes after a build.

#### 17.17.2 Configuring the PD debugger to integrate with High-performance Embedded Workshop

The following details the information required to setup the PD debugger to integrate with High-performance Embedded Workshop and launch from the external debugger option in High-performance Embedded Workshop.

External debugger selected:	
Renesas PD debugger	
Debugger settings	
<u>P</u> D debugger location:	
►	<u>B</u> rowse
PD profile <u>fi</u> le location:	
	Bro <u>w</u> se
Command line options:	
Download module:	
	B <u>r</u> owse



#### To configure the PD debugger to integrate with High-performance Embedded Workshop

- 1. Firstly, the location of the debugger executable must be specified. This may have been configured by the installation program or a project generation utility.
- 2. The second item of data is the profile file. This tells the debugger which profile file to load when it is launched. This file stores the debug setup information.
- 3. The third item of data is the command line options. This field allows additional options to be specified which can modify the behavior of the external debugger.
- 4. Finally, the location of the download module is required. This allows the High-performance Embedded Workshop to automatically switch to the debugger when the download module changes after a build.

#### 17.17.3 Configuring an external debugger to integrate with High-performance Embedded Workshop

The following details the information required to setup an external debugger which is not Hitachi Debugging Interface or the PD debugger to integrate with High-performance Embedded Workshop and launch from the external debugger option in High-performance Embedded Workshop.

External debugger selected:
Other external debugger
Debugger settings
External debugger location:
Command line options:
Download module:

#### To configure an external debugger to integrate with High-performance Embedded Workshop

- 1. Firstly, the location of the debugger executable must be specified. This may have been configured by the installation program or a project generation utility.
- 2. The second item of data is the command line options. This field allows additional options to be specified which can modify the behavior of the external debugger.
- 3. Finally, the location of the download module is required. This allows the High-performance Embedded Workshop to automatically switch to the debugger when the download module changes after a build.



# 17.18 Debugging functions dependent on the debugger

The debugging functions listed in this section are not common to all High-performance Embedded Workshop products included in tool packages. (User's manual and help information just include descriptions of these debugging functions as the previous version did.)

For details on the functions available with the debugger in use, refer to the user's manual or help of the emulator or simulator.

#### 17.18.1 Looking at labels

When the user program that includes the debugging information is loaded, the labels are registered at that time.

The Disassembly window shows the first eight characters of each label instead of the corresponding address or as a part of an instruction operand.

#### Note:

When a label value matches an operand, the corresponding instruction operand is replaced by the label. If two or more labels have the same value, the one that comes first in alphabetical order is displayed.

When an address or a value can be entered in the dialog box, the labels can also be used.

#### 17.18.1.1 Listing labels

To see a list of all the labels defined in the current debugger session, choose [View -> Symbol -> Labels] or click the View Labels toolbar button (20).

#### Window configuration



- You can view symbols, sorted either alphabetically (by ASCII code) or by address value, by clicking on the respective column heading.
- You can quickly toggle a software break at the entry point of a function by double clicking in the **BP** (breakpoint) column. Alternatively, right-click to show the pop-up menu and select Break.



#### Options

Right-clicking displays a pop-up menu containing available options.

A basic operation is allocated to the toolbar.

The **Toolbar display** and **Customize toolbar** options are also included in the pop-up menu opened by right-clicking on the toolbar.

Pop-up Menu Option	<b>Toolbar Button</b>	Macro Recording	Function
Add	°_	•	Adds a label.
Edit	20	-	Edits a label.
Delete	×	•	Deletes a label.
Delete All	×	•	Deletes all labels.
Load	<b>2</b>	•	Loads labels from a file.
Save		-	Saves labels into a file.
Save As	1	-	Saves labels into a file.
Find	<i>6</i> 4	-	Finds a label.
Fine Next	A	-	Finds the next label to match.
View Source	Image: A start of the start	-	Views the source corresponding to a label.
Print	9	-	Prints the currently displayed contents.
Toolbar display	-	-	Shows or hides the toolbar.
Customize toolbar	-	-	Customizes toolbar buttons.

#### 17.18.1.2 Adding a label

Choose Add from the pop-up menu and open the Add Label dialog box to add a label:

Add Label			? ×
<u>N</u> ame:		<b>_</b>	ОК
<u>A</u> ddress:	0000000	- 🔎	Cancel

Enter the new label name into the **Name** field and the corresponding value into the **Address** field and press OK. The **Add Label** dialog box closes and the label list is updated to show the new label. When an overloaded function or a class name is entered in the **Address** field, the **Select Function** dialog box opens for you to select a function. For details, see section 17.18.2.3, Supporting duplicate labels.

#### 17.18.1.3 Editing a label

Choose **Edit** from the pop-up menu and open the **Edit Label** dialog box to edit a label:



Edit Label	[head]	? ×
<u>N</u> ame:	head	OK
<u>A</u> ddress:	00FFB420	Cancel

Edit the label name and value as required and then press OK to save the modified version in the label list. The list display is updated to show the new label details. When an overloaded function or a class name is entered in the **Address** field, the **Select Function** dialog box opens for you to select a function. For details, see section 17.18.2.3, Supporting duplicate labels.

# 17.18.1.4 Deleting a label

To delete a label, select the label and choose **Delete** from the pop-up menu. A confirmation message box appears:

Confirma	tion Request	? ×
?	Are you sure you want t	o delete the label '_main'?
~1~	🔲 <u>D</u> on't ask this que	stion again
	Yes	No

If you click OK, the label is removed from the list and the window display is updated. If the message box is not necessary then check the **Don't ask this question again** check box.

#### 17.18.1.5 Deleting all labels

To delete all the labels from the list, choose **Delete All** from the pop-up menu. A confirmation message box appears:



If you click OK, all the labels are removed from the High-performance Embedded Workshop system's symbol table and the list display will be cleared. If the message box is not necessary then check the **Don't ask this question again** check box.

#### 17.18.1.6 Loading labels from a file

A symbol file can be loaded and merged into the High-performance Embedded Workshop's current symbol table. Choose **Load** from the pop-up menu to open the load symbols dialog box:



The dialog box operates like a standard Windows<sup>®</sup> Open file dialog box; select the file and click **Open** to start loading. The standard file extension for symbol files is ".sym".

## 17.18.1.7 Saving labels into a file

Choose **Save As** from the pop-up menu to open the save symbols dialog box. The save symbols dialog box operates like a standard Windows<sup>®</sup> Save As dialog box. Enter the name for the file in the **File name** field and click **Save** to save the High-performance Embedded Workshop's current label list to a symbol file. The standard file extension for symbol files is ".sym".

See Reference 7, Symbol File Format, for symbol file format.

Once a file is specified by the **Save As** menu, the current symbol table can be saved in the same symbol file just by choosing **Save** from the pop-up menu.

#### 17.18.1.8 Searching for a label

Choose **Find** from the pop-up menu to open the **Find Label** dialog box.

Find Label	? ×
Name:	▼ OK
Address:	💌 🗾 Cancel
Match whole word only Match case	

- 1. Enter the label name that you wish to find into the **Name** field or select a label name that you have searched before from the drop-down list box.
- 2. Enter the address of the label that you wish to find into the **Address** field or select an address that you have searched before from the drop-down list box.
- 3. To find the label having the name entered in the Name field, select the **Match whole word only** check box. When this option is not selected, the High-performance Embedded Workshop searches for any labels that partly match the name entered in the Name field (the Address field cannot be used).
- 4. When the label name must be case sensitive, select the **Match case** check box.
- 5. Click the OK button to start the search.

#### Note:

Only the label is stored by 1024 characters of the start, therefore the label name must not overlap mutually in 1024 characters or less.



#### 17.18.1.9 Searching for the next label

Choose Find Next from the pop-up menu to find the next occurrence of the label containing the text that you entered.

#### 17.18.1.10 Viewing the source code corresponding to a label

Select a label and choose **View Source** from the pop-up menu to open the source file containing the address corresponding to the label.

#### 17.18.2 Elf/Dwarf2 support

The High-performance Embedded Workshop supports the Elf/Dwarf2 object file format for debugging applications written in C/C++ and assembly language for Renesas microcomputers.

#### **Key Features:**

- Source level debugging
- C/C++ operators
- C/C99\*/C++ expression (casting, pointers, references, etc.)
- Ambiguous function names
- Overlay memory loading
- Watch locals and user defined
- Stack trace

#### Note:

\*. It is only possible when a compiler that supports C99 specifications is in use.

#### 17.18.2.1 C/C++ operators

The C/C++ language operators are available:

```
+, -, *, /, &, |, ^, ~, !, >>, <<, %, (, ), <, >, <=, >=, !=, &&, ||
Buffer_start + 0x1000
#R1 | B'10001101
((pointer + (2 * increment_size)) & H'FFFF0000) >> D'15
!(flag ^ #ER4)
```



## 17.18.2.2 C/C99/C++ expressions

#### **Expression examples:**

Object.value	Specifies direct reference of a member (C/C++)
p_Object->value	Specifies indirect reference of a member (C/C++)
Class::value	Specifies reference of a member with class (C++)
*value	Specifies a pointer (C/C++)
&value	Specifies a reference (C/C++)
array[0]	Specifies an array (C/C++)
Object.*value	Specifies reference of a member with pointer (C++)
::g_value	Specifies reference of a global variable (C/C++)
Class::function(short)	Specifies a member function (C++)
(struct STR) *value	Specifies a cast operation (C/C++)
[+ -] floating-point constant (+ -)	Specifies a complex or imaginary number. Hexadecimal floating-point constants are
floating-point constant *I	also supported as the floating-point constants in this expression (only when a compiler
	that supports C99 specifications is in use).

#### 17.18.2.3 Supporting duplicate labels

In some languages, for example in C++ overloaded functions, a label may represent more than one address. Just entering the label name is ambiguous, so the High-performance Embedded Workshop will display the **Select Function** dialog box to display overloaded functions and member functions.

Select Function		? ×
Select Function Name	Set Function Name	
Name         Addm           {Sample::change(long *)}         2006           {Sample::Sample()}         2000           {Sample::sort(long *)}         2068	Name	Address
Counter       All Function       Select Function       3       Functions       3	OK	Cancel

Select overloaded functions or member functions in the **Select Function** dialog box. Generally, only one function can be selected at one time (except for setting breakpoints, as multiple functions can be selected). This dialog box has three areas:

Select Function Name	Displays the same-name functions or member functions and their detailed information.		
	Name	Displays the names of the functions	
	Address	Displays the addresses of the functions	
Set Function Name Displays the function to be set and their detailed information.		n to be set and their detailed information.	
	Name	Displays the names of the functions	



	Address	Displays the addresses of the functions	
Counter	Displays the number of the functions having the same name.		
	All Function	Displays the number of same-name functions or member functions.	
	Select Function	Displays the number of functions displayed in the Select Function	
		Name list box.	
	Set Function	Displays the number of functions displayed in the Set Function	
		Name list box.	

#### (1) Selecting a function

Click the function you wish to select in the **Select Function Name** list box, and click the > button. You will see the selected function in the **Set Function Name** list box. To select all functions in the **Select Function Name** list box, click the >> button.

#### (2) Deselecting a function

Click the function you wish to deselect from the **Set Function Name** list box, and click the < button. To deselect all functions, click the << button. The deselected function(s) will be moved from the **Set Function Name** list box back to the **Select Function Name** list box.

#### (3) Setting a function

Click the **OK** button to set the functions displayed in the **Set Function Name** list box. The functions are set and the **Select Function** dialog box closes.

Click the **Cancel** button to close the dialog box without setting the functions.

#### 17.18.2.4 Debugging an overlay program

Programs making use of the **Overlay** function can be debugged. This section explains the settings for using the **Overlay** function.

#### (1) Displaying section group

When the **Overlay** function is used (i.e. when several section groups are assigned to the same address range), the address ranges and section groups are displayed in the **Overlay** dialog box.

Open the **Overlay** dialog box by choosing [**Debug -> Overlay**]. This dialog box has two areas: the **Address** list box and the **Section Name** list box.



Overlay		×
Address:	Section Name:	OK Cancel

The **Address** list box displays the address ranges used by the **Overlay** function. Click to select one of the address ranges in the Address list box.

The Section Name list box displays the section groups assigned to the selected address range.

Overlay		X
Address: 001000-001023	Section Name: Psect01, Psect11 Psect02, Psect12 Psect03, Psect13	OK. Cancel
		1

#### (2) Setting section group

When using the **Overlay** function, the highest-priority section group must be selected in the **Overlay** dialog box, otherwise the High-performance Embedded Workshop will operate incorrectly.

Firstly, click one of the address ranges displayed in the **Address** list box. The section groups assigned to the selected address range will then be displayed in the **Section Name** list box. Click to select the section group with the highest-priority among the displayed section groups.

Overlay		×
Address: 001000-001023 002000-00200B	Section Name: Psect01, Psect11 Psect02, Psect12 Psect03, Psect13	Cancel
		<b>V</b>

After selecting a section group, clicking the OK button stores the priority setting and closes the dialog box. Clicking the Cancel button closes the dialog box without storing the priority setting.



#### Note:

Within the address range used by the **Overlay** function, the debugging information for the section specified in the **Overlay** dialog box is referred to. Therefore, the same section of the currently loaded program must be selected in the **Overlay** dialog box.

#### 17.18.3 Looking at variables

This section describes how you can look at variables in the source program.

If you are debugging ELF/DWARF2 load modules, static member variables can be displayed together with other member variables when structures are expanded in the **Watch** or **Locals** window. However, it will take time to display the variables if the window contains a large amount of debugging information.

#### To select not to display static member variables when structures are expanded in the Watch or Locals window

- 1. Select [Setup -> Options] to open the Options dialog box.
- 2. Select the **Confirmation** tab.
- 3. Select the **Display static member on the variable expansion in the Watch/Locals** checkbox. By default, this checkbox is not selected.
- 4. A confirmation dialog box appears. Select No.



The static member variables are hidden even when structures are expanded and this will save time taken for display. (Even after you have selected **No**, it is still possible to view the static member variables by adding them to the **Watch** window.)

#### 17.18.3.1 Tooltip watch

Use this function to know the value of a variable defined in the source program. Open the editor window or disassembly window (in source mode) to view the source program and rest the mouse cursor over the variable name that you want to examine. A tooltip (pop-up window) will appear showing the watch information.

#### To use Tooltip watch

1. Select [Setup -> Options]. The Options dialog box opens.



- 2. Select the **Editor** tab.
- 3. Check the **Enable tooltip watch** check box.
- 4. Click OK.

Enable tooltip watch

#### To view a tooltip watch on the editor window or disassembly window (in source mode)

- 1. Open the editor window or disassembly window (in source mode) showing the variable that you want to examine.
- 2. Rest the mouse cursor over the variable name that you want to examine. A tooltip will appear near the variable containing basic watch information for that variable.

S.,	Source
	sort(a);
	printf("*** Sorting results
	<pre>for( i=0; i&lt;10; i++ ){</pre>
	printf("a[%d]=%ld\n",i,a
	}
	min = <u>a[0];</u>
	$\max = \{(long[10]) \{70000000\}\}$
	$\min = 0;$
	max = 0;
	change(a);
٢	min = a[9];
	max = a[0];

#### 17.18.3.2 Instant watch

Display the source file containing the variable that you want to examine on the editor window or disassembly window (in source mode).

Rest the mouse cursor over the variable name that you want to examine and choose **Instant Watch** from the pop-up menu; the **Instant Watch** dialog box will appear and display the variable at the cursor location.

Insta	ant Watch	? ×
Ŧ	- a {73FFFFB0}(long[10])	<u>Close</u>

'+' shown to the left of the variable name indicates that the information may be expanded by clicking on the variable name, and '-' indicates that the information may be collapsed. Clicking **Add** registers the variable in the **Watch** window. Clicking **Close** closes the window without registering the variable in the **Watch** window.



#### 17.18.3.3 Watch window

The High-performance Embedded Workshop allows you to open **Watch** windows, which contain a list of variables and their values.

#### (1) **Opening the Watch window**

To open a Watch window, choose [View -> Symbol -> Watch] or click on the Watch toolbar button (🔛) if it is visible. A Watch window opens.

#### Window configuration

Toolbar — 💻	ar — 🔄 R Ŗ 🗖 🅶 🌠 / 🏟 🗙 💣 📌 🔛 🐼 🧐				
Column —	Name	Value	Address	Туре	Scope
header	🖃 Ŗ Watch_Item_1		{ FFF805C0 }	(volatile long[10])	[Current Scope]
		H'00000000	{ FFF805C0 }	(volatile long)	
		H'00000000	{ FFF805C4 }	(volatile long)	
		H'00000000	{ FFF805C8 }	(volatile long)	
		H'00000000	{ FFF805CC }	(volatile long)	
		H'00000000	{ FFF805D0 }	(volatile long)	
		H'00000000	{ FFF805D4 }	(volatile long)	
		H'00000000	{ FFF805D8 }	(volatile long)	
		Н'00000000	{ FFF805DC }	(volatile long)	
		H'00000000	{ FFF805E0 }	(volatile long)	
		Н'00000000	{ FFF805E4 }	(volatile long)	
		H'00000000	{ FFF805E8 }	(volatile long)	[Current Scope]
	•				•
	Watch1 / Watch2 / Watch2	ch3 ∖ Watch4 /			

- This window shows data only when the debugging information available in the load module (\*.abs) includes the information on the C/C++ source program. No data is shown in the window if information on the source program is excluded from the debugging information due to optimization by the compiler. No variables declared as macros also can be displayed.
- Each variable can be dragged from the **Editor** or **Disassembly** window and dropped into the Watch window. You can also use the **Add Watch** menu option in the **Editor** or **Disassembly** window in the source mode to add a variable.
- The order of the variables can be changed by dragging and dropping items onto desired positions.
- If you drag an item in the **Watch** window and drop it into the **Address** column of the **Memory** window, the address of the variable will be the first address shown in the **Memory** window.
- The **R** mark indicates that the value of the variable can be updated while the user program is running. When a value is marked with a bold **R**, this value will be updated in real-time.
- Double-clicking within the **Name** column opens the **Edit Name** dialog box, in which you can change the name of the variable. Changing of the name can be recorded in a macro (Macro Recording).
- Double-clicking within the **Value** column opens the **Edit Value** dialog box, in which you can change the value of the variable. Changing of the value can be recorded in a macro (Macro Recording).



- In the Value column, the values of variables can be changed by in-place editing (Macro Recording).
- Clicking on the Name or Address column header sorts the variables by name or address.
- You can also watch variables in the selected scope. The following choices are available: **Current Scope** (valid variables within the current scope), **Global** (global variables), and scope specific to each file (static variables within the file scope).
- Double-clicking on the **Scope** column opens the **Set Scope** dialog box.
- In the **Scope** column, the scope can be changed by in-place editing.

#### Options

Right-clicking displays a pop-up menu containing available options.

A basic operation is allocated to the toolbar.

The **Toolbar display** and **Customize toolbar** options are also included in the pop-up menu opened by right-clicking on the toolbar.

Pop-up Menu	Option	Toolbar Button	Macro Recording	Description
Enable Auto U	pdate	R	•	Marks the selected variable with a bold <b>R</b> and updates the variable in real-time.
Enable Auto U	pdate All	R	•	Marks all variables with bold ${f R}$ s and updates all variables in real-time.
Disable Auto L	Jpdate	R	•	Marks the selected variable with an outlined <b>R</b> and cancels real- time update.
Disable Auto L	Jpdate All	R	•	Marks all variables with outlined $\mathbf{R}$ s and cancels real-time update.
Update Interva	ll *		-	Sets an update interval.
Record Update Value	Start Recording	R.	•	Opens the <b>Recording Settings</b> dialog box and starts recording the history of updating of values.
	Stop Recording	R	•	Stops recording the history of updating of values.
Add Watch			•	Launches the Add Watch dialog box, allowing the user to enter a variable or expression to be watched.
Edit Name		њ <mark>/</mark>	•	Launches the Edit Name dialog box, allowing the user to change the variable's name.
Edit Value		Ø	•	Launches the Edit Value dialog box, allowing the user to change the variable's value.
Set Scope		<b>6</b>	-	Sets the scope for the selected symbol.
Delete		×	•	Removes the variable indicated by the text cursor from the Watch window.
Delete All		X	•	Removes all the variables from the Watch window.
Radix	Hexadecimal	<u>16</u>	•	Displays in hexadecimal.
	Decimal	10	•	Displays in decimal.
	Octal	8	•	Displays in octal.
	Binary	2	•	Displays in binary.
	Set Initial	<sup>IS2D</sup>	-	Sets the default radix for newly added variables.



#### High-performance Embedded Workshop V.4.09

Pop-up Menu Option	Toolbar Macro	Description
	Button Record	ing
Сору	<b>e</b> gi -	Places a copy of the highlighted text into the Windows® clipboard.
Save As	-	Saves the currently displayed contents.
Go To Memory	<b>3</b>	Opens a Memory window for the address.
Toolbar display		Shows or hides the toolbar.
Customize toolbar		Customizes toolbar buttons.

#### Note:

\*. Available only when the debugger supports this function.

#### (2) Adding a variable

You can add variables to the Watch window.

#### To add a variable to a Watch window

- Editor or Disassembly window (in source mode)
  - Right-click on a variable in the window and select **Add Watch** from the pop-up menu.
  - Right-click on a variable in the window and select **Instant Watch** from the pop-up menu. Then click on the **Add** button in the **Instant Watch** dialog box (this is the instant watch feature).
  - Drag and drop a variable into the Watch window.

The default scope applies to the added variable but the selection depends on whether the program is running or not.

State of the Program	Scope
Stopped	[Current Scope]
Running	[Global]

- Watch window
  - 1. Open the **Add Watch** dialog box in either of the following ways.
    - Select the window and press the **Insert** key.
    - o Right-click within the window and select Add Watch from the pop-up menu.



Add Watch	? ×
Variable or expression:	OK
	Cancel
Scope:	
[Current Scope]	Detail
[Current Scope]	
[Global] dbsct.c:watchTest4.abs intprg.c:watchTest4.abs lowIvI.src:watchTest4.abs lowsrc.c:watchTest4.abs resetprg.c:watchTest4.abs sbrk.c:watchTest4.abs vecttbl.c:watchTest4.abs watchTest4.c:watchTest4.abs	

 Enter the name of the variable that you wish to watch in the Variable or expression field. The element number for an array entered in the Variable or expression field can be binary, octal, decimal, or hexadecimal.

Note, however, that there are the following rules on the prefix.

Element Number of the Array			Rule					
Binary			The pre	fix must be	: "Ol	o" or "0B".		
Octal			The pre	fix must be	: "0'			
Decimal			No prefi	xes are rec	quir	ed.		
Hexadecimal			The pre	fix must be	: "0	<" or "0X".		
l	ୟି a[	Ob00011110]	н'0000	0202	{	000011A4	}	(long)
G	R a[	ОВООО11110]	н'0000	0202	{	000011A4	}	(long)
G	R a[	036]	н'0000	0202	{	000011A4	}	(long)
6	ୟ a[	30]	н'0000	0202	{	000011A4	}	(long)
G	ୟ a[	Ox1e]	н'0000	0202	{	000011A4	}	(long)
G	R a[	OX1E]	н'0000	0202	{	000011A4	}	(long)

 Select the scope for the variable from the Scope drop-down list. Even when there are two or more variables with the same name, scope can be separately specified to distinguish them. The available types of scope are listed below.

Scope	Description
[Current Scope]	This scope covers all variables that can be watched from the address
	indicated by the program counter.
	This scope is not displayed while the program is running.
[Global]	This scope covers global variables in all load modules. Global variables
	declared as static cannot be watched.
File scope name: Load module name	This scope covers static variables within the specified file scope. Static variables declared within functions and global variables watched in the Global scope cannot be watched.

The default scope for newly added variables will be as follows depending on whether the program is running or not.

State of the Program	Scope
Stopped	[Current Scope]
Running	[Global]

4. Clicking on the **Detail** button opens the **Set Scope** dialog box, in which scope for newly added variables can be selected in the same way as in the **Scope** drop-down list of the **Add Watch** dialog box. Before selecting a desired file scope name in the **Set Scope** dialog box, you can also check full paths to source files or filter file scope names by the string entered in the **Filter** box. The full file paths, however, may not be displayable for some toolchains.

Variables that you have set in the Watch window are saved in the session file.

#### Note:

If the variable that you have added is a local variable that is not currently in scope, the High-performance Embedded Workshop will add it to the Watch window but its value will be 'Not available now'.

#### (3) Editing the name of a variable

You can change the name of a variable. Even if variables can be expanded (e.g. structures or pointers), however, the names of their sub items are not changeable.

#### To edit the name of a variable

Open the **Edit Name** dialog box in either of the following ways.

- Double-click within the Name column.
- Right-click on the line of a variable and select Edit Name from the pop-up menu.



Name	Value	Address	Type		Scope
日 Ŗ Watch_Item_1		{ FFF805C0 }	(volatile	long[10])	[Current Scope]
	H. 0000000	/ FFF90570 \	/wolatile		
	Edit Name		? ×	long)	
	0.11			long)	
	⊻ariable or exp	pression:	<u>0</u> K	long)	
	Watch_Item_1	1	<u>C</u> ancel	long)	
				long)	
	H'00000000	{ FFF805D8 }	(volatile	long)	
	H'00000000	{ FFF805DC }	(volatile	long)	
	H'00000000	{ FFF805E0 }	(volatile	long)	
	Н'00000000	{ FFF805E4 }	(volatile	long)	
🛛 Watch_Item_2[0]	H'00000000	{ FFF805E8 }	(volatile	long)	[Current Scope]

Enter the new name in the Value or expression field and click OK.

Edit Name	? ×
⊻ariable or expression:	<u>0</u> K
Watch_Item_1[1]	<u>C</u> ancel
Watch_Item_1[1]	<u>C</u> ancel

The **Watch** window is updated to show the new name.

×	R R 🗗 🖬 🖊 🎁 🗶 🍠 📌 🔛 😳						
	Name	Value	Address	Туре	Scope		
	R Watch Item_1[1]	H'00000000	{ FFF805C4 }	(volatile long)	[Current Scope]		
	Watch1 (Watch2 ) Wa	tch3 ) Westch4 /					
Ш		COD A watche 1					

# Note:

Names of variables cannot be changed in the following cases.

• The user program is running.



• Two or more variables are selected.

#### (4) Expanding a variable

If a variable is a pointer, array, or structure, then you will see a plus sign (+) expansion indicator to left of its name, this means that you can expand the variable. To expand a variable, click on it. The item expands to show the elements (in the case of structures and arrays) or data value (in the case of pointers) indented by one tab stop, and the plus sign changes to a minus sign (-). If the elements of the variable also contain pointers, structures, or arrays then they will also have expansion indicators next to them.

To collapse an expanded variable, click on the item again. The item's elements will collapse back to the single item and the minus sign changes back to a plus sign.

The variable can be expanded for the number of levels when a numerical key from 1 to 9 is pressed while a variable is selected.

#### (5) Editing the value of a variable

You can change the value of a variable.

#### To edit the value of a variable

Select either of the following ways.

• In-place edit in the Value column.

≝ R R 🗇 🗹 / 翰 X 🛃 🕫 🖓 🧐						
Name	Value	Address	Туре	Scope 🔺		
🗆 🖳 Watch_Item_1		{ FFF805C0 }	(volatile lo	[Current Scope]		
	н'оооооооо	{ FFF805C0 }	(volatile long)			
	H'00000000	{ FFF805C4 }	(volatile long)	-		
Watch1 🗸 Watch2 👌 Wat	ch3 👌 Watch4 /					

- Open the **Edit Value** dialog box in one of the following ways.
  - o Double-click within the Value column.
  - Press the **Enter** key on the line of a variable.
  - Right-click on the line of a variable and select Edit Value from the pop-up menu.



#### High-performance Embedded Workshop V.4.09

# 17. Debugging Facility

	? ×
[0]	
H'0000000	ОК
H'00000000	<u> </u>
	H'00000000

Enter the new value in the New Value field and click OK.

The Watch window is updated to show the new value.

#### (6) Setting the scope of variables

The Watch window has the Scope column that shows the scope for registered variables.

The scope for sub items is the same as that for their root item and is not shown in the window.

In the **Scope** column, you can select the scope for each of the variables that have been registered. Even when there are two or more variables with the same name, scope can be separately specified to distinguish them.

The available types of scope are listed below.

Scope	Description
[Current Scope]	This scope covers all variables that can be watched from the address indicated by the program counter.
	This scope is not displayed while the program is running.
[Global]	This scope covers global variables in all load modules. Global variables
	declared as static cannot be watched.
File scope name: Load module name	This scope covers static variables within the specified file scope. Static variables declared within functions and global variables watched in the Global scope cannot be watched.

#### To set scope of a variable

Select either of the following ways.

• In-place edit in the Scope column. Click on the button on the right edge of the list box and select scope from the list.



RR 🗗 🖬 🖊 🎁 🗙 🍠 📌 🔛 🐼							
Name	Value	Address	Туре	Scope			
		{ FFF805C0 }	(volatile lo	[Current Scope]			
R Watch_Item_2[0]	H'00000000	{ FFF805E8 }	(volatile long)	[Current Scope]			
Watch1 Watch2 Watch2	ch3 à Watch4 ∕			[Current Scope] [Global] dbsct.c:watchTest4.abs intprg.c:watchTest4.abs lowlv1.src:watchTest4.abs lowsrc.c:watchTest4.abs resetprg.c:watchTest4.abs sbrk.c:watchTest4.abs vecttb1.c:watchTest4.abs watchTest4.c:watchTest4.			

- Open the **Set Scope** dialog box in either of the following ways.
  - o Double-click within the **Scope** column.
  - Right-click on the line of a variable and select **Set Scope** from the pop-up menu.

Set Scope ?	×
Expression: Watch_Item_2[0]	1
, <u>S</u> cope:	
[Current Scope] [Global]	
watchTest4.abs	l
File Scope	1
<u>F</u> ilter:	
dbsct.c(C:\Workspace\watchTest4\watch	
intprg.src(C:\Workspace\watchTest4\watc lowlvl.src(C:\Workspace\watchTest4\watc	
lowsrc.c(C:\Workspace\watchTest4\watch resetprg.c(C:\Workspace\watchTest4\watc	
sbrk.c(C:\Workspace\watchTest4\watchT vecttbl.src(C:\Workspace\watchTest4\wat	
whandler.srcfC:\\Workspace\watchTest4\w	
<u> </u>	
	11

1. Select scope for the variable shown in the **Expression** field.

2. If you have selected a load module name in **Scope**, select the corresponding file scope from the list in the **File Scope** section. You can also filter scope names by the string (not case-sensitive) entered in the **Filter** box before selecting a desired scope name.

#### Note:

If the scope for a variable selected in the Watch window does not match the download module, the **Scope** column on the line of the variable is highlighted. Set the correct scope for the variable.

#### (7) Deleting a variable

To delete a variable, select it from the Watch view and choose **Delete** from the pop-up menu. The variable is deleted and the Watch view is updated.

To delete all variables, choose **Delete All** from the pop-up menu. The all variable s are deleted and the Watch view is updated.

#### (8) Modifying the radix

The radix for the selected variable display can be modified by choosing **Radix** from the pop-up menu.

The followings can be selected.

Hexadecimal	Display in hexadecimal. (Default)
Decimal	Display in decimal.
Octal	Display in octal.
Binary	Display in binary.

#### (9) Specifying the default radix

If you select a variable and click on [Radix -> Set Initial], the **Initial Radix** dialog box opens.



Select a radix from the list box. This will be applied as the default radix for newly added variables.

#### (10) Saving the Watch window contents in a file

To save the contents of the Watch window, choose **Save As** from the pop-up menu; the Save As dialog box opens. It allows the user to specify the name of a file and to save the contents of the Watch window in the file. If the **Append** 


check box is selected, the window contents are appended to the existing file, and if it is not selected, the existing file is overwritten.

## (11) Opening a Memory window

The contents of the memory area to which the selected variable is assigned can be displayed in the **Memory** window. Choosing **Go To Memory** from the pop-up menu opens the **Memory** window.

If you drag an item in the **Watch** window and drop it into the **Address** column of the **Memory** window, the address of the variable will be the first address shown in the **Memory** window.

R Ŗ 📑 🎫	🖊 🏥 🗙 🍠 📌 🛛 🐙 Value 🔋 Address	Type	Scope		Memory [SIM_10]
		(long[10])			■ ■ ■ === 16 10 ±0 8 2 dbc あ あ あ み
R [0]	H'11223344 { 7F000000	} (long)			Address +0 +4 +8 +C ASCII
	H'00000000 { 7F000004	} (long)		7	000000000 D0046002 40094001 D103301C 6002402B`.@ 0000x410 00090000 FFFFFD4 00000D54 00000000
	H'00000000 { 7F000008	} (long)			0000 0000000 0000000 0000000 0000000
	H'00000000 { 7F00000C	} (long)			I I I
	H'00000000 { 7F000010	} (long)		L -	
	H'00000000 { 7F000014	} (long)			Ϋ́
	H'00000000 { 7F000018	} (long)			
	H'00000000 { 7F00001C	} (long)			Memory [_Watch_Item_1]
	H'00000000 { 7F000020	} (long)			
	H'00000000 { 7F000024	} (long)			<u>・・・・・・・・・・・・・・・・・・・・・・・・・・・・・・・・・・・・</u>
🕞 🕨 🕹 🕹 🕹	atch2 👌 Watch3 👌 Watch4 🖊				Address +0 +4 +8 +C ASCII 7F000000 11223344 00000000 00000000 00000000 ."3D.
	n I				7F000010 00000000 00000000 00000000 000000

If two or more variables with the same name are assigned to different scope, the **Memory** window shows the memory area that can be watched from the address indicated by the program counter.

## (12) Setting real-time update

The **R** mark shown to the left of each variable indicates whether the variable is updated in real-time.

A pop-up menu containing the following options is available in the **Watch** window:

Enable Auto Update	Marks the selected variable with a bold $\mathbf{R}$ and updates the variable in real-time.
Enable Auto Update All	Marks all variables with bold $\mathbf{R}$ s and updates all variables in real-time.
Disable Auto Update	Marks the selected variable with an outlined $\mathbf{R}$ and cancels real-time update.
Disable Auto Update All	Marks all variables with outlined $\mathbf{R}$ s and cancels real-time update.

The values of variables for which real-time update has been enabled can be modified during execution of the program.

#### (13) Setting an update interval

The interval to update the display contents of the Watch window during execution of the user program can be changed.

Support for this function depends on the debugger.



## To set an update interval

- 1. Right-click within the window to open a pop-up menu.
- 2. Select Update Interval. The Update Interval Setting dialog box opens.

Update Interval Setting	? ×
Interval (10 - 10000ms	sec : 10msec unit):
100	msec
OK	Cancel

- 3. Specify an update interval in 10-ms units for Interval. The default value is 100 ms. The valid range of values depends on the debugger in use.
- 4. Click OK.

The specified update interval is applied to all panes of the Watch window.

The actual update interval may be longer than the specified value depending on the state of execution.

## (14) Recording the history of updating of values

The history of updating of the values of variables, to which automatic update has been applied in the Watch window, can be recorded into a file.

#### To start recording the history of updating of values

- 1. Right-click within the window to open a pop-up menu.
- 2. Select **Record Update Value** and then **Start Recording** from submenus. The **Recording Settings** dialog box opens.

Record Settings	? ×
Recording <u>S</u> heet:	<u>0</u> K
Watch1	<u>C</u> ancel
File <u>n</u> ame:	
	Browse
☐ <u>A</u> ppend	
☐ <u>T</u> imestamp	

3. In the **Recording Settings** dialog box, specify how to record the history.



4. Select a sheet that you wish to record the history in **Recording Sheet**. Only one sheet can be used for recording at one time.

**Recording Sheet** only lists the names of the sheets that include variables to which automatic update has been applied.

- 5. Enter the file name in **File name**. To insert a placeholder in **File name**, click the **Placeholder** button and select **Placeholder** from the pop-up menu. To browse a file, click the **Browse** button.
- 6. To append data to the specified file, select the **Append** checkbox.
- 7. Select the **Timestamp** checkbox to record timestamps (ms) in the file. This checkbox is not selected by default.
- 8. Clicking the OK button starts recording except in the following case:
- The **Append** checkbox is selected but the watched variables recorded in the file selected in **File name** and those to be recorded contained in the sheet selected in **Recording Sheet** do not match or the radixes are different.

Once recording is started, the values of variables are recorded into the file when:

- Values are automatically updated (in real-time) during execution
- Values are updated with the stop of execution
- Values are changed
- Display is refreshed by an operation in the window

## To stop recording the history of automatically updated values

- 1. Right-click within the window to open a pop-up menu.
- 2. Select **Record Update Value** and then **Stop Recording** from submenus.

Recording is automatically stopped when:

- A program is downloaded or unloaded
- Expansion of watched variables to be recorded is collapsed
- The radix for watched variables to be recorded is changed
- A new variable to be watched is added to the sheet selected for recording
- The order of watched variables to be recorded is changed
- Automatic update of watched variables to be recorded is canceled



## (15) Changing text colors

You can customize the font and text coloring for the **Watch** window through the **Format Views** dialog box (the **Text** category in the table below).

It is also possible to use the **Format Views** dialog box to specify the color for memory with a specific attribute or changed values shown in the **Value** column of the **Watch** window.

## To change the look of the Watch window

- 1. Select [Setup -> Format Views]. The Format Views dialog box opens.
- 2. Select the **Watch** item in the tree and expand it.
- 3. Select the category of objects to be customized.

Category	Objects to be Customized	Foreground color (default)	Background color (default)
Text	Text shown in the window	Black	White
Read *	Memory that has been read	Black	Green
Write *	Memory that has been written to	Black	Red
Modified	Variables whose value or position has been changed	Red	White

## Note:

\*. Support for this function depends on the debugger.

- 4. Modify the Foreground and Background color selection on the Color tabbed page as desired.
- 5. Click OK.

## (16) Selecting an encoding format

If the values of variables are characters or strings and they should be displayed as UTF-8 code, select [Setup -> Options] to open the Options dialog box. Click on the Debug tab and select UTF-8 for Encoding Format.

The Watch window now shows the values of variables as UTF-8 code. If Local Code Page (default) has been selected for Encoding Format, the values are displayed as ASCII code.

## Note:

The default font for display in the **Watch** window is **Courier New**, which is not supported by UTF-8. Select another font compatible with UTF-8 in the **Format Views** dialog box in advance.

## (17) Sorting variables

Clicking on the **Name** or **Address** column header sorts the variables by name or address. Every time you click on the column headers, the variables are alternately sorted in ascending and descending order.



High-performance Embedded Workshop V.4.09

×	🕺 R Ŗ 📑 🛃 🧨 👘 🗙 🍠 📌 🔛 🧐						
	Name		Value	Address	Туре	Scope	
	····· R	A	H'00000000	{ FFF80618 }	(volatile long)	[Current Scope]	
	····· R	В	H'00000000	{ FFF80614 }	(volatile long)	[Current Scope]	
	艮	С	H'00000000	{ FFF80610 }	(volatile long)	[Current Scope]	
	R	a	H'00000000	( FFF80624 )	(volatile long)	[Current Scope]	
	····· R	b	H'00000000	{ FFF80620 }	(volatile long)	[Current Scope]	
	····· R	с	H'00000000	{ FFF8061C }	(volatile long)	[Current Scope]	
	Watch1 (Watch2 ) Watch3 ) Watch4 /						

## 17.18.3.4 Locals window

The local variables and their values can be displayed in the Locals window.

## (1) Opening the Locals window

To open the Locals window, choose [View -> Symbol -> Locals] or click the Locals toolbar button (

## Window configuration

	Locals		×
Toolbar —	16	10 8 2 🖷	
Column header	Name	Value	Туре 🔺
rieauei	±…a	{ 73FFFFB0 }	(long[10])
	min	H'00040001 { R2 }	(long) 🚽
	····· max	H'00040001 { R2 }	(long) 🚽

- The local variables and their values can be displayed. As you debug your program, the **Locals** window will be updated. If a local variable is not initialized when defined, then the value in the **Locals** window will be undefined until a value is assigned to the local variable.
- The contents of this window are blank unless the current PC can be associated to a function containing local variables in the source files via the debugging information available in the absolute file (\*.abs).
- Double-clicking the **Value** column opens a dialog box, which allows you to change the value of a local variable. (Macro Recording)
- The value of a local variable can be changed by using in-place editing. (Macro Recording)



## Options

Right-clicking displays a pop-up menu containing available options.

A basic operation is allocated to the toolbar.

The **Toolbar display** and **Customize toolbar** options are also included in the pop-up menu opened by right-clicking on the toolbar.

Edit Value		•		Launches a dialog box to modify the selected variable's value.	
Radix	Hexadecimal	<u>16</u>	-	Displays in hexadecimal.	
	Decimal	10	-	Displays in decimal.	
	Octal	8	-	Displays in octal.	
	Binary	2	-	Displays in binary.	
Сору		<b>₽</b> ₽	-	Places a copy of the highlighted text into the Windows® clipboard.	
Toolba	r display	-	-	Shows or hides the toolbar.	
Customize toolbar		-	-	Customizes toolbar buttons.	

## (2) Editing the value of a local variable

You can change the value of a local variable in the Locals window.

#### To edit the value of a local variable

Select either of the following ways.

• In-place edit in the Value column.

Locals		×
/ <u>16</u> 1082		
Name	Value	Туре 🔺
⊡Watch_Item_1	{ 7F000000 }	(long[10])
[0]	Н'0000000	(long)
[1]	H'00000000 { 7F000004 }	(long) 👻

- Open the **Edit Value** dialog box in one of the following ways.
  - o Double-click within the Value column.
  - o Right-click on the line of a variable and select Edit Value from the pop-up menu.



## High-performance Embedded Workshop V.4.09

## 17. Debugging Facility

Edit Value		? ×
Expression:	[0]	
Current Value:	H'00000000	01
<u>N</u> ew Value:	H'00000000	<u> </u>

Enter the new value in the New Value field and click OK.

The Locals window is updated to show the new value.

## (3) Modifying the radix

The radix for the selected variable display can be modified by choosing **Radix** from the pop-up menu.

The followings can be selected.

Hexadecimal	Display in hexadecimal. (Default)
Decimal	Display in decimal.
Octal	Display in octal.
Binary	Display in binary.

## (4) Selecting an encoding format

If the values of variables are characters or strings and they should be displayed as UTF-8 code, select [Setup -> Options] to open the **Options** dialog box. Click on the **Debug** tab and select **UTF-8** for **Encoding Format**.

The **Locals** window now shows the values of variables as UTF-8 code. If **Local Code Page** (default) has been selected for **Encoding Format**, the values are displayed as ASCII code.

## Note:

The default font for display in the **Locals** window is **Courier New**, which is not supported by UTF-8. Select another font compatible with UTF-8 in the **Format Views** dialog box in advance.



# 18. Synchronized Debugging

The synchronized debugging system is designed to help with debugging multi-core devices.

It allows you to run programs on multiple CPU cores and synchronize common debug operations (Go, step, halt etc.) between the cores. The Synchronized Debug dialog box is the main entry point for configuring and initiating the synchronized debugging system.

Options related to synchronized debugging are saved in "SDO" files. These are always located in the "Sync" directory. This is defined as:

"<Local settings directory>\Renesas\HEW\hew\_001\Sync"

The local settings directory is set by Windows® and is normally:

Windows<sup>®</sup> XP operating system:

"C:\Documents and Settings\<user name>\Local Settings\Application Data"

Window Vista® or Window® 7 operating system:

"C:\Users\<user name>\AppData\Local"

# 18.1 The Synchronized Debug dialog box

Select Synchronized Debugging from the Debug menu to open the Synchronized Debug dialog box.

This can be done when High-performance Embedded Workshop initially starts up or when there is a workspace open.

There is also an option to open the dialog from the High-performance Embedded Workshop Welcome! dialog box.



## 18. Synchronized Debugging

Synchronized Debug							
Default			-	<u>N</u> ew	Import	Save <u>a</u> s	Dejete
Avaijable debugg	jers:						
Debugger ID	Status	Options	Туре	Workspace		Session	A <u>d</u> d
Main Video	Not open Not open	Debug Debug	Multi core Multi core		:e\Demo\Demo :e\Demo\Demo		Modify
VIDEO	Not open	Debug	Multicore	C. WORSPau	evpeniovpenio	. colei	<u>III</u> 955117
							<u>R</u> emove
Synchronization style       Synchronization options         None       Main         All debuggers synchronized       Master debugger:         Operations executed on any debugger will also be executed on all other debuggers.       Operations executed on all other debuggers.         B       Step         Step       Set by platform         Connection       Download modules							
Update the views in all debuggers when platform memory is changed							
Synchronized debugging mode: Parallel							

## 18.1.1 Managing configurations

Current configuration:				
Default	<u>N</u> ew	Import	Save <u>a</u> s	De <u>l</u> ete

## The Current configuration drop list

The drop list at the top of the Synchronized debug dialog will have an entry for each configuration file in the Sync directory. In order to change the settings on the dialog to a previously saved configuration you should select the configuration in the drop list. When you select a configuration the current settings will automatically be saved to the file of the previously selected configuration.

## To create new configurations

To create a new empty configuration use the following method.

1. Click the New button.



## High-performance Embedded Workshop V.4.09



- 2. Enter a name for the configuration. You can only enter characters which are valid in a file name. You are limited to 100 characters.
- 3. Click OK.

A ".SDO" file will automatically be created for the new configuration and stored in the "Sync" directory. The new configuration will be added to the configuration drop list and become selected as the current configuration. All other controls on the dialog will be reset to the default state.

## To save an existing configuration to a different filename

This is the equivalent of a "Save As" file operation. To save the current dialog settings to a new configuration use the following method:

- 1. Click the **Save as** button.
- 2. Enter a new unique name for the configuration file.
- 3. Click Save.

If the configuration was saved to the Sync directory then the configuration will be added to the drop list and selected exactly as it would for a completely new configuration, except that the dialogs controls will not be reset to default. However the file can also be saved to a different location (for example a network drive so that it can be shared with other High-performance Embedded Workshop users) in which case the configuration will not be added to the drop list.

## To import configurations

In order to import a configuration into High-performance Embedded Workshop, copy the configuration's SDO file into the Sync directory (before opening the Synchronized Debug dialog box) or follow the following method on the dialog box:

- 1. Click the **Import** button.
- 2. Browse to the configuration file you want to import.
- 3. Click Import.

The file will be copied to the Sync directory and the configuration will be added to the drop list where it will be selected and then loaded.



## To delete configurations

In order to delete a configuration, delete the configuration's file from the Sync directory (before opening the Synchronized Debug dialog box) or follow the following method:

- 1. Select the configuration you want to delete from the drop list.
- 2. Click the **Delete** button.

The configuration will be removed from the drop list. The file will be deleted from the Sync directory. The next (alphabetically) configuration in the drop list will be selected and then loaded.

## 18.1.2 Defining the synchronized Debuggers

Avaij	Avaijable debuggers:						
De	bugger ID	Status	Options	Туре	Workspace	Session	A <u>d</u> d
Mai	in	Not open	Debug	Multi core	C:\Workspace\Demo\Demo	Core0	
Vid	eo	Not open	Debug	Multi core	C:\Workspace\Demo\Demo	Core1	<u>M</u> odify
							<u>R</u> emove
1							-

## The Debuggers list

The list box in the middle of the dialog shows information on the synchronized Debuggers (essentially a debug session within a High-performance Embedded Workshop workspace).

When the dialog is first opened the list will be set to contain details of all debug sessions open in any High-performance Embedded Workshop application running on the local PC. The same also occurs when you change your configuration selection or create a new empty configuration. This is to help with multi-core configuration setup. For example, it is possible to open several sessions in separate High-performance Embedded Workshop applications, open the Synchronized Debug dialog in one of the applications and then proceed with synchronized Debugging without having to individually add each session to the Debuggers list.

A platform (or a core on a multi-core platform) cannot be synchronized without first creating a High-performance Embedded Workshop debug session for it.

## Note:

If any Debugger in the Debuggers list is currently connected to its platform then the Debuggers list will be read only. It will not be possible to add, modify or remove any Debugger. Also Debuggers that are already connected to a platform will not be automatically added to the list.



## High-performance Embedded Workshop V.4.09

Each entry in the list contains the following information:

Debugger ID:	A unique ID. This should be set to something that identifies the Debugger.			
Status:	Will be one of the following:			
	Not open:	The Debugger is not currently open in a High-performance Embedded Workshop application. It will be opened when the "Sync" button is clicked.		
	Not connected:	The Debugger is currently not connected to a platform.		
	Break:	The Debugger is connected to a platform and it is not running.		
	Running:	The Debugger is connected to a platform and it is running.		
Options:	Platform specific opt Select "Debug" for n	tion on how the Debugger will be used when synchronized debugging starts.		
Туре:	Type of platform ass	ociated with the Debugger. Will be one of the following:		
	Single core:	Normal single core platform.		
	Multi core:	Multi-core platform.		
Workspace:	The full name of the workspace file that contains the Debugger's session.			
Session:	The name of the Debugger's session within the workspace.			

## Adding a Debugger to the list

1. Click the **Add** button.

Add Debugger		? ×
Workspace file:	<u>B</u> rowse	OK Cancel
Project:		
Session:		

- 2. Enter, or browse to, a valid High-performance Embedded Workshop workspace file.
- 3. Select a project within the workspace.
- 4. Select a session within the project.
- 5. Click OK.

The combination of workspace, project and session specifies a Debugger. This is added to the list on the main dialog. The Debugger is given an initial ID based on its platform name (normal single core platform), or core name (multi-core platform).

## Note:

The specified workspace must be up to date with the current version of High-performance Embedded Workshop and not have been copied to a new location after it was last opened in High-performance Embedded Workshop. Any such



## High-performance Embedded Workshop V.4.09

workspace must be opened and closed normally in High-performance Embedded Workshop (so that the workspace files are updated) before it can be specified on this dialog.

## Changing a Debugger's ID

- 1. Select the Debugger to be changed in the list.
- 2. Click the **Modify** button.

Options	? ×
Debugger ID:	OK
Main	Cancel
<u>C</u> PU mode:	
Debug 💌	

- 3. Enter a new unique ID for the Debugger.
- 4. Click OK.

The Debugger's ID will be updated in the list.

It is not possible to modify Debugger IDs when synchronized.

## Changing a Debugger's options

There are two methods of changing a Debugger's options.

## First method:

- 1. Select the Debugger in the list.
- 2. Click the **Modify** button.
- 3. Select the new options from the drop list.
- 4. Click OK.

## Second method:

- 1. Select the Debugger in the list.
- 2. Click the Debugger's **Options** entry. This will display a drop list.





3. Select the new options from the drop list.

It is not possible to modify options when synchronized.

## Removing a Debugger from the list

- 1. Select the Debugger you want to remove.
- 2. Click the **Remove** button.

The Debugger will be removed from the list.

## 18.1.3 Setting the synchronization options

## Overview

In this context "synchronization" refers to a virtual link from one Debugger/core to another Debugger/core for a specific debug operation (e.g. Go). While debugging is synchronized, these virtual links are examined whenever a debug operation is executed. If a synchronization exists for the operation from the Debugger where the operation is executed to another Debugger then the operation will be executed on both Debuggers. The synchronizations are followed through multiple Debuggers to find out which Debuggers the operation needs to be executed on. Synchronizations can be one way or two way.

## Example 1:

Two Debuggers are synchronized with IDs "CORE0" and "CORE1". Only one synchronization exists and it is one way for the Go debug operation from "CORE0" to "CORE1". When a Go is executed in the "CORE0" Debugger the Go will occur on both Debuggers. When a Go is executed in the "CORE1" the Go will only occur on "CORE1".

## Example 2:

Four Debuggers are synchronized with IDs "CORE0", "CORE1", "CORE2" and "CORE3". Two way synchronizations exist for the Go debug operation between each Debugger and every other Debugger. When a Go is executed in any Debugger the Go will occur on all Debuggers.

By changing the configuration (SDO) file in a text editor (see <html link>) many synchronizations can be setup in as complex a pattern as required, however the GUI only allows the most commonly required styles of synchronizations to be specified.



## 18. Synchronized Debugging

Synchronization style		Synchronization options
O None		Reset
<ul> <li>All debuggers synchronized</li> </ul>	O Master de <u>b</u> ugger:	☑ <u>G</u> o
Operations executed on any debugger will also be executed on all other debuggers.	Operations executed on the master debugger will be executed on all other debuggers.	
$A \longleftrightarrow B$		☑ Step
$\uparrow$ $\checkmark$ $\checkmark$ $\uparrow$		- Set by platform
$    \times  $		Connection
		🗖 Download modules
		🗖 Initialize

## Synchronization style

Three basic styles of synchronization can be set using the dialog:

None:	No synchronizations.
All debuggers synchronized:	Two way synchronizations between each Debugger and every other Debugger for the checked operations.
Master debugger:	One way synchronizations from the specified master Debugger to every other Debugger for the checked operations. The drop-list next to this option
	specifies which Debugger will be the master Debugger.

## Synchronization options

The check boxes here show the debug operations which can be synchronized. The state of the following operations can be synchronized as required: Reset, Go, Break/Halt and Step. The following operations can also be synchronized but their state is always set by the platform being debugged: Connection, Download and Initialize. For non multi-core platforms they will always be off.

## **Platform settings**

When debugging a multi-core platform some synchronizations that are normally user configurable may be automatically enabled. If an option check box is checked and grayed out this means that the platform has specified that the operation must always be synchronized.

## Notes:

- Platform settings cannot be overridden by editing the SDO (configuration) file.
- Platform settings override your style selection. These operations will always be synchronized between all Debuggers even when "None" or "Master debugger" is selected.



## 18.1.4 Setting the memory update option

🔽 Update the views in all debuggers when platform memory is changed .

When this option is checked all High-performance Embedded Workshop views which display memory data (e.g. the memory view, the watch view etc.) in all Debuggers will update (i.e. read memory from the platform) whenever the memory is changed in any synchronized Debugger.

If the option is not set then only the memory views in the local Debugger will update when memory is changed. If memory is shared between Debuggers, a manual refresh will need to be performed in the memory related windows of the other Debuggers in order for them to display the correct memory.

Memory is usually only shared between Debuggers when debugging a multi-core platform. If memory is not shared between Debuggers (e.g. when debugging several single core platforms) then actions in one Debugger cannot change the memory in another so the state of this check box is not important.

## 18.1.5 Synchronized debugging mode

Synchronized debugging mode:	Parallel		
	Internal		
	Parallel		

## Select the mode to be used while Debugging:

Internal: The specified Debuggers will be opened in one High-performance Embedded Workshop application. This mode is only available when all the Debugger's Session's are in the same High-performance Embedded Workshop workspace.

**Parallel:** The specified Debuggers will be opened in separate High-performance Embedded Workshop applications.

#### Note:

The "Internal" mode option will only be available if all the sessions in the Available Debuggers list are from the same workspace. It is not possible to use the Internal mode option when sessions from separate workspaces are selected.

## 18.1.6 Start synchronized debugging

Once the synchronized debug configuration has been set up click the Sync button to begin debugging. If Parallel mode has been selected one Debugger will be opened in the High-performance Embedded Workshop application from which the Synchronized Debug dialog box was launched and new High-performance Embedded Workshop applications will be opened for the remaining Debuggers. If Internal mode was selected the workspace that is common to all the Debuggers will be opened in the High-performance Embedded Workshop application from which the Synchronized Debug dialog box was launched.

See section 18.2, Using High-performance Embedded Workshop when synchronized for details on how to use High-performance Embedded Workshop while synchronized.



## 18.1.7 Update synchronized debugging

If synchronized debug mode was already running when the Synchronized Debug dialog was opened, there will be an Update button instead of a Sync button. After the Update button is clicked the system will update according to the new settings that have been specified since opening the Synchronized Debug. This might include opening additional High-performance Embedded Workshop applications when using Parallel mode.

See section 18.2, Using High-performance Embedded Workshop when synchronized for details on how to use High-performance Embedded Workshop while synchronized.

## 18.1.8 Stop synchronized debugging

If already running in synchronized debug mode when the Synchronized Debug dialog was opened, the Unsync button will be enabled. In order to stop using synchronized debugging click the Unsync button. Any synchronized High-performance Embedded Workshop applications will disconnect from their platform and function independently again. The Debugger's tab will be removed form the Workspace Window. Debugger operations will no longer be synchronized. No High-performance Embedded Workshop applications will be closed down.

# 18.2 Using High-performance Embedded Workshop while synchronized

The main purpose of using High-performance Embedded Workshop in synchronized debugging mode is to have the main Debugger options synchronized between different High-performance Embedded Workshop debugging sessions, usually connected to a multi-core platform.

Many aspects of High-performance Embedded Workshop functionality are the same when using Parallel or Internal mode however some things will be different.

## 18.2.1 Common functionality

Workspace				×
Audio 10 Main Video				
Projects	🛃 Templates	🔍 Naviga	🚺 Test	Nebuggers

## Debuggers tab

This tab is added to the Workspace Window when synchronized debugging begins. It contains a separate item for each synchronized Debugger. The text of each item will be the display ID specified on the Synchronized debug dialog. The icon of each item indicates the current status of the Debugger. The following status icons are possible:





When you right click on a Debugger ID the following menu will be displayed:

Reset CPU Go Reset Go Step In Step Out Step Over Halt Program Allow Docking Hide Properties

When an operation is selected it will be executed on the selected Debugger(s) only. All synchronizations are ignored when an operation is executed from the Debuggers tab. When debugging a multi-core platform, some operations will be disabled. These are operations that are always synchronized by the hardware which means that the operation cannot be executed on the platform's cores separately.

Menu items will also be disabled if the operation is not possible on the selected Debugger. For example, if you right click on a Debugger which is running, only the "Halt Program" menu item will be enabled.

The "Properties" menu item is used to find out information on a Debugger. When selected, a dialog will be displayed containing the following items: Workspace file, workspace name, project name and session name.

The "Activate" menu item is only enabled when debugging in Internal mode. When selected this will make the selected Debugger the currently active one. See section 18.2.3, Internal mode functionality for more information.

## **Debugger operations**

The following table describes the operations that can be synchronized. The synchronized operations are determined by the selections on the Synchronized Debug dialog box. The table columns contain the following data:

<b>Operation:</b>	Name of the operation.
<b>Option</b> (s):	The check box option (or options) that must be checked on the Synchronized Debug dialog for
	the operation to be synchronized.
Method:	How the operation is executed in High-performance Embedded Workshop.
Effect:	The effect (in all synchronized Debuggers) of executing the operation.

"initial platform" means the platform connected to the Debugger where the operation was started. Sometimes the effect in this Debugger can be different from the others, depending on the operation executed.

Operation Option(s)	Method	Effect
Reset CPU Reset	"Debug->Reset CPU" menu.	Resets all synchronized platforms.
	Reset CPU toolbar button.	
	"reset" command.	
	After download when "Reset CPU after	
	download" option is enabled.	



18. Synchronized Debugging

High-performance Embedded Workshop V.4.09

Operation	Option(s)	Method	Effect
Go	Go	"Debug->Go" menu. Go toolbar button. "go" command.	Runs all synchronized platforms.
Go Until	Go	"Debug->Run" menu. After reset when "Run to main on reset" option is enabled. Editor pop-up "Go to Cursor" menu.	Runs initial platform with a specific set of temporary breakpoints. Runs all other synchronized platforms normally.
Reset Go	Reset and Go	"Debug->Reset Go" menu. Reset Go toolbar button. "go_reset" command.	Resets and then Runs all synchronized platforms (Note: This may not be the same as performing a reset and go separately, depending on your platform)
Break	Break/Halt	Breakpoint is reached.	Stops all synchronized platforms running.
Halt	Break/Halt	"Debug->Halt" Program menu. Halt toolbar button. "halt" command.	Stops all synchronized platforms running.
Step In	Step	"Debug->Step In" menu. Step In toolbar button. "step" command.	Performs step in on all synchronized platforms.
Step Over	Step	"Debug->Step Over" menu. Step Over toolbar button. "step_over" command.	Performs step over on all synchronized platforms.
Step Out	Step	"Debug->Step Out" menu. Step Out toolbar button. "step_out" command.	Performs step out on all synchronized platforms.
Connect	Connection	"Debug->Connect" menu. Connect toolbar button. "connect" command or platform specific command.	Connects to the platform in all synchronized Debuggers.
Download (file load)	Download modules	"Debug->Download Modules" menu item. Workspace tab "Download modules" folder. "file_load" command. On connection when "Download automatically on target connection" option is set. After build when "Download modules after build option is set. When a module is modified externally to High- performance Embedded Workshop.	Downloads the module to all synchronized platforms. The module filename must be the same in each Debugger.
Initialize	Initialize	"Debug->Initialize" menu. "initialize" command.	All synchronized platforms will be initialized.

Regarding stepping, the style of step performed (source or instruction) in a synchronized Debugger will be determined by the program counter, step mode and editor cursor position in that Debugger and not the Debugger where the operation was initiated. Each individual Debugger will effectively act as if the step operation had been executed locally by the normal High-performance Embedded Workshop method (e.g. toolbar button).

Regarding download, the operation will only be synchronized with Debuggers that have modules with the same filename as the module in the initiating Debugger. If a Debugger contains more than one module then only the one with the matching filename will be downloaded.



Unless specifically mentioned, all other Debugger operations, and other non Debugger High-performance Embedded Workshop operations, are possible while synchronized debugging. But they will only be executed on the initiating platform and not synchronized with other platforms.

## **Debug Settings dialog box**

While synchronized debugging the functionality of the Debug Settings dialog is different. Instead of being able to set all the normally available options it is only possible to change the Download modules which are set in each Debugger's session. It is possible to use multi select on the left side of the dialog to change the setup of multiple sessions at the same time. This allows the same module to be added easily to several sessions ready for a synchronized download.

For each module in each session you can use the "Modify" button to specify that debug information only should be downloaded. If a multi-core target is being used, with all cores sharing the same memory, it is recommended that the full memory image is downloaded in one session and debug information only is downloaded in the others.

The functionality of the buttons on the dialog are the same as when debugging normally. See section 17.1.4.3, Editing debug settings for details.



## 18.2.2 Parallel mode functionality



## **General operation**

When running in parallel mode each Debugger is a separate High-performance Embedded Workshop application. These can be used independently exactly like when debugging normally, just with the debugger operations synchronized.

## High-performance Embedded Workshop window title

The High-performance Embedded Workshop window title will start with the display ID of the Debugger open in the application. This will not change while the application is Synchronized.

#### Automatic un-sync

If an operation is performed that changes the current debug session the High-performance Embedded Workshop application will be removed from the synchronized debug system and will no longer be synchronized with the other High-performance Embedded Workshop applications.

The following operations will cause the session to change:

- New workspace.
- Open workspace.
- Close workspace.
- Set current project.
- Change session.



## 18.2.3 Internal mode functionality

Check-SP107.0 - high an increases train dief Markahan - Kite		<ul> <li>(iii)</li> </ul>
- THE RAY SHE READ, AND DRIVE SHED TOOL THE WARNE WE		
N2 ≥   D = 0 #   A   2	bn n <b> ≜odes</b> 5jma vena	אונ. י
바비아비아리아라 토 : : : : : : : : : : : : : : : : : :	도 [20 월월 26 월 26 월 27 월 28 월 28 월 29 월 29 월 29 월 29 월 29 월 29	B 2 F 2 F
BTIFIZ-Islaifsaan F		
11		
Anth         X         Q		34 A × 2日
E: I: Tenanté- : Skinair	tale Domesta Boy 82,82 43	Norma nee 8
iotnocmra jetto	000/000 ET 00.90	BIÓURÓLE STREADA
00000010 8001	87 8.0 J ( 20 8 177 83 84	MOLON MAD
Otrolines less	pre distriptions a	HOROGI BACHA
	800 876 25, 97 200 49 40, 70	1000124 receb
SCHOLTEL BLTT	nin direction	20002340 share P
ocuccuts cour-	N/F	Rest in the second seco
CONVERSE COURSES	879 87,87 082/80 876.01,80	100034/ JONE
0010020812 88113	DP 80.00118	3000314 DTML
CONTOCINET DEVICE	879 18 17, 99 879 87 89 50 50	1-00000 #ADUM 5
00000079 8363	877 375-00,00 886 887-00,00	k
00004824 0000	NOT .	
CONCEPT ADUS CONCEPTS PERC	BOV N., NO OTAVING REFORMED	Rea Value C
CONCERTS PERS	et distante	R 000000 3
ossosure sinc	80V 80 10.00	
OUNDERST TITS OFFICE ADD	NAM NET CO. RT NGA RE-Jacobrad	87 0R007-0-4
001002102 0019	NOT TO A CONTRACT OF	#4 #00:0000
0000000 0000	809 pt. 90 08/362 #8 06.10	ID FROMINI
COLOCION BUT	200-/00 200-00,00 97 88/21/218	16 CIRCUIC I: 17 Opposition
00809316 (2450)	p(x) (0) (x, in)	17 0000000 18 0000000
CONDENSE AND	80V 819 00, 81. 805 817 21.541.22	FP CU/CU/CU
CONTRACTOR ADDR	NA 87 3134133 809	Rap percente
Contraction of the second seco	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	kat cubrubru
		6 B
		Neseres 1
월 프 바 의 40 [21 [21 [25 [25 [25 [25 [25 [25 [25 [25 [25 [25	16201621815	
Destruction Constant of		
TERCHARITATION PRIME (PORCE), 2		
4 K Links States - Market - Ma	in it is not a finance of the second of the	
hale and have been a second se		fordering in a part of parts

## **General operation**

When using internal mode all Debuggers are opened in the one High-performance Embedded Workshop application. The state of the GUI can be switched (by various methods) to debug the different open Debuggers. When the GUI is switched to a Debugger the debug views will change to show data on that Debugger and any operations executed from then on will be executed on that Debugger (referred to as the "active Debugger"). This allows you to debug several platforms, or several cores on a multi-core platform, using one High-performance Embedded Workshop application.

## High-performance Embedded Workshop window title

The High-performance Embedded Workshop window title will start with the display ID of the current Debugger. This will change whenever the active Debugger is changed.

## Switching the active Debugger

There are several methods of switching the active Debugger.

Debuggers tab:	Right clicking on a Debugger in the tab and selecting "Activate" will make it the active Debugger.
Application toolbar:	When debugging in internal mode an extra drop list is added to the application toolbar. This contains the project/session combination that describes each
	synchronized Debugger. The active debugger can be changed to a specific Debugger by selecting its session in the drop list.
Shortcut key combination:	Pressing the Ctrl+Shift+N Key combination will change the active Debugger to the next one in the Debuggers tab.



## The Output Window

When using Internal mode the Output Window displays a separate "Debug" tab for each synchronized Debugger. Each tab includes the project name and session name of the Debugger it represents so that it can be identified.

Output 💌
01 01 Al At 🖞 🕼 🖉 🖻 🖬 🥊
Connected
Build Debug - Demo - Core0 Debug - Demo - Core1 Find in Files 1 A

## **Restricted operations**

All Debugger operations that change the active session are not allowed when in internal mode and will be disabled. This includes the following:

Set Current Project

**Remove Project** 

Unload Project

Change Session

**Change Configuration** 

The standard build operations can be used but they will only affect the active configuration in the active project, which cannot be changed.

If you want to build projects and configurations which are not active then you can use the Build Multiple dialog.

# 18.3 Using the command line window when synchronized

## Overview

Two new commands are provided when using synchronized debugging. These allow the Debugger that commands will be executed on to be changed so a command file executed in one Debugger can execute commands on a different Debugger.

## SET\_CORE command

## Abbreviation: SMC

**Description:** Sets the current Debugger from the perspective of the command line ECX. All debugger commands from now on will be sent to the requested Debugger.

Note: This does not effect operations invoked from the GUI.



Syntax: smc <Debugger ID>

Parameter	Туре	Description
<debugger id=""></debugger>	String	Debugger ID. If empty the perspective will be changed back to the local Debugger.

**Example:** smc "SH-X (core 1)"

From now on debugger commands will be directed to the "SH-X (core 1)" Debugger.

## SHOW\_ALL\_CORE command

#### Abbreviation: SAC

**Description:** Displays a list of available Debuggers. This command is used to get a list of valid inputs for the previous command. The list of Debuggers displayed will be the same as is displayed on the Debuggers Tab.

Syntax: sac

Parameter	Туре	Description
None		Displays a list of available Debuggers.

#### Exceptions

Some commands cannot be redirected to a different Debugger and will always be executed in the local Debugger no matter how the "SET\_CORE" command is used. The following commands have this limitation:

"ASSEMBLE"

"CLOSE\_TEST\_SUITE"

"COMPARE\_TEST\_DATA"

"MEMORY\_EDIT"

"OPEN\_TEST\_SUITE"

"REFRESH\_SESSION"

"RUN\_TEST"

"SET\_CORE"

"SHOW\_ALL\_CORE"

"SUBMIT"

## Note:

If the parameter of a command to be executed on a Debugger specified with the SET\_CORE command includes a filename, this filename must be enclosed (e.g. by {}) as shown below.

file\_load Elf/Dwarf2 {\$(CONFIGDIR)\\\$(PROJECTNAME).abs} 0x0 p byte



# 18.4 Glossary of terms

Term	Definition					
Core	An individual CPU core on a multi-core platform.					
Debugger	A combination of Workspace file, project name and session name that can be opened in a High-performance Embedded Workshop application to debug a specific target. When running in synchronized debug mode it is used as a generic term to refer to a synchronized High- performance Embedded Workshop application (when in parallel mode) or session (when in internal mode).					
Debugger ID	The string ID that uniquely identifies a Debugger within the High-performance Embedded Workshop GUI.					
Internal Debugger ID	The string ID that uniquely identifies a Debugger internally inside High-performance Embedded Workshop. Only needs to be understood when editing SDO files.					
Internal mode	Using synchronized debugging with all Debuggers open in one High-performance Embedded Workshop application.					
Local Debugger	The Debugger that the user is currently interacting with.					
Multi-core platform	A hardware or simulator device that has two or more CPU cores. The main purpose of the synchronized debugging feature is to help the user debug multi-core platforms.					
Parallel mode	Using synchronized debugging with each Debugger in a separate High-performance Embedded Workshop application.					
Platform	Hardware or simulator device that is connected to High-performance Embedded Workshop. Could be a single core platform or an individual core on a multi-core platform.					
SDO file	The synchronized debug configuration file. Contains all the information needed by High- performance Embedded Workshop to start a synchronized debug session. Saved from the Synchronized Debug dialog.					



# 19. Technical Support

# **19.1** Viewing the version information

The version information of the High-performance Embedded Workshop can be viewed.

## To view the version information

- 1. Select [Help -> About High-performance Embedded Workshop]. The About High-performance Embedded Workshop dialog box opens. Information including the copyright and version number is shown in this dialog box.
- 2. Clicking the **More**>> button shows **Installed package information**, which contains various information on the Renesas tool packages associated with the High-performance Embedded Workshop installed on your personal computer (e.g. version number).



3. Click on the **OK** button to close the dialog box.

# **19.2** Check for updates

To check for High-performance Embedded Workshop product updates or service packs

- 1. Select [Help -> Technical Support -> Check Website For Updates].
- 2. Your default web browser is invoked and defaults to the High-performance Embedded Workshop download page for your region.
- 3. Browse this area for High-performance Embedded Workshop updates to fix bugs or add new features.



# **19.3 Creating a bug report**

Occasionally you may experience some unforeseen problems with the High-performance Embedded Workshop application. If a problem does occur that results in an application crash the High-performance Embedded Workshop bug tracking program will be invoked automatically. This allows you to compile a bug report and this can then be sent to your technical support contact in a variety of ways. It is also possible to invoke this tracker program manually. This is described below:

## To create and send a High-performance Embedded Workshop bug report

- 1. Select [Help -> Technical Support -> Create Bug Report].
- 2. Detailed information is generated from your High-performance Embedded Workshop system. This may take some time. The **Submit a Bug Report** dialog box is then displayed.
- 3. Write the description of your problem in the **Please type a description of the problem you wish to report** edit box. (This item must be completed.)
- 4. Type your name in the **Name** edit box.
- 5. Type your company URL or purpose of the product being developed in the **Company website/Your application** edit box. (This item must be completed.)
- 6. Type your country or region in the Country/Region edit box. (This item must be completed.)
- 7. Once you are happy with your report, choose the method of sending the report in the **How would you like to submit the report?** drop-down list box. You can print it, e-mail it, or save it to a disk. (This selection is required.)
- 8. Then click **Submit**. This will send the report. When submitting a report via e-mail you will see a confirmation message after clicking Submit.

<sup>•</sup> Please type a description of the prob	lem you wish to report:	
		A
Name:	* Company website/Your application:	Submit
<sup>•</sup> Country/Region:	* How would you like to submit the report?	Subinit
	Print	Cancel
(Fields marked with a * are required)		

## Note:

Fields marked with an asterisk (\*) are required. You can click the **Submit** button after these fields are completed.



# Reference



# 1. Main Menus

# 1.1 File Menu Options

Menu	Menu Option	Shortcut Key		Macro Recording *	Function
File	New	CTRL+N	D	-	Creates a new document.
	Open	CTRL+O	<b>2</b>	-	Opens an existing document.
	Close	CTRL+F4	-	-	Closes the active document.
	New Workspace	-	-	-	Creates a new workspace.
	Open Workspace	-	-	•	Opens an existing workspace.
	Save Workspace	-	-	•	Saves the current workspace.
	Save Workspace As	-	-	-	Saves this workspace with a different name, or in an old format.
	Close Workspace	-	-	•	Closes the current workspace.
	New Session	-	-	•	Creates a new session.
	Import Session	-	-	•	Imports an existing session.
	Save Session	-	-	•	Saves the current session.
	Save Session As	-	-	-	Saves the current session with a new session name.
	Refresh Session	-	-	•	Reloads the session file for the current session.
	Download A New Module	-	-	•	Downloads a new module.
	Save	CTRL+S		-	Saves the active document.
	Save As	-	-	-	Saves the active document with a new file name.
	Save All	CTRL+ SHIFT+S	Ø	-	Saves all modified documents in the workspace.
	Page Setup	-	-	-	Changes the printing options.
	Print	CTRL+P	9	-	Prints the active document.
	Recent Files	-	-	-	Opens this document.
	Recent Workspaces	-	-	•	Opens this workspace.
	Recent Test Suites	-	-	-	Opens this test suite.
	Recent Downloaded Modules	-	-	•	Downloads this module.
	Exit	-	-	-	Exits High-performance Embedded Workshop.

Note:

\*. Operations with some menu options can be recorded as High-performance Embedded Workshop command-line commands by the macro-recording support facility. A macro record icon (●) in the "Macro Recording" column of a menu option indicates that this function can be recorded into a macro file. For details, see section 15.5.1, Recordable functions (common to all High-performance Embedded Workshop products).

# 1.2 Edit Menu Options

Menu Menu Option		Shortcut Key		r Macro Recording *	Function
Edit	Undo	CTRL+Z	-	-	Reverses the last editing operation.
	Redo	CTRL+Y	-	-	Repeats the last undone editing operation.



## High-performance Embedded Workshop V.4.09

1. Main Menus

u	Menu Option       Cut       Copy		Shortcut Key	Toolbar Button	Macro Recording *	Function
			CTRL+X	*	-	Removes highlighted text and places it on the Windows® clipboard.
•			CTRL+C	Ê <u>₽</u>	-	Places a copy of the highlighted text into the Windows® clipboard.
	Paste		CTRL+V	Ê	-	Copies the contents of the Windows® clipboard into the active window at the position of the insertion cursor.
	Clear		Delete	-	-	Removes highlighted text (it is not copied to the Windows® clipboard)
-	Select All		CTRL+A	-	-	Selects (i.e. highlights) the entire contents of the active window.
	Find		CTRL+F	a,a†	-	Finds text in the current file.
	Find In Files	3	F4	<b>-</b>	-	Finds text in multiple files.
	Replace		CTRL+H	-	-	Replaces text in the current file.
-	Goto Line		CTRL+G	-	-	Jumps to a line in a file.
	Match Brace	es	CTRL+SHIFT+M	<b>{+}</b>	-	Finds a matching brace.
	Bookmarks Toggle Bookmark Next Bookmark		CTRL+F2		-	Sets a bookmark at the current line or clear a bookmark at the current line.
			F2	â	-	Jumps to the next bookmark in the current file from the current line.
		Previous Bookmark	SHIFT+F2	â	-	Jumps to the previous bookmark in the current file from the current line.
		Clear All Bookmarks	-	<b>*</b>	-	Clears all bookmarks in the current file.
	Templates	Define Templates	-	Ť	-	Defines a template.
		Insert Template	CTRL+SHIFT+T	Т	-	Inserts a template.
	Toggle Breakpoint Enable/Disable Breakpoint Define Column Format Source Breakpoints		F9		•	Sets or clears a software breakpoint at the line showing the address.
•			CTRL+F9	Ð	•	Enables or disables the current software breakpoint.
			-	-	-	Sets the status of editor columns.
			CTRL+B	-	-	Opens the <b>Breakpoins</b> dialog box.
	Evaluate		-	-	-	Evaluates simple and complex expressions

#### Note:

\*. Operations with some menu options can be recorded as High-performance Embedded Workshop command-line commands by the macro-recording support facility. A macro record icon (●) in the "Macro Recording" column of a menu option indicates that this function can be recorded into a macro file. For details, see section 15.5.1, Recordable functions (common to all High-performance Embedded Workshop products).



# 1.3 View Menu Options

Menu	Menu	Option	Shortcut Key	Toolbar Button	Macro Recording *1	Saving into Test- Image File *2	Function
/iew	Differe	nces	-	-	-	-	Opens the <b>Difference</b> window.
	Map *3	3	-	MAP	-	-	Opens the map window.
	Comm	and Line	CTRL+L	$\geq$	-	-	Opens the <b>Command Line</b> window.
	TCL To	oolkit	CTRL+SHIFT+K	° <b>0</b>	-	-	See the "Tcl/Tk Additional document".
	Workspace		ALT+K	<b>-</b>	•	-	Opens the <b>Workspace</b> window.
	Output		ALT+O	▶	-	<b>!</b> *4	Opens the <b>Output</b> window.
	Status Bar		ALT+A		-	-	Toggles the status bar on and off.
	Disass	embly	CTRL+D	<b>S</b>	•	-	Opens the <b>Disassembly</b> window.
	CPU	Registers	CTRL+R	R1	•		Opens the Registers window.
		Memory	CTRL+M	<u>F</u>	•		Opens the <b>Memory</b> window.
		10	CTRL+I	1/0	•		Opens the <b>IO</b> window.
		Status	CTRL+U	ŧŧ	-		Opens the Status window.
	Graphi	c Image * <b>3</b>	CTRL+SHIFT+G		-	-	Opens the Image window.
		Waveform *3	CTRL+SHIFT+V	4	-	-	Opens the <b>Waveform</b> window.
	Code	Stack Trace	CTRL+K	<b>F</b>	-	H	Opens the <b>Stack Trace</b> window.

## Notes:

\*1. Operations in some windows can be recorded as High-performance Embedded Workshop command-line commands by the macro-recording support facility. A macro record icon (●) in the Macro Recording column of a menu option indicates that this function can be recorded into a macro file. For details, see section 15.5.1, Recordable functions (common to all High-performance Embedded Workshop products).

Note that a function of another window dependent on the debugger can also be recorded as a command even though this window is not included in the list of menus. For details, see section 15.5.2, Recordable functions (dependent on the debugger).

\*2. Data in some windows can be saved as test-image data into test-image files by the test-support facility. A save file icon (●) in the Saving into Test-Image File column of a menu option indicates that this data can be saved into a test-image file. For details, see section 16.6.1, Functions that can be saved into test-image files (common to all High-performance Embedded Workshop products).

Note that some functions of other windows dependent on the debugger can also be saved into test-image files even though these windows are not included in the list of menus. For details, see section 16.6.2, Functions that can be saved into test-image files (dependent on the debugger).

- **\*3**. Support for this function depends on the debugger.
- \*4. The data of the Build tab or the Debug can be saved.



# 1.4 **Project Menu Options**

Menu	Menu Option	Shortcut Key	Toolbar Button	Macro Recording *1	Function
Project	Set Current Project	-	-	•	Sets this project as the current project.
	Insert Project	-	-	•	Inserts a project to the workspace.
	Dependent Projects	-	-	-	Shows dependent projects.
	Edit Project Configuration *2	-	-	•	Edits the project configuration.
	Create Project Type	-	-	-	Creates a new project type.
	Add Files	-	-	-	Adds files to project.
	Remove Files	-	-	-	Removes files from project.
	File Extensions	-	-	-	Displays current project file extensions.
	Components	-	-	-	Loads/unloads components.

Notes:

\*1. Operations with some menu options can be recorded as High-performance Embedded Workshop command-line commands by the macro-recording support facility. A macro record icon (●) in the "Macro Recording" column of a menu option indicates that this function can be recorded into a macro file. For details, see section 15.5.1, Recordable functions (common to all High-performance Embedded Workshop products).

\*2. Support for this function depends on the debugger.

# 1.5 Build Menu Options

Menu	Menu Option	Shortcut Key	Toolbar Button	<sup>•</sup> Macro Recording *1	Function
Build	Toolchain	-	-	-	Sets build options.
*2	Build File	CTRL+F7	٢	•	Builds the selected file.
	Build	F7	<b>***</b>	•	Builds out of date project files.
	Build All	-	i di i	•	Builds project files, regardless of whether the project files are out of date.
	Build Multiple	-	-	•	Builds multiple projects.
	Clean Current Project	-	-	•	Cleans the current configuration in this project.
	Clean All Projects	-	Ħ	•	Cleans all configurations in all projects in this workspace.
	Update All Dependencies	-	-	-	Updates a project's dependencies.
	Stop Tool Execution	CTRL+Break	: 👗	-	Stops tool execution.
	Include/Exclude Build	-	-	-	Excludes a file from build or cancels the exclusion.
	Build Phase	-	-	-	Adds, removes and modifies a phase.
	Build Configurations	-	-	•	Selects the current configuration.
	Linkage Order	-	-	-	Customizes the High-performance Embedded Workshop linkage order.
	Generate Makefile	-	-	-	Generates a makefile.



## Notes:

- \*1. Operations with some menu options can be recorded as High-performance Embedded Workshop command-line commands by the macro-recording support facility. A macro record icon (●) in the "Macro Recording" column of a menu option indicates that this function can be recorded into a macro file. For details, see section 15.5.1, Recordable functions (common to all High-performance Embedded Workshop products).
- \*2. This menu is not displayed while a debug-only project "Debugger only xxxxxx" created by High-performance Embedded Workshop V.4.01 or later is in use.

# 1.6 Debug Menu Options

Menu	Menu Option	Shortcut Key	Toolbar Button	<sup>.</sup> Macro Recording *1	Function
Debug	Synchronized Debugging *2	-	-	-	Configures synchronized debugging.
	Debug Sessions	-	-	•	Opens the <b>Debug Sessions</b> dialog box to list, add, or remove the debug session.
	Debug Settings	-	-	-	Opens the <b>Debug Settings</b> dialog box to set the debugging conditions or download modules.
	Reset CPU	-	ĒŤ	•	Resets the CPU.
	Go	F5	Ē↓	•	Starts executing the user program at the current PC.
	Reset Go	SHIFT+F5	Ð	•	Executes the user program from the reset vector address.
	Free Go * <b>2</b>	-	≣∎	•	Runs program, ignoring any breakpoints.
	Go to Cursor	-	∎ <b>t</b>	•	Starts executing the user program at the current PC and continues until the PC equals the address indicated by the current text cursor position.
	Set PC to Cursor	-	I <sub>PC</sub>	•	Changes the value of the Program Counter (PC) to the address at the row of the text cursor.
	Run	-	-	•	Opens the <b>Run Program</b> dialog box allowing the user to enter temporary breakpoints before executing the user program.
	Display PC	CTRL+ SHIFT+Y	PC	-	Opens the editor or <b>Disassembly</b> window at the address of the PC.
	Step In	F11	<del>{</del> }}	•	Executes a block of user program before breaking.
	Step Over	F10	0	•	Executes a block of user program before breaking. If a subroutine call is reached, then the subroutine will not be entered.
	Step Out	SHIFT+F11	<b>{}</b>	•	Executes sufficient user program to reach the end of the current function.
	Step	-	-	•	Opens the <b>Step Program</b> dialog box allowing the user to modify the settings for stepping.
	Step Auto Mode	-	-	•	Steps only one source line when the editor window is active. When the <b>Disassembly</b> window is active, stepping is executed in a unit of assembly instructions.
	Assembly	-	-	•	Executes stepping in a unit of assembly instructions.



## High-performance Embedded Workshop V.4.09

Menu	Menu Option	Shortcut Key	Toolbar Button	Macro Recording *1	Function
	Source	-	-	•	Steps only one source line.
	Halt Program	-	<b>510</b> 3	•	Stops the execution of the user program.
	Initialize	-	-	•	Disconnects the debugger and connects it again.
	Connect *2	-	ļ ∎	•	Connects the debugger.
	Disconnect *2	-		•	Disconnects the debugger.
	Save Memory	-	-	•	Saves the specified memory area data to a file.
	Verify Memory *2	-	-	•	Verifies file contents against memory contents.
	Download Modules	-	-	•	Downloads the object program.
	Unload Modules	-	-	•	Unloads the object program.

Notes:

- \*1. Operations with some menu options can be recorded as High-performance Embedded Workshop command-line commands by the macro-recording support facility. A macro record icon (●) in the "Macro Recording" column of a menu option indicates that this function can be recorded into a macro file. For details, see section 15.5.1, Recordable functions (common to all High-performance Embedded Workshop products).
- \*2. Support for this function depends on the debugger.

# 1.7 Setup Menu Options

Menu	Menu	Option	Shortcut Key	Toolbar Button	Macro Recording *	Function
Setup	Custon	nize	-	-	-	Customizes the High-performance Embedded Workshop application.
	Option	S	-	-	-	Sets option of the High-performance Embedded Workshop application.
	Format	t Views	-	-	-	Configures fonts, colors, keywords and so on, for the window.
	Radix	Hex	-	<u>16</u>	•	Sets radix to hexadecimal.
		Decimal	-	<u>10</u>	•	Sets radix to decimal.
		Oct	-	8	•	Sets radix to octal.
		Bin	-	2	•	Sets radix to binary.

## Menu Menu Option Shortcut Key Toolbar Button Macro Recording \* Function

Note:

\*. Operations with some menu options can be recorded as High-performance Embedded Workshop command-line commands by the macro-recording support facility. A macro record icon (●) in the "Macro Recording" column of a menu option indicates that this function can be recorded into a High-performance Embedded Workshop macro file. For details, see section 15.5.1, Recordable functions (common to all High-performance Embedded Workshop products).



# 1.8 Tools Menu Options

Menu	Menu O	ption	Shortcut Key	Toolbar Button	Function
Tools	Administ	tration	-	-	Controls the components.
	Change	Toolchain Version	-	-	Changes a toolchain version.
	Version	Select	-	-	Selects a version control system.
	Control	Configure	-	-	Setups the version control system.
		Add to VCS	-	<b>1</b>	Adds files to Visual SourceSafe.
		Remove from VCS	-	÷	Gets a read-only copy of a file from Visual SourceSafe.
		Get from VCS	-	\$	Gets a read-only copy of a file from Visual SourceSafe.
		Check out from VCS	-	5	Checks out a writable copy of a file from Visual SourceSafe.
		Check in VCS	-	<u></u>	Checks in your edits to a file into Visual SourceSafe.
		Get VCS status	-		Views the status of a file in Visual SourceSafe.
	Launch	External Debugger	-	<b>&gt;</b> ‡	Launches an external debugger tool.
	Macros		-	-	Opens the <b>Macro</b> dialog box.
	Record I	Vacro	-	•	Starts recording of a macro.
	Play Ma	cro	-	Þ	Plays a macro.
	Stop Ma	cro	-		Stops a macro.
	System	tools	-	22	Invokes system tools (e.g. Renesas Call Walker).

Notes:

- \*1. Custom menu options to which recorded macros have been assigned are shown between Stop Macro and system tools.
- \*2. Further menu options for external tools added by customizing the **Tools** menu are shown under the menu options for system tools.

# 1.9 Test Menu Options

Menu	Menu Option	Shortcut Key	Toolbar Button	Function
Test	Create New Test Suite	-	-	Creates a new blank test suite so you can begin adding and running tests.
	Open Test Suite	-	-	Opens an existing test suite.
	Edit Test Suite	-	-	Edits the current test suite. Allows you to add and remove tests to the test suite.
	Close Test Suite	-	-	Closes the current test suite.
	Create New Test Image File	-	-	Setups and customizes the test image data saved to the test image file (*.HIF).
	Compare Test Image File	-	-	Compares test image file allows you to compare a test image file with the current High-performance Embedded Workshop system or with another test image file already on your disk. The results are then displayed in the test browser.



Menu Menu Option			Function
	Кеу	Button	
Run Tests	-	-	Runs tests allows you to run multiple tests that you have defined in the test suite and see the results of the comparisons in the test browser. There are various options to configure the test run execution.
Test Result Browser	-	-	Displays the test results for one or more test executions. It shows the pass and fails results and the detailed reason why the test failed.

# 1.10 Window Menu Options

Menu	Menu O	ption	Shortcut Key	Toolbar Button	Function
Window	Cascade	Э	-	-	Arranges all open windows so that they overlap.
	Tile Hori	zontally	-	-	Arranges all open windows horizontally.
	Tile Vertically		-	-	Arranges all open windows vertically.
	Arrange Icons		-	-	Lines up all minimized windows.
	Close A		-	-	Closes all open windows.
	Virtual desktop	Desktop Manager	-	-	Renames your configuration to a more meaningful name.
		Default 1-4	-	-	Switches desktop configurations.

# 1.11 Help Menu Options

Menu	Menu Opt	tion	Shortcut Key	Toolbar Button	Function
Help	Help Topic	C	-	-	Shows the main High-performance Embedded Workshop help window.
	Technical	Create Bug Report	-	-	Creates a High-performance Embedded Workshop bug report.
	Support	Check Website For Updates	-	-	Checks for High-performance Embedded Workshop product updates or service packs.
		License Management	-	٩	Opens the License Management dialog box. To view the License Management help, double-click \Tools\Renesas\DebugComp\Ecx\EcxLicMgr\EcxLicMgr.chm under the High-performance Embedded Workshop installation directory.
	About High-performance Embedded Workshop		-	-	Opens the <b>About High-performance Embedded Workshop</b> dialog box allowing the user to view the version of High- performance Embedded Workshop.
	Debugger	Help	-	-	Shows the help window of the emulator or simulator when the debugger is connected.


# 2. Windows

# 2. Windows

Window Name	Opened by
Differences	[View -> Difference]
Map Section Information *1	[View -> Map]
Map Symbol Information *1	[View -> Map]
Command Line	[View -> Command Line]
Console *2	[View -> TCL Toolkit]
Workspace	[View -> Workspace]
Output	[View -> Output]
Disassembly	[View -> Disassembly]
Registers	[View -> Registers]
Memory	[View -> Memory]
IO	[View -> CPU -> IO]
Status	[View -> CPU -> Status]
Image * <b>1</b>	[View -> Graphic -> Image]
Waveform *1	[View -> Graphic -> Waveform]
Stack Trace	[View -> Code -> Stack Trace]
Test Browser	[Test -> Test Results Browser]
NT /	

Notes:

**\*1**. Support for this function depends on the debugger.

\*2. See the "Tcl/Tk Additional document".



# 3. Commands

# 3.1 Command List (Alphabetic Order)

Command Name	Abbreviation	Description
!	-	Comment.
ADD_FILE	AF	Adds a file to the current project.
ASSERT	-	Checks if an expression is true or false.
AUTO_COMPLETE	AC	Switches the auto-completion.
BUILD *1	BU	Performs a build on the current project.
BUILD_ALL *1	BL	Performs a build all on the current project.
BUILD_FILE *1	BF	Performs a build on the file.
BUILD_MULTIPLE *1	BM	Performs a build on the multiple projects and configurations.
CACHE *2	-	Sets caching on or off.
CHANGE_CONFIGURATION	CC	Sets the configuration to the specified configuration name.
CHANGE_PROJECT	CP	Sets the specified project file as the current project.
CHANGE_SESSION	CS	Sets the specified session as the current session.
CLEAN	CL	Deletes intermediate and output files produced in building.
CLEAR_OUTPUT_WINDOW	COW	Clears the contents of the specified tab in the output window.
CLOSE_TEST_SUITE	CTS	Closes the current test suite.
CLOSE_WORKSPACE	CW	Closes a workspace.
COMPARE_TEST_DATA	CTD	Compares test data and create results.
CONNECT *2	CN	Connects the debugger.
DEFAULT_OBJECT_FORMAT	DO	Sets the object format to be used by default.
DISCONNECT *2	DN	Disconnects the debugger.
ERASE	ER	Clears the Command Line window.
EVALUATE	EV	Evaluates an expression.
FILE_LOAD	FL	Loads an object (program) file.
FILE_LOAD_ALL	LA	Loads all object (program) files.
FILE_SAVE	FS	Saves memory to a file.
FILE_UNLOAD	FU	Unloads an object (program) file from memory.
FILE_UNLOAD_ALL	UA	Unloads all object (program) files from memory.
FILE_VERIFY *2	FV	Verifies file contents against memory.
FREE_GO *2	FG	Runs program, ignoring any breakpoints.
GENERATE_MAKE_FILE *1	GM	Generates a build makefile for the current workspace.
GO	GO	Runs program.
GO_RESET	GR	Runs program from reset.
GO_TILL	GT	Runs program until specified addresses.
HALT	HA	Halts program.
HELP	HE	Displays help for Command Line or help on a command.
INITIALIZE	IN	Initializes the debugger.
LOG	LO	Controls command output logging.
MEMORY_COMPARE *2	MC	Compares memory contents.
MEMORY_DISPLAY	MD	Displays memory contents.
MEMORY_EDIT	ME	Modifies memory contents.



# High-performance Embedded Workshop V.4.09

Command Name	Abbreviation	Description
MEMORY_FILL	MF	Fills a block of memory.
MEMORY_FIND *2	MI	Finds a string in an area of memory.
MEMORY_MOVE	MV	Moves a block of memory.
MEMORY_TEST *2	MT	Tests a block of memory.
OPEN_TEST_SUITE	OTS	Opens a test suite.
OPEN_WORKSPACE	OW	Opens the specified workspace file.
QUIT	QU	Exits High-performance Embedded Workshop.
RADIX	RA	Sets the radix for the value.
REFRESH_SESSION	RSE	Reloads the session file.
REMOVE_FILE	REM	Removes a file from the current project.
RESET	RE	Resets the microprocessor.
RUN_TEST	RT	Runs a test.
SAVE_SESSION	SE	Saves the current session.
SAVE_WORKSPACE	SW	Saves the current workspace.
SET_DISASSEMBLY_SOFT_BREAK	SDB	Sets or deletes a software breakpoint at the disassembly level.
SET_SOURCE_SOFT_BREAK	SSB	Sets or deletes a software breakpoint at the source level.
SLEEP	-	Delays command execution.
STATE_DISASSEMBLY_SOFT_BREAK	TDB	Enables or disables a software breakpoint at the disassembly
		level.
STATE_SOURCE_SOFT_BREAK	TSB	Enables or disables a software breakpoint at the source level.
STEP	ST	Steps through program (by instructions or source lines).
STEP_MODE	SM	Sets the step mode.
STEP_OUT	SP	Steps out of the current function.
STEP_OVER	SO	Steps through program without stepping into functions.
STEP_RATE	SR	Sets rate of stepping.
SUBMIT	SU	Executes a file of commands.
TCL	-	Turns TCL commands on or off.
TOOL_INFROMATION	ТО	Outputs the tool information.
UPDATE_ALL_DEPENDENCIES *1	UD	Updates the current projects build dependencies.

Notes:

**\*1**. Available only when there is a toolchain installed.

\*2. Support for this command depends on the debugger.

For the syntax of each command, refer to the online help.



# 3.2 Command List (Listed by Function)

# High-performance Embedded Workshop Application Control Commands

Command Name	Abbreviation	Description
ADD_FILE	AF	Adds a file to the current project.
CHANGE_CONFIGURATION	CC	Sets the configuration to the specified configuration name.
CHANGE_PROJECT	СР	Sets the specified project file as the current project.
CHANGE_SESSION	CS	Sets the specified session as the current session.
CLEAR_OUTPUT_WINDOW	COW	Clears the contents of the specified tab in the output window.
CLOSE_WORKSPACE	CW	Closes the specified workspace file.
EVALUATE	EV	Evaluates an expression.
OPEN_WORKSPACE	OW	Opens the specified workspace file.
QUIT	QU	Exits High-performance Embedded Workshop.
RADIX	RA	Sets the radix for the value.
REFRESH_SESSION	RSE	Reloads the session file.
REMOVE_FILE	REM	Removes a file from the current project.
SAVE_SESSION	SE	Saves the current session.
SAVE_WORKSPACE	SW	Saves the current workspace.
TOOL_INFORMATION	ТО	Outputs the tool information.

#### Build Commands (Available only when there is a toolchain installed)

Command Name	Abbreviation	Description
BUILD	BU	Performs a build on the current project.
BUILD_ALL	BL	Performs a build all on the current project.
BUILD_FILE	BF	Performs a build on the file.
BUILD_MULTIPLE	BM	Performs a build on the multiple projects and configurations.
CLEAN	CL	Deletes intermediate and output files produced in building.
GENERATE_MAKE_FILE	GM	Generates a build makefile for the current workspace.
UPDATE_ALL_DEPENDENCIES	UD	Updates the current projects build dependencies.

# **Command Line Operating Commands**

Command Name	Abbreviat	ion Description
!	-	Comment.
ASSERT	-	Checks if an expression is true or false.
AUTO_COMPLETE	AC	Switches the auto-completion.
ERASE	ER	Clears the Command Line window.
HELP	HE	Displays help for Command Line or help on a command.
LOG	LO	Controls command output logging.
SLEEP	-	Delays command execution.
SUBMIT	SU	Executes a file of commands.
TCL	-	Turns TCL commands on or off.



## **Test Supporting Facilities Commands**

Command Name	Abbreviation	Description
CLOSE_TEST_SUITE	CTS	Closes the current test suite.
COMPARE_TEST_DATA	CTD	Compares test data and create results.
OPEN_TEST_SUITE	OTS	Opens a test suite.
RUN_TEST	RT	Runs a test.

## Execution Commands (Available when the debugger is connected)

Command Name	Abbreviation	Description
FREE_GO *	FG	Runs program, ignoring any breakpoints.
GO	GO	Runs program.
GO_RESET	GR	Runs program from reset.
GO_TILL	GT	Runs program until specified addresses.
HALT	HA	Halts program.
RESET	RE	Resets the microprocessor.
STEP	ST	Steps through program (by instructions or source lines).
STEP_MODE	SM	Sets the step mode
STEP_OUT	SP	Steps out of the current function.
STEP_OVER	SO	Steps through program without stepping into functions.
STEP_RATE	SR	Sets rate of stepping.
Note:		

Note:

\*. Support for this command depends on the debugger.

# Memory Operation Commands (Available when the debugger is connected)

Command Name	Abbreviat	ion Description
CACHE *	-	Sets caching on or off.
FILE_LOAD	FL	Loads an object (program) file.
FILE_LOAD_ALL	LA	Loads all object (program) files.
FILE_SAVE	FS	Saves memory to a file.
FILE_UNLOAD	FU	Unloads an object file from memory.
FILE_UNLOAD_ALL	UA	Unloads all object (program) files from memory.
FILE_VERIFY *	FV	Verifies file contents against memory.
MEMORY_COMPARE *	MC	Compares memory contents.
MEMORY_DISPLAY	MD	Displays memory contents.
MEMORY_EDIT	ME	Modifies memory contents.
MEMORY_FILL	MF	Fills a block of memory.
MEMORY_FIND *	MI	Finds a string in an area of memory.
MEMORY_MOVE	MV	Moves a block of memory.
MEMORY_TEST *	MT	Tests a block of memory.
N7 /		

Note:

\*. Support for this command depends on the debugger.



## Software Break Setting Commands (Available when the debugger is connected)

Command Name	Abbreviation	Description
SET_DISASSEMBLY_SOFT_BREAK	SDB	Sets or deletes a software breakpoint at the disassembly level.
SET_SOURCE_SOFT_BREAK	SSB	Sets or deletes a software breakpoint at the source level.
STATE_DISASSEMBLY_SOFT_BREAK	TDB	Enables or disables a software breakpoint at the disassembly
		level.
STATE_SOURCE_SOFT_BREAK	TSB	Enables or disables a software breakpoint at the source level.

## Other Debugging Commands (Available when the debugger is connected)

Command Name	Abbreviati	on Description
CONNECT *	CN	Connects the debugger.
DISCONNECT *	DN	Disconnects the debugger.
INITIALIZE	IN	Initializes the debugger.
NT /		

Note:

\*. Support for this command depends on the debugger.

For the syntax of each command, refer to the online help.



# 4. Regular Expressions

The High-performance Embedded Workshop editor allows you to include special characters in search strings when performing a find, replace or find in files operation.

These characters are listed in the table below and explained underneath.

For example, 12p matches ''eop'', "tip' but not ''trap'.         This character matches any number of occurrences (0 or more) of any character except a newline. Thus, this character will not match across new lines. The * character will match as few occurrences as are necessary to make the rest of the pattern match.         For example, 1*0 matches the 'too'' of 'too'', the "tro'' of "trowel" and the "ty o'' of "sporty orange" but not "smart orange" because the * character does not match across a new line.         t       This character matches the tab character.         Example 1: \t8       Finds every occurrence of a tab character followed by an 8.         Example 2: initt       Finds every occurrence of a tab character following "init".         ]       This matches any one character or a range of single characters listed within the brackets. Brackets cannot be nested.         [-] specifies a range of characters e.g. [a-2] or [0-9]. The beginning character in the range must have a lower ASCII value than the ending character of the range.         [-] matches a single characters if it is not any one of the characters between [~ and ]. This pattern also matches newline characters, unless the newline character is included within the brackets.         Example 2: [<>?]         Finds every uppercase vowel.         Example 2: [<>?]         Finds a uiteral <, > or ?.         Example 2: [<>?]         Finds a uiteral <, > or ?.         Example 3: [A-Za-z0-9.]         Matches an upper or lowercase letter, a digit or an underscore.	Character	Function
This character matches any number of occurrences (0 or more) of any character except a newline. Thus, this character will not match across new lines. The * character will match as few occurrences as are necessary to make the rest of the pattern match.         For example, it'o matches the "to" of "too", the "tro" of "trowel" and the "ty o" of "sporty orange" but not "smart orange" because the * character does not match across a new line.         t       This character matches the tab character.         Example 1: \text{VB}       Finds every occurrence of a tab character followed by an 8.         Example 2: init!       Finds every occurrence of a tab character following "init".         ]       This matches any one character or a range of single characters listed within the brackets. Brackets cannot be nested.         [-] specifies a range of characters e.g. [a-z] or [0-9]. The beginning character in the range must have a lower ASCI value than the ending character of the range.         [-] matches a single characters. unless the newline character is included within the brackets.         Example 1: [AEIOU]         Finds every uppercase vowel.         Example 2: [ ]</td Finds a literal <, > or ?.         Example 4: [-0-9]         Matches an upper or lowercase letter, a digit or an underscore.         Example 5: [\tens         Example 6: [\tens         Matches a space, a tab or newline.         Example 7: [\tens         Example 6: [\tens         Matches a space	?	This character matches any single character, except the newline character.
<ul> <li>this character mill not match across new lines. The * character will match as few occurrences as are necessary to make the rest of the pattern match.</li> <li>For example, t*o matches the "to" of "too", the "tro" of "trowel" and the "ty o" of "sporty orange" but not "smart orange" because the * character does not match across a new line.</li> <li>t This character matches the tab character.</li> <li>Example 1: \text{triangle} the character of a tab character followed by an 8.</li> <li>Example 2: initt</li> <li>Finds every occurrence of a tab character following "init".</li> <li>This matches any one character or a range of single characters listed within the brackets. Brackets cannot be nested.</li> <li>[-] specifies a range of characters e.g. [a-z] or [0-9]. The beginning character in the range must have a lower ASCII value than the ending character of the range.</li> <li>[-] matches a single character if it is not any one of the characters between [- and ]. This pattern also matches newline characters, unless the newline character is included within the brackets.</li> <li>Example 1: [AEIOU]</li> <li>Finds every uppercase vowel.</li> <li>Example 2: [c-?]</li> <li>Finds a literal (_&gt; or ?.</li> <li>Example 2: [c-?]</li> <li>Matches an upper or lowercase letter, a digit or an underscore.</li> <li>Example 4: [-0-9]</li> <li>Matches a space, a tab or newline.</li> <li>Example 6: [\text{II}]</li> <li>Matches a literal [] is placed after \.</li> </ul>		For example, t?p matches "top", "tip" but not "trap".
<ul> <li>necessary to make the rest of the pattern match.</li> <li>For example, t*o matches the "to" of "too", the "tro" of "trowel" and the "ty o" of "sporty orange" but not "smart orange" because the * character does not match across a new line.</li> <li>t</li> <li>This character matches the tab character.</li> <li>Example 1: \t8</li> <li>Finds every occurrence of a tab character followed by an 8.</li> <li>Example 2: initt</li> <li>Finds every occurrence of a tab character following "init".</li> <li>]</li> <li>This matches any one character or a range of single characters listed within the brackets. Brackets cannot be nested.</li> <li>[-] specifies a range of characters e.g. [a-z] or [0-9]. The beginning character in the range must have a lower ASCII value than the ending character of the range.</li> <li>[-] matches a single character; if it is not any one of the characters between [- and ]. This pattern also matches newline characters, unless the newline character is included within the brackets.</li> <li>Example 1: [AEIOU]</li> <li>Finds every uppercase vowel.</li> <li>Example 3: [A-Za-z0-9_]</li> <li>Matches an upper or lowercase letter, a digit or an underscore.</li> <li>Example 4: [-0-9]</li> <li>Matches a space, a tab or newline.</li> <li>Example 6: [\1]</li> <li>Matches a singer expression override character. If the character following the backslash is a regular expression character, it is treated as a normal character. The backslash is ignored if it is followed by a normal (non-regular expression) character.</li> </ul>	*	This character matches any number of occurrences (0 or more) of any character except a newline. Thus,
For example, t*o matches the "to" of "too", the "tro" of "trowel" and the "ty o" of "sporty orange" but not "smart orange" because the * character does not match across a new line.         t       This character matches the tab character.         Example 1: 1t8       Finds every occurrence of a tab character followed by an 8.         Example 2: initit       Finds every occurrence of a tab character following "init".         ]       This matches any one character or a range of single characters listed within the brackets. Brackets cannot be nested.         [-] specifies a range of characters e.g. [a-z] or [0-9]. The beginning character in the range must have a lower ASCII value than the ending character of the range.         [-] matches a single characters, unless the newline characters between [~ and ]. This pattern also matches newline characters, unless the newline character is included within the brackets.         Example 2: [<>?]         Finds a literal <, > or ?.         Example 3: [A-Za-20-9_]         Matches an upper or lowercase letter, a digit or an underscore.         Example 4: [-0-9]         Matches any character except a digit.         Example 5: [ ttn]         Matches a space, a tab or newline.         Example 6: [N]         Matches a literal ] if ] is placed after \.         This is the regular expression override character. If the character following the backslash is a regular expression character, it is treated as a normal character. The backslash is ignored if it is followed by a normal (non-re		this character will not match across new lines. The * character will match as few occurrences as are
orange" because the * character does not match across a new line.         t       This character matches the tab character.         Example 1: \t8         Finds every occurrence of a tab character followed by an 8.         Example 2: init\t         Finds every occurrence of a tab character following "init".         ]         This matches any one character or a range of single characters listed within the brackets. Brackets cannot be nested.         [-] specifies a range of characters e.g. [a-z] or [0-9]. The beginning character in the range must have a lower ASCII value than the ending character of the range.         [-] matches a single character if it is not any one of the characters between [~ and ]. This pattern also matches newline characters, unless the newline character is included within the brackets.         Example 1: [AEIOU]         Finds every uppercase vowel.         Example 3: [A-Za-20-9_]         Matches an upper or lowercase letter, a digit or an underscore.         Example 4: [-0-9]         Matches any character except a digit.         Example 4: [-0-9]         Matches a space, a tab or newline.         Example 6: [\U]         Matches a space, a tab or newline.         Example 6: [\U]         Matches a space, a tab or newline.         Example 6: [\U]         Matches a literal ] if ] is placed after \.         This is the regular expression ove		necessary to make the rest of the pattern match.
t       This character matches the tab character.         Example 1: \t8         Finds every occurrence of a tab character followed by an 8.         Example 2: initt         Finds every occurrence of a tab character following "init".         This matches any one character or a range of single characters listed within the brackets. Brackets cannot be nested.         [-] specifies a range of characters e.g. [a-z] or [0-9]. The beginning character in the range must have a lower ASCII value than the ending character of the range.         [-] matches a single character if it is not any one of the characters between [- and ]. This pattern also matches newline characters, unless the newline character is included within the brackets.         Example 1: [AEIOU]         Finds every uppercase vowel.         Example 3: [A-Za-20-9]         Matches an upper or lowercase letter, a digit or an underscore.         Example 4: [-0-9]         Matches any character except a digit.         Example 5: [\thn]         Matches a space, a tab or newline.         Example 6: [\]         Matches a space, a tab or newline.         Example 6: [\]         Matches a space, a tab or newline.         Example 6: [\]         Matches a space, a tab or newline.         Example 6: [\]         Matches a space, it is reated as a normal character. If the character following the backslash is a regular expression character, it is		For example, t*o matches the "to" of "too", the "tro" of "trowel" and the "ty o" of "sporty orange" but not "smart
Example 1: \t8 Finds every occurrence of a tab character followed by an 8. Example 2: init\t Finds every occurrence of a tab character following "init". ] This matches any one character or a range of single characters listed within the brackets. Brackets cannot be nested. [-] specifies a range of characters e.g. [a-z] or [0-9]. The beginning character in the range must have a lower ASCII value than the ending character of the range. [-] matches a single characters, unless the newline characters between [- and ]. This pattern also matches newline characters, unless the newline character is included within the brackets. Example 1: [AEIOU] Finds every uppercase vowel. Example 2: [->?] Finds a literal <, > or ?. Example 3: [A-Za-z0-9_] Matches an upper or lowercase letter, a digit or an underscore. Example 4: [-0-9] Matches any character except a digit. Example 5: [\t\n] Matches a space, a tab or newline. Example 6: [\]] Matches a space, a tab or newline. Example 6: [\] This is the regular expression override character. If the character following the backslash is a regular expression character, it is treated as a normal character. The backslash is ignored if it is followed by a normal (non-regular expression) character.		orange" because the * character does not match across a new line.
Finds every occurrence of a tab character following "init".         Example 2: init\t         Finds every occurrence of a tab character following "init".         This matches any one character or a range of single characters listed within the brackets. Brackets cannot be nested.         [-] specifies a range of characters e.g. [a-z] or [0-9]. The beginning character in the range must have a lower ASCII value than the ending character of the range.         [-] matches a single character if it is not any one of the characters between [~ and ]. This pattern also matches newline characters, unless the newline character is included within the brackets.         Example 1: [AEIOU]         Finds every uppercase vowel.         Example 2: [<>?]         Finds a literal <, > or ?.         Example 3: [A-Za-z0-9_]         Matches an upper or lowercase letter, a digit or an underscore.         Example 4: [-0-9]         Matches any character except a digit.         Example 5: [\thn]         Matches a literal   if ] is placed after \.         This is the regular expression override character. If the character following the backslash is a regular expression character, it is treated as a normal character. The backslash is ignored if it is followed by a normal (non-regular expression) character.	\t	This character matches the tab character.
Example 2: init\t         Finds every occurrence of a tab character following "init".         ]       This matches any one character or a range of single characters listed within the brackets. Brackets cannot be nested.         [-] specifies a range of characters e.g. [a-z] or [0-9]. The beginning character in the range must have a lower ASCII value than the ending character of the range.         [-] matches a single character; unless the newline characters between [- and ]. This pattern also matches newline characters, unless the newline character is included within the brackets.         Example 1: [AEIOU]         Finds every uppercase vowel.         Example 2: [ ]</td Finds a literal <, > or ?.         Example 3: [A-Za-z0-9_]         Matches an upper or lowercase letter, a digit or an underscore.         Example 4: [-0-9]         Matches any character except a digit.         Example 5: [ \thn]         Matches a space, a tab or newline.         Example 6: [\]]         Matches a literal ] if ] is placed after \.         This is the regular expression override character. If the character following the backslash is a regular expression character, it is treated as a normal character. The backslash is ignored if it is followed by a normal (non-regular expression) character.		Example 1: \t8
Finds every occurrence of a tab character following "init".         I       This matches any one character or a range of single characters listed within the brackets. Brackets cannot be nested.         [-] specifies a range of characters e.g. [a-z] or [0-9]. The beginning character in the range must have a lower ASCII value than the ending character of the range.         [-] matches a single character if it is not any one of the characters between [~ and ]. This pattern also matches newline characters, unless the newline character is included within the brackets.         Example 1: [AEIOU]         Finds every uppercase vowel.         Example 2: [<>?]         Finds a literal <, > or ?.         Example 3: [A-Za-z0-9_]         Matches an upper or lowercase letter, a digit or an underscore.         Example 4: [-0-9]         Matches a space, a tab or newline.         Example 5: [1\th]         Matches a space, a tab or newline.         Example 6: [\]]         Matches a literal ] if ] is placed after \.		Finds every occurrence of a tab character followed by an 8.
<ul> <li>This matches any one character or a range of single characters listed within the brackets. Brackets cannot be nested.</li> <li>[-] specifies a range of characters e.g. [a-z] or [0-9]. The beginning character in the range must have a lower ASCII value than the ending character of the range.</li> <li>[-] matches a single character if it is not any one of the characters between [~ and ]. This pattern also matches newline characters, unless the newline character is included within the brackets.</li> <li>Example 1: [AEIOU]</li> <li>Finds every uppercase vowel.</li> <li>Example 2: [&lt;&gt;?]</li> <li>Finds a literal &lt;, &gt; or ?.</li> <li>Example 3: [A-Za-z0-9_]</li> <li>Matches an upper or lowercase letter, a digit or an underscore.</li> <li>Example 4: [-0-9]</li> <li>Matches any character except a digit.</li> <li>Example 5: [\thn]</li> <li>Matches a space, a tab or newline.</li> <li>Example 6: [\thn]</li> <li>Matches a literal ] if ] is placed after \.</li> </ul>		Example 2: init\t
<ul> <li>be nested.</li> <li>[-] specifies a range of characters e.g. [a-z] or [0-9]. The beginning character in the range must have a lower ASCII value than the ending character of the range.</li> <li>[-] matches a single character if it is not any one of the characters between [~ and ]. This pattern also matches newline characters, unless the newline character is included within the brackets.</li> <li>Example 1: [AEIOU]</li> <li>Finds every uppercase vowel.</li> <li>Example 2: [&lt;&gt;?]</li> <li>Finds a literal &lt;, &gt; or ?.</li> <li>Example 3: [A-Za-z0-9_]</li> <li>Matches an upper or lowercase letter, a digit or an underscore.</li> <li>Example 4: [~0-9]</li> <li>Matches any character except a digit.</li> <li>Example 5: [ \tn]</li> <li>Matches a space, a tab or newline.</li> <li>Example 6: [\]]</li> <li>Matches a literal ] if ] is placed after \.</li> </ul>		Finds every occurrence of a tab character following "init".
<ul> <li>[-] specifies a range of characters e.g. [a-z] or [0-9]. The beginning character in the range must have a lower ASCII value than the ending character of the range.</li> <li>[-] matches a single character if it is not any one of the characters between [~ and ]. This pattern also matches newline characters, unless the newline character is included within the brackets.</li> <li>Example 1: [AEIOU]</li> <li>Finds every uppercase vowel.</li> <li>Example 2: [&lt;&gt;?]</li> <li>Finds a literal &lt;, &gt; or ?.</li> <li>Example 3: [A-Za-z0-9_]</li> <li>Matches an upper or lowercase letter, a digit or an underscore.</li> <li>Example 4: [~0-9]</li> <li>Matches any character except a digit.</li> <li>Example 5: [ \t\n]</li> <li>Matches a space, a tab or newline.</li> <li>Example 6: [\]]</li> <li>Matches a literal ] if ] is placed after \.</li> </ul>	[]	
ASCII value than the ending character of the range. [-] matches a single character if it is not any one of the characters between [~ and ]. This pattern also matches newline characters, unless the newline character is included within the brackets. Example 1: [AEIOU] Finds every uppercase vowel. Example 2: [<>?] Finds a literal <, > or ?. Example 3: [A-Za-z0-9_] Matches an upper or lowercase letter, a digit or an underscore. Example 4: [~0-9] Matches any character except a digit. Example 5: [ \tn] Matches a space, a tab or newline. Example 6: [\]] Matches a literal ] if ] is placed after \. This is the regular expression override character. If the character following the backslash is a regular expression character, it is treated as a normal character. The backslash is ignored if it is followed by a normal (non-regular expression) character.		be nested.
<ul> <li>[-] matches a single character if it is not any one of the characters between [~ and ]. This pattern also matches newline characters, unless the newline character is included within the brackets.</li> <li>Example 1: [AEIOU]</li> <li>Finds every uppercase vowel.</li> <li>Example 2: [&lt;&gt;?]</li> <li>Finds a literal &lt;, &gt; or ?.</li> <li>Example 3: [A-Za-z0-9_]</li> <li>Matches an upper or lowercase letter, a digit or an underscore.</li> <li>Example 4: [~0-9]</li> <li>Matches any character except a digit.</li> <li>Example 5: [ \t\n]</li> <li>Matches a space, a tab or newline.</li> <li>Example 6: [\]]</li> <li>Matches a literal ] if ] is placed after \.</li> </ul>		
<ul> <li>matches newline characters, unless the newline character is included within the brackets.</li> <li>Example 1: [AEIOU]</li> <li>Finds every uppercase vowel.</li> <li>Example 2: [&lt;&gt;?]</li> <li>Finds a literal &lt;, &gt; or ?.</li> <li>Example 3: [A-Za-z0-9_]</li> <li>Matches an upper or lowercase letter, a digit or an underscore.</li> <li>Example 4: [~0-9]</li> <li>Matches any character except a digit.</li> <li>Example 5: [ \t\n]</li> <li>Matches a space, a tab or newline.</li> <li>Example 6: [\]]</li> <li>Matches a literal ] if ] is placed after \.</li> </ul> This is the regular expression override character. If the character following the backslash is a regular expression character, it is treated as a normal character. The backslash is ignored if it is followed by a normal (non-regular expression) character.		
Example 1: [AEIOU]         Finds every uppercase vowel.         Example 2: [<>?]         Finds a literal <, > or ?.         Example 3: [A-Za-zO-9_]         Matches an upper or lowercase letter, a digit or an underscore.         Example 4: [~0-9]         Matches any character except a digit.         Example 5: [ \t\n]         Matches a space, a tab or newline.         Example 6: [\]]         Matches a literal ] if ] is placed after \.         This is the regular expression override character. If the character following the backslash is a regular expression character, it is treated as a normal character. The backslash is ignored if it is followed by a normal (non-regular expression) character.		
Finds every uppercase vowel.         Example 2: [<>?]         Finds a literal <, > or ?.         Example 3: [A-Za-z0-9_]         Matches an upper or lowercase letter, a digit or an underscore.         Example 4: [~0-9]         Matches any character except a digit.         Example 5: [ \t\n]         Matches a space, a tab or newline.         Example 6: [\]]         Matches a literal ] if ] is placed after \.         This is the regular expression override character. If the character following the backslash is a regular expression character, it is treated as a normal character. The backslash is ignored if it is followed by a normal (non-regular expression) character.		
Example 2: [ ]<br Finds a literal <, > or ?. Example 3: [A-Za-z0-9_] Matches an upper or lowercase letter, a digit or an underscore. Example 4: [~0-9] Matches any character except a digit. Example 5: [ \t\n] Matches a space, a tab or newline. Example 6: [\]] Matches a literal ] if ] is placed after \. This is the regular expression override character. If the character following the backslash is a regular expression character, it is treated as a normal character. The backslash is ignored if it is followed by a normal (non-regular expression) character.		
Finds a literal <, > or ?.         Example 3: [A-Za-z0-9_]         Matches an upper or lowercase letter, a digit or an underscore.         Example 4: [~0-9]         Matches any character except a digit.         Example 5: [ \t\n]         Matches a space, a tab or newline.         Example 6: [\]]         Matches a literal ] if ] is placed after \.         This is the regular expression override character. If the character following the backslash is a regular expression character, it is treated as a normal character. The backslash is ignored if it is followed by a normal (non-regular expression) character.		
Example 3: [A-Za-z0-9_] Matches an upper or lowercase letter, a digit or an underscore. Example 4: [~0-9] Matches any character except a digit. Example 5: [ \t\n] Matches a space, a tab or newline. Example 6: [\]] Matches a literal ] if ] is placed after \. This is the regular expression override character. If the character following the backslash is a regular expression character, it is treated as a normal character. The backslash is ignored if it is followed by a normal (non-regular expression) character.		
Matches an upper or lowercase letter, a digit or an underscore. Example 4: [~0-9] Matches any character except a digit. Example 5: [ \t\n] Matches a space, a tab or newline. Example 6: [\]] Matches a literal ] if ] is placed after \. This is the regular expression override character. If the character following the backslash is a regular expression character, it is treated as a normal character. The backslash is ignored if it is followed by a normal (non-regular expression) character.		
Example 4: [~0-9] Matches any character except a digit. Example 5: [ \t\n] Matches a space, a tab or newline. Example 6: [\]] Matches a literal ] if ] is placed after \. This is the regular expression override character. If the character following the backslash is a regular expression character, it is treated as a normal character. The backslash is ignored if it is followed by a normal (non-regular expression) character.		
Matches any character except a digit. Example 5: [\t\n] Matches a space, a tab or newline. Example 6: [\]] Matches a literal ] if ] is placed after \. This is the regular expression override character. If the character following the backslash is a regular expression character, it is treated as a normal character. The backslash is ignored if it is followed by a normal (non-regular expression) character.		
Example 5: [ \t\n] Matches a space, a tab or newline. Example 6: [\]] Matches a literal ] if ] is placed after \. This is the regular expression override character. If the character following the backslash is a regular expression character, it is treated as a normal character. The backslash is ignored if it is followed by a normal (non-regular expression) character.		
Matches a space, a tab or newline. Example 6: [\]] Matches a literal ] if ] is placed after \. This is the regular expression override character. If the character following the backslash is a regular expression character, it is treated as a normal character. The backslash is ignored if it is followed by a normal (non-regular expression) character.		
Example 6: [\]] Matches a literal ] if ] is placed after \. This is the regular expression override character. If the character following the backslash is a regular expression character, it is treated as a normal character. The backslash is ignored if it is followed by a normal (non-regular expression) character.		
Matches a literal ] if ] is placed after \. This is the regular expression override character. If the character following the backslash is a regular expression character, it is treated as a normal character. The backslash is ignored if it is followed by a normal (non-regular expression) character.		-
This is the regular expression override character. If the character following the backslash is a regular expression character, it is treated as a normal character. The backslash is ignored if it is followed by a normal (non-regular expression) character.		
expression character, it is treated as a normal character. The backslash is ignored if it is followed by a normal (non-regular expression) character.	1	
normal (non-regular expression) character.	/	
Searches for every occurrence of an asterisk.		
Example 2: \\		
Searches for every occurrence of a backslash.		



# 5. Placeholders

This section describes how to use the placeholders, a feature provided by several of the High-performance Embedded Workshop components.

# 5.1 What is a placeholder?

A placeholder is a special string, inserted into text, which is replaced at some subsequent time for the actual value. For example, one of the High-performance Embedded Workshop placeholders is \$(FULLFILE) which represents a file with a full path.

Suppose that you have an editor in c:\myedit\myeditor.exe, which can accept the file to be edited as a parameter. When invoking the editor (for example, you may want to open the file 'FILE1.C' from the directory 'c:\files'), the following shortcut could be made:

c:\myedit\myeditor.exe c:\files\FILE1.C

However, what happens if you want to open any file through this editor? The problem is that the command above is specific to 'c:\files\file1.c'. What we want to be able to do is to tell the High-performance Embedded Workshop to use the editor specified but to open the file that we have chosen at that time. To do this, you can substitute the specific name of the file for a general **Placeholder**:

```
c:\myedit\myeditor.exe $(FULLFILE)
```

Now whenever the High-performance Embedded Workshop launches the editor with a file, it knows that it has to replace the \$(FULLFILE) placeholder with the file you have selected.

# 5.2 Inserting a placeholder

To insert a placeholder, select in any of the following operations.

## Example 1

- 1. Place the insertion cursor at the point you would like to insert the placeholder.
- 2. Click the placeholder button. A pop-up menu will be displayed which lists all of the placeholders that are valid for the associated edit box.
- 3. Select the desired placeholder from the pop-up menu. The equivalent placeholder will be inserted into the edit box.



Filename:	Configuration directory
	Project directory
	Workspace directory
	HEW installation directory
	Workspace name
	Project name
	Configuration name

# Example 2

- 1. Select the required placeholder other than "Custom directory" from the drop-down list box and specify a subdirectory relative to the directory shown by the placeholder.
- 2. If you select "Custom directory", specify an absolute directory path in the Sub-Directory field.

<u>R</u> elative to :	
Configuration directory	•
Sub-Directory :	

### Example 3

- 1. Place the insertion cursor at the point you would like to insert the placeholder.
- 2. Select the required placeholder from the drop-down list box.
- 3. Click Insert.

<u>P</u> laceholder:		
Configuration directory	•	<u>I</u> nsert

## Example 4

1. Alternatively, if you know the placeholder already, type it into the field directly. Ensure that you type the placeholder name in uppercase and that it is preceded by '\$(' and followed by ')'.

This is correct:

\$(FILEDIR)

These are incorrect:

\$(Filedir)

\$( FILEDIR )

\$FILEDIR



# 5.3 Available placeholders

Placeholder	Meaning	Expanded placeholder (example)
\$(FULLFILE)	Filename (including full path)	c:\workspace\project\file.src
\$(FILEDIR)	File directory	c:\workspace\project
\$(FILENAME)	Filename (excluding path, including extension)	file.src
\$(FILELEAF)	Filename (excluding path and extension)	file
\$(EXTENSION)	File extension	src
\$(WORKSPDIR)	Workspace directory	c:\workspace
\$(WORKSPNAME)	Workspace name	workspace
\$(PROJDIR)	Project directory	c:\workspace\project
\$(PROJECTNAME)	Project name	project
\$(CONFIGDIR)	Configuration directory	c:\workspace\project\debug
\$(CONFIGNAME)	Configuration name	debug
\$(HEWDIR)	High-performance Embedded Workshop installation directory	c:\Program Files\Renesas\Hew
\$(TCINSTALL)	Toolchain install directory (on the options	c:\Program
	setting dialogs and the New Build Phase dialog (step 3))	Files\Renesas\Hew\Tools\Renesas\Sh\9_0_1_1
\$(TOOLDIR)	Tool installation directory (Tools Administration	) c:\Program
		Files\Renesas\Hew\Tools\Renesas\Sh\9_0_1_1
\$(TEMPDIR)	Temp directory	c:\temp
\$(WINDIR)	Windows® directory	c:\windows
\$(WINSYSDIR)	Windows® system directory	c:\windows\system
\$(EXEDIR)	Command directory	v:\vc\win32
\$(USERNAME)	User login (Version control)	JHARK
\$(PASSWORD)	User password (Version control)	214436
\$(VCDIR)	"Virtual" version control directory	"c:\project" is mapped to "x:\vc\project".
\$(COMMENT)	Comment (Version control)	"Please Enter Comment" dialog box opens.
\$(LINE)	Line number of an error/warning	12

The table below lists the available placeholders and their meanings, along with an example of their use.

In the table above, we are assuming that:

- a file path is "c:\workspace\project\file.src".
- a workspace named "workspace" is located at "c:\workspace".
- a project named "project" is located at "c:\workspace\project".
- a configuration named "debug" has a configuration directory located at "c:\workspace\project\debug".
- HEW2.EXE is installed in "c:\Program Files\Renesas\Hew".
- a \*.HRF file of a toolchain (i.e. compiler, assembler, linker) is located in "c:\Program Files\Renesas\Hew\Tools\Renesas\Sh\9\_0\_1\_1". This is referred to as \$(TCINSTALL) on the options setting dialogs and the New Build Phase dialog (step 3) of the Build menu and as \$(TOOLDIR) on the Tools Administration dialog box.



- the Windows® operating system is installed in "c:\windows" and the Windows® system directory is located at "c:\windows\system".
- a version control executable path is "v:\vc\win32\ss.exe"; a user name and its password to login to the version control system are "JHARK" and "214436" respectively; \$(COMMENT) is specified in a command line to the version control executable; "c:\project" is mapped to "x:\vc\project" on the Projects tab of the Version Control Setup dialog box, which is invoked via [Tools -> Version Control -> Configure].
- an error of compiler or assembler occurred at line 12.

### Note:

Not all of the placeholders are relevant in every field. For example, the \$(LINE) placeholder has no meaning when specifying a dependent file's location. \$(USERNAME), \$(PASSWORD), \$(VCDIR), and \$(COMMENT) placeholders are acceptable only in version control. If you enter a placeholder into an edit field where it is not acceptable you will be informed.

# 5.4 Placeholder tips

Placeholders are there to allow you to create flexible paths to the various files used by the system.

- If there is a placeholder pop-up menu () next to an edit field into which you are about to enter a path or file, you should consider how you can use a placeholder to make that path or file definition flexible.
- If you use several configurations, then the \$(CONFIGDIR) placeholder is very useful to ensure that files can be written to and from the current configuration's directory.
- Wherever possible, use a placeholder. They can always be removed or added later so don't be afraid to experiment.



# 6. I/O File Format

High-performance Embedded Workshop formats the **IO** window based on information it finds in an I/O Register definition file. When you select a debugger, High-performance Embedded Workshop will look for a "*<device>*.IO" file corresponding to the selected device and load it if it exists. This file is a formatted text file that describes the I/O modules and the address and size of their registers. You can edit this file, with a text editor, to add support for memory mapped registers or peripherals you may have specific to your application e.g. registers in an ASIC device mapped into the microcomputer's address space.

## File format

Each module name must be defined in the **Modules** definition section and the numbering of each module must be sequential. Each module corresponds to a register definition section and within the section each entry defines an I/O register.

The **BaseAddress** definition is for devices where the location of I/O registers moves in the address space depending on the CPU mode. In this case, the **BaseAddress** value is the base address of the I/O registers in one specific mode and the addresses used in the register definitions are the address locations of the registers in the same mode. When the I/O register file is actually used, the **BaseAddress** value is subtracted from the defined register address and the resultant offset added to the relevant base address for the selected mode.

Each module has a section that defines the registers forming it along with an optional dependency, the dependency is checked to see if the module is enabled or not. Each register name must be defined in the section and the numbering of each register must be sequential. The dependency is entered in the section as dep=<reg><bit><value>.

- 1. <reg> is the register id of the dependency.
- 2. <bit> is the bit position within the register.
- 3. <value> is the value that the bit must be for the module to be enabled.

The [Register] definition entry is entered in the format id=<name> <address> [<size> [<absolute>[<format>[<bitfields>]]]].

- 1. <name> register name to be displayed.
- 2. <address> address of the register.
- 3. <size> which may be B, W or L for byte, word, or long word (default is byte).
- 4. <absolute> which can be set to A if the register is at an absolute address. This is only relevant if the I/O area address range moves about on the CPU in different modes. In this case, if a register is defined as absolute the base address offset calculation is not performed and the specified address is used directly.
- 5. <format> Format for register output. Valid values are H for Hexadecimal, D for decimal, and B for binary.
- 6. <br/>section defining the bits within the register.

Bitfield sections define the bits within a register each entry is of the type bit<no>=<name>.



- 1.  $\langle no \rangle$  is the bit number.
- 2. <name> is a symbolic name of the bit.

Comment lines are allowed and must start with a ";" character.



bit3=WE1B

# 7. Symbol File Format

In order for High-performance Embedded Workshop to be able to understand and decode the symbol file correctly, the file must be formatted as a Pentica-B file:

- 1. The file must be a plain ASCII text file.
- 2. The file must start with the word "BEGIN".
- 3. Each symbol must be on a separate line with the value first, in hexadecimal terminated by an "H", followed by a space then the symbol text.
- 4. The file must end with the word "END".

Example:

BEGIN

11FAH Symbol\_name\_1

11FCH Symbol\_name\_2

11FEH Symbol\_name\_3

1200H Symbol\_name\_4

END

## Note:

Support for this function depends on the debugger.



# 8. Keyboard Shortcuts

All major commands in the High-performance Embedded Workshop application can be driven by the keyboard. Below is a list of all keyboard commands in the application.

Function key	Кеу	Function
ALT	0-9	Play a macro assigned to keyboard shortcut.
ALT	А	Toggle status bar.
ALT	К	Open the workspace window.
ALT	0	Open the output window.
ALT	F4	Exit the application.
ALT	BACKSPACE	Undo (alternative in the editor to CTRL+Z)
CTRL	0-9	Reserved for use for template insertion.
CTRL	А	Select all in the editor. May also work in other windows.
CTRL	В	Open the Breakpoints dialog box.
CTRL	С	Copy the .selected items to the clipboard.
CTRL	D	Open the disassembly window.
CTRL	F	Open the Find dialog box.
CTRL	G	Go to source line.
CTRL	Н	Open the Replace dialog box.
CTRL	I	Open the IO window.
CTRL	К	Open the stack trace window.
CTRL	L	Open the command line window.
CTRL	Μ	Open the memory window.
CTRL	Ν	Create a new source file in the editor.
CTRL	0	Open the open file dialog box.
CTRL	Р	Print the document.
CTRL	R	Open the registers window.
CTRL	S	Save the current file.
CTRL	U	Open the status window.
CTRL	V	Paste the clipboard contents.
CTRL	Х	Cut the selection.
CTRL	Υ	Redo the action.
CTRL	Z	Undo the last action.
CTRL	F2	Toggle bookmark.
CTRL	F3	Search for the next occurrence of the text chosen in the editor or open the Find dialog box.
CTRL	F4	Close the editor window.
CTRL	F6	Switch windows in the editor.
CTRL	F7	Build an individual file.
CTRL	F9	Enable or disable a breakpoint on the current line.
CTRL	SPACEBAR	Re-shown list members as a pop-up window in the editor.
CTRL	BACKSPACE	Delete current word.
CTRL	BREAK	Stop tool execution.
CTRL	INSERT	Copy the selection to the clipboard.
CTRL	ТАВ	Switch windows in the editor.



# High-performance Embedded Workshop V.4.09

Function key	Кеу	Function
CTRL	HOME	Return the cursor in the editor to the beginning of the current file.
CTRL	END	Send the cursor in the editor to the end of the current file.
CTRL	UP ARROW	Scroll window up and leave the cursor in the same place in the editor.
CTRL	DOWN ARROW	Scroll window down and leave the cursor in the same place in the editor.
CTRL	LEFT ARROW	Move the cursor to the previous word in the editor.
CTRL	<b>RIGHT ARROW</b>	Move the cursor to the next word in the editor.
CTRL+ALT	1-4	Switch virtual desktop.
CTRL+ALT	PAGE UP	Move to next tab. For example, the output or workspace window.
CTRL+ALT	PAGE DOWN	Move to previous tab. For example, the output or workspace window.
CTRL+SHIFT	8	Show white space and tab characters in the editor.
CTRL+SHIFT	G	Open the Image window.
CTRL+SHIFT	К	Open TCL toolkit.
CTRL+SHIFT	L	Line deletion in editor.
CTRL+SHIFT	Μ	Find a matching brace.
CTRL+SHIFT	N	Change the active Debugger to the next one in the Debuggers tab of the
		Workspace window. (Synchronized Debugging Facility)
CTRL+SHIFT	S	Save all modified documents in the workspace.
CTRL+SHIFT	Т	Insert a template.
CTRL+SHIFT	U	Change into a uppercase letter all the texts chosen in the editor.
CTRL+SHIFT	V	Open the Waveform window.
CTRL+SHIFT	Y	Display PC position.
CTRL+SHIFT	F3	Search for the previous occurrence of the text chosen in the editor or open the
		Find dialog box.
CTRL+SHIFT	SPACEBAR	Re-shown the function name and parameter as a pop-up window when the
		first open bracket is entered in the editor.
CTRL+SHIFT	ТАВ	Switch windows in the editor.
CTRL+SHIFT	HOME	Select from the cursor to the beginning of the file.
CTRL+SHIFT	END	Select from the cursor to the end of the file.
CTRL+SHIFT	LEFT ARROW	Select the previous word in the editor.
CTRL+SHIFT	RIGHT ARROW	Select the next word in the editor.
SHIFT	F1	Display context-sensitive help.
SHIFT	F2	Previous bookmark.
SHIFT	F3	Search for the previous occurrence of the specified text.
SHIFT	F5	Run from reset.
SHIFT	F8	Displays the editor that generated the previous build error or warning.
SHIFT	F10	Show context menu. Same as right-click pop-up menu.
SHIFT	F11	Step out of the current function.
SHIFT	F12	Refresh all windows.
SHIFT	DELETE	Cut the selection in the editor. (alternative in the editor to CTRL+X)
SHIFT	ESC	Hide the output window.
SHIFT	TAB	Move the tab back in the editor.
SHIFT	HOME	Selects from the cursor to the beginning of the current line in the editor.
SHIFT	END	Selects from the cursor to the end of the current line in the editor.
SHIFT	UP ARROW	Select the next line up.
SHIFT	DOWN ARROW	Select the next line down.



# High-performance Embedded Workshop V.4.09

Function key	Кеу	Function
SHIFT+ALT	BACKSPACE	Redo the action (alternative in the editor to CTRL+Y)
None	F1	Display context sensitive help.
None	F2	Jump to the next bookmark.
None	F3	Search for the next of the specified text.
None	F4	Find text in multiple files.
None	F5	Start or continue the program.
None	F7	Build the project.
None	F8	Display the editor that generated the next build error or warning.
None	F9	Set or remove a breakpoint on the current line.
None	F10	Step over the next statement
None	F11	Step into the next statement.
None	F12	Refresh the contents of the window.
None	DELETE	Clear.
None	INSERT	Toggle insert and overwrite mode.
None	ENTER	Carriage return in the editor.
None	ТАВ	Insert a tab in the editor.
None	ESC	Stop running the program.
None	HOME	Move the cursor to the beginning of the current line in the editor.
None	END	Move cursor to the end of the current line in the editor.
None	PAGE UP	Move the page in the editor up.
None	PAGE DOWN	Move the page in the editor down.
None	UP ARROW	Move cursor up in the editor.
None	DOWN ARROW	Move cursor down in the editor.
None	LEFT ARROW	Move cursor left in the editor.
None	RIGHT ARROW	Move cursor right in the editor.
None	ALT + Mouse select	Select column in the editor.

Note:

Support for this function depends on the debugger.



# 9. Drag and Drop in the Debugger

When using the High-performance Embedded Workshop debugger it is possible for each debug component to interact with the others. This can be achieved simply by dragging objects from one view to another.

## Some examples are listed below

- 1. It is possible to drag a label from the labels view onto other debug views. So for example if you drag a label onto the disassembly window it will scroll to the address that the label is located at.
- 2. It is possible to drag a watch variable from the editor into the watch window. This adds the watch variable to the window.
- 3. Dragging a function name from the editor into the disassembly should jump the disassembly view to the label location.



# **10. Using Labels to View Your Code**

Labels are a useful way of navigating through your debug module. It is possible to use labels in any edit field that allows addresses. If you enter a label in such a field then the built in evaluator will check the label and then convert it to an address. This allows you to enter evaluations such as "\_main+100" or "\_MyFunction+100".

This means that any times that labels are used the addresses which will be evaluated are not fixed. This is especially useful if you are using a command line batch file to set a number of breakpoints. The command line batch file might always need to set a breakpoint on a certain function and this can be achieved by using a label.

Using the label allows the code to change without affecting the batch file contents.

High-performance Embedded Workshop V.3.1 onwards also supports a number of easy ways to use label completion.



The dialog box above also shows an example of a label pick list. This store the last twenty entries made into address fields throughout the entire High-performance Embedded Workshop application. This means if you are entering a label multiple times it should be much faster and efficient if you use this recently used address field list. This control is available for all instances of the address edit field where the input is evaluated.

There is the browse button (1997) right next to the edit field where an address can be entered.

If you click this button, the **Select Label** dialog box will be displayed. A label can be selected from the current list of labels in this dialog box.



Select Label		? ×
<u>F</u> ilter:		OK
Label 🔸	Address	Cancel
add	00006558	
_allzero	00005234	
alocbuf	00005250	
calcnpw	000052CC	
CLOSEALL	000024DC	
cmpeq64	000045D0	
cmpge64	000045E8	
cmpgt64u	00004608	
cmpne64	00004620	
conv64i	00004640	
conv64u	00004640	
convi64	00004644	
convu64	0000465C	
ctype	000068A4	
div64	00005E70	
div64u	00004670	
divls	00004400	
divlu	00004408	
duchek	00006060	
errno	700005C0	
flclose	000027C4	
flopen	00002888	
flshbuf	00004BA4	
fmtout	00002A04	
free	000050AC	<b>-</b>

In this dialog box, the labels are initially listed in alphabetical order and their addresses are displayed on the right. If you click a column header (of names or addresses), the labels will be sorted by the label name or the address value.

The **Filter** edit box can be used to help you search for a specific label in the list. When the contents of the edit box are changed the label list is updated to display only the labels that contain the string entered (not case sensitive). So, in the above dialog example, if you entered "conv6" as the filter the list box would be set to only contain "\_\_\_conv64i" and "\_\_\_conv64u". This is a useful feature for when you have a large number of labels in your project.



# 11. Integrated Toolbars in a Components View

The High-performance Embedded Workshop V.4.0 onwards has the capability to include a toolbar in a views client area. This toolbar allows the views functionality to be accessed quickly from this integrated toolbar.

Various views in the High-performance Embedded Workshop system have this functionality. One example is the **Difference** view. This is shown below:

	Difference	X
Toolbar—	🖽 🛤 🛱 🐞 🕈 🛓	
	C:\Workspace\DemoSH4\DemoSH4\resetprg.c	C:\Workspace\DemoSH1\DemoSH1\resetprg.c
	○ extern void INTHandlerPRG(void)	37
	<pre>&gt;&gt; void PowerON_Reset(void);</pre>	28
	<pre>&gt;&gt; void Manual_Reset(void);</pre>	29
		<pre>void PowerON_Reset_PC(void);</pre>
		void Manual_Reset_PC(void); 🧮
	<pre>30 void main(void);</pre>	<pre>30 void main(void);</pre>
	🗉 #ifdefcplusplus	» #ifdefcplusplus
	30 }	30 }
	33 #endif	33 #endif
		•

The toolbar allows access to key Difference view features.

It is also possible to customize the toolbar further. This can be achieved via the pop-up menu of the toolbar or the component itself. If you right click on the toolbar itself the following menu is displayed:

🗸 Toolbar	
Customize	

The top menu option named "Toolbar" switches showing/hiding of the toolbar (the toolbar is hidden in the figure below).

Difference	×
C:\Workspace\DemoSH4\DemoSH4\resetprg.c	C:\Workspace\DemoSH1\DemoSH1\resetprg.c
<pre>&gt;? extern void INTHandlerPRG(void) &gt;* void PowerON_Reset(void); &gt;* void Manual_Reset(void);</pre>	27 28 29
	<pre>void PowerON_Reset_PC(void); void Manual_Reset_PC(void);</pre>
<sup>30</sup> void main(void);	<pre>30 void main(void);</pre>
#ifdef cplusplus	* #ifdef cplusplus
30 }	30 }
33 #endif	33 #endif
34	34
#ifdef culusulus	u #ifdef cnlusnlus ▼



The final menu option named "Customize" launches the **Customize Toolbar** dialog box. This is shown below:

Customize Toolbar		? ×
A <u>v</u> ailable toolbar buttons: Separator	Current toolbar buttons: Compare files ignore white space Add → Find	<u>O</u> K <u>C</u> ancel
	<ul> <li>&lt;- Bemove</li> <li>Find next occurrence of searc</li> <li>Find previous occurrence of searc</li> <li>Find previous difference</li> <li>Find next difference</li> </ul>	Move Up Move Down

This dialog box allows you to modify the displayed buttons and change the ordering. The list on the left named **Available toolbar buttons** displays all toolbar buttons not currently in use on the toolbar. The list named **Current toolbar buttons** displays all of the toolbars currently added to the components' toolbar.

#### To add the currently selected buttons to a toolbar

- 1. Select the toolbar button you wish to add from the Available toolbar buttons list.
- 2. Press the **Add** button.
- 3. Click OK.



## To move the currently selected buttons

- 1. Select the toolbar you wish to move in the **Current toolbar buttons** list.
- 2. Click Move Up or Move Down until it is in the desired position.
- 3. Click OK.

## To remove the currently selected buttons from a toolbar

- 1. Select the toolbar button you wish to remove from the **Current toolbar buttons** list.
- 2. Press the **Remove** button. The toolbar is added to the **Available toolbar buttons** list.
- 3. Click OK.



# 12. To Build in Toolchain for High-performance Embedded Workshop V.1.x

When a project created in High-performance Embedded Workshop V.1.x is used without upgrading to new toolchain that has been registered in High-performance Embedded Workshop V.2.x onwards, the toolchain for the old version must be registered. Select the 'HRF' file for the old toolchain with the **Register** button by selecting [**Tools -> Administration**]. Build can be executed on High-performance Embedded Workshop V.2.x onwards by using the old toolchain.

However, note that a workspace which has been opened in High-performance Embedded Workshop V.2.x onwards cannot be opened in High-performance Embedded Workshop V.1.x.

In High-performance Embedded Workshop V.2.x onwards, a new project for the old version cannot be created. When a project for the old version in High-performance Embedded Workshop V.1.x is created, use High-performance Embedded Workshop V.1.x.

Select HEW	Registration File	? ×
Look in: 🔁	6_0_a 🔽 🗧 🖆 🎫 -	
🗋 BIN		
🔁 hew		
include		
🔳 sh_60a0.h	<u> </u>	
File <u>n</u> ame:	sh_60a0.hrf Selec	xt
Files of <u>type</u> :	HEW Registration Files (*.hrf)	el



# 13. HMAKE User Guide

# 13.1 Command line

The following section describes the command line that should be used to execute the hmake program on a file using none or more of the available options.

#### **Basic structure**

The command line must be of the following syntax:

hmake <make file you wish to execute> <parameter list>

If a file is specified without an extension then ".mak" will be appended to it. The parameter list may include none or more of the parameters listed in the following section. The parameters list may appear before the make file name if you wish. Each parameter must be separated by at least one white space character. Parameters are not case sensitive. If no parameters are given and no file is given then help information will be displayed.

### Exit codes

If there are any syntax errors in the make file being executed or if any process executed whilst running the make file returns an invalid error code then hmake will exit with code 1. Otherwise hmake will exit with code 0 (See below for file syntax and how to specify exit code conditions).

#### Parameters

The following table shows the available parameters and their function:

Parameter	Function
/A	Execute all commands regardless of input/output file status. Equivalent to a Build All.
/N	Use status of input/output files to calculate what commands need to be executed (as normal) and
	then display the commands but do not execute them.
/?	Displays help info.

# 13.2 File syntax

There are four basic types of statement used in a hmake file, the variable declaration the description block, the comment and the message command. These can be combined in any order to produce a hmake file but a variable must be declared in a variable declaration before it is used in a description block or other variable declaration. The first "all" statement used in nmake files is not required in a hmake file. Commands are executed in order, as they appear in the make file.

Note: the " $\rightarrow$ " character is used to show were a tab character must be used in order to keep the make file syntactically correct.



#### Variable declarations

A variable declaration declares a variable which can then be used in any statement throughout the rest of the hmake file. A declaration has the following syntax:

<variable name> = <value>

Any number of white space characters are allowed between the variable name and the '=' sign and the value and the '=' sign. The value may be split over several lines using a '\' character. If the value contains '\' characters within the main text then these are taken literally. Only '\' characters followed by a new line are considered to indicate a value wrapping over more than one line.

There follows some examples of valid variable declarations:

EXECUTABLE = c:\dir\prog.exe OUTPUT = c:\dir2\file1.out INPUT = c:\dir2\file1.c DEPEND = c:\dir2\file2.h \ c:\dir2\file3.h \ c:\dir2\file4.h

In order to use a variable later in the hmake file write the variable name with "\$(" added to the front and ")" added to the back. The variable name (along with the "\$()" characters) will be substituted with the variables value. For examples of this see later under description blocks. Only alphanumeric characters and underscore characters are allowed in variable names. It is possible to use a variable inside the declaration of a different variable but all variables must be declared before they are used.

# 13.3 Description blocks

#### **Basic outline**

A description block specifies one or more targets, zero or more dependants and a list of commands which should be executed if the newest dependent is newer than the newest target. If none of the targets exist and/or none of the dependants exist then the commands will always be executed. It is not necessary to specify any dependants if you wish the commands to always be executed. A description block has the following syntax:

```
<\!\!target1\!\!><\!\!target2\!\!>\ldots:<\!\!dependent1\!\!><\!\!dependent2\!\!>\ldots
```

- $\rightarrow$  <command1>
- $\rightarrow$  <command2>
- $\rightarrow \dots$
- $\rightarrow$  <commandn>



## High-performance Embedded Workshop V.4.09

Any number of white space characters are allowed between the last target and the ':' character and the first dependant and the ':' character. No white space is allowed before the first target. Each target and each dependant must be separated by at least one white space character. A tab character must be present at the start of a line containing a command. Variables may be used in a description block using the syntax specified above under variable declarations.

There follows some examples of valid description blocks (one of which uses the variable specified above under variable declarations):

c:\dir1\file1.obj : c:\dir1\file1.c c:\dir1\file1.h

 $\rightarrow$  gcc c:\dir1\file1.c

\$(OUTPUT) : \$(INPUT) \$(DEPEND)

 $\rightarrow$  \$(EXECUTABLE) \$(INPUT)

#### Special commands

There are two special commands which can be used in a description block. The "cd" command changes the current directory and the "set" command sets an environment variable which will then be in use for the duration of the make file execution. Both are used in the same way as the DOS equivalents.

There follows some examples of valid description blocks which use these commands:

- CHANGEDIR :
- $\rightarrow$  cd c:\dir1\dir2

SETENV:

- $\rightarrow$  set VAR1=value1
- $\rightarrow$  set VAR2=value2
- $\rightarrow$  set VAR3=value3

It does not matter that CHANGEDIR and SETENV are not file names. They will be treated as files that do not exist and so the commands will always be executed.

#### Sub command files

If you wish hmake to generate a sub command file for you then the command part of the description block should be specified as follows (this replaces <commandn> above):



- $\rightarrow$  <command start> <<
- $\rightarrow$  <sub command1>,
- $\rightarrow$  <sub command2>,

```
\rightarrow ...
```

- $\rightarrow$  <sub commandn>
- <<<command end>

This will generate a sub command file, in the Windows® temporary directory, which will contain the lines <sub command1>, <sub command2> etc. This command file will be deleted once the make process has completed. The name of the command file will be substituted for all the text between the two "<<". You do not have to worry about the name of the sub command file. This is generated by hmake. For example:

c:\dir1\file1.obj : c:\dir1\file1.c c:\dir1\file1.h

- $\rightarrow$  gcc @"<<
- $\rightarrow \ \mbox{-c} \ \mbox{-o} \ \mbox{c:\dir1\file1.c}$
- <<''

If the sub command file generated has the name "c:\temp\hmk111.cmd" then the following would be executed by hmake (assuming c:\dir1\file1.obj is out of date):

```
gcc @"c:\temp\hmk111.cmd"
```

The command file (c:\temp\hmk111.cmd) would contain:

```
-c -o c:\dir1\file1.obj c:\dir1\file1.c
```

It is possible to include more than one command in the description block and to use combinations of the standard, and sub command file commands.



# 13.4 Comments

A '#' character signifies a comment. When this character appears as the first character on a line the rest of the line (up until the next new line character) is ignored. There follows examples of valid comments:

# My hmake file

# Variable declaration

OUTPUT= c:\dir1\file1.obj

# Descriptor

 $(OUTPUT) : c:\dir1\file1.c c:\dir1\file1.h$ 

 $\rightarrow$  set VAR1=value1

 $\rightarrow$  gcc c:\dir1\file1.c

A comment must occupy its own line in the hmake file. It is not possible to put comments on the end of other statements.

# 13.5 Message commands

The message command is used to output a line of text to standard out whilst a make file is executing. These text lines will be output in the order they appear in the make file, in amongst output from any executables being executed as appropriate. No buffering of output text will take place. A message command has the following syntax:

!MESSAGE <text to output>

A new line character is assumed to come after the last character in <text to output>. Any white space between !MESSAGE and <text to output> will be ignored. There follows an example of a valid message command:

!MESSAGE Executing C Compiler



High-performance Embedded Workshop V.4.09 User's Manual					
Publication Date:	Feb 01, 2011	Rev.1.00			
Published by:	Renesas Electronics Corporation				
Edited by: Microcomputer Tool Development Department Renesas Solutions Corp.					



**Renesas Electronics Corporation** 

#### SALES OFFICES

http://www.renesas.com

Refer to "http://www.renesas.com/" for the latest and detailed information. Renesas Electronics America Inc. 2880 Scott Boulevard Santa Clara, CA 95050-2554, U.S.A. Tel: +1-408-588-6000, Fax: +1-408-588-6130 Renesas Electronics Canada Limited 1101 Nicholson Road, Newmarket, Ontario L3Y 9C3, Canada Tel: +1-905-898-5441, Fax: +1-905-898-3200 Renesas Electronics Europe Limited Dukes Meadow, Millboard Road, Bourne End, Buckinghamshire, SL8 5FH, U.K Tel: +44-1628-585-100, Fax: +44-1628-585-900 Renesas Electronics Europe GmbH Arcadiastrasse 10, 40472 Düsseldorf, Germany Tel: +49-211-65030, Fax: +49-211-6503-1327 Renesas Electronics (China) Co., Ltd. 7th Floor, Quantum Plaza, No.27 ZhiChunLu Haidian District, Beijing 100083, PR.China Tel: +86-10-8235-1155, Fax: +86-10-8235-7679 Renesas Electronics (Shanghai) Co., Ltd. 7th Floor, Quantum Plaza, No.27 ZhiChunLu Haidian District, Beijing 100083, PR.China Tel: +86-21-5877-1818, Fax: +86-21-6887-7858 / -7898 Renesas Electronics (Shanghai) Co., Ltd. 7th Floor, Quantum Plaza, No.27 ZhiChunLu Haidian District, Beijing 100083, PR.China Tel: +86-21-5877-1818, Fax: +86-21-6887-7858 / -7898 Renesas Electronics (Shanghai) Co., Ltd. 7th Floor, Quantum Plaza, No.27 ZhiChunLu Haidian District, Beijing 100083, PR.China Tel: +86-21-5877-1818, Fax: +86-21-6887-7858 / -7898 Renesas Electronics (Shanghai) Co., Ltd. 7th Floor, Quantum Plaza, Bax: +86-21-6887-7858 / -7898 Renesas Electronics Toing Guimited Unit 1001-1613, 16/F, Tower 2, Grand Century Place, 193 Prince Edward Road West, Mongkok, Kowloon, Hong Kong Tel: +886-28175-9900, Fax: +886-28175-9670 Renesas Electronics Singapore Pte. Ltd. 1 harbourfront Avenue, #06-10, keppel Bay Tower, Singapore 098632 Tel: +682-28175-9900, Fax: +65-278-8001 Renesas Electronics Korea Co., Ltd. 11 harbourfront Avenue, #06-10, keppel Bay Tower, Singapore 098632 Tel: +602-7955-990, Fax: +65-278-801 Renesas Electronics Korea Co., Ltd. 11 F, Samik Lavied' or Bidg,, 720-2 Yeoksam-Dong, Kangnam-Ku, Seoul 135-080, Korea 

# High-performance Embedded Workshop V.4.09 User's Manual



R20UT0372EJ0100